UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2004

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to ____.

Commission file number 1-13300

CAPITAL ONE FINANCIAL CORPORATION

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

1680 Capital One Drive, McLean, Virginia (Address of principal executive offices) 54-1719854 (I.R.S. Employer Identification No.)

> 22102 (Zip Code)

(703) 720-1000

(Registrant's telephone number, including area code)

(Not Applicable)

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗵 No 🗆

As of June 30, 2004 there were 241,511,250 shares of the registrant's Common Stock, par value \$.01 per share, outstanding.

CAPITAL ONE FINANCIAL CORPORATION FORM 10-Q INDEX

June 30, 2004

			Page
PART I.	FINANC	TAL INFORMATION	
	Item 1.	<u>Financial Statements (unaudited):</u>	
		Condensed Consolidated Balance Sheets	3
		Condensed Consolidated Statements of Income	4
		Condensed Consolidated Statements of Changes in Stockholders' Equity	5
		Condensed Consolidated Statements of Cash Flows	6
		Notes to Condensed Consolidated Financial Statements	7
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	14
	Item 3.	Quantitative and Qualitative Disclosure of Market Risk	49
	Item 4.	Controls and Procedures	49
PART II.	OTHER	INFORMATION	
	Item 1.	Legal Proceedings	50
	Item 2.	Changes in Securities, Uses of Proceeds and Issuer Purchases of Equity Securities	50
	Item 6.	Exhibits and Reports on Form 8-K	50
		Signatures	51
		_	

Part I. Financial Information

Item 1. Financial Statements (unaudited)

CAPITAL ONE FINANCIAL CORPORATION

Condensed Consolidated Balance Sheets

(Dollars in thousands, except per share data) (unaudited)

	June 30, 2004	December 31, 2003
Assets:		
Cash and due from banks	\$ 346,978	\$ 382,212
Federal funds sold and resale agreements	1,082,939	1,010,319
Interest-bearing deposits at other banks	290,242	587,751
Cash and cash equivalents	1,720,159	1,980,282
Securities available for sale	8,946,836	5,866,628
Consumer loans	34,551,343	32,850,269
Less: Allowance for loan losses	(1,425,000)	(1,595,000)
Net loans	33,126,343	31,255,269
Accounts receivable from securitizations	3,972,754	4,748,962
Premises and equipment, net	868,203	902,600
Interest receivable	234,348	214,295
Other	1,201,018	1,315,670
Total assets	\$ 50,069,661	\$46,283,706
Liabilities:		
Interest-bearing deposits	\$ 24,178,756	\$22,416,332
Senior and subordinated notes	7,727,810	7,016,020
Other borrowings	7,885,340	7,796,613
Interest payable	256,293	256,015
Other	2,800,405	2,746,915
Total liabilities	42,848,604	40,231,895
Stockholders' Equity:		
Preferred stock, par value \$.01 per share; authorized 50,000,000 shares, none issued or outstanding	—	—
Common stock, par value \$.01 per share; authorized 1,000,000,000 shares; 242,819,065 and 236,352,914 shares issued		
as of June 30, 2004 and December 31, 2003, respectively	2,428	2,364
Paid-in capital, net	2,348,401	1,937,302
Retained earnings	4,924,023	4,078,508
Cumulative other comprehensive income (loss)	(4,367)	83,158
Less: Treasury stock, at cost; 1,307,815 and 1,310,582 shares as of June 30, 2004 and December 31, 2003, respectively	(49,428)	(49,521)
Total stockholders' equity	7,221,057	6,051,811
Total liabilities and stockholders' equity	\$ 50,069,661	\$46,283,706

See Notes to Condensed Consolidated Financial Statements.

CAPITAL ONE FINANCIAL CORPORATION

Condensed Consolidated Statements of Income (Dollars in thousands, except per share data) (unaudited)

		nths Ended ne 30	Six Months Ended June 30			
	2004	2003	2004	2003		
Interest Income:						
Consumer loans, including past-due fees	\$ 1,019,076	\$ 960,124	\$ 2,054,093	\$ 1,973,406		
Securities available for sale	76,081	47,895	139,797	90,826		
Other	56,789	62,261	122,787	112,614		
Total interest income	1,151,946	1,070,280	2,316,677	2,176,846		
Interest Expense:						
Deposits	244,978	220,640	484,490	429,948		
Senior and subordinated notes	124,809	99,120	249,227	197,699		
Other borrowings	71,142	68,257	139,921	132,132		
Total interest expense	440,929	388,017	873,638	759,779		
Net interest income	711,017	682,263	1,443,039	1,417,067		
Provision for loan losses	242,256	387,097	485,924	762,948		
110413101110111011103513				/ 02,340		
Net interest income after provision for loan losses	468,761	295,166	957,115	654,119		
Non-Interest Income:						
Servicing and securitizations	868,041	742,696	1,785,710	1,472,385		
Service charges and other customer-related fees	368,469	402,970	722,962	844,196		
Interchange	117,329	89,141	222,924	174,492		
Other	42,225	75,815	107,602	124,152		
Total non-interest income	1,396,064	1,310,622	2,839,198	2,615,225		
Non-Interest Expense:						
Salaries and associate benefits	419,695	374,245	844,087	772,712		
Marketing	253,838	270,555	508,985	512,251		
Communications and data processing	108,191	112,456	225,297	224,508		
Supplies and equipment	74,582	87,680	162,903	171,492		
Occupancy	70,494	42,755	109,213	86,329		
Other	302,012	263,865	603,223	558,196		
Total non-interest expense	1,228,812	1,151,556	2,453,708	2,325,488		
Income before income taxes	636,013	454,232	1,342,605	943,856		
Income taxes	228,626	168,066	484,412	349,227		
Net income	\$ 407,387	\$ 286,166	\$ 858,193	\$ 594,629		
Desis cominge and the	¢ 174	¢ 1.00	¢	¢ 2.00		
Basic earnings per share	\$ 1.74	\$ 1.28	\$ 3.68	\$ 2.66		
Diluted earnings per share	\$ 1.65	\$ 1.23	\$ 3.48	\$ 2.58		
Dividends paid per share	\$ 0.03	\$ 0.03	\$ 0.05	\$ 0.05		

See Notes to Condensed Consolidated Financial Statements.

CAPITAL ONE FINANCIAL CORPORATION

Condensed Consolidated Statements of Changes in Stockholders' Equity

(Dollars in thousands, except per share data) (unaudited)

	Common Stock						umulative Other	-	Total
	Shares	Amount	Paid-in Capital	Deferred Compensation	Retained Earnings		prehensive ome (Loss)	Treasury Stock	Stockholder's Equity
Balance, December 31, 2002	227,073,162	\$2,271	\$ 1,806,440	\$ (101,970)	\$2,966,948	\$	(15,566)	\$ (34,952)	\$4,623,171
Comprehensive income:									
Net income					594,629				594,629
Other comprehensive income,									
net of income tax:									
Unrealized gains on									
securities, net of									
income taxes of									
\$13,621							25,509		25,509
Foreign currency									
translation adjustments							16,353		16,353
Unrealized gains on cash									
flow hedging									
instruments, net of							14007		14007
income taxes of \$8,414							14,327		14,327
							EC 400		
Other comprehensive income							56,189		56,189
Comprehensive income					(10.001)				650,818
Cash dividends - \$.05 per share					(12,061)			(1.1.5.1.1)	(12,061)
Purchase of treasury stock								(14,214)	(14,214)
Issuances of common and restricted	601 106	-	25 417	(2,205)					22.020
stock, net	691,196	7	25,417	(3,385)					22,039
Exercise of stock options	662,952	6	15,682	10 212					15,688
Amortization of deferred compensation			070	19,312					19,312
Other items, net			973						973
Palance June 20, 2002	228,427,310	\$2,284	\$ 1,848,512	\$ (86,043)	\$3,549,516	\$	40,623	\$ (49,166)	\$5,305,726
Balance, June 30, 2003	220,427,310	Ф 2,204	р 1,040,312	\$ (80,043)	\$5,549,510	ф	40,023	\$(49,100)	\$ 5,505,720
Balance, December 31, 2003	236,352,914	\$2,364	\$ 2,164,435	\$ (227,133)	\$4,078,508	\$	83,158	\$ (49,521)	\$6,051,811
Comprehensive income:									
Net income					858,193				858,193
Other comprehensive loss, net of									
income tax:									
Unrealized losses on									
securities, net of									
income tax benefit of									
\$72,431							(126,991)		(126,991)
Foreign currency									
translation adjustments							9,095		9,095
Unrealized gains on cash									
flow hedging									
instruments, net of									
income taxes of									
\$17,962							30,371		30,371
							105		
Other comprehensive loss							(87,525)		(87,525)
Comprehensive income									770,668
Cash dividends - \$.05 per share					(12,678)				(12,678)
Issuances of common and restricted		(4)	000	44 405				00	10.050
Issuances of common and restricted stock, net of forfeitures	(73,015)	(1)	980	11,187				93	12,259
Issuances of common and restricted stock, net of forfeitures Exercise of stock options	(73,015) 6,539,166	(1) 65	980 334,014					93	334,079
Issuances of common and restricted stock, net of forfeitures Exercise of stock options Amortization of deferred compensation				11,187 44,725				93	
Issuances of common and restricted stock, net of forfeitures Exercise of stock options Amortization of deferred compensation Common stock issuable under			334,014					93	334,079 44,725
Issuances of common and restricted stock, net of forfeitures Exercise of stock options Amortization of deferred compensation						_		93	334,079
Issuances of common and restricted stock, net of forfeitures Exercise of stock options Amortization of deferred compensation Common stock issuable under			334,014		\$4,924,023	\$	(4,367)	93 \$ (49,428)	334,079 44,725

See Notes to Condensed Consolidated Financial Statements.

CAPITAL ONE FINANCIAL CORPORATION Condensed Consolidated Statements of Cash Flows (Dollars in thousands) (unaudited)

		ths Ended ie 30
	2004	2003
Operating Activities:		
Net income	\$ 858,193	\$ 594,629
Adjustments to reconcile net income to cash provided by operating activities:		
Provision for loan losses	485,924	762,948
Depreciation and amortization, net	197,401	186,165
Impairment of long-lived assets	34,703	
Losses (gains) on securities available for sale	20,941	(9,564)
Gains on sales of automobile loans	(24,405)	(39,312
Stock plan compensation expense	64,918	20,576
(Increase) decrease in interest receivable	(20,053)	16,995
Decrease (increase) in accounts receivable from securitizations	774,166	(479,255)
Decrease in other assets	254,221	366,917
Increase (decrease) in interest payable	278	(5,245)
Increase (decrease) in other liabilities	83,884	(471,966)
increase (decrease) in other natimites	03,004	(471,900)
Net cash provided by operating activities	2,730,171	942,888
Investing Activities:		
Purchases of securities available for sale	(5,011,254)	(2,221,002)
Proceeds from maturities of securities available for sale	706,085	576,866
Proceeds from sales of securities available for sale	984,134	677,344
Proceeds from sale of automobile loans	600,917	1,445,291
Proceeds from securitization of consumer loans	5,946,510	4,797,861
Net increase in consumer loans	(9,134,567)	(6,795,790
Recoveries of loans previously charged off	230,616	178,536
Additions of premises and equipment, net	(119,130)	(118,122)
Net cash used for investing activities	(5,796,689)	(1,459,016)
Financing Activities:		
Net increase in interest-bearing deposits	1,762,424	2,495,916
Net increase (decrease) in other borrowings	88,599	(127,784)
Issuances of senior notes	998,190	1,092,144
Maturities of senior notes	(295,000)	(678,690
Issuances (purchases) of treasury stock	93	(3,714
Dividends paid	(12,678)	(12,061)
Net proceeds from issuances of common stock	12,166	11,539
Proceeds from exercise of stock options	252,601	15,688
Net cash provided by financing activities	2,806,395	2,793,038
Increase in cash and cash equivalents	(260,123)	2,276,910
Cash and cash equivalents at beginning of period	1,980,282	918,778
Cash and cash equivalents at end of period	\$ 1,720,159	\$ 3,195,688

See Notes to Condensed Consolidated Financial Statements.

CAPITAL ONE FINANCIAL CORPORATION Notes to Condensed Consolidated Financial Statements (Dollars in thousands, except per share data) (unaudited)

Note A: Significant Accounting Policies

Business

The condensed consolidated financial statements include the accounts of Capital One Financial Corporation (the "Corporation") and its subsidiaries. The Corporation is a holding company whose subsidiaries market a variety of financial products and services to consumers. The principal subsidiaries are Capital One Bank (the "Bank"), which offers credit card products, Capital One, F.S.B. (the "Savings Bank"), which offers consumer lending (including credit cards) and deposit products, and Capital One Auto Finance, Inc. ("COAF") which offers primarily automobile financing products. The Corporation and its subsidiaries are collectively referred to as the "Company."

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by GAAP for complete consolidated financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Actual results could differ from these estimates.

Operating results for the three and six months ended June 30, 2004 are not necessarily indicative of the results for the year ending December 31, 2004.

The notes to the consolidated financial statements contained in the Annual Report on Form 10-K for the year ended December 31, 2003 should be read in conjunction with these condensed consolidated financial statements.

All significant intercompany balances and transactions have been eliminated.

Stock-Based Compensation

In December 2003, the Company adopted the expense recognition provisions of SFAS No. 123, *Accounting for Stock Based Compensation*, ("SFAS 123"), prospectively to all awards granted, modified, or settled after January 1, 2003. Typically, awards under the Company's plans vest over a three year period. Therefore, cost related to stock-based compensation included in net income for 2003 and 2004 is less than that which would have been recognized if the fair value method had been applied to all awards since the original effective date of SFAS 123. The effect on net income and earnings per share if the fair value based method had been applied to all outstanding and unvested awards in each period is presented in the table below. The fair value of options was estimated at the date of grant using the Black-Scholes option-pricing model and is amortized into expense over the options' vesting period.

	F	or the Th Ended		hs	For the Six Months Ended June 30			
Pro Forma Information	200	04	2	2003	:	2004		2003
Net income, as reported	\$407	,387	\$28	36,166	\$8	58,193	\$5	94,629
Stock-based employee compensation expense included in reported net income	17	,966		6,661		35,261		13,431
Stock-based employee compensation expense determined under fair value based method ⁽¹⁾	(50	,184)	(4	44,209)	(1	00,815)	(86,813)
	_	_	_		_		_	
Pro forma net income	\$375	,169	\$24	48,618	\$ 7	92,639	\$5	21,247
Earnings per share:								
Basic – as reported	\$	1.74	\$	1.28	\$	3.68	\$	2.66
Basic – pro forma	\$	1.60	\$	1.11	\$	3.40	\$	2.33
Diluted – as reported	\$	1.65	\$	1.23	\$	3.48	\$	2.58
Diluted – pro forma	\$	1.51	\$	1.11	\$	3.20	\$	2.26

⁽¹⁾ Includes amortization of compensation expense for current year stock option grants and prior year stock option grants over the stock options' vesting period.

The fair value of the options granted during the three and six months ended June, 2004 and 2003 was estimated at the date of grant using a Black-Scholes optionpricing model with the weighted average assumptions described below.

	For the Three Ended Ju		For the Six Months Ended June 30		
Assumptions	2004 ⁽¹⁾	2003	2004 ⁽¹⁾	2003	
Dividend yield	.15%	.24%	.15%	.29%	
Volatility factors of expected market price of stock	47%	55%	58%	55%	
Risk-free interest rate	2.50%	2.73%	2.22%	2.77%	
Expected option lives (in years)	2.2	4.5	2.8	4.5	

⁽¹⁾ Stock options granted during the three and six months ended June 30, 2004 consisted primarily of grants issued upon exercises of vested stock options through stock swaps in accordance with stock option agreements.

Note B: Segments

The Company maintains three distinct operating segments: U.S. Card, Auto Finance, and Global Financial Services. The U.S. Card segment consists of domestic credit card lending activities. The Auto Finance segment primarily consists of automobile financing activities. The Global Financial Services segment is comprised of international lending activities (including credit card lending), installment lending, small business lending, patient financing, and other investment businesses. The U.S. Card, Auto Finance and Global Financial Services segments are considered reportable segments based on quantitative thresholds applied to the managed loan portfolio for reportable segments provided by SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*, and are disclosed separately. The Other caption includes the Company's liquidity portfolio, emerging businesses not included in the reportable segments, investments in external companies, and various non-lending activities. The Other caption also includes the net impact of transfer pricing, certain unallocated expenses and gains/losses related to the securitization of assets.

Management decision making is performed on a managed portfolio basis. An adjustment to reconcile the managed financial information to the reported financial information in the consolidated financial statements is provided. This adjustment reclassifies a portion of net interest income, non-interest income and provision for loan losses into non-interest income from servicing and securitization.

The Company maintains its books and records on a legal entity basis for the preparation of financial statements in conformity with GAAP. The following tables present information prepared from the Company's internal management information system, which is maintained on a line of business level through allocations from the consolidated financial results.

		For the Three Months Ended June 30, 2004											
		U.S. Card		Auto Finance		Global Financial Services	Other		Total Managed		curitization djustments		Total Reported
Net interest income	\$	1,124,099	\$	195,974	\$	338,192	\$ (72,795)	\$	1,585,470	\$	(874,453)	\$	711,017
Non-interest income		816,034		22,666		185,488	(12,890)		1,011,298		384,766		1,396,064
Provision for loan losses		519,569		54,908		159,001	(1,535)		731,943		(489,687)		242,256
Non-interest expenses		820,424		81,345		295,117	31,926		1,228,812				1,228,812
Income tax provision (benefit)		216,051		29,659		23,471	(40,555)		228,626		_		228,626
Net income (loss)	\$	384,089	\$	52,728	\$	46,091	\$ (75,521)	\$	407,387	\$	_	\$	407,387
			_					_					
Loans receivable	\$ 4	45,247,444	\$	9,383,432	\$	18,722,812	\$ 13,664	\$	73,367,352	\$ (38,816,009)	\$	34,551,343

		For the Three Months Ended June 30, 2003											
		U.S. Card		Auto Finance		Global Financial Services	Other		Total Managed		curitization djustments		Total Reported
Net interest income	\$	1,072,943	\$	180,890	\$	255,612	\$(51,928)	\$	1,457,517	\$	(775,254)	\$	682,263
Non-interest income		856,028		37,551		147,165	5,268		1,046,012		264,610		1,310,622
Provision for loan losses		671,154		77,977		132,673	15,937		897,741		(510,644)		387,097
Non-interest expenses		822,644		70,620		231,711	26,581		1,151,556		_		1,151,556
Income tax provision (benefit)		161,014		25,842		12,872	(31,662)		168,066				168,066
									<u> </u>				
Net income (loss)	\$	274,159	\$	44,002	\$	25,521	\$ (57,516)	\$	286,166	\$		\$	286,166
	_		_		_			_		_		_	
Loans receivable	\$	39,318,185	\$	7,379,815	\$	14,045,765	\$ (7,956)	\$	60,735,809	\$ (33,887,231)	\$	26,848,578

		For the Six Months Ended June 30, 2004											
		U.S. Card		Auto Finance		Global Financial Services	Other		Total Managed		ecuritization Adjustments		Total Reported
Net interest income	\$	2,324,676	\$	385,172	\$	670,081	\$(117,381)	\$	3,262,548	\$	(1,819,509)	\$	1,443,039
Non-interest income		1,585,090		46,096		362,814	31,834		2,025,834		813,364		2,839,198
Provision for loan losses		1,054,848		135,090		312,438	(10,307)		1,492,069		(1,006,145)		485,924
Non-interest expenses		1,650,349		165,878		57 4,9 77	62,504		2,453,708				2,453,708
Income tax provision (benefit)		433,645		46,908		48,454	(44,595)		484,412		_		484,412
Net income (loss)	\$	770,924	\$	83,392	\$	97,026	\$ (93,149)	\$	858,193	\$	_	\$	858,193
	_		_					_		_		_	
Loans receivable	\$	45,247,444	\$	9,383,432	\$	18,722,812	\$ 13,664	\$	73,367,352	\$	(38,816,009)	\$	34,551,343

		For the Six Months Ended June 30, 2003										
	U.S. Card	Auto Finance	Global Financial Services	Other	Total Managed	Securitization Adjustments	Total Reported					
Net interest income	\$ 2,181,655	\$ 355,303	\$ 496,957	\$ (68,447)	\$ 2,965,468	\$ (1,548,401)	\$ 1,417,067					
Non-interest income	1,723,539	52,561	290,706	7,139	2,073,945	541,280	2,615,225					
Provision for loan losses	1,369,010	209,959	269,155	(78,055)	1,770,069	(1,007,121)	762,948					
Non-interest expenses	1,611,927	138,316	458,829	116,416	2,325,488	_	2,325,488					
Income tax provision (benefit)	341,975	22,048	19,186	(33,982)	349,227	_	349,227					
Net income (loss)	\$ 582,282	\$ 37,541	\$ 40,493	\$ (65,687)	\$ 594,629	\$ —	\$ 594,629					
Loans receivable	\$ 39,318,185	\$ 7,379,815	\$ 14,045,765	\$ (7,956)	\$ 60,735,809	\$ (33,887,231)	\$ 26,848,578					

During the three months ended June 30, 2004, the Company recognized \$56.0 million in early termination and facility consolidation charges related to cost reduction initiatives. Of this amount, \$49.9 million was allocated to the U.S. Card segment, \$5.1 million was allocated to the Global Financial Services segment, \$0.5 million was allocated to the Auto Finance segment, and \$0.5 million was held in the Other category.

During the three months ended June 30, 2004 and 2003, the Company sold auto loans of \$322.7 million and \$1.3 billion, respectively. These transactions resulted in allocated pre-tax income for the Auto Finance segment of \$12.5 million and \$26.0 million during the three months ended June 30, 2004 and 2003, respectively.

During the six months ended June 30, 2004 and 2003, the Company sold auto loans of \$582.4 million and \$1.4 billion, respectively. These transactions resulted in allocated pre-tax income of \$25.8 million and \$30.2 million for the Auto Finance segment during the six months ended June 30, 2004 and 2003, respectively.

In July of 2004, the Company provided notice to terminate its forward flow auto receivables agreement; however, the Company plans to continue to sell auto receivables through other channels.

Note C: Capitalization

In June 2004, the Bank issued \$500.0 million of five-year 5.00% fixed rate bank notes under the Senior and Subordinated Global Bank Note Program.

In June 2004, the Company terminated its Domestic Revolving and Multicurrency Credit Facilities and replaced them with a new revolving credit facility ("Credit Facility") providing for an aggregate of \$750.0 million in unsecured borrowings from various lending institutions to be used for general corporate purposes. The Credit Facility is available to the Corporation, the Bank, the Savings Bank, and Capital One Bank (Europe), plc. The Corporation's availability has been increased to \$500.0 million under the Credit Facility. All borrowings under the Credit Facility are based upon varying terms of London Interbank Offering Rate ("LIBOR").

In February 2004, the Bank issued \$500.0 million of ten-year 5.125% fixed rate bank notes under the Senior and Subordinated Global Bank Note Program.

Note D: Comprehensive Income

Comprehensive income for the three months ended June 30, 2004 and 2003, respectively was as follows:

		onths Ended ıne 30
	2004	2003
Comprehensive Income:		
Net income	\$ 407,387	\$ 286,789
Other comprehensive (loss) income, net of tax	(151,775)	65,979
Total comprehensive income	\$ 255,612	\$ 352,768

Note E: Earnings Per Share

The following table sets forth the computation of basic and diluted earnings per share:

	Three	Three Months Ended June 30		Six Months Ended June 30	
	2004	2003 ⁽¹⁾	2004	2003 ⁽¹⁾	
Numerator:					
Net income	\$ 407,38	\$ 286,166	\$ 858,193	\$ 594,629	
Denominator:					
Denominator for basic earnings per share - Weighted-average shares	234,73	2 223,691	233,377	223,277	
Effect of dilutive securities:					
Stock options	10,34	1 8,111	10,813	6,791	
Restricted stock	2,51	8 750	2,312	375	
Dilutive potential common shares	12,85	9 8,861	13,125	7,166	
Denominator for diluted earnings per share - Adjusted weighted-average shares	247,59	1 232,552	246,502	230,443	
Basic earnings per share	\$ 1.7	4 \$ 1.28	\$ 3.68	\$ 2.66	
Diluted earnings per share	\$ 1.6	5 \$ 1.23	\$ 3.48	\$ 2.58	

⁽¹⁾ In the fourth quarter 2003, the Company adopted the expense recognition provisions of SFAS 123, under the prospective method for all awards granted, modified or settled after January 1, 2003. Amounts related to the three and six months ended June 30, 2003, have been restated in accordance with SFAS 123.

Note F: Goodwill

The following table provides a summary of the goodwill.

	Auto Finance	Global Financial Services	Total
Balance at December 31, 2003	\$ 218,957	\$136,978	\$355,935
Impairment loss	—	(3,848)	(3,848)
Foreign currency translation	—	70	70
		<u> </u>	
Balance at June 30, 2004	\$ 218,957	\$133,200	\$352,157

In March 2004, the Company recognized a \$3.8 million impairment loss on goodwill related to certain international operations. This impairment was recorded in other non-interest expense in the consolidated income statement.

Note G: Impairment of Long-Lived Assets

In June 2004, the Company approved a plan to liquidate certain buildings as part of its facility consolidation efforts. The Company expects to complete the sale within one year. The buildings met the held-for-sale criteria of SFAS No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets* and as such, an impairment charge of \$34.7 million was recognized to write down the properties to fair value. Depreciation expense is no longer being recognized for these assets. As of June 30, 2004, the carrying value of these assets was \$44.2 million.

Note H: Consolidation of Variable Interest Entities

In June 2004, the Company established and consolidated Capital One Appalachian LLC ("COAL") which qualifies as a variable interest entity under the requirements of FASB Interpretation No. 46(R), *Consolidation of Variable Interest Entiries, an Interpretation of ARB No. 51* ("FIN 46"). COAL purchased a limited interest in a partnership from a third party that operates a facility which produces a coal-based synthetic fuel that qualifies for tax credits pursuant to Section 29 of the Internal Revenue Code. COAL paid \$2.1 million in cash and agreed to a fixed note payable of \$26.4 million and additional quarterly variable payments based on the amount of tax credits generated by the partnership from June 2004 through the end of 2007. COAL has an ongoing commitment to fund the losses of the partnership to maintain its 24.9% minority ownership interest, and make fixed and variable payments to the third party. The Corporation has guaranteed COAL's commitments to both the partnership and the third party. COAL's equity investment in the partnership was included in Other Assets at June 30, 2004.

Note I: Commitments and Contingencies

Securities Litigation

Beginning in July 2002, the Corporation was named as a defendant in twelve putative class action securities cases. All twelve actions were filed in the United States District Court for the Eastern District of Virginia. Each complaint also named as "Individual Defendants" several of the Corporation's executive officers.

On October 1, 2002, the Court consolidated these twelve cases. Pursuant to the Court's order, Plaintiffs filed an amended complaint on October 17, 2002, which alleged that the Corporation and the Individual Defendants violated Section 10(b) of the Exchange Act, Rule 10b-5 promulgated thereunder, and Section 20(a) of the Exchange Act. The amended complaint asserted a class period of January 16, 2001, through July 16, 2002, inclusive. The amended complaint alleged generally that, during the asserted class period, the Corporation misrepresented the adequacy of its capital levels and loan loss allowance relating to higher risk assets. In addition, the amended complaint alleged generally that the Corporation failed to disclose that it was experiencing serious infrastructure deficiencies and systemic computer problems as a result of its growth.

On December 4, 2002, the Court granted defendants' motion to dismiss plaintiffs' amended complaint with leave to amend. Pursuant to that order, plaintiffs filed a second amended complaint on December 23, 2002, which asserted the same class period and alleged violations of the same statutes and rule. The second amended complaint also added a new Individual Defendant and asserted violations of GAAP. Defendants moved to dismiss the second amended complaint on January 8, 2003, and plaintiffs filed a motion on March 6, 2003, seeking leave to amend their complaint. On April 10, 2003, the Court granted defendants' motion to dismiss plaintiffs' second amended complaint, denied plaintiffs' motion for leave to amend, and dismissed the consolidated action with prejudice. Plaintiffs appealed the Court's order, opinion, and judgment to the United States Court of Appeals for the Fourth Circuit on May 8, 2003, and briefing on the appeal concluded in September 2003. Oral argument was held on February 25, 2004.

The Company believes that it has meritorious defenses with respect to this case and intends to defend the case vigorously. At the present time, management is not in a position to determine whether the resolution of this case will have a material adverse effect on either the consolidated financial position of the Company or the Company's results of operations in any future reporting period.

Other Pending and Threatened Litigation

In addition, the Company is also commonly subject to various pending and threatened legal actions relating to the conduct of its normal business activities. In the opinion of management, the ultimate aggregate liability, if any, arising out of any such pending or threatened legal actions will not be material to the consolidated financial position or results of operations of the Company.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

CAPITAL ONE FINANCIAL CORPORATION Management's Discussion and Analysis of Financial Condition and Results of Operations (Dollars in thousands) (yields and rates presented on an annualized basis)

Introduction

Capital One Financial Corporation (the "Corporation") is a holding company whose subsidiaries market a variety of products and services to consumers using its Information-Based Strategy ("IBS"). The Corporation's principal subsidiaries are Capital One Bank (the "Bank"), which offers consumer credit card products, Capital One, F.S.B. (the "Savings Bank"), which offers consumer lending products (including credit cards) and deposit products, and Capital One Auto Finance, Inc. ("COAF"), which offers automobile and other motor vehicle financing products. The Corporation and its subsidiaries are hereafter collectively referred to as the "Company." As of June 30, 2004, the Company had 46.6 million accounts and \$73.4 billion in managed consumer loans outstanding and was one of the largest providers of MasterCard and Visa credit cards in the world.

The Company's profitability is affected by the net interest income and non-interest income generated on earning assets, consumer usage patterns, credit quality, levels of marketing expense and operating efficiency. The Company's revenues consist primarily of interest income on consumer loans (including past-due fees) and securities and non-interest income consisting of servicing income on securitized loans, fees (such as annual membership, cash advance, cross-sell, interchange, overlimit and other fee income, collectively "fees") and gains on the securitization of loans. Loan securitization transactions qualifying as sales under accounting principles generally accepted in the United States ("GAAP") remove the loan receivables from the consolidated balance sheet. However, the Company continues to own and service the related accounts. The Company generates earnings from its managed loan portfolio that includes both on-balance sheet and off-balance sheet loans. Interest income, fees, and recoveries in excess of the interest paid to investors and charge-offs generated from off-balance loans are recognized as servicing and securitization income.

The Company's primary expenses are the cost of funding assets, provision for loan losses, operating expenses (including salaries and associate benefits), marketing expenses and income taxes. Marketing expenses (e.g., advertising, printing, credit bureau costs and postage) to implement the Company's product strategies are incurred and expensed prior to the acquisition of new accounts while the resulting revenues are recognized over the life of the acquired accounts. Revenues recognized are a function of the response rate of the initial marketing program, usage and attrition patterns, credit quality of accounts, product pricing and effectiveness of account management programs.

Significant Accounting Policies

See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003, Part I, Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations" for a summary of the Company's significant accounting policies.

Off-Balance Sheet Arrangements

Off-Balance Sheet Securitizations

The Company actively engages in off-balance sheet securitization transactions of loans for funding purposes. The Company receives the proceeds from third party investors for securities issued from the Company's securitization vehicles which are collateralized by transferred receivables from the Company's portfolio. Securities outstanding totaling \$38.4 billion as of June 30, 2004, represent undivided interests in the pools of consumer loan receivables that are sold in underwritten offerings or in private placement transactions.

The securitization of consumer loans has been a significant source of liquidity for the Company. Maturity terms of the existing securitizations vary from 2004 to 2019 and, for revolving securitizations, have accumulation periods during which principal payments are aggregated to make payments to investors. As payments on the loans are accumulated and are no longer reinvested in new loans, the Company's funding requirements for such new loans increase accordingly. The Company believes that it has the ability to continue to utilize off-balance sheet securitization arrangements as a source of liquidity, however, a significant reduction or termination of the Company's off-balance sheet securitizations could require the Company to draw down existing liquidity and/or to obtain additional funding through the issuance of secured borrowings or unsecured debt, the raising of additional deposits or the slowing of asset growth to offset or to satisfy liquidity needs.

Recourse Exposure

The credit quality of the receivables transferred is supported by credit enhancements, which may be in various forms including interest-only strips, subordinated interests in the pool of receivables, cash collateral accounts, cash reserve accounts and accrued interest and fees on the investor's share of the pool of receivables. Some of these credit enhancements are retained by the seller and are referred to as retained residual interests. The Company's retained residual interests are generally restricted or subordinated to investors' interests and their value is subject to substantial credit, repayment and interest rate risks on transferred assets if the off-balance sheet loans are not paid when due. The investors and the trusts only have recourse to the retained residual interests, not the Company's assets. See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003, Part I, Item 8 "Financial Statements and Supplementary Data—Notes to the Consolidated Financial Statements.—Note R" for quantitative information regarding retained interests.

Collections and Amortization

Collections of interest and fees received on securitized receivables are used to pay interest to investors, servicing and other fees, and are available to absorb the investors' share of credit losses. For revolving securitizations, amounts collected in excess of that needed to pay the above amounts are remitted to the Company. For amortizing securitizations, amounts collected in excess of the amount that is used to pay the above amounts, are generally remitted to the Company, but may be paid to investors in further reduction of their outstanding principal. See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003, Part I, Item 8 "Financial Statements and Supplementary Data—Notes to the Consolidated Financial Statements—Note R" for quantitative information regarding revenues, expenses and cash flows that arise from securitization transactions.

Securitization transactions may amortize earlier than scheduled due to certain early amortization triggers, which would accelerate the need for funding. Additionally, early amortization would have a significant impact on the ability of the Bank and Savings Bank to meet regulatory capital adequacy requirements as all off-balance sheet loans experiencing such early amortization would be recorded on the balance sheet and accordingly would require incremental regulatory capital. As of June 30, 2004, no early amortization events related to its off-balance sheet securitizations have occurred.

Funding Commitments Related to Synthetic Fuel Tax Credit Transaction

In June 2004, the Corporation established and consolidated Capital One Appalachian LLC ("COAL"). COAL is a special purpose entity established to invest a 24.9% minority ownership interest in a limited



partnership. The partnership was established to operate a facility which produces a coal-based synthetic fuel that qualifies for tax credits pursuant to Section 29 of the Internal Revenue Code. COAL purchased its interest in the partnership from a third party paying \$2.1 million in cash and agreeing to pay an estimated \$115.0 million comprised of fixed note payments, variable payments and the funding of its 24.9% share of the operating losses of the partnership. Actual total payments will be based on the amount of tax credits generated by the partnership from June of 2004 through the end of 2007. In exchange, COAL will receive an estimated \$137.7 million in tax benefits resulting from a combination of deductions, allocated partnership operating losses, and tax credits. The Corporation has guaranteed COAL's commitments to both the partnership and the third party.

Reconciliation to GAAP Financial Measures

The Company's consolidated financial statements prepared in accordance with GAAP are referred to as its "reported" financial statements. Loans included in securitization transactions which qualified as sales under GAAP have been removed from the Company's "reported" balance sheet. However, servicing fees, finance charges, and other fees, net of charge-offs, and interest paid to investors of securitizations are recognized as servicing and securitizations income on the "reported" income statement.

The Company's "managed" consolidated financial statements reflect adjustments made related to effects of securitization transactions qualifying as sales under GAAP. The Company generates earnings from its "managed" loan portfolio which includes both the on-balance sheet loans and off-balance sheet loans. The Company's "managed" income statement takes the components of the servicing and securitizations income generated from the securitized portfolio and distributes the revenue and expense to appropriate income statement line items from which it originated. For this reason, the Company believes the "managed" consolidated financial statements and related managed metrics to be useful to stakeholders.

	As of and	As of and for the Three Months Ended June 30, 2004			
(Dollars in thousands)	Total Reported	Securitization Adjustments ⁽¹⁾	Total Managed ⁽²⁾		
Income Statement Measures					
Net interest income	\$ 711,017	\$ 874,453	\$ 1,585,470		
Non-interest income	1,396,064	(384,766)	1,011,298		
Total revenue	2,107,081	489,687	2,596,768		
Provision for loan losses	242,256	489,687	731,943		
Net charge-offs	309,787	489,687	799,474		
Balance Sheet Measures					
Consumer loans	\$ 34,551,343	\$ 38,816,009	\$ 73,367,352		
Total assets	50,069,661	38,247,397	88,317,058		
Average consumer loans	33,290,487	39,036,733	72,327,220		
Average earning assets	45,705,396	37,199,690	82,905,086		
Average total assets	50,020,115	38,452,658	88,472,773		
Delinquencies	1,350,969	1,405,289	2,756,258		

(1) Includes adjustments made related to the effects of securitization transactions qualifying as sales under GAAP and adjustments made to reclassify to "managed" loans outstanding the collectible portion of billed finance charge and fee income on the investors' interest in securitized loans excluded from loans outstanding on the "reported" balance sheet in accordance with Financial Accounting Standards Board Staff Position, "Accounting for Accrued Interest Receivable Related to Securitized and Sold Receivables under FASB Statement 140, Accounting for Transfers and Servicing of Financial Assets and Extinguishments of Liabilities," issued in April 2003.

⁽²⁾ The managed loan portfolio does not include auto loans which have been sold in whole loan sale transactions where the Company has retained servicing rights.

Earnings Summary

Net income was \$407.4 million, or \$1.65 per share, for the three months ended June 30, 2004 compared with net income of \$286.2 million, or \$1.23 per share, for the three month period ended June 30, 2003. This represents 42% net income growth and 34% earnings per share growth. The growth in earnings for the second quarter 2004 was primarily driven by an increase in the managed consumer loan portfolio and a reduction in the provision for loan losses, offset in part by a decrease in service charges and other customer-related fees, a decrease in other non-interest income and an increase in operating expenses.

Managed loans consist of the Company's reported loan portfolio combined with the off-balance sheet securitized loan portfolio. The Company has retained servicing rights for its securitized loans and receives servicing fees in addition to the excess spread generated from the off-balances sheet loan portfolio. Average managed loans increased \$12.4 billion, or 21%, to \$72.3 billion for the three months ended June 30, 2004 compared with \$59.9 billion for the three months ended June 30, 2003.

The managed net interest margin for the three months ended June 30, 2004, decreased to 7.65% from 8.64% for the three months ended June 30, 2003. This decrease is due to a reduction in the managed earning asset yields. Managed loan yields decreased 129 basis points to 12.80% for the second quarter 2004, from 14.09% for the same period in 2003. The decrease in managed loan yields resulted from the shift in the mix of the managed loan portfolio to higher credit quality, lower yielding loans. In addition, the Company built its average liquidity portfolio by \$3.1 billion to \$10.6 billion for the three months ended June 30, 2004, from \$7.5 billion for the same period in the prior year. This placed additional downward pressure on managed earning asset yields as the yield on the liquidity portfolio is significantly lower than the yield on consumer loans.

For the three months ended June 30, 2004, the provision for loan losses decreased \$144.8 million to \$242.3 million, from \$387.1 million for the same period in the prior year. The decrease in the provision for loan losses reflects the reduction in net charge-offs combined with a reduction in the allowance for loan losses as a result of improving delinquency rates and lower forecasted charge-offs for the reported portfolio at June 30, 2004. The improvements in the Company's credit quality metrics are a result of the change in composition of the reported loan portfolio to a higher concentration of higher credit quality loans, improved collection experience and improved economic conditions compared with the same period of the prior year.

Services charges and other customer-related fees decreased \$34.5 million, to \$368.5 million for the three months ended June 30, 2004, from \$403.0 million for the same period in the prior year. The decrease is primarily the result of the shift in the mix of the reported loan portfolio to higher credit quality, lower fee-generating loans and ongoing product modifications.

Other non-interest income decreased \$33.6 million to \$42.2 million for the three months ended June 30, 2004 when compared to the same period in the prior year. This decrease is primarily the result of \$20.8 million in losses on the sale of securities for the three months ended June 30, 2004 compared with \$0.2 million in losses in the same period of the prior year. The rebalancing of the liquidity investment portfolio during the three month period ending June 30, 2004, was in reaction to changes in interest rates.

For the three months ended June 30, 2004, operating expenses increased \$94.0 million, or 11%, compared with the same period in the prior year. Included in the operating expenses for the three months ended June 30, 2004 is \$56.0 million in employee termination and facility consolidation charges related to cost reduction initiatives addressed in the Non-Interest Expense section below. Although operating expenses increased, annualized operating expenses as a percentage of average managed loans for the three month period ended June 30, 2004 fell 49 basis points to 5.39% from 5.88% for the same period in the prior year. This reduction reflects the continued improvement of the Company's operating efficiencies.

Net income for the six months ended June 30, 2004, was \$858.2 million, or \$3.48 per share, compared to \$594.6 million, or \$2.58 per share, for the same period in 2003. This 44% increase in net income is primarily the result of an increase in managed earning assets and a reduction in the provision for loan losses, offset in part by an increase in non-interest expenses during the six months ended June 30, 2004, compared to the same period in the prior year. Each component is discussed in further detail in subsequent sections of this analysis.

Consolidated Statements of Income

Finance Charge and Fee Revenue Recognition

The Company recognizes earned finance charges and fee income on loans according to the contractual provisions of the credit arrangements. When the Company does not expect full payment of finance charges and fees, it does not accrue the estimated uncollectible portion as income (hereafter the "suppression amount"). To calculate the suppression amount, the Company first estimates the uncollectible portion of finance charge and fee receivables using a formula based on historical account migration patterns and current delinquency status. This formula is consistent with that used to estimate the allowance related to expected principal losses on reported loans. The suppression amount is calculated by adding any current period change in the estimate of the uncollectible portion of finance charge and fee receivables to the amount of finance charges and fees charged-off (net of recoveries) during the period. The Company subtracts the suppression amount from the total finance charges and fees billed during the period to arrive at total reported revenue.

The amount of finance charges and fees suppressed were \$263.5 million and \$497.3 million for the three months ended June, 2004 and 2003, respectively and \$549.0 million and \$1.0 billion for the six months ended June 30, 2004 and 2003, respectively. The reduction in the suppression amount, among other factors, was driven by the shift in mix of the Company's loan portfolio to a higher concentration of higher credit quality, lower fee generating loans, product diversification and improving economic conditions. These factors drove a reduction in total finance charges and fees billed during the period and increased the likelihood of collectibility. Together, the lower volume of total finance charges and fees billed and the higher expectations of collectibility drove the reduction in the amount of finance charges and fees suppressed. Actual payment experience could differ significantly from management's assumption, resulting in higher or lower future finance charge and fee income.

Net Interest Income

Net interest income is comprised of interest income and past-due fees earned and deemed collectible from the Company's consumer loans, securities income, less interest expense on borrowings, which includes interest-bearing deposits, borrowings from senior and subordinated notes and other borrowings.

The change in composition of the loan portfolio, described in the Earnings Summary above, also impacted net interest income and margins. Reported net interest income for the three month period ended June 30, 2004 was \$711.0 million, compared with \$682.3 million for the same period in the prior year. Reported net interest income increased slightly to \$1.4 billion for the six months ended June 30, 2004 compared with the six months ended June 30, 2003. The slight increase in net interest income is primarily the result of a decrease in the cost of funds largely offset by decreases in earning asset yields. The reported net interest margin decreased 130 and 162 basis points to 6.22% and 6.43% from 7.52% and 8.05% for the three and six months ended June 30, 2004, respectively, compared to the same periods in the prior year. The

decrease in net interest margin was primarily due to a decrease in loan yield. The reported loan yield decreased 193 and 209 basis points to 12.24% and 12.42% for the three and six months ended June 30, 2004, respectively, compared to 14.17% and 14.51% for the three and six months ended June 30, 2003. The yield on consumer loans decreased due to the change in composition of the reported loan portfolio to a higher concentration of higher credit quality, lower yielding loans compared with the same period of the prior year. In addition, the Company increased its average liquidity portfolio by \$3.1 billion, or 41%, compared with the same period in the prior year. The yield on liquidity portfolio assets is significantly lower than those on consumer loans and served to reduce the overall earning asset yields.

Table 1 provides average balance sheet data, and an analysis of net interest income, net interest spread (the difference between the yield on earning assets and the cost of interest-bearing liabilities) and net interest margin for the three and six months ended June 30, 2004 and 2003.

TABLE 1 - STATEMENTS OF AVERAGE BALANCES, INCOME AND EXPENSE, YIELDS AND RATES

			Three Months I	Ended June 30	ed June 30				
		2004			2003				
(Dollars in thousands)	Average Balance	Income/ Expense	Yield/ Rate	Average Balance	Income/ Expense	Yield/ Rate			
Assets:									
Earning assets									
Consumer loans ⁽¹⁾									
Domestic	\$29,765,214	\$ 920,758	12.37%	\$24,525,992	\$ 881,922	14.38%			
International	3,525,273	98,318	11.16%	2,575,050	78,202	12.15%			
IIIternational	5,525,275	90,510	11.10%	2,375,030	/6,202	12.1370			
Total	33,290,487	1,019,076	12.24%	27,101,042	960,124	14.17%			
Securities available for sale	9,291,237	76,081	3.28%	5,386,070	47,895	3.56%			
Other	3,231,237	7 0,001	0.2070	5,500,070	,000	0.0070			
Domestic	2,358,697	43,018	7.30%	3,079,492	54,291	7.05%			
International	764,975	13,771	7.20%	731,318	7,970	4.36%			
Total	3,123,672	56,789	7.27%	3,810,810	62,261	6.54%			
1000				5,010,010					
Total earning assets	45,705,396	\$ 1,151,946	10.08%	36,297,922	\$ 1,070,280	11.79%			
Cash and due from banks	648,479	\$ 1,101,010	1010070	242,861	¢ 1,070,200	111/0/0			
Allowance for loan losses	(1,494,236)			(1,634,498)					
Premises and equipment, net	907,957			773,126					
Other	4,252,519			3,998,825					
Total assets	\$ 50,020,115			\$39,678,236					
Liabilities and Equity:									
Interest-bearing liabilities									
Deposits									
Domestic	\$22,260,973	\$ 224,213	4.03%	\$18,029,732	\$ 202,134	4.48%			
International	1,687,181	20,765	4.92%	1,148,422	18,506	6.45%			
Total	23,948,154	244,978	4.09%	19,178,154	220,640	4.60%			
10181	25,940,134	244,970	4.09%	19,170,134	220,040	4.00%			
Senior notes	7,380,437	124,809	6.76%	5,533,693	99,120	7.16%			
Other borrowings	· ;===; :=:	,		-,,					
Domestic	8,487,287	71,137	3.35%	6,681,627	68,248	4.09%			
International		,				3.08%			
International	740	5	2.70%	1,170	9	5.00%			
Total	8,488,027	71,142	3.35%	6,682,797	68,257	4.09%			
Total interest-bearing liabilities	39,816,618	\$ 440,929	4.43%	31,394,644	\$ 388,017	4.94%			
		φ 440,525	4,4370		\$ 500,017	4.5470			
Other	3,260,219			3,135,528					
Total liabilities	43,076,837			34,530,172					
Equity	6,943,278			5,148,064					
Total liabilities and equity	\$ 50,020,115			\$39,678,236					
Total habilities and equity	\$ 50,020,115		_	\$ 39,070,230					
Net interest spread			5.65%			6.85%			
Interest income to average earning assets			10.08%			11.79%			
Interest expense to average earning assets			3.86%			4.27%			
Not interact margin			6 220/			7 500/			
Net interest margin			6.22%			7.52%			

⁽¹⁾ Interest income includes past-due fees of approximately \$189,100 and \$188,057 for the three months ended June 30, 2004 and 2003, respectively.

TABLE 1 - STATEMENTS OF AVERAGE BALANCES, INCOME AND EXPENSE, YIELDS AND RATES

		Six Months Ended June 30					
		2004		2003			
(Dollars in thousands)	Average Balance	Income/ Expense	Yield/ Rate	Average Balance	Income/ Expense	Yield/ Rate	
Assets:							
Earning assets							
Consumer loans ⁽¹⁾							
Domestic	\$29,696,924	\$ 1,868,526	12.58%	\$24,496,856	\$ 1,791,916	14.63%	
International	3,387,082	185,567	10.96%	2,711,168	181,490	13.39%	
Total	33,084,006	2,054,093	12.42%	27,208,024	1,973,406	14.51%	
Securities available for sale	8,195,094	139,797	3.41%	4,904,480	90,826	3.70%	
Other							
Domestic	2,755,160	91,134	6.62%	2,533,869	98,491	7.77%	
International	874,209	31,653	7.24%	580,778	14,123	4.86%	
Total	3,629,369	122,787	6.77%	3,114,647	112,614	7.23%	
Total earning assets	44,908,469	\$ 2,316,677	10.32%	35,227,151	\$ 2,176,846	12.36%	
Cash and due from banks	575,532	•))-		372,955	* , -,		
Allowance for loan losses	(1,544,068)			(1,676,490)			
Premises and equipment, net	912,085			777,313			
Other	4,007,547			4,301,009			
Total assets	\$48,859,565			\$39,001,938			
Liabilities and Equity: Interest-bearing liabilities							
Deposits Domestic	\$21,794,369	\$ 442,452	4.06%	\$17,461,274	\$ 396,972	4.55%	
International	1,676,064	442,432	5.02%	1,101,252	32,976	5.99%	
Total	23,470,433	484,490	4.13%	18,562,526	429,948	4.63%	
			<u> </u>			= 200/	
Senior notes	7,325,663	249,227	6.80%	5,422,310	197,699	7.29%	
Other borrowings							
Domestic	8,160,121	139,902	3.43%	6,843,779	132,104	3.86%	
International	916	19	4.15%	1,673	28	3.35%	
Total	8,161,037	139,921	3.43%	6,845,452	132,132	3.86%	
Total interest-bearing liabilities	38,957,133	\$ 873,638	4.49%	30,830,288	\$ 759,779	4.93%	
Other	3,209,165			3,185,442			
Total liabilities	42,166,298			34,015,730			
Equity	6,693,267			4,986,208			
Total liabilities and equity	\$48,859,565			\$39,001,938			
Net interest spread			5.83%			7.43%	
Interest income to average earning assets			10.32%			12.36%	
Interest income to average earning assets			3.89%			4.31%	
Net interest margin			6.43%			8.05%	
זיכר חונדכזנ וומוצוו			0.4370			0.05%	

⁽¹⁾ Interest income includes past-due fees of approximately \$396,345 and \$404,682 for the six months ended June 30, 2004 and 2003, respectively.

Interest Variance Analysis

Net interest income is affected by changes in the average interest rate generated on earning assets and the average interest rate paid on interest-bearing liabilities. In addition, net interest income is affected by changes in the volume of earning assets and interest-bearing liabilities. Table 2 sets forth the dollar amount of the increases and decreases in interest income and interest expense resulting from changes in the volume of earning assets and interest-bearing liabilities and from changes in yields and rates.

TABLE 2 - INTEREST VARIANCE ANALYSIS

		Three Months Ende June 30, 2004 vs. 200			3	
		Change	due to ⁽¹⁾		Change	due to ⁽¹⁾
(Dollars in thousands)	Increase (Decrease)	Volume	Yield/Rate	Increase (Decrease)	Volume	Yield/Rate
Interest Income:						
Consumer loans						
Domestic	\$ 38,836	\$619,487	\$(580,651)	\$ 76,610	\$650,439	\$(573,829)
International	20,116	58,298	(38,182)	4,077	78,627	(74,550)
Total	58,952	691,498	(632,546)	80,687	730,327	(649,640)
Securities available for sale	28,186	52,757	(24,571)	48,971	69,457	(20,486)
Other						
Domestic	(11,273)	(22,883)	11,610	(7,357)	18,974	(26,331)
International	5,801	383	5,418	17,530	8,909	8,621
Total	(5,472)	(37,921)	32,449	10,173	28,174	(18,001)
Total interest income	81,666	849,269	(767,603)	139,831	985,720	(845,889)
Interest Expense:						
Deposits						
Domestic	22,079	130,070	(107,991)	45,480	150,513	(105,033)
International	2,259	24,763	(22,504)	9,062	23,249	(14,187)
Total	24,338	152,174	(127,836)	54,542	171,190	(116,648)
Senior notes	25,689	59,971	(34,282)	51,528	87,715	(36,187)
Other borrowings						
Domestic	2,889	60,610	(57,721)	7,798	42,294	(34,496)
International	(4)	(3)	(1)	(9)	(23)	14
Total	2,885	60,601	(57,716)	7,789	42,273	(34,484)
Total interest expense	52,912	270,766	(217,854)	113,859	288,802	(174,943)
Net interest income ⁽¹⁾	\$ 28,754	\$ 582,273	\$(553,519)	\$ 25,972	\$673,420	\$(647,448)

(1) The change in interest due to both volume and rates has been allocated in proportion to the relationship of the absolute dollar amounts of the change in each. The changes in income and expense are calculated independently for each line in the table. The totals for the volume and yield/rate columns are not the sum of the individual lines.

Servicing and Securitization Income

Servicing and securitization income represents servicing fees, excess spread and other fees relating to consumer loan receivables sold through securitization and other sale transactions, as well as gains and losses resulting from securitization transactions and fair value adjustments of the retained interests. Servicing and securitizations income increased \$125.3 million, or 17%, to \$868.0 million for the three months ended June 30, 2004, from \$742.7 million for the same period in 2003. Servicing and

securitizations income increased \$313.3 million, or 21%, to \$1.8 billion for the six months ended June 30, 2004, from \$1.5 billion for the same period in 2003. This increase was primarily the result of a 19% increase in the average off-balance sheet loan portfolio for both the three and six months ended June 30, 2004, compared with the same period in the prior year.

Service Charges and Other Customer-Related Fees

Service charges and other customer-related fees decreased by \$34.5 million, or 9%, to \$368.5 million for the three months ended June 30, 2004, compared with \$403.0 million in the same period during 2003, and decreased by \$121.2 million or 14%, to \$723.0 million for the six months ended June 30, 2004, from \$844.2 million during the same period in 2003. The decrease is primarily the result of lower overlimit and annual membership fees generated on the reported loan portfolio resulting from the shift in the mix of the reported portfolio towards higher credit quality, less fee intensive products and ongoing product modifications.

Interchange Income

Interchange increased \$28.2 million and \$48.4 million, or 32% and 28%, to \$117.3 million and \$222.9 million for the three and six months ended June 30, 2004, respectively, compared to \$89.1 million and \$174.5 million for the same periods in the prior year. This increase is primarily attributable to growth in the reported loan portfolio and increased rates from Mastercard and Visa. Total interchange income is net of \$26.9 and \$54.8 million of costs related to the Company's rewards programs for the three and six months ended June 30, 2004, compared with \$25.9 million and \$47.7 for the three and six months ended June 30, 2003, respectively.

Other Non-Interest Income

Other non-interest income includes, among other items, gains and losses on sales of securities, gains and losses associated with hedging transactions, service provider revenue generated by the Company's patient finance business, gains on the sale of auto loans and income earned related to purchased charged-off loan portfolios.

Other non-interest income decreased \$33.6 million and \$16.6 million, or 44% and 13%, to \$42.2 million and \$107.6 million for the three and six months ended June 30, 2004, respectively, compared with the same period in 2003. The decrease in other non-interest income for the three months ended June 30, 2004, is primarily the result of a \$20.6 million increase in realized losses on the sale of securities and a \$24.0 million decrease in auto loan sale gains, partially offset by a \$14.5 million increase in income earned from purchased charged-off loan portfolios. The decrease in other non-interest income for the six month period ended June 30, 2004, is primarily attributable to a \$20.9 million loss from the sale of securities in 2004 versus a \$9.6 million gain in the same period of 2003 and an \$14.9 million decrease in auto loan sale gains, partially offset by a \$29.4 million increase in income earned from purchased charge-off loan portfolios when compared with the same periods in the prior year.

Non-Interest Expense

In June 2004, the Company recognized charges related to corporate-wide cost reduction initiatives. The cost reduction initiatives will serve to reduce future operating expenses. The Company recognized \$56.0 million of pre-tax charges (\$21.3 million related to employee severance and \$34.7 million related to facility consolidation) during the three month period ended June 30, 2004. Additional estimated charges of \$60 million to \$100 million are expected to be incurred in the second half of 2004.

Non-interest expense, which consists of marketing and operating expenses, increased \$77.3 million and \$128.2 million, or 7% and 6%, to \$1.2 billion and \$2.5 billion for the three and six months ended June 30, 2004, respectively. Marketing expense decreased \$16.7 million and \$3.3 million, or 6% and 1%, to \$253.8 million and \$509.0 million for the three and six months ended June 30, 2004, compared with the same periods in the prior year. Operating expenses increased \$94.0 million and \$131.5 million for the three and six months ended June 30, 2004, respectively, compared with the same periods in the prior year. The increases in operating expenses were primarily the result of increases in salaries and associate benefits, occupancy and credit and recovery efforts. Salaries and associate benefits increased \$45.5 million and \$71.4 million for the three and six months ended June 30, 2004, respectively, and occupancy increased \$27.7 million and \$22.9 million for the same periods primarily as the result of charges taken during the second quarter 2004 for employee termination benefits and facility consolidation related to the corporate-wide cost reduction initiatives. Credit and recovery costs increased \$19.1 million and \$54.4 million for the three and six months ended June 30, 2004, respectivels. Although operating expenses increased, operating expenses as a percentage of average managed loans for the three and six months ended June 30, 2004 fell 49 and 66 basis points to 5.39% and 5.42% compared to the three and six month periods ended June 30, 2003, respectively. This reduction reflects the continued improvement of the Company's operating efficiencies.

Income Taxes

The Company's income tax rate was 36% and 37% for the three months ended June 30, 2004 and 2003, respectively. The decrease was primarily due to increased profitability in the Company's International businesses in lower taxed territories and ongoing tax planning activities. The effective rate includes both state and federal income tax components.

Managed Consumer Loan Portfolio

The Company's managed consumer loan portfolio is comprised of on-balance sheet and off-balance sheet loans.

The Company analyzes its financial performance on a managed consumer loan portfolio basis. The managed consumer loan portfolio includes securitized loans for which the Company has retained significant risks and potential returns. Table 3 summarizes the Company's managed consumer loan portfolio.

TABLE 3 - MANAGED CONSUMER LOAN PORTFOLIO

	Three Months	Ended June 30
(Dollars in thousands)	2004	2003
Period-End Balances:		
Reported consumer loans:		
Domestic	\$ 31,080,395	\$ 24,156,037
International	3,470,948	2,692,541
Total	34,551,343	26,848,578
Securitization adjustments ⁽¹⁾ :		
Domestic	33,671,814	30,518,384
International	5,144,195	3,368,847
Total	38,816,009	33,887,231
Managed consumer loan portfolio:		
Domestic	64,752,209	54,674,421
International	8,615,143	6,061,388
Total	\$ 73,367,352	\$ 60,735,809
Average Balances:		
Reported consumer loans:		
Domestic	\$ 29,765,214	\$ 24,525,992
International	3,525,273	2,575,050
Total	33,290,487	27,101,042
Securitization adjustments ⁽¹⁾ :		
Domestic	34,221,627	29,760,892
International	4,815,106	3,053,863
Total	39,036,733	32,814,755
Managed consumer loan portfolio:		
Domestic	63,986,841	54,286,884
International	8,340,379	5,628,913
Total	\$ 72,327,220	\$ 59,915,797

25

-

		hs Ended e 30
(Dollars in thousands)	2004	2003
Average Balances:		
Reported consumer loans:		
Domestic	\$ 29,696,924	\$ 24,496,856
International	3,387,082	2,711,168
Total	33,084,006	27,208,024
Securitization adjustments ⁽¹⁾ :		
Domestic	33,892,007	29,594,303
International	4,761,741	2,782,261
Total	38,653,748	32,376,564
Managed consumer loan portfolio:		
Domestic	63,588,931	54,091,159
International	8,148,823	5,493,429
Total	\$ 71,737,754	\$ 59,584,588
		·

(1) Includes adjustments made related to the effects of securitization transactions qualifying as sales under GAAP and adjustments made to reclassify to "managed" loans outstanding the collectible portion of billed finance charge and fee income on the investors' interest in securitized loans excluded from loans outstanding on the "reported" balance sheet in accordance with Financial Accounting Standards Board Staff Position, "Accrued Interest Receivable," issued in April 2003.

Table 4 indicates the impact of the consumer loan securitizations on average earning assets, net interest margin and loan yield for the periods presented. The Company intends to continue to securitize consumer loans.

TABLE 4 COMPARISON OF MANAGED AND REPORTED OPERATING DATA AND RATIOS

Three Months Ended June 30			Six Months Ended June 30		
(Dollars in thousands)	2004	2003	2004	2003	
Reported:					
Average earning assets	\$45,705,396	\$36,297,922	\$44,908,469	\$35,227,151	
Net interest margin ⁽¹⁾	6.22%	7.52%	6.43%	8.05%	
Loan yield	12.24%	14.17%	12.42%	14.51%	
Managed:					
Average earning assets	\$82,905,086	\$67,450,839	\$81,700,160	\$66,034,100	
Net interest margin ⁽¹⁾	7.65%	8.64%	7.99%	8.98%	
Loan yield	12.80%	14.09%	13.16%	14.30%	

 $^{\left(1\right) }$ Net interest margin is equal to net interest income divided by average earning assets.

Revenue Margin

The Company's products are designed with the objective of maintaining strong risk-adjusted returns and providing diversification across the credit spectrum and consumer lending products. Management believes that a comparable measure for external analysis is the Company's managed revenue margin (based on average managed earning assets.)

The Company has aggressively marketed lending products to high credit quality consumers to take advantage of favorable long-term risk-adjusted returns of this consumer type. In addition, the Company continues to diversify its products beyond U.S. consumer credit cards. While these products typically consist of lower yielding loans compared with those previously made, they provide favorable impacts on managed charge-offs, operating expenses and marketing as a percentage of average managed earning assets.

Table 5 provides income statement data and ratios for the Company's reported and managed consumer loan portfolio. The causes of increases and decreases in the various components of revenue are discussed in sections previous to this analysis.

TABLE 5 – REVENUE MARGIN

		Six Months Ended June 30	
2004	2003	2004	2003
\$ 711,017	\$ 682,263	\$1,443,039	\$1,417,067
1,396,064	1,310,622	2,839,198	2,615,225
\$2,107,081	\$1,992,885	\$4,282,237	\$4,032,292
6.22%	7.52%	6.43%	8.05%
12.22%	14.44%	12.64%	14.85%
18.44%	21.96%	19.07%	22.90%
\$1.585.470	\$1.457.517	\$3,262,548	\$2,965,468
1,011,298	1,046,012	2,025,834	2,073,945
\$2,596,768	\$2,503,529	\$5,288,382	\$5,039,413
7 659/	0.640/	7.000/	8.98%
			6.28%
4.00%	0.21%	4.90%	0.28%
12.53%	14.85%	12.95%	15.26%
	2004 2004 \$ 711,017 1,396,064 \$ 2,107,081 6.22% 12.22% 18.44% \$ 1,585,470 1,011,298 \$ 2,596,768 7.65% 4.88%	\$ 711,017 \$ 682,263 1,396,064 1,310,622 \$ 2,107,081 \$ 1,992,885 6.22% 7.52% 12.22% 14.44% 18.44% 21.96% \$ 1,585,470 \$ 1,457,517 1,011,298 1,046,012 \$ 2,596,768 \$ 2,503,529 7.65% 8.64% 4.88% 6.21%	June 30 June 2004 2003 2004 \$ 711,017 \$ 682,263 \$1,443,039 1,396,064 1,310,622 2,839,198 \$ 2,107,081 \$1,992,885 \$4,282,237 6.22% 7.52% 6.43% 12.22% 14.44% 12.64% 18.44% 21.96% 19.07% \$ 1,585,470 \$1,457,517 \$3,262,548 1,011,298 1,046,012 2,025,834 \$ 2,596,768 \$2,503,529 \$5,288,382 7.65% 8.64% 7.99% 4.88% 6.21% 4.96%

⁽¹⁾ As a percentage of average earning assets.

Asset Quality

The asset quality of a portfolio is generally a function of the initial underwriting criteria used, levels of competition, account management activities and demographic concentration, as well as general economic conditions. The Company's credit risk profile is managed to maintain strong risk adjusted returns and diversification across the full credit spectrum and in each of its consumer lending products. Certain customized consumer lending products have, in some cases, higher delinquency and charge-off rates. The costs associated with higher delinquency and charge-off rates are considered in the pricing of individual products.

Delinquencies

The Company's loan portfolio is comprised of predominantly small balance loans spread across the full credit spectrum. The Company believes delinquencies to be a primary indicator of loan portfolio credit quality at a point in time. Table 6 shows the Company's consumer loan delinquency trends for the periods presented on a reported and managed basis. The entire balance of an account is contractually delinquent if the minimum payment is not received by the payment due date. Delinquencies not only have the potential to impact earnings if the account charges off, but they also result in additional costs in terms of the personnel and other resources dedicated to resolving the delinquencies.

TABLE 6 - DELINQUENCIES

		June 30			
	2004		2003		
(Dollars in thousands)	Loans	% of Total Loans	Loans	% of Total Loans	
Reported:					
Loans outstanding	\$ 34,551,343	100.00%	\$ 26,848,578	100.00%	
Loans delinquent:					
30-59 days	684,513	1.98%	712,616	2.65%	
60-89 days	295,066	0.85%	344,582	1.29%	
90-119 days	168,609	0.49%	198,384	0.74%	
120-149 days	116,887	0.34%	141,032	0.53%	
150 or more days	85,894	0.25%	110,198	0.40%	
Total	\$ 1,350,969	3.91%	\$ 1,506,812	5.61%	
Loans delinquent by geographic area:					
Domestic	1,273,255	4.10%	1,406,694	5.82%	
International	77,714	2.24%	100,118	3.72%	
Managed:					
Loans outstanding	\$ 73,367,352	100.00%	\$ 60,735,809	100.00%	
Loans delinquent:					
30-59 days	1,206,122	1.64%	1,270,107	2.10%	
60-89 days	609,552	0.83%	693,164	1.14%	
90-119 days	405,484	0.55%	449,523	0.74%	
120-149 days	296,918	0.41%	326,795	0.54%	
150 or more days	238,182	0.33%	264,537	0.43%	
Total	\$ 2,756,258	3.76%	\$ 3,004,126	4.95%	

Consumer loan delinquency rate decreases principally reflect a continued shift in the mix of the loan portfolio towards higher credit quality assets, improvements in collection experience and improved economic conditions.

Net Charge-Offs

Net charge-offs include the principal amount of losses (excluding accrued and unpaid finance charges, fees and fraud losses) less current period principal recoveries. The Company charges off credit card loans at 180 days past the due date, and generally charges off other consumer loans at 120 days past the due date or upon repossession of collateral. Costs to recover previously charged-off accounts are recorded as collection expenses in non-interest expense.

For the three and six months ended June 30, 2004, the reported net charge-off rate decreased 271 and 265 basis points to 3.72% and 3.94%, compared with 6.43% and 6.59% for the three and six months ended June 30, 2003, respectively. For the three and six months ended June 30, 2004, the managed net charge-off rate decreased 190 and 177 basis points to 4.42% and 4.62%, compared with 6.32% and 6.39% for the same periods in the prior year, respectively. The decrease in both the reported and managed net charge-off rates principally relates to the shift in the mix of the loan portfolio towards higher credit quality loans, loan diversification beyond U.S. Card, improvements in collections and improved economic conditions. Table 7 shows the Company's net charge-offs for the three and six month periods presented on a reported and managed basis.

TABLE 7 – NET CHARGE-OFFS

	Three Mon June		Six Months Ended June 30	
(Dollars in thousands)	2004	2003	2004	2003
Reported:				
Average loans outstanding	\$33,290,487	\$27,101,042	\$33,084,006	\$27,208,024
Net charge-offs	309,787	435,634	652,178	897,091
Net charge-offs as a percentage of average loans outstanding	3.72%	6.43%	3.94%	6.59%
Managed:				
Average loans outstanding	\$72,327,220	\$59,915,797	\$71,737,754	\$59,584,588
Net charge-offs	799,474	946,278	1,658,323	1,904,212
Net charge-offs as a percentage of average loans outstanding	4.42%	6.32%	4.62%	6.39%

Allowance For Loan Losses

The allowance for loan losses is maintained at an amount estimated to be sufficient to absorb probable losses, net of principal recoveries (including recovery of collateral), inherent in the existing reported loan portfolio. The provision for loan losses is the periodic cost of maintaining an adequate allowance. Management believes that, for all relevant periods, the allowance for loan losses was adequate to cover anticipated losses in the total reported consumer loan portfolio under then current conditions, met applicable legal and regulatory guidance and was consistent with GAAP. There can be no assurance as to future credit losses that may be incurred in connection with the Company's consumer loan portfolio, nor can there be any assurance that the loan loss allowance that has been established by the Company will be sufficient to absorb such future credit losses. The allowance is a general allowance applicable to the reported consumer loan portfolio. The amount of allowance necessary is determined primarily based on a migration analysis of delinquent and current accounts and forward loss curves. In evaluating the sufficiency of the allowance for loan losses, management also takes into consideration the following factors: recent trends in delinquencies and charge-offs including bankrupt, deceased and recovered amounts; forecasting uncertainties and size of credit risks; the degree of risk inherent in the composition of the loan portfolio; economic conditions; legal and regulatory guidance; credit evaluations and underwriting policies; seasonality; and the value of collateral supporting the loans.

Table 8 sets forth the activity in the allowance for loan losses for the periods indicated. See "Asset Quality," "Delinquencies" and "Net Charge-Offs" for a more complete analysis of asset quality.

		Three Months Ended June 30		Six Months Ended June 30		
(Dollars in thousands)	2004	2003	2004	2003		
Balance at beginning of period	\$1,495,000	\$1,635,000	\$1,595,000	\$ 1,720,000		
Provision for loan losses:						
Domestic	207,540	347,409	432,820	695,906		
International	34,716	39,688	53,104	67,042		
Total provision for loan losses	242,256	387,097	485,924	762,948		
Other	(2,469)	3,537	(3,746)	4,143		
Charge-offs:						
Domestic	(389,188)	(482,980)	(815,641)	(996,801)		
International	(36,155)	(37,096)	(67,153)	(78,826)		
Total charge-offs	(425,343)	(520,076)	(882,794)	(1,075,627)		
Recoveries:						
Domestic	103,493	74,796	208,549	160,739		
International	12,063	9,646	22,067	17,797		
Total recoveries	115,556	84,442	230,616	178,536		
Net charge-offs	(309,787)	(435,634)	(652,178)	(897,091)		
Balance at end of period	\$1,425,000	\$1,590,000	\$1,425,000	\$ 1,590,000		
Allowance for loan losses to loans at period-end	4.12%	5.92%	4.12%	5.92%		
Allowance for loan losses by geographic distribution:						
Domestic	\$1,301,264	\$1,493,406	\$1,301,264	\$ 1,493,406		
International	123,736	96,594	123,736	96,594		

For the three and six months ended June 30, 2004, the provision for loan losses decreased to \$242.3 million and \$485.9 million, respectively, from \$387.1 million and \$762.9 million for the three and six months ended June 30, 2003, respectively. While the Company's reported loan portfolio increased to \$34.6 billion at June 30, 2004 from \$26.8 billion at June 30, 2003, the impact of the loan growth to the allowance was more than offset by the loan growth being concentrated in higher credit quality loans, an improvement in collection experience, and improved economic conditions. Evidence of the improvement in these factors can be seen in the 30-plus day reported delinquency rate which was 3.91% at June 30, 2004, down 170 basis points from 5.61% at June 30, 2003.

Reportable Segments

The Company manages its business by three distinct operating segments: U.S. Card, Auto Finance and Global Financial Services. The U.S. Card, Auto Finance and Global Financial Services segments are considered reportable segments based on quantitative thresholds applied to the managed loan portfolio for reportable segments provided by SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*. Management decision making is performed on a managed portfolio basis, and such information about reportable segments is provided on a managed basis.

The Company maintains its books and records on a legal entity basis for the preparation of financial statements in conformity with GAAP. The following table presents information prepared from the Company's internal management information system, which is maintained on a line of business level through allocations from the consolidated financial results.

	U.S.	Card	Auto F	inance	Global Finar	icial Services	
	Months	or the Three s Ended ne 30	Months	or the Three 5 Ended e 30	Months	or the Three 5 Ended e 30	
(Dollars in thousands)	2004	2003	2004	2003	2004	2003	
Loans receivable	\$45,247,444	\$39,318,185	\$9,383,432	\$7,379,815	\$18,722,812	\$14,045,765	
Net income	384,089	274,159	52,728	44,002	46,091	25,521	
Net charge-off rate	5.19%	7.63%	2.53%	4.22%	3.43%	3.95%	
30+ Delinquency rate	3.95%	5.42%	5.59%	6.97%	2.50%	2.81%	
	U.S.	Card	Auto F	inance	Global Finar	ncial Services	
	As of and for the Six Months Ended June 30		As of and for the Six Months Ended June 30		As of and for the Six Months Ended June 30		
(Dollars in thousands)	2004	2003	2004	2003	2004	2003	
Net income	\$ 770,924	\$ 582,282	\$ 83,392	\$ 37,541	\$ 97,026	\$ 40,493	

TABLE 9 - REPORTABLE SEGMENTS - MANAGED BASIS

U.S. Card Segment

The U.S. Card segment consists of domestic credit card lending activities. Total U.S. Card segment loans increased 15% to \$45.2 billion at June 30, 2004, compared with \$39.3 billion at June 30, 2003. The loan growth in this segment reflects, among other things, the Company's continued success in applying IBS. The contribution to net income from the U.S. Card segment increased \$109.9 million, or 40%, to \$384.1 million for the three months ended June 30, 2004 and \$188.6 million, or 32%, to \$770.9 million for the six months ended June 30, 2004 compared with the same periods in the prior year. The increase in net income is attributable to the improving credit quality of the portfolio, partially offset by \$31.9 million in after tax charges related to cost reduction initiatives allocated to the U.S. Card segment.

Net charge-offs of the U.S. Card segment loans decreased \$154.2 million, or 21%, while average U.S. Card segment loans for the three months ended June 30, 2004, grew \$6.4 billion, or 16%, compared with the same period in the prior year. For the three months ended June 30, 2004, the U.S. Card segment's net charge-off rate was 5.19%, compared with 7.63% for the same period in 2003. This decrease was due to the addition of higher credit quality loans to the loan portfolio and improved economic conditions.

The 30-plus day delinquency rate for the U.S. Card segment was 3.95% as of June 30, 2004, down 147 basis points from 5.42% as of June 30, 2003. The decrease in delinquencies is due to the addition of higher credit quality loans to the portfolio, improvements in collection experience, and improved economic conditions.

Auto Finance Segment

The Auto Finance segment primarily consists of automobile financing activities. Total Auto Finance segment loans outstanding increased 27% to \$9.4 billion at June 30, 2004, compared with \$7.4 billion at June 30, 2003. The increase in auto loans outstanding was the result of additional prime loans originated through the internet. For the three months ended June 30, 2004, the net income contribution from the Auto Finance segment increased \$8.7 million, or 20% for the three months ended June 30, 2004 and \$45.9 million, or 122% for the six months ended June 30, 2004 when compared to the same period in the prior year. This increase in the net income contribution was the result of the growth in the loan portfolio and improving credit loss experience.

During the three months ended June 30, 2004 and 2003, the Company sold \$322.7 million and \$1.3 billion of auto loans, respectively. These transactions resulted in allocated pre-tax income for the Auto Finance segment of \$12.5 million and \$26.0 million for the three months ended June 30, 2004 and 2003, respectively. During the six months ended June 30, 2004 and 2003, the Company sold \$582.4 million and \$1.4 billion, respectively, of auto loans. These transactions resulted in allocated pre-tax income for the Auto Finance segment of \$25.8 million and \$30.2 million, respectively. In July of 2004, the Company provided notice to terminate its forward flow auto receivables agreement; however, the Company plans to continue to sell auto receivables through other channels.

Net charge-offs of Auto Finance segment loans decreased \$24.9 million, or 30%, while average Auto Finance segment loans for the three months ended June 30, 2004 grew \$1.3 billion, or 16%, compared with the same period in the prior year. For the three months ended June 30, 2004, the Auto Finance segment's net charge-off rate was 2.53% compared with 4.22% for the prior year. The decrease was primarily driven by the shift to higher credit quality loans as well as improvements in used car values and operating improvements.

The 30-plus day delinquency rate for the Auto Finance segment was 5.59% as of June 30, 2004, down 138 basis points from 6.97% as of June 30, 2003. The decrease in delinquencies was the result of a higher mix of prime loans and improved delinquency performance on non-prime loans.

Global Financial Services Segment

The Global Financial Services segment consists of international lending activities (including credit card lending), installment lending, small business lending, patient financing and other investment businesses. Total Global Financial Services segment loans increased 33% to \$18.7 billion at June 30, 2004, compared with \$14.0 billion at June 30, 2003. The increase in total loans reflects the Company's successful efforts to diversify its loan portfolio and the impact of changes in foreign currency rates. Net income contribution from the Global Financial Services segment for the three months ended June 30, 2004, increased \$20.6 million, to \$46.1 million, and increased \$56.5 million to \$97.0 million for the six months ended June 30, 2004, compared with the same period in the prior year. The three and six month periods ending June 30, 2004, included \$3.3 million in after tax charges related to cost reduction initiatives. The improvement in the Global Financial Services segment's financial performance was due to the maturation of many of the Company's diversification businesses in the U.S., U.K. and Canada.

Net charge-offs of Global Financial Services segment loans increased \$23.1 million, or 17%, while average Global Financial Services segment loans for the three months ended June 30, 2004 grew \$4.7 billion, or 35%, compared with the same period in the prior year. For the three months ended June 30, 2004 the Global Financial Services segment's net charge-off rate was 3.43% compared with 3.95% for the same period in the prior year. The decrease was driven primarily by the continued shift to higher credit quality loans and an overall improvement in credit quality.

The 30-plus day delinquency rate for the Global Financial Services segment was 2.50% as of June 30, 2004, down 31 basis points from 2.81% as of June 30, 2003. The Global Financial Services segment's delinquencies decreased primarily as a result of the addition of higher credit quality loans to the portfolio and improving economic conditions.

Funding

The Company has established access to a variety of funding alternatives in addition to securitization of its consumer loans. Table 10 illustrates the Company's unsecured funding sources and its collateralized revolving credit facility.

TABLE 10 - FUNDING AVAILABILITY AS OF JUNE 30, 2004

(Dollars or dollar equivalents in millions)	Effective/ Issue Date	Availa	ability ⁽¹⁾⁽⁵⁾	Ou	tstanding	Final Maturity ⁽⁴⁾
Senior and Subordinated Global Bank Note Program ⁽²⁾	1/03	\$	1,800	\$	5,736	
Senior Domestic Bank Note Program ⁽³⁾	4/97		—	\$	232	—
Credit Facility	6/04	\$	750			6/07
Collateralized Revolving Credit Facility	_	\$	3,440	\$	910	—
Corporation shelf registration	7/02	\$	1,948		N/A	

(1) All funding sources are non-revolving except for the Credit Facility and the Collateralized Revolving Credit Facility. Funding availability under the credit facilities is subject to compliance with certain representations, warranties and covenants. Funding availability under all other sources is subject to market conditions.

(2) The notes issued under the Global Senior and Subordinated Bank Note Program may have original terms of thirty days to thirty years from their date of issuance. This program was updated in April 2004.

(3) The notes issued under the Senior Domestic Bank Note Program have original terms of one to ten years. The Senior Domestic Bank Note Program is no longer available for issuances.

(4) Maturity date refers to the date the facility terminates, where applicable.

(5) Availability does not include unused conduit capacity related to securitization structures of \$9.2 billion at June 30, 2004.

The Senior and Subordinated Global Bank Note Program gives the Bank the ability to issue securities to both U.S. and non-U.S. lenders and to raise funds in U.S. and foreign currencies. The Senior and Subordinated Global Bank Note Program had \$5.7 billion outstanding at June 30, 2004. Under the Senior and Subordinated Global Bank Note Program, the Bank issued \$500.0 million of five-year 5.00% fixed rate bank notes in June 2004. Prior to the establishment of the Senior and Subordinated Global Bank Note Program, the Bank issued senior unsecured debt through an \$8.0 billion Senior Domestic Bank Note Program, of which \$231.5 million was outstanding at June 30, 2004. The Bank did not renew the Senior Domestic Bank Note Program for future issuances following the establishment of the Senior and Subordinated Global Bank Note Program.

In June 2004, the Company terminated its Domestic Revolving and Multicurrency Credit Facilities and replaced them with a new revolving credit facility ("Credit Facility") providing for an aggregate of \$750.0 million in unsecured borrowings from various lending institutions to be used for general corporate purposes. The Credit Facility is available to the Corporation, the Bank, the Savings Bank, and Capital One Bank (Europe), plc. The Corporation's availability has been increased to \$500.0 million under the Credit Facility. All borrowings under the Credit Facility are based upon varying terms of London Interbank Offering Rate ("LIBOR").

In April 2002, COAF entered into a revolving warehouse credit facility collateralized by a security interest in certain auto loan assets (the "Collateralized Revolving Credit Facility"). As of June 30, 2004, the Collateralized Revolving Credit Facility had the capacity to issue up to \$4.4 billion in secured notes. The Collateralized Revolving Credit Facility has multiple participants each with a separate renewal date. The facility does not have a final maturity date. Instead, each participant may elect to renew the commitment for another set period of time. Interest on the facility is based on commercial paper rates.

As of June 30, 2004, the Corporation had one effective shelf registration statement under which the Corporation from time to time may offer and sell senior or subordinated debt securities, preferred stock, common stock, common equity units and stock purchase contracts.

The Company continues to expand its retail deposit gathering efforts through both direct and broker marketing channels. The Company uses its IBS capabilities to test and market a variety of retail deposit origination strategies, including via the Internet, as well as to develop customized account management programs. As of June 30, 2004, the Company had \$24.2 billion in interest-bearing deposits of which \$11.8 billion represented large denomination certificates of \$100 thousand or more, with original maturities up to ten years.

Table 11 shows the maturities of domestic time certificates of deposit in denominations of \$100 thousand or greater (large denomination CDs) as of June 30, 2004.

TABLE 11 - MATURITIES OF LARGE DENOMINATION CERTIFICATES-\$100,000 OR MORE

	June 30, 2	June 30, 2004	
(Dollars in thousands)	Balance	Percent	
Three months or less	\$ 1,247,387	10.54%	
Over 3 through 6 months	1,090,327	9.22	
Over 6 through 12 months	1,851,748	15.65	
Over 12 months through 10 years	7,641,749	64.59	
Total	\$ 11,831,211	100.00%	

Supervision and Regulation

On July 6, 2004, the Board of Governors of the Federal Reserve System approved the Corporation's application pursuant to section 3(a)(1) of the Bank Holding Company Act of 1956, as amended (the "BHC Act") (12 U.S.C. § 1842(a)(1)) to become a bank holding company ("BHC") as a result of the Bank's proposal to amend its Virginia charter to remove existing restrictions on its activities and thereby permit the Bank to engage in the full range of lending, deposit-taking and other activities permissible under Virginia and federal banking laws and regulations. The Corporation also filed a notice with the Federal Reserve Bank of Richmond pursuant to 12 U.S.C. § 1843(c)(8) to retain its nonbanking subsidiaries, including the Savings Bank and COAF, upon its conversion to a BHC. The Corporation seeks to effect this change to create a more efficient corporate structure and rationalize its funding base. The Corporation will register as a BHC with the Federal Reserve Bank of Richmond and become subject to the requirements of the BHC Act, including limiting its nonbanking activities to those that are permissible for a BHC. Such activities include those that are so closely related to banking as to be incident thereto such as consumer lending and other activities that have been approved by the Federal Reserve Bank of Richmond by regulation or order. Certain servicing activities are also permissible for a BHC if conducted for or on behalf of the BHC or any of its affiliates. Impermissible activities for BHCs include activities that are related to commerce such as retail sales of nonfinancial products. The Corporation does not engage in any significant activities impermissible for a BHC and therefore, does not anticipate a significant change in its activities as a result of this proposal.

Basel Committee

On May 11, 2004, the Basel Committee on Banking Supervision (the "Committee") announced that it has achieved consensus on the new Basel Capital Accord ("Basel II"), which proposes establishment of a new framework of capital adequacy for banking organizations; the Committee published the text of the framework on June 26, 2004. Despite the release of the Basel II framework, it is not clear at this time whether and in what manner the new accord will be adopted by bank regulators with respect to banking organizations that they supervise and regulate. Although the Committee's stated intent is that Basel II will not change the overall amount of capital in the global banking system, adoption of the proposed new accord could require individual banking organizations, including the Company, to increase the minimum level of capital held. The Company will continue to closely monitor regulatory action on this matter and assess the potential impact on the Company.

Enterprise Risk Management

Risk is an inherent part of the Company's business and activities. The Company has an ongoing Enterprise Risk Management ("ERM") program designed to ensure appropriate and comprehensive oversight and management of risk. The ERM program operates at all levels in the Company: first, at the most senior levels with the Board of Directors and senior management committees that oversee risk and risk management practices; second, in the centralized departments headed by the Chief Enterprise Risk Officer and the Chief Credit Officer that establish risk management methodologies, processes and standards; and third, in the individual business areas throughout the Company which own the management of risk and perform ongoing identification, assessment and response to risks. The Company's Corporate Audit Services department also assesses risk and the related quality of internal controls and quality of risk management through its audit activities. To facilitate the effective management of risk, the Company utilizes a risk and control framework that includes eight categories of risk: credit, liquidity, market, operational, legal, strategic, reputation and compliance. For additional information on the Company's ERM program, see the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003, Part I, Item 1, "Enterprise Risk Management."

Capital Adequacy

The Bank and the Savings Bank are subject to capital adequacy guidelines adopted by the Federal Reserve Board (the "Federal Reserve") and the Office of Thrift Supervision (the "OTS") (collectively, the "regulators"), respectively. The capital adequacy guidelines and the regulatory framework for prompt corrective action require the Bank and the Savings Bank to maintain specific capital levels based upon quantitative measures of their assets, liabilities and off-balance sheet items.

The most recent notifications received from the regulators categorized the Bank and the Savings Bank as "well-capitalized." As of June 30, 2004, there were no conditions or events since these notifications that management believes would have changed either the Bank or the Savings Bank's capital category.

	Regulatory Filing Basis Ratios	Applying Subprime Guidance Ratios	Minimum for Capital Adequacy Purposes	To Be "Well-Capitalized" Under Prompt Corrective Action Provisions
June 30, 2004				
Capital One Bank				
Tier 1 Capital	14.70%	11.75%	4.00%	6.00%
Total Capital	18.95	15.35	8.00	10.00
Tier 1 Leverage	11.60	11.60	4.00	5.00
Capital One, F.S.B.				
Tier 1 Capital	15.40%	12.56%	4.00%	6.00%
Total Capital	16.69	13.84	8.00	10.00
Tier 1 Leverage	14.13	14.13	4.00	5.00
June 30, 2003				
Capital One Bank				
Tier 1 Capital	15.08%	10.98%	4.00%	6.00%
Total Capital	20.31	15.04	8.00	10.00
Tier 1 Leverage	12.70	12.70	4.00	5.00
Capital One, F.S.B.				
Tier 1 Capital	15.39%	11.87%	4.00%	6.00%
Total Capital	17.09	13.47	8.00	10.00
Tier 1 Leverage	14.53	14.53	4.00	5.00

The Bank and Savings Bank treat a portion of their loans as "subprime" under the "Expanded Guidance for Subprime Lending Programs" ("Subprime Guidelines") issued by the four federal banking agencies and have assessed their capital and allowance for loan losses accordingly. Under the Subprime Guidelines, the Bank and Savings Bank each exceeds the requirements for a "well-capitalized" institution as of June 30, 2004.

For purposes of the Subprime Guidelines, the Company has treated as "subprime" all loans in the Bank's and the Savings Bank's targeted "subprime" programs to customers either with a FICO score of 660 or below or with no FICO score. The Bank and the Savings Bank hold on average 200% of the total risk-based capital charge that would otherwise apply to such assets. This results in higher levels of regulatory capital at the Bank and the Savings Bank. As of June 30, 2004, approximately \$4.7 billion, or 17.8%, of the Bank's, and \$2.5 billion, or 18.1%, of the Savings Bank's, on-balance sheet assets were treated as "subprime" for purposes of the Subprime Guidelines.

The reduction in the Bank's capital ratios at June 30, 2004, compared with June 30, 2003, resulted from the growth in the managed loan portfolio and dividends made to the Corporation. The Company currently expects to operate each of the Bank and Savings Bank in the future with a total risk-based capital ratio of at least 12%. The Corporation has a number of alternatives available to meet any additional regulatory capital needs of the Bank and the Savings Bank, including substantial liquidity held at the Corporation and available for contribution.

In August 2000, the Bank received regulatory approval and established a subsidiary bank in the United Kingdom. In connection with the approval of its former branch office in the United Kingdom, the Company committed to the Federal Reserve that, for so long as the Bank maintains a branch or subsidiary bank in the United Kingdom, the Company will maintain a minimum Tier 1 Leverage ratio of 3.0%. As of June 30, 2004 and 2003, the Company's Tier 1 Leverage ratio was 14.25% and 12.84%, respectively.

Additionally, federal banking law limits the ability of the Bank and Savings Bank to transfer funds to the Corporation. As of June 30, 2004, retained earnings of the Bank and the Savings Bank of \$646.5 million and \$587.5 million, respectively, were available for payment of dividends to the Corporation without prior approval by the regulators. The Savings Bank, however, is required to give the OTS at least 30 days advance notice of any proposed dividend and the OTS, in its discretion, may object to such dividend.

Dividend Policy

Although the Company expects to reinvest a substantial portion of its earnings in its business, the Company also intends to continue to pay regular quarterly cash dividends on its common stock. The declaration and

payment of dividends, as well as the amount thereof, are subject to the discretion of the Board of Directors of the Company and will depend upon the Company's results of operations, financial condition, cash requirements, future prospects and other factors deemed relevant by the Board of Directors. Accordingly, there can be no assurance that the Corporation will declare and pay any dividends. As a holding company, the ability of the Corporation to pay dividends is dependent upon the receipt of dividends or other payments from its subsidiaries. Applicable banking regulations and provisions that may be contained in borrowing agreements of the Corporation or its subsidiaries may restrict the ability of the Corporation's subsidiaries to pay dividends to the Corporation or the ability of the Corporation to pay dividends to its stockholders.

Business Outlook

This business outlook section summarizes the Company's expectations for earnings for 2004, and its primary goals and strategies for continued growth. The statements contained in this section are based on management's current expectations. Certain statements are forward looking, and therefore actual results could differ materially. Factors that could materially influence results are set forth throughout this section and below, under, "Risk Factors."

Earnings Goals

The Company expects fully diluted earnings per share results of between \$5.60 and \$5.90 in 2004, which represents an increase of between 15% and 22% growth over its earnings of \$4.85 per share (fully diluted) in 2003.

The Company's 2004 earnings per share estimate is based on its expectations for continued strong earnings in its U.S. Card segment and an increasing earnings contribution from its Auto Finance and Global Financial Services segments. The Company anticipates its percentage of managed loan growth rate in 2004 to be in the mid-teens, with a bias towards lower loss assets and a higher growth rate in its diversification businesses than its U.S. Card segment.

The Company's earnings are a function of its revenues (net interest income and non-interest income), consumer usage, payment and attrition patterns, the credit quality and growth rate of its earning assets (which affects fees, charge-offs and provision expense) and the Company's marketing and operating expenses. Specific factors likely to affect the Company's 2004 earnings are the portion of its loan portfolio it holds in lower loss assets, changes in consumer payment behavior, the competitive, legal and regulatory environment and the level of investments in its diversification businesses.

The Company expects to achieve these results based on the continued success of its business strategies and its current assessment of the competitive, regulatory and funding market environments that it faces (each of which is discussed elsewhere in this document), as well as the expectation that the geographies in which the Company competes will not experience significant consumer credit quality erosion, as might be the case in an economic downturn or recession.

Managed Revenue Margin

The Company expects its managed revenue margin (defined as managed net interest income plus managed non-interest income divided by managed average earning assets) to be modestly lower over time as a result of the Company's expected continued bias in its loan portfolio towards lower loss assets, as well as the higher expected growth rate of the Company's Auto Finance and GFS segments. As discussed more fully below, these lower loss assets typically have higher average balances than the loans in the Company's current portfolio. As a result, the Company expects its charge-offs, operating expenses and marketing expenses to be lower in 2004 than in 2003 when measured as a percentage of average managed loans outstanding.

Marketing Investment

The Company expects its marketing expense in 2004 to be higher than in 2003, subject to opportunities it perceives in the competitive market. The Company believes the branded franchise that it is building strengthens and enables its IBS and mass customization strategies across product lines. The Company cautions, however, that an increase or decrease in marketing expense or brand awareness does not necessarily correlate to a comparable increase or decrease in outstanding balances or accounts due to, among other factors, the long-term nature of brand building, customer attrition and utilization patterns, and shifts over time in targeting consumers and/or products that have varying marketing acquisition costs.

The Company expects to vary its marketing across its various products depending on the competitive dynamics of the various markets in which it participates. The Company expects to adjust its marketing allocations to target specific product lines that it believes offer attractive response rates and opportunities from time to time.

Due to the nature of competitive market dynamics and therefore the limited periods of opportunity identified by the Company's testing processes, marketing expenditures may fluctuate significantly from quarter to quarter. However, the Company expects its bias toward lower loss assets will lead to a gradual decline through 2004 in its marketing costs as a percentage of average managed loans, despite the Company's expectation that absolute marketing costs will increase.

Operating Cost Trends

The Company measures operating efficiency using a variety of metrics which vary by specific department or business unit. Nevertheless, the Company believes that overall annual operating costs as a percentage of managed loans (defined as all non-interest expense less marketing, divided by average managed loans) is an appropriate gauge of the operating efficiency of the enterprise as a whole. As the Company continues its bias towards lower loss assets and more diversified businesses, the Company expects operating costs as a percentage of its average managed loans to decline over time as a result of efficiency gains related to, among other things, servicing higher balance, higher credit quality accounts. In addition, the Company expects additional severance and facilities closing charges of between \$60 million and \$100 million to be taken in the second half of 2004.

Loan Growth

The Company expects managed loans to grow at a percentage rate in the mid-teens in 2004. It expects to achieve this growth by delivering moderate growth in its U.S. Card segment, coupled with stronger growth in its Auto Finance and Global Financial Services segments. Additionally, the Company is targeting a mid-teens percentage loan growth rate in 2005.

Delinquencies and Charge-offs

The Company's charge-off rate improved during the second quarter of 2004 due to its ongoing bias towards lower loss assets, seasonality, and improving economic conditions. The Company expects its charge-off rate to stabilize in the 4% to 4.5% range in the second half of 2004 and into 2005, with seasonal variations.

The Company's delinquency rate also improved during the second quarter of 2004 due to similar factors. Generally, fluctuations in delinquency levels can have several effects, including changes in the amounts of past due and over-limit fees assessed (lower delinquencies typically cause lower assessments), changes to the non-accrued amounts for finance charges and fees (lower delinquencies typically decrease non-accrued

amounts), increased or decreased collections expenses, and/or changes in the allowance for loan losses and the associated provision expenses. The Company's allowance for loan losses in a given period is a function of charge-offs in the period, the delinquency status of those loans and other factors, such as the Company's assessment of general economic conditions and the amount of outstanding loans added to the reported balance sheet during the period.

The Company expects to build its allowance for loan losses in the second half of 2004. The outlook is based on current and expected charge-off and delinquency rates, as well as expected on-balance sheet loan growth, an on-going bias towards lower loss and diversified assets, and on a continuation of current economic conditions. This outlook is sensitive to general economic conditions, employment trends, and bankruptcy trends, in addition to the Company's on balance sheet loan growth results.

Return on Managed Assets

The Company expects that its return on managed assets will be around 1.6% for the full year of 2004. However, the Company expects that this metric will vary quarter-to-quarter throughout 2004. Additionally, the Company is targeting a return on managed assets of 1.6% in 2005.

Impact of Attrition

The Company's earnings are also sensitive to the level of customer and/or balance attrition that it experiences. Fluctuation in attrition levels can occur due to the level of competition within the industries in which the Company competes, as well as competition from outside of the Company's industries, such as consumer debt consolidation that may occur during a period of significant mortgage refinancing.

The Company's Core Strategy: IBS

The Company's core strategy has been, and is expected to continue to be, to apply its proprietary IBS to its consumer lending business and other financial products. The Company continues to seek to identify new product and new market opportunities, and to make investment decisions that are informed by the Company's intensive testing and analysis. The Company's objective is to continue diversifying its consumer finance activities, which may include expansion into additional geographic markets, other consumer loan products, and/or the retail branch banking business.

The Company's lending products and other products are subject to intense competitive pressures that management anticipates will continue to increase as the lending markets mature, and that could affect the economics of decisions that the Company has made or will make in the future in ways that it did not anticipate, test or analyze.

U.S. Card Segment

The Company's U.S. Card segment consisted of \$45.2 billion of U.S. consumer credit card receivables as of June 30, 2004, marketed to consumers across the full credit spectrum. The Company's strategy for its U.S. Card segment is to offer compelling, value-added products to its customers, such as Lifestyles and Rewards credit cards. The Company expects balanced growth across the various credit risk segments of its credit card portfolio in the second half of 2004.

The competitive environment is currently intense for credit card products. Industry mail volume has increased substantially in recent years, resulting in declines in response rates to the Company's new customer solicitations. Additionally, competition has increased the attrition levels in the Company's existing portfolio. Despite this intense competition, the Company continues to believe that its capabilities will enable it to originate new credit card accounts that exceed the Company's return on investment requirements.

The Company continues to test new credit card products. While the Company continues to market an array of products to customers with a broad range of risk profiles, the Company plans to focus more of its marketing towards those customers with strong credit histories. Products targeted to high credit quality consumers tend to be more expensive for the Company to originate, and produce revenues and balances more slowly than credit card products marketed to customers with weaker credit histories.

The Company's credit card products marketed to consumers with less established or higher risk credit profiles continue to experience steady mail volume and increased pricing competition. These products generally feature higher annual percentage rates, lower credit lines, and annual membership fees. These products produce revenues more quickly than higher credit quality loans.

Additionally, since these borrowers are generally viewed as higher risk, they tend to be more likely to pay late or exceed their credit limit, which results in additional fees assessed to their accounts. The Company's strategy has been, and is expected to continue to be, to offer competitive annual percentage rates and annual membership, late and overlimit fees on these accounts.

Auto Finance Segment

This segment consisted of \$9.4 billion of U.S. auto receivables as of June 30, 2004, marketed across the credit spectrum, via direct and indirect marketing channels. The Company expects to increase its auto loan portfolio more quickly in prime and direct marketed products than through other products or channels in 2004. The Company sold \$0.3 billion of automobile receivables in the second quarter of 2004, and expects to sell additional auto finance receivables in the second half of 2004.

In the fourth quarter of 2002, the Company entered into a forward flow agreement with a purchaser to sell non-prime auto receivables originated through the Company's network of automobile dealers. These assets are sold at a premium, servicing released with no recourse. Loans sold under this agreement are originated using the Company's underwriting policies. During July of 2004, the Company gave notice to terminate this forward flow agreement; however, the Company plans to continue to sell auto receivables through other channels.

The Company expects that in 2004 the Auto Finance segment will continue to grow as the Company continues to diversify its loan portfolio.

Global Financial Services Segment

This segment primarily consisted of \$18.7 billion of installment loans and small business receivables originated within the U.S. and credit card receivables and installment loans originated outside of the U.S., primarily in the U.K. and Canada, as of June 30, 2004.

The products contained within the Global Financial Services segment play a key role in the asset diversification strategy of the Company, and thus the Company expects the Global Financial Services segment will continue to grow loan balances at a faster pace than the U.S. Card segment.

Risk Factors

This Quarterly Report on Form 10-Q contains forward-looking statements. We also may make written or oral forward-looking statements in our periodic reports to the Securities and Exchange Commission on

Forms 10-K and 8-K, in our annual report to shareholders, in our proxy statements, in our offering circulars and prospectuses, in press releases and other written materials and in statements made by our officers, directors or employees to third parties. Statements that are not historical facts, including statements about our beliefs and expectations, are forward-looking statements. Forward-looking statements include information relating to our future earnings per share, growth in managed loans outstanding, product mix, segment growth, managed revenue margin, funding costs, operations costs, employment growth, marketing expense, delinquencies and charge-offs. Forward-looking statements also include statements using words such as "expect," "anticipate," "hope," "intend," "plan," "believe," "estimate" or similar expressions. We have based these forward-looking statements on our current plans, estimates and projections, and you should not unduly rely on them.

Forward-looking statements are not guarantees of future performance. They involve risks, uncertainties and assumptions, including the risks discussed below. Our future performance and actual results may differ materially from those expressed in these forward-looking statements. Many of the factors that will determine these results and values are beyond our ability to control or predict. We undertake no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. You should carefully consider the factors discussed below in evaluating these forward-looking statements.

This section highlights specific risks that could affect our business and us. Although we have tried to discuss key factors, please be aware that other risks may prove to be important in the future. New risks may emerge at any time and we cannot predict such risks or estimate the extent to which they may affect our financial performance. In addition to the factors discussed elsewhere in this report, among the other factors that could cause actual results to differ materially are the following:

We Face Strategic Risks in Sustaining Our Growth and Pursuing Diversification

Our growth strategy is threefold. First, we seek to continue to grow our domestic credit card business. Second, we desire to continue to build and grow our automobile finance business. Third, we hope to continue to diversify our business, both geographically and in product mix, by growing our lending business, including credit cards, internationally, principally in the United Kingdom and Canada, and by identifying, pursuing and expanding new business opportunities, such as installment lending and small business lending. Our ability to grow is driven by the success of our fundamental business plan and our revenue may be adversely affected by our continuing bias toward lower loss assets (because of the potentially lower margins on such accounts), the level of our investments in new businesses or regions and our ability to successfully apply IBS to new businesses. This risk has many components, including:

- Customer and Account Growth. As a business driven by customer finance, our growth is highly dependent on our ability to retain existing customers and
 attract new ones, grow existing and new account balances, develop new market segments and have sufficient funding available for marketing activities to
 generate these customers and account balances. Our ability to grow and retain customers is also dependent on customer satisfaction, which may be adversely
 affected by factors outside of our control, such as postal service and other marketing and customer service channel disruptions and costs.
- Product and Marketing Development. Difficulties or delays in the development, production, testing and marketing of new products or services, which may be
 caused by a number of factors including, among other things, operational constraints, regulatory and other capital requirements and legal difficulties, will
 affect the success of such products or services and can cause losses arising from the costs to develop unsuccessful products and services, as well as decreased
 capital availability. In addition, customers may not accept the new products and services offered.

- Competition. As explained in more detail below, we face intense competition from many other providers of credit cards and other consumer financial products and services. The competition affects not only our existing businesses, but also our ability to grow these businesses, to develop new opportunities, and to make new acquisitions. As we continue to have a bias toward lower loss assets in our portfolio and to diversify beyond U.S. consumer credit cards, pricing competition, in particular, may make such growth and diversification difficult or financially impractical to achieve. See "We Face Intense Competition in All of Our Markets" below.
- International Risk. Part of our diversification strategy has been to grow internationally. Our growth internationally faces additional challenges, including
 limited access to information, differences in cultural attitudes toward credit, changing regulatory and legislative environments, political developments,
 exchange rates and differences from the historical experience of portfolio performance in the United States and other countries.

We Face Intense Competition in All of Our Markets

We face intense competition from many other providers of credit cards and other consumer financial products and services. In particular, in our credit card activities, we compete with international, national, regional and local bank card issuers, with other general purpose credit or charge card issuers, and to a certain extent, issuers of smart cards and debit cards and providers of other types of financial services (such as home equity lines and other products). We face similar competitive markets in our auto financing, small business lending and installment loan activities as well as in our international markets. Thus, the cost to acquire new accounts will continue to vary among product lines and may rise. In addition, the Gramm-Leach-Bliley Financial Services Modernization Act of 1999 (the "GLB Act"), which permits greater affiliations between banks, securities firms and insurance companies, may increase competition in the financial services industry, including in the credit card business. Increased competition has resulted in, and may continue to cause, a decrease in credit card response rates and reduced productivity of marketing dollars invested in certain lines of business. Other credit card issuers based on price (primarily interest rates and fees), credit limit and other product features, customer loyalty is limited. We may lose entire accounts, or may lose account balances, to competing card issuers. Our auto financing and installment product salso face intense competition on the basis of price. Customer attrition from any or all of our products, together with any lowering of interest rates or fees that we might implement to retain customers, could reduce our revenues and therefore our earnings. We expect that competition will continue to grow more intense with respect to most of our products, including the products we offer internationally.

In addition, some of our competitors may be substantially larger than we are, which may give those competitors advantages, including a more diversified product and customer base, operational efficiencies and more versatile technology platforms. These competitors may also consolidate with other financial institutions in ways that enhance these advantages.

We Face Risk From Economic Downturns

Delinquencies and credit losses in the consumer finance industry generally increase during economic downturns or recessions. Likewise, consumer demand may decline during an economic downturn or recession. Accordingly, an economic downturn (either local or national), can hurt our financial performance as accountholders default on their loans or, in the case of credit card accounts, carry lower balances. Furthermore, because our business model is to lend across the credit spectrum, we make loans to lower credit quality customers. These customers generally have higher rates of charge-offs and delinquencies than do higher credit quality customers. Additionally, as we increasingly market our cards internationally, an economic downturn or recession outside the United States also could hurt our financial performance.

Reputational Risk and Social Factors May Impact our Results

Our ability to originate and maintain accounts is highly dependent upon consumer perceptions of our financial health and business practices. To this end, we carefully monitor internal and external developments for areas of potential reputational risk and have established a Corporate Reputation Committee, a committee of senior management, to assist in evaluating such risks in our business practices and decisions. We have also aggressively pursued a campaign to enhance our brand image and awareness in recent years. Adverse developments in our brand campaign or in any of the areas described above, however, could damage our reputation in both the customer and funding markets, leading to difficulties in generating and maintaining accounts as well as in financing them. Adverse impacts on our reputation may also create difficulties with our regulators.

A variety of social factors may cause changes in credit card and other consumer finance use, payment patterns and the rate of defaults by accountholders and borrowers. These social factors include changes in consumer confidence levels, the public's perception of the use of credit cards and other consumer debt, and changing attitudes about incurring debt and the stigma of personal bankruptcy and consumer concerns about the practices of certain lenders. Our goal is to manage these risks through our underwriting criteria and product design, but these tools may not be sufficient to protect our growth and profitability during a sustained period of economic downturn or recession or a material shift in social attitudes.

We May Face Limited Availability of Financing, Variation in Our Funding Costs and Uncertainty in Our Securitization Financing

In general, the amount, type and cost of our funding, including financing from other financial institutions, the capital markets and deposits, directly impacts our expense in operating our business and growing our assets and therefore, can positively or negatively affect our financial results. A number of factors could make such financing more difficult, more expensive or unavailable on any terms both domestically and internationally (where funding transactions may be on terms more or less favorable than in the United States), including, but not limited to, financial results and losses, changes within our organization, specific events that adversely impact our reputation, changes in the activities of our business partners, disruptions in the capital markets, specific events that adversely impact the financial services industry, counter-party availability, changes affecting our assets, our corporate and regulatory structure, interest rate fluctuations, ratings agencies actions, general economic conditions and accounting and regulatory changes and relations. Our funding risks are also higher due to our lower unsecured debt rating compared to other banking institutions and the proportion of certain accounts in our loan portfolio viewed by some as "subprime." In addition, our ability to raise funds is strongly affected by the general state of the U.S. and world economies, and may become increasingly difficult due to economic and other factors.

The securitization of consumer loans, which involves the legal sale of beneficial interests in consumer loan balances, is one of our major sources of funding. The consumer asset-backed securitization market in the United States currently exceeds \$1.5 trillion, with approximately \$298.0 billion issued in 2004. As of June 30, 2004, we had \$45.1 billion of securitization funding outstanding, comprising 56% of our total managed liabilities. Despite the size and relative stability of these markets and our position as a leading issuer, if these markets experience difficulties we may be unable to securitize our loan receivables or to do so at favorable pricing levels. Factors affecting our ability to securitize our loan receivables or to do so at

favorable pricing levels include, in addition to the above factors, the overall credit quality of our securitized loans, the stability of the market for securitization transactions, and the legal, regulatory, accounting and tax environments governing securitization transactions. If we were unable to continue to securitize our loan receivables at current levels, we would use our investment securities and money market instruments in addition to alternative funding sources to fund increases in loan receivables and meet our other liquidity needs. The resulting change in our current liquidity sources could potentially subject us to certain risks. These risks would include an increase in our cost of funds, an increase in the reserve for possible credit losses and the provision for possible credit losses as more loans would remain on our consolidated balance sheet, and lower loan growth, if we were unable to find alternative and cost-effective funding sources. Also, if we could not continue to remove the loan receivables from the balance sheet we would possibly need to raise additional capital to support loan and asset growth and potentially provide additional credit enhancement.

In addition, the occurrence of certain events may cause the securitization transactions to amortize earlier than scheduled, which would accelerate the need for additional funding. This early amortization could, among other things, have a significant effect on the ability of the Bank and the Savings Bank to meet the capital adequacy requirements as all off-balance sheet loans experiencing such early amortization would have to be recorded on the balance sheet. See page 52 in Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity Risk Management" contained in the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003.

We May Experience Changes in Our Debt Ratings

In general, ratings agencies play an important role in determining, by means of the ratings they assign to issuers and their debt, the availability and cost of wholesale funding. We currently receive ratings from several ratings entities for our secured and unsecured borrowings. As private entities, ratings agencies have broad discretion in the assignment of ratings. A rating below investment grade typically reduces availability and increases the cost of market-based funding, both secured and unsecured. A debt rating of Baa3 or higher by Moody's Investors Service, or BBB- or higher by Standard & Poor's and Fitch Ratings, is considered investment grade. Currently, all three ratings agencies rate the unsecured senior debt of the Bank as investment grade. Two of the three ratings agencies rate the unsecured senior debt of the Corporation investment grade, with Standard & Poor's assigning a rating of BB+, or one level below investment grade. The following chart shows ratings for Capital One Financial Corporation and Capital One Bank as of June 30, 2004. As of that date, the ratings outlooks were as follows:

	Standard & Poor's	Moody's	Fitch
Capital One Financial Corporation	BB+	Baa3	BBB
Capital One Financial Corporation - Outlook	Positive	Stable	Stable
Capital One Bank	BBB-	Baa2	BBB+
Capital One Bank - Outlook	Positive	Stable	Stable

Because we depend on the capital markets for funding and capital, we could experience reduced availability and increased cost of funding if our debt ratings were lowered. This result could make it difficult for us to grow at or to a level we currently anticipate. The immediate impact of a ratings downgrade on other sources of funding, however, would be limited, as deposit funding and pricing is not generally determined by corporate debt ratings. The Savings Bank is authorized to engage in a full range of deposit-taking activities, but our ability to use deposits as a source of funding is generally regulated by federal laws and regulations. Likewise, our credit facility does not contain covenants that could be triggered by a ratings downgrade, although the pricing of any borrowings under this facility is linked to these ratings.

We compete for funding with other banks, savings banks and similar companies. Some of these institutions are publicly traded. Many of these institutions are substantially larger, have more capital and other resources and have better debt ratings than we do. In addition, as some of these competitors consolidate with other financial institutions, these advantages may increase. Competition from these institutions may increase our cost of funds. Events that disrupt capital markets and other factors beyond our control could also make our funding sources more expensive or unavailable.

We Face Exposure from Our Unused Customer Credit Lines

Because we offer our customers credit lines, the full amount of which is most often not used, we have exposure to these unfunded lines of credit. These credit lines could be used to a greater extent than our historical experience would predict. If actual use of these lines were to materially exceed predicted line usage, we would need to raise more funding than anticipated in our current funding plans. It could be difficult to raise such funds, either at all, or at favorable rates.

Our Accounts and Loan Balances Can Be Volatile

Changes in our aggregate accounts or consumer loan balances and the growth rate and composition thereof, including changes resulting from factors such as shifting product mix, amount of actual marketing expenses and attrition of accounts and loan balances, can have a material adverse effect on our financial results. The number of accounts and aggregate total of loan balances of our consumer loan portfolio (including the rate at which it grows) will be affected by a number of factors, including the level of our marketing investment, how we allocate such marketing investment among different products, the rate at which customers transfer their accounts and loan balances to us or away from us to competing lenders. Such accounts and loan balances are also affected by our desire to avoid unsustainable growth rates, and general economic conditions, which may increase or decrease the amount of spending by our customers and affect their ability to repay their loans, and other factors beyond our control.

We Face Risk Related to the Strength of our Operational and Organizational Infrastructure

Our ability to grow is also dependent on our ability to build or acquire the necessary operational and organizational infrastructure, manage expenses as we expand, and recruit management and operations personnel with the experience to run an increasingly complex business. Similar to other large corporations, operational risk can manifest itself at Capital One in many ways, such as errors related to failed or inadequate processes, faulty or disabled computer systems, fraud by employees or persons outside the Company and exposure to external events. In addition, we outsource some of our operational functions to third parties; these third parties may experience similar errors or disruptions that could adversely impact us and over which we may have limited control. We are also subject to business interruptions arising from events either partially or completely beyond our control such as disruption in the U.S. Postal Service that could adversely impact our response rates and consumer payments. Failure to build and maintain the necessary operational infrastructure can lead to risk of loss of service to customers, legal actions or noncompliance with applicable laws or regulatory standards. Although we have devoted and will continue to devote resources to building and maintaining our operational infrastructure, including our system of internal control, there can be no assurance that we will not suffer losses from operational risks in the future.

We May Experience Increased Delinquencies and Credit Losses

Like other credit card lenders and providers of consumer financing, we face the risk that our customers will not repay their loans. A customer's failure to repay is generally preceded by missed payments. In some instances, a customer may declare bankruptcy prior to missing payments, although this is not generally the

case. Customers who declare bankruptcy frequently do not repay credit card or other consumer loans. Where we have collateral, we attempt to seize it when customers default on their loans. The value of the collateral may not equal the amount of the unpaid loan and we may be unsuccessful in recovering the remaining balance from our customers. Rising delinquencies can require us to increase our allowance for loan losses and thus hurt our overall financial performance. In addition, rising delinquencies and rising rates of bankruptcy are often precursors of future charge-offs. High charge-off rates may hurt our overall financial performance if we are unable to raise revenue to compensate for these losses, may adversely impact the performance of our securitizations, and may increase our cost of funds.

Our ability to assess the credit worthiness of our customers may diminish. We market our products to a wide range of customers including those with less experience with credit products and those with a history of missed payments. We select our customers, manage their accounts and establish prices and credit limits using proprietary models and other techniques designed to accurately predict future charge-offs. Our goal is to set prices and credit limits such that we are appropriately compensated for the credit risk we accept for both high and low risk customers. We face a risk that the models and approaches we use to select, manage, and underwrite our customers may become less predictive of future charge-offs due to changes in the competitive environment or in the economy. Intense competition, a weak economy, or even falling interest rates can adversely affect our actual charge-offs and our ability to accurately predict future charge-offs. These factors may cause both a decline in the ability and willingness of our customers to repay their loans and an increase in the frequency with which our lower risk customers defect to more attractive, competitor products. In our auto finance business, declining used-car prices reduce the value of our collateral and can adversely affect charge-offs. We attempt to mitigate these risks by adopting a conservative approach to our predictions of future charge-offs. Nonetheless, there can be no assurance that we will be able to accurately predict charge-offs, and our failure to do so may adversely affect our profitability and ability to grow.

The trends that have caused the reduction of charge-offs over the course of 2003 and the first half of 2004 may not continue. During that time, we increased the proportion of lower-risk borrowers in our portfolio and increased the proportion of lower risk asset classes, like auto loans, relative to credit cards. In addition, in 2003, our managed loan portfolio grew 19%. Especially in the credit card business, higher growth rates cause lower charge-offs. This is primarily driven by lower charge-offs in the first six to eight months of the life of a pool of new accounts. Finally, although the U.S. economy has been improving, there can be no assurance that these trends will continue in the future.

We hold an allowance for expected losses inherent in our existing reported loan portfolio as provided for by the applicable accounting rules. There can be no assurance, however, that such allowances will be sufficient to account for actual losses. We record charge-offs according to accounting practices consistent with accounting and regulatory guidelines and rules. These rules could change and cause our charge-offs to increase for reasons unrelated to the underlying performance of our portfolio. Unless offset by other changes, this could reduce our profits.

We Face Market Risk of Interest Rate and Exchange Rate Fluctuations

Like other financial institutions, we borrow money from institutions and depositors, which we then lend to customers. We earn interest on the consumer loans we make, and pay interest on the deposits and borrowings we use to fund those loans. Changes in these two interest rates affect the value of our assets and liabilities. If the rate of interest we pay on our borrowings increases more than the rate of interest we earn on our loans, our net interest income, and therefore our earnings, could fall. Our earnings could also be hurt if the rates on our consumer loans fall more quickly than those on our borrowings.

However, our goal is generally to maintain an interest rate neutral or "matched" position, where interest rates and exchange rates on loans and borrowings or foreign currencies go up or down by the same amount

and at the same time so that interest rate and exchange rate changes for loans or borrowings or foreign currencies will not affect our earnings. The financial instruments and techniques we use to manage the risk of interest rate and exchange rate fluctuations, such as asset/liability matching and interest rate and exchange rate swaps and hedges and some forward exchange contracts, may not always work successfully or may not be available at a reasonable cost. Furthermore, if these techniques become unavailable or impractical, our earnings could be subject to volatility and decreases as interest rates and exchange rates change.

Changes in interest rates also affect the balances our customers carry on their credit cards and affect the rate of pre-payment for installment loan products. When interest rates fall, there may be more low-rate product alternatives available to our customers. Consequently, their credit card balances may fall and pre-payment rates may rise. We can mitigate this risk by reducing the interest rates we charge or by refinancing installment loan products. However, these changes can reduce the overall yield on our portfolio if we do not adequately provide for them in our interest rate hedging strategies. When interest rates rise, there are fewer low-rate alternatives available to customers. Consequently, credit card balances may rise (or fall more slowly) and pre-payment rates on installment lending products may fall. In this circumstance, we may have to raise additional funds at higher interest rates. In our credit card business, we could, subject to legal and competitive constraints, mitigate this risk by increasing the interest rates we charge, although such changes may increase opportunities for our competitors to offer attractive products to our customers and consequently increase customer attrition from our portfolio. See Item 2 "Management's Discussion and Analysis of Financial Condition and Results of Operations—Interest Rate Risk Management" contained in the Annual Report on Form 10-K for the year ended December 31, 2003.

We Face the Risk of a Complex and Changing Regulatory and Legal Environment

Due to our significant reliance on certain contractual relationships, including our funding providers, as well as our unique corporate structure and heavily regulated industry, we face a risk of loss due to legal contracts, aspects of or changes in our legal structure and changes in laws and regulations. We also are subject to an array of banking, consumer lending and deposit laws and regulations that apply to almost every element of our business. Failure to comply with these laws and regulations could result in financial, structural and operational penalties, including receivership. In addition, efforts to comply with these laws and regulations may increase our costs and/or limit our ability to pursue certain business opportunities. See "Supervision and Regulation" above.

Federal and state laws and rules, as well as accounting rules and rules to which we are subject in foreign jurisdictions in which we conduct business, significantly limit the types of activities in which we may engage. For example, federal and state consumer protection laws and rules, and laws and rules of foreign jurisdictions where we conduct business, limit the manner in which we may offer and extend credit. From time to time, the U.S. Congress, the states and foreign governments consider changing these laws and may enact new laws or amend existing laws to regulate further the consumer lending industry. Such new laws or rules could limit the amount of interest or fees we can charge, restrict our ability to collect on account balances, or materially affect us or the banking or credit card industries in some other manner. Additional federal, state and foreign consumer protection legislation also could seek to expand the privacy protections afforded to customers of financial institutions and restrict our ability to share or receive customer information.

The laws governing bankruptcy and debtor relief, in the U.S. or in foreign jurisdictions in which we conduct business, also could change, making it more expensive or more difficult for us to collect from our customers. Congress has recently considered, and the House of Representatives has passed, legislation that would change the existing federal bankruptcy laws. One intended purpose of this legislation is to increase the collectibility of unsecured debt; however, it is not clear whether or in what form Congress may adopt this legislation and we cannot predict how the final version of this legislation may affect us, if passed into law.

In addition, banking regulators possess broad discretion to issue or revise regulations, or to issue guidance, which may significantly impact us. For example, in 2001, regulators restricted the ability of two of our competitors to provide further credit to higher risk customers due principally to supervisory concerns over rising charge-off rates and capital adequacy. We cannot, however, predict whether and how any new guidelines issued or other regulatory actions taken by the banking regulators will be applied to the Bank or the Savings Bank or the resulting effect on the Corporation, the Bank or the Savings Bank. In addition, certain state and federal regulators are considering or have approved rules affecting certain practices of "subprime" mortgage lenders. There can also be no assurance that these regulators will not also consider or approve additional rules with respect to "subprime" credit card lending or, if so, how such rules would be applied to or affect the Corporation, the Bank or the Savings Bank.

In addition, existing laws and rules in the U.S., at the state level, and in the foreign jurisdictions in which we conduct operations, are complex. If we fail to comply with them, we may not be able to collect our loans in full, or we might be required to pay damages or penalties to our customers. For these reasons, new or changes in existing laws or rules could hurt our profits.

Fluctuations in Our Expenses and Other Costs May Hurt Our Financial Results

Our expenses and other costs, such as human resources and marketing expenses, directly affect our earnings results. Many factors can influence the amount of our expenses, as well as how quickly they grow. For example, further increases in postal rates or termination of our negotiated service arrangement with the United States Postal Service could raise our costs for postal service, which is a significant component of our expenses for marketing and for servicing our 46.6 million accounts as of June 30, 2004. As our business develops, changes or expands, additional expenses can arise from asset purchases, structural reorganization, a reevaluation of business strategies and/or expenses to comply with new or changing laws or regulations. Other factors that can affect the amount of our expenses include legal and administrative cases and proceedings, which can be expensive to pursue or defend. In addition, changes in accounting policies can significantly affect how we calculate expenses and earnings.

Item 3. Quantitative and Qualitative Disclosure of Market Risk

The information called for by this item is provided under the Corporation's Annual Report on Form 10-K for the year ended December 31, 2003, Item 7A "Quantitative and Qualitative Disclosures about Market Risk." No material changes have occurred during the six month period ended June 30, 2004.

Item 4. Controls and Procedures

The Corporation carried out an evaluation, under the supervision and with the participation of the Corporation's management, of the effectiveness of the design and operation of the Corporation's disclosure controls and internal controls and procedures as of June 30, 2004 pursuant to Exchange Act Rules 13a-14 and 13a-15. These controls and procedures for financial reporting are the responsibility of the Corporation's management. Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the Company's disclosure controls and procedures are effective in alerting them in a timely manner to material information relating to the Corporation (including consolidated subsidiaries) required to be included in the Corporation's periodic filings with the Securities and Exchange Commission. The Corporation has established a Disclosure Committee consisting of members of senior management to assist in this evaluation.

Part II Other Information

Item 1. Legal Proceedings

The information required by Item 1 is included in Part 1 of this Quarterly Report under the heading "Notes to Condensed Consolidated Financial Statements – Note I – Commitments and Contingencies."

Item 2. Changes in Securities, Uses of Proceeds and Issuer Purchases of Equity Securities

Period	(a) Total Number of Shares Purchased ⁽¹⁾	(b) rage Price per Share	(c) Total Number of Shares Purchased as Part of Publicly Announced Plans	(d) Maximum Number of Shares that May Yet Be Purchased Under the Plans
April 1-30, 2004	7,440	\$ 70.07	N/A	N/A
May 1-31, 2004	8,222	\$ 64.58	N/A	N/A
June 1-30, 2004	18,287	\$ 64.91	N/A	N/A
Total	33,949	\$ 65.96	N/A	N/A

⁽¹⁾ Shares purchased represent share swaps made in connection with stock option exercises.

Item 6. Exhibits and Reports on Form 8-K

- (a) Exhibits:
 - 4.1 Copy of 5.00% Notes, due 2009, of Capital One Bank
 - 10.1 Capital One Financial Corporation, Voluntary Non-Qualified Deferred Compensation Plan, dated May 28, 2004
 - 10.2 Capital One Financial Corporation, 2004 Stock Incentive Plan (incorporated by reference to Annex B to the Company's Proxy Statement on Schedule 14, filed March 17, 2004)
 - 10.3 Revolving Credit Facility Agreement, dated June 29, 2004 by and between Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B. and Capital One Bank (Europe), plc, as borrowers and JPMorgan Chase Bank, an Administrative Agent.
 - 31.1 Certification of Richard D. Fairbank
 - 31.2 Certification of Gary L. Perlin
 - 32.1 Certification* of Richard D. Fairbank
 - 32.2 Certification* of Gary L. Perlin
- (b) Reports on Form 8-K:

On April 21, 2004, the Company furnished under Item 9 – "Regulation FD Disclosure" of Form 8-K on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics — March 2004, for the month ended March 31, 2004.

On April 21, 2004, the Company furnished under Item 5 – "Other Events" and filed under Item 7 – "Financial Statements, Proforma Financial Information and Exhibits" and Item 12 – "Results of Operations and Financial Condition" of Form 8-K on Exhibit 99.1, a copy of its earnings press release for the first quarter of 2004 that was issued April 21, 2004. Additionally, the Company furnished the information in Exhibit 99.2, First Quarter Earnings Presentation for the quarter ended March 31, 2004.

On May 11, 2004, the Company furnished under Item 9 – "Regulation FD Disclosure" of Form 8-K on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics — April 2004, for the month ended April 30, 2004.

On May 17, 2004, the Company furnished under Item 9 – "Regulation FD Disclosure" of Form 8-K on Exhibit 99.1 its press release dated May 17, 2004.

On June 9, 2004, the Company furnished under Item 9 – "Regulation FD Disclosure" of Form 8-K on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics — May 2004, for the month ended May 31, 2004.

^{*} Information in this 10-Q furnished herewith shall not be deemed to be "filed" for the purposes of Section 18 of the 1934 Act or otherwise subject to the liabilities of that section.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: August 5, 2004

CAPITAL ONE FINANCIAL CORPORATION

(Registrant)

/s/ GARY L. PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer (Principal Financial Officer and duly authorized officer of the Registrant)

(to Offering Circular dated April 29, 2004)

Capital One Bank

(a Bank organized pursuant to the Laws of Virginia)

Global Bank Notes

Issue of US\$500,000,000 5.000% Senior Global Bank Notes due 2009

UNDER THE U.S.\$8,000,000,000 GLOBAL BANK NOTE PROGRAM

This document constitutes the Pricing Supplement relating to the issue of Notes described herein. Terms used herein shall be deemed to be defined as such for the purposes of the conditions set forth in the Offering Circular dated April 29, 2004. This Pricing Supplement is supplemental to and must be read in conjunction with such Offering Circular.

1.	Issuer:	Capital One Bank
2.	(i) Series Number:	1
	(ii) Tranche Number:	Not Applicable
3.	Specified Currency	USD
4.	Aggregate Principal Amount:	\$500,000,000
5.	Original Issue Date and Interest Commencement Date:	June 9, 2004
6.	Stated Maturity Date:	June 15, 2009
7.	Status of the Notes:	Senior
8.	Interest Basis:	5.000 per cent. Fixed Rate
9.	Redemption/Payment Basis:	Redemption at par
10.	Change of Interest or Redemption/Payment Basis:	Not Applicable
11.	Issue Price:	99.740 per cent. of the aggregate principal amount of the Notes
12.	Default Rate (if other than Interest Rate):	Not Applicable (Required only for listed issues)
13.	Authorized Denominations:	Minimum denominations of \$100,000 and integral multiples of \$1,000 in excess thereof
14.	Listing:	None
15.	Method of distribution:	Syndicated
PROVISIONS RELATING TO INTEREST (IF ANY) PAYABLE		

16.	Fixed Rate Note Provisions:	Applicable
	(i) Interest Rate(s):	5.000 per cent. per annum payable semi-annually in arrears
	(ii) Interest Payment Date(s):	June 15 and December 15 (or the next Business Day) in each year, commencing on December 15, 2004 up to and including the Stated Maturity Date
	(iii) Day Count Convention:	30/360
	(iv) Interest Determination Date(s):	Not Applicable
	(v) Other terms relating to the method of calculating interest for Fixed Rate Notes:	None
17.	Floating Rate Note Provisions:	Not Applicable
18.	Original Issue Discount Note (including Zero Coupon Note) Provisions:	Not Applicable
19.	Index/Formula Linked Interest Note Provisions:	Not Applicable
20.	Dual Currency Note Provisions:	Not Applicable
PRO	VISIONS RELATING TO REDEMPTION	
21.	Redeemable at Option of Issuer:	Not Applicable
22.	Repayable at Option of Holders:	Not Applicable
GEN	ERAL PROVISIONS APPLICABLE TO THE NOTES	
23.	Form of Notes:	
	(i) Bearer Notes:	Not Applicable
	(ii) Registered Notes:	Applicable
	— Registrar:	JPMorgan Chase Bank

— Transfer Agent:

- Record Dates:
- 24. Partly Paid Notes: amount of each payment comprising the Issue Price and date on which each payment is to be made and consequences (if any) of failure to pay, including the right of the Issuer to forfeit the Notes and interest due on late payment:

2

JPMorgan Chase Bank

Not Applicable

the related Interest Payment Date

15th calendar day (whether or not a Business Day) immediately preceding

25.	Installment Notes:		
	(i) Installment amount(s):	Not Applicable	
	(ii) Installment date(s):	Not Applicable	
26.	Other terms or specified conditions:	Certain of the Distribution Agents may make the notes available for distribution on the Internet through a third-party system operated by Market Axess Inc., an Internet-based communications technology provider. Market Axess Inc. is providing the system as a conduit for communications between such Distribution Agents and their respective customers and is not a party to any transactions. Market Axess Inc. is a registered broker-dealer and will receive compensation from the relevant Distribution Agent based on transactions conducted through the system. The relevant Distribution Agents will make the notes available to their respective customers through the Internet on the same terms as distributions of the notes made through other channels.	
27.	Talons for future Coupons or Receipts to be attached to Definitive Bearer Notes (and dates on which such Talons mature):	Not Applicable	
28.	Details of any additional or different Paying Agents, Registrars, London Issuing Agents, Transfer Agents:	Not Applicable	
DISTI	RIBUTION		
29.	(i) If syndicated, names of Distribution Agents:	Deutsche Bank Securities Inc.\$200,000,000J.P. Morgan Securities Inc.\$200,000,000Barclays Capital Inc.\$25,000,000Credit Suisse First Boston LLC\$25,000,000HSBC Securities (USA) Inc.\$25,000,000Wachovia Capital Markets, LLC\$25,000,000	
	(ii) Stabilization Manager (if any):	Deutsche Bank Securities Inc. The Stabilization Manager or any other person acting for the Stabilization Manager may over-allot or effect transactions with a view to supporting the market price of the Notes at a level higher than that which might otherwise prevail for a limited period. However, there may be no obligation on the Stabilization Manager any agent of the Stabilization Manager to do this. Such stabilizatio if commenced, may be discontinued at any time and must be terminated after a limited period. Such stabilization, if any, must comply with all applicable laws, regulations and rules.	

30. If non-syndicated, name of Distribution Agent:

3

Not Applicable

OPERATIONAL INFORMATION			
32.	CUSIP Code:	14040EHL9	
33.	ISIN Code:	US14040EHL92	
34.	Common Code:	019427285	
35.	Clearing System(s):	DTC, Euroclear and Clearstream, Luxembourg through DTC	
36.	Delivery:	Delivery against payment	
37.	Redenomination applicable:	Redenomination not applicable	
38.	38. "Business Day" definition (if other than as defined in the Offering		
	Circular):	Not Applicable	
39.	Governing Law:	New York	
[Remainder of Page Intentionally Left Blank]			

4

31.

Additional selling restrictions:

Not Applicable

RESPONSIBILITY

The Issuer accepts responsibility for the information contained in this Pricing Supplement. Signed on behalf of the Issuer:

By: /s/ Tom Feil

Tom Feil, Vice President, Capital Markets and Assistant Treasurer Duly authorized THIS NOTE HAS NOT BEEN AND WILL NOT BE REGISTERED WITH THE UNITED STATES SECURITIES AND EXCHANGE COMMISSION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS.

THIS NOTE IS A DTC GLOBAL NOTE WITHIN THE MEANING OF THE GLOBAL AGENCY AGREEMENT REFERRED TO HEREINAFTER.

UNLESS THIS CERTIFICATE IS PRESENTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITORY TRUST COMPANY (55 WATER STREET, NEW YORK, NEW YORK) (THE "*DEPOSITARY*") TO CAPITAL ONE BANK (THE "*BANK*") OR ITS AGENT FOR REGISTRATION OF TRANSFER, EXCHANGE OR PAYMENT, AND ANY CERTIFICATE ISSUED UPON REGISTRATION OF TRANSFER OF, OR IN EXCHANGE FOR, OR IN LIEU OF, THIS CERTIFICATE IS REGISTERED IN THE NAME OF CEDE & CO. OR IN SUCH OTHER NAME AS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITARY (AND ANY PAYMENT HEREON IS MADE TO CEDE & CO OR TO SUCH OTHER ENTITY AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITARY), ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL INASMUCH AS THE REGISTERED OWNER HEREOF, CEDE & CO., HAS AN INTEREST HEREIN.

TRANSFERS OF THIS DTC GLOBAL NOTE SHALL BE LIMITED TO TRANSFERS IN WHOLE, BUT NOT IN PART, TO NOMINEES OF THE DEPOSITARY OR TO A SUCCESSOR THEREOF OR SUCH SUCCESSOR'S NOMINEE.

THIS NOTE IS A DIRECT, UNCONDITIONAL, UNSECURED AND UNSUBORDINATED GENERAL OBLIGATION OF THE BANK. THIS NOTE RANKS *PARI PASSU* WITH ALL OTHER UNSECURED AND UNSUBORDINATED INDEBTEDNESS OF THE BANK, EXCEPT OBLIGATIONS, INCLUDING DEPOSIT LIABILITIES AND OTHER OBLIGATIONS THAT ARE SUBJECT TO ANY PRIORITY OR PREFERENCE UNDER APPLICABLE LAW.

THIS NOTE DOES <u>NOT</u> EVIDENCE A DEPOSIT AND IS <u>NOT</u> INSURED BY THE FEDERAL DEPOSIT INSURANCE CORPORATION ("*FDIC*") OR ANY OTHER INSURER.

THIS NOTE IS ISSUABLE ONLY IN MINIMUM DENOMINATIONS OF U.S. \$100,000 AND INTEGRAL MULTIPLES OF U.S. \$1,000 IN EXCESS THEREOF. EACH OWNER OF A BENEFICIAL INTEREST IN THIS NOTE IS REQUIRED TO HOLD A BENEFICIAL INTEREST OF U.S. \$100,000 PRINCIPAL AMOUNT AND INTEGRAL MULTIPLES OF U.S. \$1,000 IN EXCESS THEREOF OF THIS NOTE AT ALL TIMES.

CAPITAL ONE BANK GLOBAL BANK NOTE (Registered Global Note)

ORIGINAL ISSUE DATE:	June 9, 2004	PRINCIPAL AMOUNT: U.S.\$ 500,000,000
		SPECIFIED CURRENCY:
MATURITY DATE:	June 15, 2009	⊠ U.S. dollar
⊠ FIXED RATE NOTE	5.000%	Other:

FLOATING RATE NOTE

CAPITAL ONE BANK, a bank organized under the laws of the Commonwealth of Virginia (the "*Bank*"), for value received, hereby promises to pay to CEDE & CO, or registered assigns, the principal amount specified above as adjusted in accordance with Schedule 1 hereto, on the Maturity Date specified above (except to the extent redeemed or repaid prior to the Maturity Date) and to pay interest thereon (i) in accordance with the provisions set forth on the reverse hereof under the caption "Fixed Rate Interest Provisions," if this Note is designated as a "Fixed Rate Note" above, or (ii) in accordance with the provisions set forth on the reverse hereof under the caption "Floating Rate Interest Provisions," if this Note is designated as a "Fixed Rate Note" above, or (iii) naccordance with the provisions set forth on the reverse hereof under the caption "Floating Rate Interest Provisions," if this Note is designated as a "Fixed Rate Note" above, or (ii) in accordance with the provisions set forth on the reverse hereof under the caption "Floating Rate Interest Provisions," if this Note is designated as a "Floating Rate Note" above, in each case as such provisions may be modified or supplemented by the terms and provisions set forth in the Pricing Supplement attached hereto (the "*Pricing Supplement*"), and (to the extent that the payment of such interest shall be legally enforceable) to pay interest at the Default Rate per annum specified in the Pricing Supplement on any overdue principal and premium, if any, and on any overdue installment of interest. The interest so payable, and punctually paid or duly provided for, on any Interest Payment Date specified in the Pricing Supplement will be paid to the person in whose name this Note (or any predecessor Note) is registered at the close of business on the fifteenth calendar day (whether or not a Business Day (as defined on the reverse hereof)) next preceding the applicable Interest Payment Date (unless otherwise specified in the Pricing Supplement) (each, a "*Record Date"*);

This Note is one of a duly authorized issue of the Bank's notes due from 30 days to 30 years or more from date of issue (the "*Notes*"). The Notes are issued and to be issued in accordance with the Global Agency Agreement, dated as of April 29, 2004 (the "*Global Agency Agreement*"), among the Bank and JPMorgan Chase Bank as paying agent (the "*Domestic Paying Agent*") and as registrar (the "*Registrar*"), JPMorgan Chase Bank, London Branch, as paying agent (the "*London Paying Agent*") and as issuing agent (the "*London Issuing Agent*") and J.P. Morgan Bank Luxembourg S.A. as transfer agent (the "*Transfer Agent*") and as paying agent (the "*London Issuing Agent*") and J.P. Morgan Bank Luxembourg Agent, the "*Paying Agents*", and each individually, a "*Paying Agent*") and Kredietbank S.A. Luxembourgeoise as listing agent (the "*Listing Agent*"). The terms Domestic Paying Agent, Registrar, London Paying Agent, London Issuing Agent, Transfer Agent and Listing Agent shall include any additional or successor agents appointed in such capacities by the Bank.

The Bank shall cause to be kept at the office of the Registrar a register (the register maintained in such office or any other office or agency of the Registrar, herein referred to as the "*Note Register*") in which, subject to such reasonable regulations as it may prescribe, the Bank shall provide for the registration of Notes issued in registered form and of transfers of such Notes. The Bank has initially appointed JPMorgan Chase Bank, acting through its principal office at JPMorgan Chase Bank, Institutional Trust Services, 4 New York Plaza, 15th Floor, New York, New York 10004, as "Registrar" for the purpose of registering Notes issued in registered form and transfers of such Notes. The Bank reserves the right to rescind such designation at any time, and to transfer such function to another bank or financial institution.

The transfer of this Note is registrable in the Note Register, upon surrender of this Note for registration of transfer at the office or agency of the Registrar or any transfer agent maintained for that purpose, accompanied by a written instrument of transfer in form satisfactory to the Registrar (or such transfer agent) duly executed by, the registered holder hereof in person or its attorney thereunto duly authorized in writing.

Payment of principal, premium, if any, and interest with respect to this Note at Maturity to be made in U.S. Dollars, will be made in immediately available funds upon surrender of this Note at a specified office of a Paying Agent; *provided*, that this Note is surrendered to such Paying Agent in time for such Paying Agent to make such payment in immediately available funds in accordance with its normal procedures. Payments of interest on this Note (other than at Maturity) will be made by wire transfer to such account as has been appropriately designated to a Paying Agent by the person entitled to such payments.

Reference is made to the further provisions of this Note set forth on the reverse hereof and in the Pricing Supplement, which further provisions shall for all purposes have the same effect as if set forth at this place. In the event of any conflict between the provisions contained herein or on the reverse hereof and the provisions contained in the Pricing Supplement attached hereto, the latter shall control. References herein to "this Note," "hereof," "herein" and comparable terms shall include the Pricing Supplement attached hereto.

Unless the certificate of authentication hereon has been executed by the Registrar, by manual signature of an authorized signatory, this Note shall not be valid or obligatory for any purpose.

This Note shall be governed by, and construed in accordance with, the laws of the State of New York, without regard to the conflicts of law principles thereof.

IN WITNESS WHEREOF, the Bank has caused this Note to be duly executed.

CAPITAL ONE BANK

By: /s/ Stephen Linehan

Name: Stephen Linehan Title: Senior Vice President and Treasurer

Dated:

REGISTRAR'S CERTIFICATE OF AUTHENTICATION

This is one of the Notes referred to in the within-mentioned Global Agency Agreement.

JPMORGAN CHASE BANK, as Registrar

By: /s/ Indira Javier

Name: Indira Javier Title: Trust Administrator

[ATTACH PRICING SUPPLEMENT.]

1. <u>Authorized Denomination</u>.

This Note, and any Note issued in exchange or substitution herefor or in place hereof, or upon registration of transfer, exchange or partial redemption or repayment of this Note, may be issued only in an Authorized Denomination specified in the Pricing Supplement.

2. Payment of Principal, Premium, if any, and Interest.

(a) Unless otherwise provided herein or in the Pricing Supplement and subject to the provisions with respect to DTC Global Notes in clause (b) below, the principal, premium, if any, and interest with respect to this Note are payable in the Specified Currency indicated on the face hereof (or, if such Specified Currency is not at the time of such payment legal tender for the payment of public and private debts, in such other coin or currency of the country which issued such Specified Currency as at the time of such payment is legal tender for the payment of debts).

(b) Unless otherwise provided herein or in the Pricing Supplement, if this Note is a DTC Global Note and the Specified Currency indicated on the face hereof is other than U.S. dollars:

(i) Any such amounts paid by the Bank will be converted by JPMorgan Chase Bank, as Exchange Rate Agent (or such other agent as may be specified in the Pricing Supplement), which for these purposes shall act as currency exchange agent (the "*Exchange Rate Agent*"), into U.S. dollars for payment to the holder of this Note.

(ii) Any U.S. dollar amount to be received by the holder of this Note will be based on the Exchange Rate Agent's bid quotation as of 11:00 a.m., London time, on the second day on which banks are open for business in London, New York City and McLean, Virginia, preceding the applicable payment date, for the purchase of U.S. dollars with the Specified Currency for settlement on such payment date of the aggregate amount of the Specified Currency payable to all holders of Notes denominated other than in U.S. dollar scheduled to receive U.S. dollar payments. If such bid quotation is not available, the Exchange Rate Agent will obtain a bid quotation from a leading foreign exchange bank in London or New York City selected by the Exchange Rate Agent for such purchase. If no such bids are available, payment of the aggregate amount due to the holder of this Note on the payment date will be made in the Specified Currency, subject to the other provisions of this Note relating to payment in such Specified Currency. All currency exchange costs will be borne by the holder of this Note by deductions from such payments. All determinations referred to above made by the Exchange Rate Agent shall, in the absence of manifest error, be conclusive for all purposes and binding upon the holder of this Note.

(iii) If DTC, acting on instructions from any beneficial owner of this Note, notifies the Paying Agents in writing prior to 5:00 P.M., New York City time, on the fifth Business Day following any Record Date in the case of interest and the tenth calendar

day prior to any payment date for the payment of principal, of such beneficial owner's election to receive payments with respect to this Note in the Specified Currency, together with wire transfer instructions to an account denominated in the Specified Currency, payments from and including the related payment date will be made to such beneficial owner in the Specified Currency by wire transfer of same day funds in accordance with the relevant wire transfer instructions on the relevant payment date. If the Paying Agents have not been so notified, payments will be made in U.S. dollar as provided under clauses (i) and (ii) above. Any such notification from DTC to the Paying Agents shall remain in effect with respect to any further payments of principal, premium if any, and interest with respect to this Notes to such beneficial owner, unless and until DTC, acting on instructions from such beneficial owner, notifies the Paying Agents on or prior to 5:00 P.M., New York City time, on the fifth Business Day following any Record Date in the case of interest and the tenth calendar day prior to any payment date for the payment of principal, of such beneficial owner's revocation of its election. Any beneficial owner must rely on the procedures of DTC and, if such person is not a participant in DTC, on the procedures of the participant therein through which the beneficial owner owns its interest, with respect to making and revoking its election to receive payments in the Specified Currency.

(c) Subject to any fiscal or other laws and regulations applicable thereto in the place of payment, payments on registered Notes to be made in a Specified Currency other than the U.S. Dollar and payments on bearer Notes will be made by transfer to an account in the Specified Currency (which, in the case of a payment in Yen to a non-resident of Japan, shall be a non-resident account) maintained by the payee with, or by a check in the Specified Currency drawn on, a bank (which, in the case of a payment in Yen to a non-resident of Japan, shall be an authorized foreign exchange bank) in the Principal Financial Center (as defined below) of the Specified Currency (or such other location designated in the Pricing Supplement); *provided, however*, that a check may not be delivered to an address in, and an amount may not be transferred to an account at a bank in, the United States of America or its possessions by any office or agency of the Bank or any Paying Agent.

(d) If the principal, premium, if any, and interest with respect to this Note pursuant to the terms of this Note is payable in a Specified Currency and such Specified Currency has been replaced by another currency (a "*Replacement Currency*"), the Bank will be entitled to satisfy its obligations to the holder of this Note by making such payments of principal, premium, if any, and interest with respect to this Note, at the option of the Bank, in the Replacement Currency or in U.S. dollars, at a rate of exchange which takes into account the conversion, at the rate prevailing on the most recent date on which official conversion rates were quoted or set by the national government or other authority responsible for issuing the Replacement Currency, from the Specified Currency to the Replacement Currency and, if necessary, the conversion of the Replacement Currency into U.S. Dollars at the rate prevailing on the date of such conversion.

(e) If the principal, premium, if any, and interest with respect to this Note pursuant to the terms of this Note is payable in a Specified Currency other than the U.S. dollar and such Specified Currency is not available due to the imposition of exchange controls or other circumstances beyond the control of the Bank, the Bank will be entitled to satisfy its obligations to the holder of this Note by making such payments of principal, premium, if any, and interest

with respect to this Note in U.S. dollars (until, in the sole discretion of the Bank, the Specified Currency is again available) on the basis of the most recently available bid quotation from a leading foreign exchange bank in London or New York City selected by the Exchange Rate Agent, for the purchase of U.S. dollars with the Specified Currency for settlement on such payment date of the aggregate amount of the Specified Currency payable to all holders of Notes denominated other than in the U.S. dollar scheduled to receive U.S. dollar payments. Any payment made under such circumstances in U.S. dollars, where the payment is required to be made in the Specified Currency, will not constitute an "Event of Default" with respect to this Note.

(f) If the Specified Currency for this Notes is a domestic currency of a state that is or subsequent to the Original Issue Date of this Note becomes a European Union member state and "Redenomination" is specified as applicable in the Pricing Supplement, and, subsequent to the Original Issue Date, such state participates in the European monetary union, this Note may be redenominated in euro. In addition, this Note may be redenominated as a matter of law whether or not the Pricing Supplement provides for redenomination.

(g) If this Note is in registered form, this Note may be presented or surrendered for payment, and notices, designations or requests in respect of payments with respect to this Note may be served, at the office or agency of any Paying Agent maintained for that purpose. The Bank may at any time rescind any designation of a Paying Agent, appoint any additional or successor Paying Agents or approve a change in the office through which a Paying Agent acts.

(h) If this Note is in bearer form, this Note and any related receipts and coupons shall become void to the extent not presented for payment within a period of two years after the date on which a payment of principal or interest shall have become due and payable and monies sufficient therefor shall have been made available for payment.

3. Interest Rate Provisions.

(a) The Bank will pay interest on each Interest Payment Date specified in the Pricing Supplement and on the Maturity Date or any Redemption Date (as defined below), if any, or Holder's Optional Repayment Date (as defined below), if any (each such Maturity Date, Redemption Date and Holder's Optional Repayment Date and the date on which the principal or an installment of principal is due and payable by declaration of acceleration as provided herein being hereinafter referred to as a "Maturity" with respect to the principal repayable on such date), commencing, unless otherwise specified in the Pricing Supplement, on the first Interest Payment Date next succeeding the Original Issue Date specified on the face hereof (or if this is a registered Note, if the Original Issue Date is between a Record Date and the Interest Payment Date immediately following such Record Date, on the Second Interest Payment Date following the Original Issue Date) at the rate specified in the Pricing Supplement, until the principal hereof is paid or duly made available for payment.

(b) Payments of interest hereon will include interest accrued from and including the most recent Interest Payment Date to which interest on this Note (or any predecessor Note) has been paid or duly provided for (or, if no interest has been paid or duly provided for, from and

including the Original Issue Date) to but excluding the relevant Interest Payment Date or Maturity, as the case may be (each such period, an "Interest Period").

(c) Fixed Rate Interest Provisions

If this Note is designated as a "Fixed Rate Note" on the face hereof:

(i) Unless otherwise specified in the Pricing Supplement, if this Note is denominated in U.S. dollars and (1) the Maturity Date specified on the face hereof falls more than one year from the Original Issue Date, then interest payments for this Note shall be computed and paid on the basis of a 360-day year of twelve 30-day months, or (2) the Maturity Date specified on the face hereof falls one year or less from the Original Issue Date, then interest payments for this Note shall be computed and paid on the basis of the actual number of days in the Interest Period divided by 360.

(ii) Unless otherwise specified in the Pricing Supplement, interest on Fixed Rate Notes denominated other than in U.S. dollars will be computed on the basis of the "Actual/Actual (ISMA)" Fixed Day Count Convention.

"Actual/Actual (ISMA)" Fixed Day Count Convention means:

(A) in the case of Fixed Rate Notes where the number of days in the relevant period from and including the most recent Interest Payment Date (or, if none, from and including the Interest Commencement Date, which unless otherwise specified in the Pricing Supplement shall be the Original Issue Date) to but excluding the relevant payment date (the "Accrual Period") is equal to or shorter than the Determination Period (as defined below) during which the Accrual Period ends, the number of days in such Accrual Period divided by the product of (1) the number of days in such Determination Period and (2) the number of Determination Periods that would occur in one calendar year, assuming interest were payable in respect of the whole of that year; or

(B) in the case of Fixed Rate Notes where the Accrual Period is longer than the Determination Period during which the Accrual Period ends, the sum of:

(1) the number of days in such Accrual Period falling in the Determination Period in which the Accrual Period begins divided by the product of (x) the number of days in such Determination Period and (y) the number of Determination Periods that would occur in one calendar year, assuming interest were payable in respect of the whole of that year; and

(2) the number of days in such Accrual Period falling in the next Determination Period divided by the product of (x) the number of days in such Determination Period and (y) the number of Determination Periods that would occur in one calendar year, assuming interest were payable in respect of the whole of that year.

"Determination Period" means the period from and including an Interest Determination Date to but excluding the next Interest Determination Date (including, where either the Interest Commencement Date or the final Interest Payment Date is not an Interest Determination Date, the period commencing on the first Interest Determination Date prior to, and ending on the first Interest Determination Date falling after, such date.

"Interest Determination Date" means each date specified in the Pricing Supplement or, if none is specified, each Interest Payment Date.

(iii) Unless otherwise provided herein or in the Pricing Supplement, if any Interest Payment Date or the Maturity of this Note falls on a day which is not a Business Day, the related payment of principal, premium, if any, and interest with respect to this Note shall be made on the next succeeding Business Day with the same force and effect as if made on the date such payments were due, and no interest shall accrue on the amount so payable for the period from and after such Interest Payment Date or the Maturity, as the case may be.

(d) Floating Rate Interest Provisions

If this Note is designated as a "Floating Rate Note" on the face hereof:

(i) Unless otherwise specified herein or in the Pricing Supplement, if any Interest Payment Date (or other date which is subject to adjustment in accordance with a Business Day Convention specified in the Pricing Supplement) in respect of this Note (other than an Interest Payment Date at Maturity) would otherwise fall on a day that is not a Business Day, then, if the Business Day Convention specified in the Pricing Supplement is:

(A) the "Floating Rate Convention," such Interest Payment Date (or other date) shall be postponed to the next succeeding Business Day unless it would thereby fall into the next calendar month, in which event (A) such Interest Payment Date (or other date) shall be brought forward to the immediately preceding Business Day and (B) each subsequent Interest Payment Date (or other date) shall be the last Business Day in the month which falls the number of months or other period specified as the Interest Payment Period in the Pricing Supplement after the preceding applicable Interest Payment Date (or other date) occurred; or

(B) the "Following Business Day Convention," such Interest Payment Date (or other date) shall be postponed to the next succeeding Business Day; or

(C) the "Modified Following Business Day Convention," such Interest Payment Date (or other date) shall be postponed to the next succeeding Business Day unless it would thereby fall into the next calendar month, in which event such Interest Payment Date (or other date) shall be brought forward to the immediately preceding Business Day; or

(D) the "Preceding Business Day Convention," such Interest Payment Date (or other date) shall be brought forward to the immediately preceding Business Day.

If the Maturity of this Note falls on a day that is not a Business Day, the related payment of principal, premium, if any, and interest with respect to this Note will be made on the next succeeding Business Day with the same force and effect as if made on the date such payment was due, and no interest shall accrue on the amount so payable for the period from and after such Maturity.

(ii) If "<u>ISDA Rate</u>" is specified in the Pricing Supplement in connection with the determination of the rate of interest on this Note, the rate of interest on this Note for each Interest Period will be the relevant ISDA Rate (as defined below) plus or minus the Margin, if any, specified in the Pricing Supplement. Unless otherwise specified in the Pricing Supplement, "ISDA Rate" means, with respect to any Interest Period, the rate equal to the Floating Rate that would be determined by the Calculation Agent or other person specified in the Pricing Supplement pursuant to an interest rate swap transaction if the Calculation Agent or that other person were acting as calculation agent for that swap transaction in accordance with the terms of an agreement in the form of the Interest Rate and Currency Exchange Agreement published by the International Swaps and Derivatives Association, Inc. ("*ISDA*") (the "*ISDA Agreement*") and evidenced by a Confirmation (as defined in the ISDA Agreement) incorporating the 2000 ISDA Definitions as in effect on the Original Issue Date and under which:

(A) the Floating Rate Option is as specified in the Pricing Supplement;

(B) the Designated Maturity is the period specified in the Pricing Supplement;

(C) the relevant Reset Date is either (i) if the applicable Floating Rate Option is based on the London inter-bank offered rate ("LIBOR") for a currency or on the Euro-zone inter-bank offered rate for a currency ("EURIBOR"), the first day of that Interest Period or (ii) in any other case, as specified in the Pricing Supplement;

(D) the Interest Commencement Date is the Effective Date; and

(E) all other terms are as specified in the Pricing Supplement.

As used in this paragraph, "Floating Rate," "Floating Rate Option," "Designated Maturity," "Reset Date" and "Effective Date" have the meanings ascribed to those terms in the 2000 ISDA Definitions.

(iii) If "<u>Reference Rate Determination</u>" is specified in the Pricing Supplement in connection with the determination of the rate of interest on this Note:

(A) This Note will bear interest at a rate per annum equal to the Initial Interest Rate specified in the Pricing Supplement until the Initial Interest Reset

Date specified in the Pricing Supplement and thereafter at a rate per annum determined as follows:

1. If this Note is designated as a "*Regular Floating Rate Note*" in the Pricing Supplement or if no designation is made for Interest Calculation in the Pricing Supplement, then, except as described below (or in the Pricing Supplement), this Note shall bear interest at the rate determined by reference to the applicable Interest Rate Basis or Bases specified in the Pricing Supplement (i) plus or minus the applicable Spread, if any, and/or (ii) multiplied by the applicable Spread Multiplier, if any, specified and applied in the manner described in the Pricing Supplement. Commencing on the Initial Interest Reset Date specified in the Pricing Supplement, the rate at which interest on this Note shall be payable shall be reset as of each Interest Reset Date specified in the Pricing Supplement; *provided, however*, that the interest rate in effect for the period from and including the Original Issue Date to but excluding the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified in the Initial Interest Reset Date specified for the period from and including the Original Issue Date to but excluding the Initial Interest Reset Date will be the Initial Interest Reset Date specified in the Initial Interest Reset Date will be the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified in the Initial Interest Reset Date specified in the Initial Interest Reset Date will be the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date will be the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date will be the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date will be the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified to but excluding the Initial Interest Reset Date specified to but e

2. If this Note is designated as a "*Floating Rate/Fixed Rate Note*" in the Pricing Supplement, then, except as described below or in the Pricing Supplement, this Note shall bear interest at the rate determined by reference to the applicable Interest Rate Basis or Bases specified in the Pricing Supplement (i) plus or minus the applicable Spread, if any, and/or (ii) multiplied by the applicable Spread Multiplier, if any, specified and applied in the manner described in the Pricing Supplement. Commencing on the Initial Interest Reset Date, the rate at which interest on this Note shall be payable shall be reset as of each Interest Reset Date specified in the Pricing Supplement; *provided, however*, that (i) the interest rate in effect for the period from the Original Issue Date to but excluding the Initial Interest Reset Date shall be the Initial Interest Rate and (ii) the interest rate in effect commencing on the Fixed Rate Commencement Date specified in the Pricing Supplement to but excluding Maturity shall be the Fixed Interest Rate, if such a rate is specified in the Pricing Supplement, or if no such Fixed Interest Rate is so specified, the interest rate in effect hereon on the Business Day immediately preceding the Fixed Rate Commencement Date.

3. If this Note is designated as an "*Inverse Floating Rate Note*" in the Pricing Supplement, then, except as described below or in the Pricing Supplement, this Note shall bear interest equal to the Fixed Interest Rate specified in the Pricing Supplement minus the rate determined by reference to the applicable Interest Rate Basis or Bases specified in the Pricing Supplement (i) plus or minus the applicable Spread, if any, and/or (ii) multiplied by the applicable Spread Multiplier, if any, specified and applied in the manner described in the Pricing Supplement; *provided, however*, that, unless otherwise specified in the Pricing Supplement, the interest rate hereon will not be less than zero percent. Commencing on the Initial Interest Reset Date, the rate at which interest on this Note shall be

payable shall be reset as of each Interest Rate Reset Date specified in the Pricing Supplement; *provided, however*, that the interest rate in effect for the period from and including the Original Issue Date to but excluding the Initial Interest Reset Date shall be the Initial Interest Rate.

Except as provided above, if "Reference Rate Determination" is specified in the Pricing Supplement in connection with the determination of the rate of interest on this Note, the interest rate in effect on each day in an Interest Reset Period shall be the rate determined as of the Interest Determination Date immediately preceding the Interest Reset Date on which such Interest Reset Period commenced. Each Interest Rate Basis shall be the rate determined in accordance with the applicable provision below. If any Interest Reset Date (which term includes the term Initial Interest Reset Date unless the context otherwise requires) would otherwise be a day that is not a Business Day, such Interest Reset Date shall be adjusted in accordance with the Business Day Convention specified in the Pricing Supplement.

(B) Unless otherwise specified in the Pricing Supplement, the "Interest Determination Date":

(1) with respect to the CD Rate, the CMT Rate, the Commercial Paper Rate, the Federal Funds Rate and the Prime Rate shall be the second Business Day preceding each Interest Reset Date;

(2) with respect to EURIBOR shall be the second day on which the Trans-European Automated Real-Time Gross Settlement Express Transfer System (or any successor thereto) is open immediately preceding each Interest Reset Date;

(3) with respect to LIBOR shall be the second London Banking Day (as defined below) preceding each Interest Reset Date;

(4) with respect to the Treasury Rate shall be the day in the week in which the related Interest Reset Date falls on which day Treasury Bills (as defined below) are normally auctioned (Treasury Bills are normally sold at auction on Monday of each week, unless that day is a legal holiday, in which case the auction is normally held on the following Tuesday, except that such auction may be held on the preceding Friday; *provided,however*, that if an auction is held on the Friday of the week preceding the related Interest Reset Date, the related Interest Determination Date shall be such preceding Friday; and *provided, further*, that if an auction shall fall on any Interest Reset Date, then the Interest Reset Date shall instead be the first Business Day following such auction.

(5) if the interest rate of this Note is determined with reference to two or more Interest Rate Bases as specified in the Pricing Supplement, shall be the latest Business Day which is at least two Business Days prior

to such Interest Reset Date on which each Interest Rate Basis is determinable.

Each Interest Rate Basis shall be determined on such date, and the applicable interest rate shall take effect on the Interest Reset Date.

(C) Unless otherwise specified in the Pricing Supplement, the "*Calculation Date*" pertaining to any Interest Determination Date will be the earlier of (i) the tenth calendar day after such Interest Determination Date or, if such day is not a Business Day, the next succeeding Business Day and (ii) the Business Day immediately preceding the applicable Interest Payment Date or Maturity Date, as the case may be. All calculations in respect of determining the interest rate applicable to this Note (other than any calculations made by the Exchange Rate Agent) shall be made by the Calculation Agent specified, in the Pricing Supplement or such successor thereto as is duly appointed by the Bank. The determination of any interest rate by the Calculation Agent shall, in the absence of manifest error, be final and conclusive for all purposes and binding upon the holder hereof.

(D) *Determination of CD Rate*. If an Interest Rate Basis for this Note is the CD Rate, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the CD Rate shall be determined as of the applicable Interest Determination Date (a "*CD Rate Interest Determination Date*") as:

(1) the rate on that CD Rate Interest Determination Date for negotiable United States dollar certificates of deposit having the Index Maturity specified in the Pricing Supplement as published in H.15(519) (as hereinafter defined) under the heading "CDs (secondary market);" or

(2) if the rate referred to in clause (1) above is not so published in H.15(519) prior to 3:00 P.M., New York City time, on the related Calculation Date, then the CD Rate shall be the rate on that CD Rate Interest Determination Date for negotiable United States dollar certificates of deposit having the Index Maturity specified in the Pricing Supplement as published in H.15 Daily Update (as hereinafter defined), or another recognized electronic source used for the purpose of displaying that rate, under the heading "CDs (secondary market);" or

(3) if the rate referred to in clauses (1) and (2) above is not yet so published in H.15(519), H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, then the CD Rate for that CD Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of the secondary market offered rates, as of 10:00 A.M., New York City time, on that CD Rate Interest Determination Date, of three leading nonbank dealers of negotiable United States dollar certificates of deposit in The City of New York (which may include the Distribution Agents and any of their respective affiliates) selected by the

Calculation Agent for negotiable United States dollar certificates of deposit of major United States money market banks for negotiable certificates of deposit with a remaining maturity closest to the Index Maturity specified in the Pricing Supplement in an amount that is representative for a single transaction in that market at that time; or

(4) if fewer than three dealers so selected by the Calculation Agent are quoting as described in clause (3) above, the CD Rate on that CD Rate Interest Determination Date shall be the CD Rate in effect immediately prior to that CD Rate Interest Determination Date.

"H.15(519)" means the weekly statistical release designated as such, or any successor publication, published by the Board of Governors of the Federal Reserve System (the "Federal Reserve Board").

"H.15 Daily Update" means the daily update of H.15(519), available through the web site of the Federal Reserve Board at http://www.federalreserve.gov/releases/h15/update, or any successor site or publication.

(E) <u>Determination of CMT Rate</u>. If an Interest Rate Basis for this Note is the CMT Rate, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the CMT Rate shall be determined by the Calculation Agent as of the applicable Interest Determination Date (a "CMT Rate Interest Determination Date") in accordance with the following provisions:

(1) If "CMT Moneyline Telerate Page 7051" is the specified CMT Moneyline Telerate Page in the Pricing Supplement:

(a) the CMT Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as published in H.15(519) under the heading "Treasury Constant Maturities," for such CMT Rate Interest Determination Date as such yield is displayed on Moneyline Telerate, Inc. (or any successor service) on page 7051 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 7051"); or

(b) if the rate referred to in subclause (a) above is not so displayed on Moneyline Telerate Page 7051, then the CMT Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as published in H.15(519) under the heading

"Treasury Constant Maturities" for such CMT Rate Interest Determination Date; or

(c) if the rate referred to in subclause (b) above is not so published in H.15(519), then the CMT Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the yield for United States Treasury Securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as may then be published by either the Federal Reserve Board or the United States Department of the Treasury that the Calculation Agent determines to be comparable to the rate which would otherwise have been published in H.15(519); or

(d) if neither the Federal Reserve Board nor the United States Department of the Treasury publishes a yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement for that CMT Rate Interest Determination Date as referred to in subclause (c) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be a yield-to-maturity based on the arithmetic mean of the secondary market bid prices at approximately 3:30 P.M., New York City time, on such CMT Rate Interest Determination Date of three leading primary United States government securities dealers in The City of New York (each, a "Reference Dealer") selected by the Calculation Agent (from five such Reference Dealers and eliminating the highest quotation (or, in the event of equality, one of the lowest)) for United States Treasury securities with an original maturity equal to the Index Maturity specified in the Pricing Supplement, a remaining term to maturity no more than 1 year shorter than such Index Maturity and in a principal amount that is representative for a single transaction in such securities in such market at such time; or

(e) if fewer than five but more than two such prices are provided as requested pursuant to subclause (d) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be based on the arithmetic mean of the bid prices obtained and neither the highest nor lowest of such quotations shall be eliminated; or

(f) if fewer than three prices are provided as requested pursuant to subclause (d) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be a yield-to-maturity based on the arithmetic mean of the secondary market bid prices as of approximately 3:30 P.M., New York City time, on that CMT Rate

Interest Determination Date of three Reference Dealers selected by the Calculation Agent (from five such Reference Dealers and eliminating the highest quotation (or, in the event of equality, one of the highest) and the lowest quotation (or, in the event of equality, one of the lowest)) for United States Treasury securities with an original maturity longer than the Index Maturity specified in the Pricing Supplement, a remaining term to maturity closest to such Index Maturity and in a principal amount that is representative for a single transaction in such securities in such market at such time; or

(g) if fewer than five but more than two such prices are provided as requested pursuant to subclause (f) above, then the CMT Rate on such CMT Rate Interest Determination Date shall be based on the arithmetic mean of the bid prices obtained and neither the highest nor the lowest of such quotations shall be eliminated; or

(h) if fewer than three such prices are provided as requested pursuant to subclause (f) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be the CMT Rate in effect immediately prior to that CMT Rate Interest Determination Date.

For purposes of subclause (f) above, if two such United States Treasury securities with an original maturity greater than the Index Maturity have remaining terms to maturity equally close to the Index Maturity, the quotes for the Treasury security with the shorter original term to maturity will be used.

(2) If "CMT Moneyline Telerate Page 7052" is the specified CMT Moneyline Telerate Page in Pricing Supplement:

(a) the CMT Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the one-week or one-month, as specified in the Pricing Supplement, average yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as published in H.15(519) opposite the heading "Treasury Constant Maturities," for the week or month, as applicable, ended immediately preceding the week or month, as applicable, in which such CMT Rate Interest Determination Date falls, as such yield is displayed on Moneyline Telerate, Inc. (or any successor service) on page 7052 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 7052"); or

(b) if the rate referred to in subclause (a) above is not so displayed on the Moneyline Telerate Page 7052, then the CMT

Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the one-week or one-month, as applicable, average yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as published in H.15(519) opposite the heading "Treasury Constant Maturities," for the week or month, as applicable, ending immediately preceding the week or month, as applicable, in which such CMT Rate Interest Determination Date falls; or

(c) if the rate referred to in subclause (b) above is not so published in H.15(519), then the CMT Rate on that CMT Rate Interest Determination Date shall be a percentage equal to the one-week or one-month, as applicable, average yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement as may then be published by the Federal Reserve Bank of New York for the week or month, as applicable, ended immediately preceding the week or month, as applicable, in which such CMT Rate Interest Determination Date falls; or

(d) if the Federal Reserve Bank of New York does not publish a one-week or one-month, as applicable, average yield for United States Treasury securities at "constant maturity" having the Index Maturity specified in the Pricing Supplement for the applicable week or month as referred to in subclause (c) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be a yield-to-maturity based on the arithmetic mean of the secondary market bid prices at approximately 3:30 P.M., New York City time, on such CMT Rate Interest Determination Date of three Reference Dealers selected by the Calculation Agent (from five such Reference Dealers and eliminating the highest quotation (or, in the event of equality, one of the highest) and the lowest quotation (or, in the event of equality, one of the lowest)) for United States Treasury securities with an original maturity equal to the Index Maturity specified in the Pricing Supplement, a remaining term to maturity of no more than 1 year shorter than such Index Maturity and in a principal amount that is representative for a single transaction in such securities in such market at such time; or

(e) if fewer than five but more than two such prices are provided as requested pursuant to subclause (d) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be based on the arithmetic mean of the bid prices obtained and neither the highest nor lowest of such quotations shall be eliminated; or

(f) if fewer than three prices are provided as requested pursuant to subclause (d) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be a yield-to-maturity based on the arithmetic mean of the secondary market bid prices as of approximately 3:30 P.M., New York City time, on such CMT Rate Interest Determination Date of three Reference Dealers selected by the Calculation Agent (from five such Reference Dealers and eliminating the highest quotation (or, in the event of equality, one of the highest) and the lowest quotation (or, in the event of equality, one of the lowest)) for United States Treasury securities with an original maturity longer than the Index Maturity specified in the Pricing Supplement, a remaining term to maturity closest to such Index Maturity and in a principal amount that is representative for a single transaction in such securities in such market at such time;

(g) if fewer than five but more than two such prices are provided as requested pursuant to subclause (f) above, then the CMT Rate on that CMT Rate Interest Determination Date shall be based on the arithmetic mean of the bid prices obtained and neither the highest nor lowest of such quotations shall be eliminated; or

(h) if fewer than three such prices are provided as requested pursuant to subclause (f) above, the CMT Rate on that CMT Rate Interest Determination Date shall be the CMT Rate in effect immediately prior to that CMT Rate Interest Determination Date.

For purposes of subclause (f) above, if two United States Treasury securities with an original maturity greater than the Index Maturity have remaining terms to maturity equally close to the Index Maturity, the quotes for the Treasury security with the shorter original term to maturity will be used.

(F) <u>Determination of Commercial Paper Rate</u>. If an Interest Rate Basis for this Note is the Commercial Paper Rate, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the Commercial Paper Rate shall be determined as of the applicable Interest Determination Date (a "Commercial Paper Rate Interest Determination Date") as:

(1) the Money Market Yield (as defined hereinafter) on that Commercial Paper Rate the Money Market Yield (as defined hereinafter) on that Commercial Paper Rate Interest Determination Date of the rate for commercial paper having the Index Maturity specified in the Pricing Supplement as published in H.15(519) under the caption "Commercial Paper-Nonfinancial;" or

(2) if the rate referred to in clause (1) above is not so published in H.15(519) by 3:00 P.M., New York City time, on the related Calculation Date, then the Commercial Paper Rate shall be the Money Market Yield on that Commercial Paper Rate Interest Determination Date of the rate for commercial paper having the Index Maturity specified in the Pricing Supplement as published in H.15 Daily Update, or another recognized electronic source used for the purpose of displaying such rate, under the heading "Commercial Paper-Nonfinancial;" or

(3) if the rate referred to in clauses (1) and (2) above is not yet so published in H.15(519), H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, then the Commercial Paper Rate on that Commercial Paper Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be the Money Market Yield of the arithmetic mean of the offered rates at approximately 11:00 A.M., New York City time, on such Commercial Paper Rate Interest Determination Date of three leading dealers of United States dollar commercial paper in The City of New York (which may include the Distribution Agents and any of their respective affiliates) selected by the Calculation Agent for commercial paper having the Index Maturity specified in the Pricing Supplement placed for industrial issuers whose bond rating is "Aa", or the equivalent, from a nationally recognized statistical rating organization; or

(4) if fewer than three dealers so elected by the Calculation Agent are quoting as described in clause (3) above, the Commercial Paper Rate on that Commercial Paper Rate Interest Determination Date shall be the Commercial Paper Rate in effect immediately prior to that Commercial Paper Rate Interest Determination Date.

"Money Market Yield" means a yield (expressed as a percentage) calculated in accordance with the following formula:

Money Market Yield =

 $\frac{D \times 360}{360 - (D \times M)} \times 100$

where "D" refers to the applicable per annum rate for commercial paper quoted on a bank discount basis and expressed as a decimal, and "M" refers to the actual number of days in the applicable Interest Reset Period.

(G) <u>Determination of EURIBOR</u>. If an Interest Rate Basis for this Note is EURIBOR, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, EURIBOR shall be determined by the Calculation Agent as of the applicable Interest Determination Date (a "EURIBOR Interest Determination Date"), as:

(1) the rate for deposits in euro as sponsored, calculated and published jointly by the European Banking Federation and ACI — The Financial Market Association, or any company established by the joint sponsors for purposes of compiling and publishing those rates, having the Index Maturity specified in the Pricing Supplement, commencing on the applicable Interest Reset Date, as that rate is displayed on Moneyline Telerate, Inc., or any successor service, on page 248 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 248") as of 11:00 A.M., Brussels time, on that EURIBOR Interest Determination Date; or

(2) if the rate referred to in clause (1) above is not so published or is not displayed on Moneyline Telerate Page 248, by 11:00 A.M., Brussels time, on that EURIBOR Interest Determination Date, then EURIBOR for that EURIBOR Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of at least two quotations obtained by the Calculation Agent after requesting the principal Euro-zone (as defined hereinafter) offices of four major banks in the Euro-zone interbank market, in the European interbank market, to provide the Calculation Agent with its offered quotation for deposits in euro for the period of the Index Maturity specified in the Pricing Supplement, commencing on the applicable Interest Reset Date, to prime banks in the Euro-zone interbank market at approximately 11:00 A.M., Brussels time, on that EURIBOR Interest Determination Date and in a principal amount not less than the equivalent of U.S.\$1,000,000 in euros that is representative for a single transaction in euro in such market at such time; or

(3) if fewer than two quotations referred to in clause (2) above are so provided, then EURIBOR for that EURIBOR Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of the rates quoted at approximately 11:00 A.M., Brussels time, on that EURIBOR Interest Determination Date by four major banks in the Euro-zone selected by the Calculation Agent for loans to leading European banks in euro having the Index Maturity specified in the Pricing Supplement, commencing on the applicable Interest Reset Date and in a principal amount not less than the equivalent of U.S.\$1,000,000 in euros that is representative for a single transaction in euro in such market at such time; or

(4) if fewer than four banks so selected by the Calculation Agent are quoting as described in clause (3) above, EURIBOR on that EURIBOR Interest Determination Date shall be EURIBOR in effect immediately prior to that EURIBOR Interest Determination Date.

"Euro-zone" means the region comprised of member states that adopt the single currency in accordance with the Treaty on European Union of 1992, as amended.

(H) <u>Determination of Federal Funds Rate</u>. If an Interest Rate Basis for this Note is the Federal Funds Rate, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the Federal Funds Rate shall be determined as of the applicable Interest Determination Date (a "Federal Funds Rate Interest Determination Date") as:

(1) the rate on that Federal Funds Rate Interest Determination Date for United States dollar federal funds as published in H.15(519) under the heading "Federal Funds (Effective)," as such rate is displayed on Moneyline Telerate, Inc. (or any successor service) on page 120 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 120"); or

(2) if the rate referred to in clause (1) is not so published in H.15(519) or is not displayed on Moneyline Telerate Page 120 by 3:00 P.M., New York City time, on the related Calculation Date, then the Federal Funds Rate shall be the rate on that Federal Funds Rate Interest Determination Date for United States dollar federal funds as published in H.15 Daily Update, or another recognized electronic source used for the purpose of displaying such rate, under the caption "Federal Funds (Effective);" or

(3) if the rate referred to in clauses (1) and (2) is not displayed on Moneyline Telerate Page 120 or is not yet so published in H.15(519), H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, then the Federal Funds Rate on that Federal Funds Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of the rates for the last transaction in overnight United States dollar federal funds arranged by three leading brokers of United States dollar federal funds transactions in The City of New York (which may include the Distribution Agents and any of their respective affiliates) selected by the Calculation Agent, prior to 9:00 A.M., New York City time, on such Federal Funds Rate Interest Determination Date; or

(4) if fewer than three brokers so selected by the Calculation Agent are quoting as described in clause (3) above, the Federal Funds Rate on that Federal Funds Rate Interest Determination Date shall be the Federal Funds Rate in effect immediately prior to that Federal Funds Rate Interest Determination Date.

(I) **Determination of LIBOR**. If an Interest Rate Basis for this Note is LIBOR, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, LIBOR shall be determined by the Calculation Agent as of the applicable Interest Determination Date (a *"LIBOR Interest Determination Date"*), as:

(1) either (a) if "LIBOR Moneyline Telerate" is specified in the Pricing Supplement or if neither "LIBOR Reuters" nor "LIBOR Moneyline Telerate" is specified in the Pricing Supplement as the method for calculating LIBOR, the rate for deposits in the Designated LIBOR Currency having the Index Maturity specified in such Pricing Supplement, commencing on the applicable Interest Reset Date, that appears on the Designated LIBOR Page (as defined hereinafter) as of 11:00 A.M., London time, on such LIBOR Interest Determination Date; or (b) if "LIBOR Reuters" is specified in the Pricing Supplement, then LIBOR shall be calculated by the Calculation Agent and shall be the arithmetic mean of the offered rates (unless the Designated LIBOR Page (as defined hereinafter) by its terms provides only for a single rate, in which case such single rate shall be used) for deposits in the Designated LIBOR Currency having the Index Maturity specified in such Pricing Supplement, commencing on the applicable Interest Reset Date, that appear (or, if only a single rate is required as aforesaid, appears) on the Designated LIBOR Page as of 11:00 A.M., London time, on such LIBOR Interest Determination Date; or

(2) if fewer than two offered rates appear, or no rate appears, as the case may be, on the Designated LIBOR Page as required pursuant to clause (1) above, then the Calculation Agent shall request the principal London offices of each of four major reference banks (which may include the Distribution Agents and any of their respective affiliates) in the London interbank market, as selected by the Calculation Agent, to provide the Calculation Agent with its offered quotation for deposits in the Designated LIBOR Currency for the period of the Index Maturity specified in the Pricing Supplement, commencing on the applicable Interest Reset Date, to prime banks in the London interbank market at approximately 11:00 A.M., London time, on such LIBOR Interest Determination Date and in a principal amount that is representative for a single transaction in the Designated LIBOR Currency in such market at such time. If at least two such quotations are so provided, then LIBOR on such LIBOR Interest Determination Date shall be the arithmetic mean of such quotations calculated by the Calculation Agent; or

(3) if fewer than two quotations referred to in clause (2) above are so provided, then LIBOR on such LIBOR Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of the rates quoted at approximately 11:00 A.M., in the applicable Principal Financial Center (as defined hereinafter), on such LIBOR Interest Determination Date by three major banks (which may include the Distribution Agents and any of their respective affiliates) in such Principal Financial Center selected by the Calculation Agent for loans to leading European banks in the Designated LIBOR Currency having the Index Maturity specified in the Pricing Supplement and in a principal amount

that is representative for a single transaction in the Designated LIBOR Currency in such market at such time; or

(4) if fewer than three banks so selected by the Calculation Agent are quoting as described in clause (3) above, LIBOR on that LIBOR Interest Determination Date shall be LIBOR in effect immediately prior to that LIBOR Interest Determination Date.

"Designated LIBOR Currency" means the currency specified in the Pricing Supplement as to which LIBOR shall be calculated or, if no such currency is specified in the Pricing Supplement, United States dollars.

"Designated LIBOR Page" means (a) if "LIBOR Reuters" is specified in the Pricing Supplement, the display on the Reuters Monitor Money Rates Service (or any successor service) on the page specified in the Pricing Supplement (or any other page as may replace such page on such service) for the purpose of displaying the London interbank rates of major banks for the Designated LIBOR Currency, or (b) if "LIBOR Moneyline Telerate" is specified in the Pricing Supplement or neither "LIBOR Reuters" nor "LIBOR Moneyline Telerate" is specified in the Pricing Supplement as the method for calculating LIBOR, the display on Moneyline Telerate, Inc. (or any successor service) on the page specified in the Pricing Supplement (or any other page as may replace such page on such service) for the purpose of displaying the London interbank rates of major banks for the Designated LIBOR Currency.

"Principal Financial Center" means (i) the capital city of the country issuing the specified currency or (ii) the capital city of the country to which the Designated LIBOR Currency, if applicable, relates, except, in each case (i) and (ii), that (x) with respect to United States dollars, Australian dollars, Canadian dollars, New Zealand dollars, South African rand and Swiss francs, the "Principal Financial Center" shall be The City of New York, Sydney, Toronto, Wellington, Johannesburg and Zurich, respectively and (y) with respect to euros, the Principal Financial Center may be any principal financial center in the Euro-zone, as selected by the Calculation Agent.

"London Banking Day" means any day (other than a Saturday or Sunday) on which dealings in deposits in the Designated LIBOR Currency are transacted in the London interbank market.

(J) <u>Determination of Prime Rate</u>. If an Interest Rate Basis for this Note is the Prime Rate, as specified in the Pricing Supplement, unless the Pricing Supplement specifies otherwise, the Prime Rate shall be determined as of the applicable Interest Determination Date (a "*Prime Rate Interest Determination Date*") as:

(1) the rate on that Prime Rate Interest Determination Date as such rate is published in H.15(519) under the heading "Bank Prime Loan;" or

(2) if the rate referred to in clause (1) above is not so published in H.15(519) by 3:00 P.M., New York City time, on the related Calculation Date, then the Prime Rate shall be the rate on that Prime Rate Interest Determination Date as such rate is published in H.15 Daily Update, or another recognized electronic source used for the purpose of displaying such rate, under the heading "Bank Prime Loan;" or

(3) if the rate referred to in clauses (1) and (2) above is not yet so published in H.15(519), H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, the Prime Rate for that Prime Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be the arithmetic mean of the rates of interest publicly announced by each bank that appears on the Reuters Screen US PRIME 1 Page (as defined hereinafter) as such bank's prime rate or base lending rate as of 11:00 A.M., New York City time, on that Prime Rate Interest Determination Date; or

(4) if fewer than four such rates referred to in clause (3) above so appear on the Reuters Screen US PRIME 1 Page for that Prime Rate Interest Determination Date, by 3:00 p.m., New York City time, on the related Calculation Date, the Prime Rate shall be calculated by the Calculation Agent and shall be the arithmetic mean of the prime rates or base lending rates quoted on the basis of the actual number of days in the year divided by a 360-day year by three major banks (which may include any affiliate of any Distribution Agent) in The City of New York selected by the Calculation Agent as of the close of business on such Prime Rate Interest Determination Date; or

(5) if less than three banks so selected by the Calculation Agent are quoting as described in clause (4) above, the Prime Rate on that Prime Rate Interest Determination Date shall be the Prime Rate in effect on such Prime Rate Interest Determination Date.

"Reuters Screen US PRIME 1 Page" means the display on the Reuters Monitor Money Rates Service (or any successor service) on the "US PRIME 1 Page" (or such other page as may replace the US PRIME 1 page on such service) for the purpose of displaying prime rates or base lending rates of major United States banks.

(K) <u>Determination of Treasury Rate</u>. If an Interest Rate Basis for this Note is the Treasury Rate, as specified in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the Treasury Rate shall be determined as of the applicable Interest Determination Date (a "*Treasury Rate Interest Determination Date*") as:

(1) the rate from the auction held on such Treasury Rate Interest Determination Date (the "Auction") of direct obligations of the United States ("Treasury Bills") having the Index Maturity specified in the Pricing Supplement under the caption "INVESTMENT RATE" on the display on Moneyline Telerate, Inc. (or any successor service) on page 56 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 56") or page 57 (or any other page as may replace such page on such service) ("Moneyline Telerate Page 57"); or

(2) if the rate referred to in clause (1) above is not so displayed on Moneyline Telerate Page 56 or Moneyline Telerate Page 57 by 3:00 P.M., New York City time, on the related Calculation Date, then the Treasury Rate on that Treasury Rate Interest Determination Date shall be the Bond Equivalent Yield (as defined hereinafter) of the auction rate for such Treasury Bills having the Index Maturity specified in the Pricing Supplement as published in H.15 Daily Update, or another recognized electronic source used for the purpose of displaying such rate, under the heading "U.S. Government Securities/Treasury Bills/Auction High;" or

(3) if the rate referred to in clause (2) above is not so published in H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, then the Treasury Rate on that Treasury Rate Interest Determination Date shall be the Bond Equivalent Yield of the auction rate of such Treasury Bills having the Index Maturity specified in the Pricing Supplement as announced by the United States Department of the Treasury; or

(4) if the rate referred to in clause (3) above is not so announced by the United States Department of the Treasury on such Calculation Date or if no such Auction is held, then the Treasury Rate on that Treasury Rate Interest Determination Date shall be the Bond Equivalent Yield of the rate on that Treasury Rate Interest Determination Date of Treasury Bills having the Index Maturity specified in the Pricing Supplement as published in H.15(519) under the heading "U.S. Government Securities/Treasury Bills/Secondary Market;" or

(5) if the rate referred to in clause (4) above is not yet so published in H.15(519) by 3:00 P.M., New York City time, on the related Calculation Date, then the Treasury Rate on that Treasury Rate Interest Determination Date shall be the rate on such Treasury Rate Interest Determination Date of such Treasury Bills having the Index Maturity specified in the Pricing Supplement as published in H.15 Daily Update, or such other recognized electronic source used for the purpose of displaying such rate, under the heading "U.S. Government Securities/Treasury Bills/Secondary Market;" or

(6) if the rate referred to in clause (4) and (5) above is not yet so published in H.15(519), H.15 Daily Update or another recognized electronic source by 3:00 P.M., New York City time, on the related Calculation Date, then the Treasury Rate on such Treasury Rate Interest Determination Date shall be calculated by the Calculation Agent and shall be the Bond Equivalent Yield of the arithmetic mean of the secondary market bid rates, as of approximately 3:30 P.M., New York City time, on such Treasury Rate Interest Determination Date, of three leading primary United States government securities dealers (which may include the Distribution Agents or any of their respective affiliates) selected by the Calculation Agent, for the issue of Treasury Bills with a remaining maturity closest to the Index Maturity specified in the Pricing Supplement; or

(7) if fewer than three dealers so selected by the Calculation Agent are quoting as described in clause (6) above, then the Treasury Rate on that Treasury Rate Interest Determination Date shall be the Treasury Rate in effect on such Treasury Rate Interest Determination Date."Bond Equivalent Yield" means a yield (expressed as a percentage) calculated in accordance with the following formula:

Bond Equivalent Yield = $D \ge N$ $360 - (D \ge M)$ X 100

where "D" refers to the applicable per annum rate for Treasury Bills quoted on a bank discount basis and expressed as a decimal, "N" refers to 365 or 366, as the case may be, and "M" refers to the actual number of days in the applicable Interest Reset Period.

(iv) Unless otherwise specified in the Pricing Supplement, if this Note is designated a "Floating Rate Note" on the face hereof, accrued interest hereon shall be an amount calculated by multiplying the face amount hereof by an accrued interest factor. Such accrued interest factor shall be computed by adding the interest factor calculated for each day in the period for which accrued interest is being calculated. Unless otherwise specified in the Pricing Supplement, the interest factor for each such day shall be computed on the basis of a 360-day year of twelve 30-day months if the Day Count Convention specified in the Pricing Supplement is "30/360" for the period specified thereunder, or by dividing the applicable per annum interest rate by 360 if the Day Count Convention specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/360" for the period specified in the Pricing Supplement is "Actual/Actual" for the period specified thereunder. If no Day Count Convention is specified in the Pricing Supplement, the interest factor for each day in the relevant Interest Period shall be computed, if an Interest Rate Basis specified in the Pricing Supplement is the CMT Rate or Treasury Rate or if this Note is denominated in Sterling, as if "Actual/Actual" had been specified thereon and, in all other cases, as if

"Actual/360" had been specified therein. Unless otherwise specified in the Pricing Supplement, if interest on this Note is to be calculated with reference to two or more Interest Rate Bases as specified in the Pricing Supplement, the interest factor will be calculated in each period in the same manner as if only one of the applicable Interest Rate Bases applied as specified in the Pricing Supplement.

(v) All percentages resulting from any calculation on this Note shall be rounded, if necessary, to the nearest one hundred-thousandth of a percentage point, with five one-millionths of a percentage point rounded upward (*e.g.*, 9.876545% (or 0.09876545) shall be rounded to 9.87655% (or 0.0987655%) and 9.876544% (or 0.09876544) shall be rounded to 987654% (or 0.09876544), and all dollar amounts used in or resulting from such calculation shall be rounded to the nearest cent or, if the Specified Currency is other than dollars, to the nearest unit (with one-half cent or unit being rounded upward).

(vi) At the request of the holder hereof, the Calculation Agent shall provide to the holder hereof the interest rate hereon then in effect and, if determined, the interest rate which shall become effective as a result of a determination made for the next Interest Period with respect to this Note.

(vii) Notwithstanding the foregoing, the interest rate hereon shall not be greater than the Maximum Interest Rate, if any, or less than the Minimum Interest Rate, if any, specified in the Pricing Supplement. In addition to any Maximum Interest Rate applicable hereto pursuant to the above provisions, the interest rate on this Note will in no event be higher than the maximum rate permitted by New York law, as the same may be modified by United States law of general application. Under present New York law, the maximum rate of interest is 25% per annum on a simple basis, with certain exceptions. The limit may not apply to Floating Rate Notes in which \$2,500,000 or more has been invested.

4. <u>Redemption Provisions</u>.

(a) <u>Redemption at the Option of the Bank</u>

(i) Unless otherwise specified in the Pricing Supplement, this Note will not be subject to any sinking fund.

(ii) This Note may be redeemed by the Bank either in whole or in part on and after the Initial Redemption Date, if any, specified in the Pricing Supplement. If no Initial Redemption Date is specified in the Pricing Supplement, this Note may not be redeemed prior to the Maturity Date except as provided below in the event that any Additional Amounts (as defined below) are required to be paid by the Bank with respect to this Note. On and after the Initial Redemption Date, if any, this Note may be redeemed in increments of US\$1,000 (or, if the Specified Currency of this Note is not U.S. dollars, the authorized integral multiple in excess of the minimum denomination specified in the Pricing Supplement or as otherwise provided in the Pricing Supplement or required by applicable laws and regulations for currencies other than the U.S. Dollar)

at the option of the Bank at the applicable Redemption Price (as defined below) together with unpaid interest accrued hereon at the applicable rate borne by this Note to the date of redemption (each such date, a "*Redemption Date*"), on written notice to the holder hereof given by or on behalf of the Bank not more than 60 and not less than 30 calendar days prior to the Redemption Date (unless otherwise specified in the Pricing Supplement); *provided*, *however*, that, in the event of redemption of this Note in part only, the unredeemed portion thereof shall be at least an Authorized Denomination specified in the Pricing Supplement. In the event of redemption of this Note in part only, either a new Note for the unredeemed portion hereof shall be issued in the name of the holder hereof upon the surrender of this Note, or, where applicable, the aggregate principal amount of this Note shall be reduced by the amount so redeemed by an appropriate notation made on the schedule attached hereto for such notations.

The "Redemption Price" shall initially be the Initial Redemption Percentage, specified in the Pricing Supplement, of the principal amount of this Note to be redeemed and shall decline at each anniversary of the Initial Redemption Date, specified in the Pricing Supplement, by the Annual Redemption Percentage Reduction, if any, specified in the Pricing Supplement, of the principal amount to be redeemed until the Redemption Price is 100% of such principal amount.

Unless otherwise specified in the Pricing Supplement, if less than the entire outstanding principal amount of this Note is to be redeemed, the Notes to be so redeemed shall be selected by the Registrar by lot or in any usual manner approved by it. Unless otherwise specified in the Pricing Supplement, if this is a Floating Rate Note, the redemption date shall be an Interest Payment Date.

(iii) Notwithstanding clause (ii) above, if this Note is a Subordinated Note, to the extent then required under or pursuant to applicable capital regulations, this Note may not be redeemed at the option of the Bank prior to the Maturity Date pursuant to clause (ii) above, without the prior written consent of the Federal Reserve Board. If then required under applicable capital regulations, the Bank shall apply to the Federal Reserve Board for such prior written consent.

(b) Repayment at the Option of the Holder

(i) This Note may be subject to repayment at the option of the holder hereof in accordance with the terms hereof on any Holder's Optional Repayment Date(s), if any, specified in the Pricing Supplement. If no Holder's Optional Repayment Date is specified in the Pricing Supplement, this Note will not be repayable at the option of the holder hereof prior to the Maturity Date.

(ii) On any Holder's Optional Repayment Date, if any, specified in the Pricing Supplement, this Note will be repayable in whole or in part in increments of US\$1,000 (or, if the Specified Currency of this Note is not U.S. dollars, the authorized integral multiple in excess of the minimum denomination specified in the Pricing Supplement or as otherwise provided in the Pricing Supplement or required by applicable laws and regulations for currencies other than the U.S. Dollar) at the option of the holder hereof at the repayment price equal to 100% of the principal amount to be repaid, together with

accrued and unpaid interest hereon payable to the date of repayment; *provided, however*, that, in the event of repayment of this Note in part only, the unrepaid portion hereof shall be at least an Authorized Denomination specified in the Pricing Supplement. For this Note to be repaid in whole or in part at the option of the holder hereof on a Holder's Optional Repayment Date, this Note must be received, together with the form entitled "Option to Elect Repayment" duly completed attached hereto, by the Domestic Paying Agent or the London Paying Agent (as appropriate in accordance with such attached form) at the address set forth on such form (or at such other address which the Bank shall from time to time notify the holders of the Notes) not more than 60 and not less than 30 days prior to such Holder's Optional Repayment Date. In the event of repayment of this Note in part only, either a new Note for the unrepaid portion hereof shall be issued in the name of the holder hereof upon the surrender hereof, or, where applicable, the aggregate principal amount of this Note shall be reduced by an appropriate notation made on the schedule attached hereto for such notations. Exercise of such repayment option by the holder hereof shall be irrevocable.

(iii) Notwithstanding clause (ii) above, if this Note is a Subordinated Note, to the extent then required under or pursuant to applicable capital regulations, this Note may not be repaid at the option of the holder pursuant to clause (ii) above, without the prior written consent of the Federal Reserve Board. If then required under applicable capital regulations, the Bank shall apply to the Federal Reserve Board for such prior written consent.

5. <u>Additional Amounts</u>.

(a) All payments of principal, premium, if any and interest with respect to this Note will be made without withholding or deduction at source for, or on account of, any present or future taxes, fees, duties, assessments or governmental charges of whatever nature imposed or levied by the United States or any political subdivision or taxing authority thereof or therein, unless such withholding or deduction is required by (i) the laws (or any regulations or rulings promulgated thereunder) of the United States or any political subdivision or taxing authority thereof or therein or (ii) an official position regarding the application, administration, interpretation or enforcement of any such laws, regulations or rulings including, without limitation, a holding by a court of competent jurisdiction or by a taxing authority in the United States or any political subdivision thereof). If a withholding or deduction at source is required, the Bank will, subject to certain limitations and exceptions (set forth below), pay to the holder hereof on behalf of an owner of a beneficial interest herein (an "*Owner*") who is a United States Alien (as defined below) such additional amounts ("*Additional Amounts*") as may be necessary so that every net payment of principal, premium, if any, or interest made to the holder hereof on behalf of such Owner, after such withholding or deduction, will not be less than the amount provided for in this Note; *provided*, *however*, that the Bank shall not be required to make any payment of Additional Amounts for or on account of:

(i) any tax, fee, duty, assessment or other governmental charge which would not have been imposed but for (1) the existence of any present or former connection between such Owner (or between a fiduciary, settlor, beneficiary, member or shareholder of, or possessor of a power over, such Owner, if such Owner is an estate, trust,

partnership or corporation) and the United States, including, without limitation, such Owner (or such fiduciary, settlor, beneficiary, member, shareholder or possessor) being or having been a citizen or resident thereof or being or having been present or engaged in trade or business therein or having or having had a permanent establishment therein, or (2) the presentation of this Note for payment on a date more than 15 days after the date on which such payment became due and payable or the date on which payment thereof is duly provided for, whichever occurs later;

(ii) any estate, inheritance, gift, sales, transfer, personal property or similar tax, assessment or other governmental charge;

(iii) any tax, fee, duty, assessment or other governmental charge imposed by reason of such Owner's past or present status as a personal holding company, foreign personal holding company, passive foreign investment company or controlled foreign corporation with respect to the United States or as a corporation which accumulates earnings to avoid United States federal income tax;

(iv) any tax, fee, duty, assessment or other governmental charge which is payable otherwise than by withholding from payments of principal or interest with respect to this Note;

(v) any tax, fee, duty, assessment or other governmental charge imposed on interest received by anyone who owns (actually or constructively) 10% or more of the total combined voting power of all classes of stock of the Bank;

(vi) any tax, fee, duty, assessment or other governmental charge required to be withheld by any Paying Agent from any payment of principal, premium, if any, or interest with respect to this Note, if such payment can be made without such withholding by any other Paying Agent with respect to this Note;

(vii) any tax, fee, duty, assessment or other governmental charge which would not have been imposed but for the failure to comply with certification, information or other reporting requirements concerning the nationality, residence, identity or connection with the United States of the holder hereof or of such Owner, if such compliance is required by statute or by regulation of the United States Treasury Department as a precondition to relief or exemption from such tax, assessment or other governmental charge;

(viii) any tax, assessment or other governmental charge imposed as a result of such holder of this Notes being a bank receiving payments on an extension of credit made pursuant to a loan agreement entered into in the ordinary course of business;

(ix) any tax, assessment or other governmental charge required to be imposed or withheld on a payment to an individual and such deduction or withholding is required to be made pursuant to any European Union Directive on the taxation of savings or any law implementing or complying with, or introduced in order to conform to, such Directive; or

(x) any combination of items (i), (ii), (iii), (iv), (v), (vi), (vii), (viii) and (ix);

nor shall Additional Amounts be paid to any holder of this Note on behalf of any Owner who is a fiduciary or partnership or other than the sole Owner to the extent a beneficiary or settlor with respect to such fiduciary or a member of such partnership or Owner would not have been entitled to payment of the Additional Amounts had such beneficiary, settlor, member or Owner been the sole Owner of this Note.

As used herein, the term "United States Alien" means any person who, for United States federal income tax purposes, is a foreign corporation, a nonresident alien individual, a nonresident alien fiduciary of a foreign estate or trust, or a foreign partnership one or more of the members of which is, for United States federal income tax purposes, a foreign corporation, a nonresident alien individual or a nonresident alien fiduciary of a foreign estate or trust.

(b) If this Note is in bearer form and the Bank shall determine, based upon a written opinion of independent counsel selected by the Bank, that any payment made outside the United States by the Bank or any of its Paying Agents of the full amount of the next scheduled payment of either principal (and premium, if any) or interest due with respect to this Note would, under any present or future laws or regulations of the United States affecting taxation or otherwise, be subject to any certification, information or other reporting requirements of any kind, the effect of which requirements is the disclosure to the Bank, any of its Paying Agents or any governmental authority of the nationality, residence or identity (as distinguished from status as a United States Alien) of any Owner of this Note who is a United States Alien (other than such requirements which (i) would not be applicable to a payment made to a custodian, nominee or other agent of the Owner, or which can be satisfied by such a custodian, nominee or other agent certifying to the effect that such Owner is a United States Alien; provided, however, in each case that payment by such custodian, nominee or agent to such Owner is not otherwise subject to any requirements referred to in this sentence, (ii) are applicable only to payment by a custodian, nominee or other agent of the Owner to or on behalf of such Owner, or (iii) would not be applicable to a payment made by any other paying agent of the Bank), the Bank shall redeem this Note as a whole but not in part at a redemption price equal to the principal amount hereof (or, if this is an Original Issue Discount Note, the Amortized Face Amount (as defined herein) hereof determined as of the date of redemption), together, if appropriate, with accrued interest to, but excluding, the date fixed for redemption, such redemption to take place on such date not later than one year after notice of such determination has been given as described herein. If the Bank becomes aware of an event that might give rise to such certification, information or other reporting requirements, the Bank shall, as soon as practicable, solicit advice of independent counsel selected by the Bank to establish whether such certification, information or other reporting requirements will apply and, if such requirements will, in the written opinion of such counsel, apply, the Bank shall give prompt notice of such determination (a "Tax Notice") stating in such notice the effective date of such certification, information or other reporting requirements and, if applicable, the date by which the redemption shall take place. Notwithstanding the foregoing, the Bank shall not redeem this Note if the Bank, based upon the written opinion of independent counsel selected by the Bank, shall subsequently determine not less than 30 days prior to the date fixed for redemption that subsequent payments would not be subject to any such requirements, in which case the Bank shall give prompt notice of such determination and any earlier redemption notice shall thereby be revoked and of no further effect.

Notwithstanding the foregoing, if and so long as the certification information or other reporting requirements referred to in the preceding paragraph would be fully satisfied by payment of a withholding tax, backup withholding tax or similar charge, the Bank may elect prior to giving the Tax Notice to have the provisions described in this paragraph apply in lieu of the provisions described in the preceding paragraph, in which case the Tax Notice shall state the effective date of such certification, information or reporting requirements and that the Bank has elected to pay Additional Amounts rather than redeem this Note. In such event, the Bank will also pay as Additional Amounts such sums as may be necessary so that every net payment made following the effective date of such certification, information or reporting requirements outside the United States by the Bank or any of its Paying Agents of principal, premium, if any, or interest due with respect to this Note to the bearer hereof who certifies to the effect that the beneficial owner of this Note is a United States Alien (provided that such certification shall not have the effect of communicating to the Bank or any of its Paying Agents or any governmental authority the nationality, residence or identity of such beneficial owners) after deduction or withholding for or on account of such withholding, backup withholding tax or similar charge (other than a withholding, backup withholding tax or similar charge which (i) is imposed as a result of certification, information or other reporting requirements referred to in the second parenthetical clause of the first sentence of the preceding paragraph or (ii) is imposed as a result of the fact that the Bank or any of its Paying Agents has actual knowledge that the bearer hereof or any beneficial owner of this Note is not a United States Alien but is within the category of persons, corporations or other entities described in Section 5(a)(i)(1) above, or (iii) is imposed as a result of presentation of this Note for payment more than 15 days after the date on which such payment becomes due and payable or on which payment thereof is duly provided for, whichever occurs later), will not be less than the amount provided for in this Note to be then due and payable. In the event the Bank elects to pay such Additional Amounts, the Bank will have the right, at its sole option, at any time, to redeem this Note, as a whole but not in part, at a redemption price equal to the principal amount hereof (or, if this is an Original Issue Discount Note, the Amortized Face Amount hereof determined as of the date of redemption), together, if appropriate, with accrued interest to the date fixed for redemption including any Additional Amounts required to be paid under this paragraph. If the Bank has made the determination described in the preceding paragraph with respect to certification, information or other reporting requirements applicable to interest only and subsequently makes a determination in the manner and of the nature referred to in such preceding paragraph with respect to such requirements applicable to principal, the Bank will redeem this Note in the manner and on the terms described in the preceding paragraph (except as provided below), unless the Bank elects to have the provisions of this paragraph apply rather than the provisions of the immediately preceding paragraph. If in such circumstances this Note is to be redeemed, the Bank will be obligated to pay Additional Amounts with respect to interest, if any, accrued to the date of redemption. If the Bank has made the determination described in the preceding paragraph and subsequently makes a determination in the manner and of the nature referred to in such preceding paragraph that the level of withholding applicable to principal or interest has been increased, the Bank will redeem this Note in the manner and on the terms described in the preceding paragraph (except as provided below), unless the Bank elects to have the provisions of this paragraph apply rather than the provisions of the immediately preceding paragraph. If in such circumstances this Note is to be redeemed, the Bank will be obligated to pay Additional Amounts with respect to the original level of withholding on principal and interest, if any, accrued to the date of redemption.

Notwithstanding the foregoing, if this is a Subordinated Note, to the extent then required under or pursuant to applicable capital regulations, this Note may not be redeemed by the Bank prior to their Stated Maturity Date under the circumstances described above without the prior written consent of the Federal Reserve Board.

(c) Whenever in this Note there is mentioned, in any context, the payment of the principal, premium, if any, or interest with respect to this Note, such mention shall be deemed to include mention of the payment of Additional Amounts provided for herein to the extent that, in such context, Additional Amounts are, were or would be payable in respect hereof pursuant to the provisions of this Note and express mention of the payment of Additional Amounts (if applicable) in any provisions hereof shall not be construed as excluding Additional Amounts in those provisions hereof where such express mention is not made.

(d) Except as specifically provided in the Pricing Supplement (i) neither the Bank nor any Paying Agent shall be required to make, any payment with respect to any tax, fee, duty, assessment or other governmental charge imposed by any government or a political subdivision or taxing authority thereof or therein; (ii) a Paying Agent on behalf of the Bank shall have the right, but not the duty, to withhold from any amounts otherwise payable to a holder of this Note such amount as is necessary for the payment of any such taxes, fees, duties, assessments or other governmental charges; and (iii) if such an amount is withheld, the amount payable to the holder of this Note shall be the amount otherwise payable reduced by the amount so withheld.

6. <u>Redemption for Tax Reasons</u>.

(a) The Bank may redeem this Note in whole, but not in part, at any time at a redemption price equal to the principal amount hereof (or, if this is an Original Issue Discount Note, the Amortized Face Amount hereof determined as of the date of redemption), together, if appropriate, with accrued interest to but excluding the date fixed for redemption, if the Bank shall determine, based upon a written opinion of independent counsel selected by the Bank, that (1) as a result of any change in or amendment to the laws (or any regulations or rulings promulgated thereunder) of the United States (or of any political subdivision or taxing authority thereof or therein affecting taxation) or the relevant taxing jurisdiction (or any political subdivision or taxing authority thereof or or official interpretation of such laws, regulations or rulings, which amendment or change is effective on or after the Original Issue Date of this Notes, the Bank would be required to pay Additional Amounts on the occasion of the next payment due with respect to such Note.

(b) If this Note is in bearer form, it is subject to redemption as set forth under Section 5 (Additional Amounts) above.

(c) Notice of intention to redeem this Note, in whole but not in part, pursuant to clause (a) above shall be given at least once not less than 30 days and not more than 60 days prior to the date fixed for redemption, *provided that* no such notice of redemption shall be given earlier than 90 days prior to the effective date of such change or amendment and that at the time notice of such redemption is given, such obligation to pay such Additional Amounts remains in effect and cannot be avoided by the Bank's taking reasonable measures available to it. From and after any redemption date, if monies for the redemption of this Note shall have been made available for redemption on such redemption date, this Note shall cease to bear interest (and, if

this Note is a definitive bearer Note, any Coupons appertaining hereto (whether or not attached) maturing after the redemption date shall become void and no payment shall be made in respect thereof), and the only right of the holder of this Note shall be to receive payment of the principal amount hereof (or, if this is an Original Issue Discount Note, the Amortized Face Amount hereof) and, if appropriate, all unpaid interest accrued to such redemption date.

Unless otherwise specified in the Pricing Supplement, if this is a Floating Rate Note, the redemption date shall be an Interest Payment Date.

Notwithstanding the foregoing, if this Note is a Subordinated Note, to the extent then required under or pursuant to applicable capital regulations, this Note may not be redeemed by the Bank prior to the Maturity Date pursuant to this Section 6 (Redemption for Tax Reasons), without the prior written consent of the Federal Reserve Board.

7. Events of Default, Acceleration of Maturity.

(a) Senior Notes

(i) If this Note is a Senior Note, the occurrence of any of the following events shall constitute an "Event of Default" with respect to this Note:

(A) default in the payment of any interest (including any Additional Amounts) with respect to this Note when due, which continues for 30 calendar days;

(B) default, in the payment of any principal, or premium, if any, with respect to this Note when due whether at maturity or upon early redemption or otherwise;

(C) the entry by a court or other governmental agency or body having jurisdiction in the premises of:

(1) a decree or order for relief in respect of the Bank in a case or proceeding under any applicable United States federal or state bankruptcy, insolvency, reorganization or other similar law; or

(2) a decree or order appointing a conservator, receiver, liquidator, assignee, trustee, sequestrator or any other similar official of the Bank, or of substantially all of the property of the Bank, or ordering the winding up or liquidation of the affairs of the Bank,

and in either case (1) and (2) such decree or order for relief or any such other decree or order shall have remained in force undischarged or unstayed for a period of 60 consecutive days; or

(D) the commencement by the Bank of a voluntary case or proceeding under any applicable United States federal or state bankruptcy, insolvency, reorganization or other similar law or of any other case or proceeding to be adjudicated as bankrupt or insolvent, or the consent by the Bank to the entry of a

decree or order for relief in an involuntary case or proceeding under any applicable United States federal or state bankruptcy, insolvency, reorganization or other similar law, or the filing by the Bank of a petition or answer or consent seeking reorganization or relief under any applicable United States federal or state bankruptcy, insolvency, reorganization or similar law, or the consent by the Bank to the filing of such petition or to the appointment of or taking possession by a custodian, receiver, liquidator, assignee, trustee, sequestrator or similar official of the Bank or of substantially all of the property of the Bank, or the making by the Bank of an assignment for the benefit of creditors, or the taking of corporate action by the Bank in furtherance of any such action.

(ii) If an Event of Default shall occur and be continuing with respect to this Senior Note, the holder of this Senior Note may declare the principal amount of, and accrued interest and premium, if any, on, this Note due and payable immediately by written notice to the Bank. Upon such declaration and notice, such principal amount, accrued interest and premium, if any, shall become immediately due and payable. Any Event of Default with respect to this Senior Note may be waived by the holder thereof.

(b) Subordinated Notes

(i) If this Note is a Subordinated Note, the occurrence of any of the following events shall constitute an "Event of Default" with respect to this Subordinated Note:

(A) the Bank shall consent to the appointment of a receiver or other similar official (other than a conservator) in any liquidation, insolvency or similar proceeding with respect to the Bank or all or substantially all of the property of the Bank; or

(B) a court or other governmental agency or body having jurisdiction in the premises shall enter a decree or order for the appointment of a receiver or other similar official (other than a conservator) in any liquidation, insolvency or similar proceeding with respect to the Bank or all or substantially all of the property of the Bank, and such decree or order shall have remained in force undischarged or unstayed for a period of 60 days.

(ii) If an Event of Default shall occur and be continuing, the holder of this Subordinated Note may declare the principal amount of, premium, if any, and accrued interest on, this Subordinated Note due and payable immediately by written notice to the Bank. Upon such declaration and notice, such principal amount (and premium, if any) and accrued interest shall become immediately due and payable. Any Event of Default with respect to this Subordinated Note may be waived by the holder of this Note.

(iii) The indebtedness of the Bank evidenced by this Note, including the principal, premium, if any and interest (including any Additional Amounts), shall be, and the registered holder of this Note (or, if this Note is in bearer form, the bearer of this Note), by its acceptance hereof, agrees that the indebtedness of the Bank evidenced by this Note, including the principal (and premium, if any) and interest (including any

Additional Amounts) is, unsecured and subordinate and junior in right of payment to the Bank's obligations to its depositors, the Bank's obligations under bankers' acceptances and letters of credit, and the Bank's obligations to its other creditors, including its obligations to any Federal Reserve Bank and the Federal Deposit Insurance Corporation (the "*FDIC*") (except for obligations to the FDIC arising under provisions of Section 1815(e) of Title 12 of the United States Code) and any rights acquired by the FDIC as a result of loans made by the FDIC to the Bank or the purchase or guarantee of any of its assets by the FDIC pursuant to the provisions of Section 1823(c) or (d) of Title 12 of the United States Code, whether outstanding a the time this Note is issued or thereafter incurred other than any obligations, if any, which by their express terms rank on parity with, or junior to, the Subordinated Notes. In the event of any insolvency proceeding, receivership, conservatorship, reorganization, readjustment of debt, marshalling of assets and liabilities or similar proceedings or any liquidation or winding up of or relating to the Bank, whether voluntary or involuntary, all such obligations (except obligations which by their express terms rank on parity with, or junior to, the Subordinated Notes) the principal, premium, if any, or interest with respect to this Note. In the event of any such obligations (except obligations which by their express terms rank on parity with, or junior to, the Subordinated Notes) shall be entitled to be paid in full before any payment shall be made on account of the principal, premium, if any, or interest with respect to this Note. In the event of any obligations of the Bank ranking on a parity with his Note, shall be entitled to be paid from the remaining assets of the Bank the unpaid principal, premium, if any, and interest with respect to this Note in account of any capital stock or any obligations, before any payment or other distribution, whether in cash, property, or otherwis

This Note and any Subordinated Notes issued subsequently hereto pursuant to the Offering Circular, will rank *pari passu* among themselves and *pari passu*, in the event of a liquidation or similar proceeding with respect to the Bank, whether voluntary or involuntary, with all other present or future unsecured subordinated debt obligations of the Bank, except any unsecured subordinated debt which may be expressly stated to be subordinated to this Note.

Notwithstanding any other provisions of this Note, including specifically those set forth in the paragraphs relating to subordination, Events of Default and covenants of the Bank, it is expressly understood and agreed that the FDIC or any receiver or conservator of the Bank shall have the right in the performance of his legal duties, and as part of any transaction or plan of reorganization or liquidation designed to protect or further the continued existence of the Bank or the rights of any parties or agencies with an interest in, or claim against, the Bank or its assets, to transfer or direct the transfer of the obligations represented by this Note to any state bank, national banking association, or bank holding company selected by such official which shall expressly assume the obligation of the due and punctual payment of the unpaid principal, premium, if any, and interest with respect to this Note and the due and punctual performance of all covenants and conditions contained herein; and that the completion of such transfer and assumption shall serve to supersede and void any default, acceleration or subordination which may have occurred, or which may occur due or related to such transaction, plan, transfer or assumption, pursuant to the provisions of this Note, and shall serve to return the holder hereof to

the same position, other than for substitution of the obligor, it would have occupied had no default, acceleration or subordination occurred; except that any interest and principal (and premium, if any) previously due, other than by reason of acceleration, and not paid shall, in the absence of a contrary agreement by the holder of this Note, be deemed to be immediately due and payable as of the date of such transfer and assumption, together with interest from its original due date at the rate provided for herein.

This Note contains no limitation on the amount of senior debt, deposit liabilities or other obligations that rank senior to this Note that may be hereafter incurred or assumed by the Bank.

8. <u>Miscellaneous</u>.

(a) Original Issue Discount Note

Notwithstanding anything to the contrary contained herein, if this Note is identified as an Original Issue Discount Note in the Pricing Supplement, unless otherwise specified in the Pricing Supplement, the amount payable to the holder of this Note in the event of redemption, repayment or acceleration of maturity will be equal to (i) the Amortized Face Amount (as defined below) determined as of the date of such redemption, repayment or acceleration of Maturity, plus (ii) with respect to any redemption of this Note (other than as provided above in the event that Additional Amounts are required to be paid by the Bank with respect to this Note), the Initial Redemption Percentage specified in the Pricing Supplement (as adjusted by the Annual Redemption Percentage Reduction specified in the Pricing Supplement, if any) minus 100% multiplied by the Issue Price specified in the Pricing Supplement, net of any portion of such Issue Price which has been paid prior to the date of redemption, or the portion of the Issue Price (or the net amount) proportionate to the portion of the unpaid principal amount to be redeemed, plus (iii) any accrued interest to the date of such event the payment of which would constitute qualified stated interest payments within the meaning of U.S. Treasury Regulation 1. 1273-1 (c) under the Code. The *"Amortized Face Amount"* shall mean an amount equal to (i) the Issue Price plus (ii) the aggregate portions of the OID (as specified in the Pricing Supplement which shall theretofore have accrued pursuant to Section 1272 of the Code (without regard to Section 1272(a)(7) of the Code) from the Original Issue Date of this Note to the date of determination, minus (iii) any amount considered as part of the "stated redemption price at maturity" of this Note which has been paid from the Original Issue Date to the date of determination.

(b) Business Day

As used herein, "Business Day" means, unless otherwise specified in the Pricing Supplement, a day which is both (i) a day (other than a Saturday or a Sunday) on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign currency deposits) in The City of New York, McLean, Virginia, and London; and (ii) either (a) if this is a Note denominated in a Specified Currency other than euro, a day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign european day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign european day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign european day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign european day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign currency deposits) in the principal financial center of the country of the relevant Specified Currency (if other than the City of New York or London) or (b) if this is a Note denominated in euro, a day (other than a Saturday or a Sunday) on which the Trans-European Automated Real-Time Gross Settlement Express Transfer (TARGET) System is open. As used herein, "London Business Day" means

any day (other than a Saturday or Sunday) on which commercial banks and foreign exchange markets settle payments in London.

(c) Action by Holders

Any action by the holder of this Note shall bind all future holders of this Note, and of any Note issued in exchange or substitution hereof or in place hereof, in respect of anything done or permitted by the Bank or by the Paying Agents in pursuance of such action.

(d) Issue of Replacement Notes

In case this Note shall at any time become mutilated, defaced, destroyed, lost or stolen, the London Issuing Agent or the Registrar, as applicable, shall issue a replacement Note of like tenor and principal amount, having a serial number not contemporaneously outstanding, in exchange and substitution for the mutilated or defaced Note or in lieu of this Note if it is destroyed, lost or stolen, *provided* the applicant therefor prior to the issuance of such replacement Note shall have (i) furnished it with such evidence (including the serial number of the Note) and such indemnity as the Bank and the Registrar or the London Issuing Agent, as the case may be, may require, (ii) in the case of any mutilated or defaced Note, surrendered such Note, and (iii) paid such costs as may be incurred in connection with the preparation and delivery of a new Note. If any Note which has matured or for which a redemption date or repayment date has occurred or which is about to mature or to be redeemed or repaid, shall become mutilated, defaced, destroyed, lost or stolen, the Bank may, instead of issuing a replacement Note, pay or authorize the payment of the same (without surrender thereof except in the case of a mutilated or defaced Note) upon compliance by the holder with the provisions of this paragraph.

(e) No Recourse Against Shareholder, Employee, Agent, Officer or Director of Bank

No recourse shall be had for the payment of principal, premium, if any, or interest with respect to this Note for any claim based hereon, or otherwise in respect hereof, against any shareholder, employee, agent, officer or director, as such, past, present or future, of the Bank or of any successor organization, either directly or through the Bank or any successor organization, whether by virtue of any constitution, statute or rule of law or by the enforcement of any assessment or penalty or otherwise, all such liability being, by the acceptance hereof and as part of the consideration for the issue hereof, expressly waived and released.

(f) Amendments

The Notes are issued in accordance with the Global Agency Agreement. The Notes, and any receipts or interest coupons appertaining thereto, may be amended by the Bank, and the Global Agency Agreement may be amended by the parties thereto, (i) for the purpose of curing any ambiguity, or of curing, correcting or supplementing any defective provision contained therein, (ii) to make any further modifications of the terms of the Global Agency Agreement necessary or desirable to allow for the issuance of any additional Notes (which modifications shall not be materially adverse to holders of outstanding Notes) or (iii) in any manner which the Bank (and, in the case of the Global Agency Agreement, the parties thereto) may deem necessary or desirable and which shall not materially adversely affect the interests of the holders of the Notes, or any receipts, talons or interest coupons appertaining thereto, to all of which each holder of Notes, receipts, talons or interest coupons shall, by acceptance thereof, be deemed to have

consented; *provided, however*, that no such modification or amendment may, without the consent of the holder of each outstanding Note affected thereby, (1) change the Maturity Date with respect to any Note or reduce or cancel the amount payable at Maturity; (2) reduce the amount payable or modify the payment date for any interest with respect to any Note or vary the method of calculating the rate of interest with respect to any Note; (3) reduce any Minimum Interest Rate and/or Maximum Interest Rate with respect to any Note; (4) modify the currency in which payments under any Note and/or any receipts, coupons or talons appertaining thereto are to be made; (5) change the obligation of the Bank to pay Additional Amounts with respect to Notes, receipts, talons or coupons; or (6) reduce the percentage in principal amount of outstanding Notes the consent of the holders of which is necessary to modify the provisions of the Notes or to waive any future compliance or past default. Any instrument given by or on behalf of any holder of a Note in connection with any consent to any such modifications, amendments or waivers to the Global Agency Agreement or the provisions of the Notes, receipts, talons or coupons, whether or not notation of such modifications, amendments or waivers to the Global Agency Agreement or the provisions of the Notes, receipts, talons or coupons. It will not be necessary for the consent of the holders of Notes to approve the particular form of any proposed amendment, but it shall be sufficient if such consent shall approve the substance thereof.

(g) <u>General</u>

No provision of this Note shall alter or impair the obligation of the Bank, which is absolute and unconditional, to pay principal, premium, if any, interest and any Additional Amounts with respect to this Note in accordance with its terms.

Except as provided in the Pricing Supplement, no service charge shall be made to a holder of this Note for any transfer or exchange of this Note, but the Bank or Registrar may require payment of a sum sufficient to cover any stamp tax or other tax, duty, assessment governmental charge that may be imposed in connection therewith.

Subject to the terms of the Global Agency Agreement, if this Note is in registered form, the Bank, Domestic Paying Agent, Registrar, London Paying Agent, Luxembourg Paying Agent, Transfer Agent and Listing Agent (collectively, together with any successors thereto, the "*Agents*") or any representative or agent of the Bank or the Agents may treat the holder in whose name this Note is registered as the owner hereof for all purposes, whether or not this Note be overdue, and neither the Bank, the Agents nor any such representative or agent shall be affected by notice to the contrary except as required by applicable law.

All notices to the Bank under this Note shall be in writing and addressed to the Bank at Capital One Bank, 1680 Capital One Drive, McLean, Virginia 22102, USA, Attention: Treasurer; telephone: (703) 720-1000; and facsimile: (703) 720-2156 or to such other address of the Bank as the Bank may notify the holders of the Notes.

ABBREVIATIONS

The following abbreviations, when used in the inscription on the within Note, shall be construed as though they were written out in full according to applicable laws or regulations.

 TEN COM
 –
 as tenants in common

 TEN ENT
 –
 as tenants by the entireties

 JT TEN
 –
 as joint tenants with right of survivorship and not as tenants in common

 UNIF GIFT
 MIN
 ACT –
 Custodian

 (Cust)
 (Minor)

under Uniform Gifts to Minors Act

State

Additional abbreviations may also be used though not in the above list.

ASSIGNMENT

FOR VALUE RECEIVED, the undersigned hereby sell(s), assign(s) and transfer(s) unto

PLEASE INSERT SOCIAL SECURITY NUMBER OR OTHER IDENTIFYING NUMBER OF ASSIGNEE

(Please print or typewrite name and address, including postal zip code, of assignee)

the within Note and all rights thereunder, and hereby

irrevocably constitutes and appoints

to transfer said Note in the Note Register and on the books of the Bank, with full power of substitution in the premises.

Dated:

NOTICE: The signature to this assignment must correspond with the name as written upon the within Note in every particular, without alteration or enlargement or any change whatsoever.

Signature Guarantee

NOTICE: The signature(s) should be guaranteed by an eligible guarantor institution (i.e., banks, stockbrokers, savings and loan associations, and credit unions with membership in an approved signature guarantee medallion program), pursuant to Rule 17Ad-15 under the Securities Exchange Act of 1934.

Schedule 1

SCHEDULE OF TRANSFERS AND EXCHANGES

The following increases and decreases in the principal amount of this Note have been made:

Date of Transfer	Increase (Decrease) in Principal Amount of this Note Due to Transfer Among Global Notes	Principal Amount of this Note After Transfer	Notation made by or on behalf of the Bank
	4	8	

CAPITAL ONE FINANCIAL CORPORATION

VOLUNTARY NON-QUALIFIED DEFERRED

COMPENSATION PLAN

EFFECTIVE MAY 15, 2004

Section	Page	
1.	DEFINITIONS	1
2.	PURPOSE	5
3.	PARTICIPATION	5
4.	DEFERRAL ELECTION	5
5.	EFFECT OF NO ELECTION	7
6.	INVESTMENT FUNDS	7
7.	DISTRIBUTIONS	7
8.	HARDSHIP DISTRIBUTIONS	9
9.	COMPANY'S OBLIGATION	10
10.	CONTROL BY PARTICIPANT	10
11.	CLAIMS AGAINST PARTICIPANT'S BENEFIT	10
12.	AMENDMENT OR TERMINATION	11
13.	ADMINISTRATION	11
14.	NOTICES	11
15.	WAIVER	11
16.	CONSTRUCTION	11

(i)

CAPITAL ONE FINANCIAL CORPORATION

VOLUNTARY NON-QUALIFIED DEFERRED COMPENSATION PLAN

1. <u>**DEFINITIONS</u>**. The following definitions apply to this Plan and to any related documents.</u>

- (a) <u>Accounts</u> means a Participant's Deferral Account.
- (b) <u>Administrator</u> means the Company. As Plan Administrator, the Company has delegated responsibility for the day-to-day administration of the Plan to the Committee.
- (c) Base Salary means the Participant's annual rate of base pay, without regard to bonuses or any other amount reported as compensation income.
- (d) Beneficiary or Beneficiaries means a person or persons or other entity that a Participant designates on a Beneficiary Designation Form to receive Benefit payments pursuant to Plan Section 7(c). If a Participant does not execute a valid Beneficiary Designation Form, or if the designated Beneficiary or Beneficiaries fail to survive the Participant or otherwise fail to take the Benefit, the Participant's Beneficiary or Beneficiaries shall be the first of the following persons who survive the Participant: a Participant's spouse (the person legally married to the Participant when the Participant dies); the Participant's children in equal shares. If none of these persons survive the Participant, the Beneficiary shall be the Participant's estate.
- (e) **Beneficiary Designation Form** means the form, written or electronic, that a Participant uses to name the Participant's Beneficiary or Beneficiaries.
- (f) **Benefit** means collectively, a Participant's Deferred Benefit.
- (g) **Board** means the Board of Directors of Capital One Financial Corporation.
- (h) **Bonus** means the discretionary annual cash incentive based performance bonus that may be paid to Company employees each year by the Company for attaining certain Company and/or individual performance goals.
- (i) <u>Capital One Company</u> means any organization under common control (as described in Code Sections 414(b) and (c)) with the Company or any organization that is a member of an affiliated service group (as described in Code Section 414(m)) of which the Company is a member, and any other organization required to be aggregated with the Company pursuant to Code Section 414(o).

(1)

(j) <u>**Change of Control**</u> means the occurrence of any of the following events:

(i) The acquisition by an individual, entity or "group" (within the meaning of Section 13(d)(3) or 14(d)(2) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 20% (or, if such shares are purchased from the Company, 40%) or more of either (A) the then outstanding shares of common stock of the Company (the "Outstanding Company Common Stock") or (B) the combined voting power of the then outstanding voting securities of the Company or any of its Subsidiaries, or any employee benefit plan (or related trust) sponsored or maintained by the Company or any of its Subsidiaries or (y) any corporation with respect to which, immediately following such acquisition, more than 60% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Company Voting Securities immediately prior to such acquisition in substantially the same proportion as their ownership, immediately prior to such acquisition, of the Outstanding Company Common Stock and Company Voting Securities, as the case may be, shall not constitute a Change of Control; or

(ii) Individuals who constitute the Board as of January 17, 2004 (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided that any individual becoming a director subsequent to January 17, 2002 whose appointment to fill a vacancy or to fill a new Board position or whose nomination for election by the Company's shareholders was approved by a vote of at least a majority of the directors then comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election of the Directors of the Company (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act); or

(iii) Approval by the shareholders of the Company of a reorganization, merger or consolidation (a "Business Combination"), in each case, with respect to which all or substantially all of the individuals and entities who were the respective beneficial owners of the Outstanding Company Common Stock and Company Voting Securities immediately prior to such Business Combination do not in the aggregate, immediately following such Business Combination, beneficially own, directly or indirectly, more than 60% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such Business Combination in substantially the same proportion as their ownership immediately prior to such Business

(2)

Combination of the Outstanding Company Common Stock and Company Voting Securities, as the case may be; or

(iv) (A) a complete liquidation or dissolution of the Company or (B) sale or other disposition of all or substantially all of the assets of the Company other than to a corporation with respect to which, immediately following such sale or disposition, more than 60% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, in the aggregate by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Company Voting Securities immediately prior to such sale or disposition in substantially the same proportion as their ownership of the Outstanding Company Common Stock and Company Voting Securities, as the case may be, immediately prior to such sale or disposition.

(k) Code means the Internal Revenue Code of 1986, as amended.

(r)

- (l) **Committee** means the Company's Employee Benefits Committee.
- (m) <u>**Company**</u> means Capital One Financial Corporation and any Capital One Company that is designated by the Committee as covered by this Plan, and any successor business by merger, purchase, or otherwise that maintains the Plan.
- (n) <u>Compensation</u> means the Base Salary and Bonus for services preformed for the Company that is currently includable in gross income. Compensation does not include stock, stock options, and gains from stock option grants. The Committee may determine whether to include or exclude an item of income from Compensation.
- (o) **Deferral** means the amount of Compensation that a Participant has elected to defer under a Deferral Election Form.
- (p) <u>Deferral Account</u> means a bookkeeping record established for each Participant who is eligible to receive a Deferred Benefit. A Deferral Account shall be established only for purposes of measuring a Deferred Benefit and not to segregate assets or to identify assets that may be used to satisfy a Deferred Benefit. A Deferral Account shall be credited with that amount of a Participant's Compensation deferred according to a Participant's Deferral Election Form. A Deferral Account also shall be credited periodically with deemed investment gain or loss under Plan Section 7.
- (q) Deferral Election Form means the form, written or electronic, that a Participant uses to elect to defer Compensation pursuant to Plan Section 4.
 - **Deferred Benefit** means the benefit available to a Participant who has executed a valid Deferral Election Form.

(3)

- (s) **Disability** or **Disabled** means, with respect to a Participant, that the Participant has a total and permanent disability, as determined by the Committee, in accordance with the provisions of the Company's long term disability program.
- (t) Distribution Election Form means a form, written or electronic, that a Participant uses to establish the duration of the deferral of Compensation and the frequency of payments of a Benefit. If a Participant does not execute a valid Distribution Election Form, the distribution of a Benefit shall be governed by Plan Section 5.
- (u) <u>Election Date</u> means the date by which an Executive must submit a valid Deferral Election Form for Compensation. For each Plan Year, the Election Date shall be January 1 unless the Committee sets a different Election Date for either Base Salary or Bonus, or as provided in Plan Section 4.
- (v) Excess Savings Plan means the Capital One Financial Corporation Excess Savings Plan.
- (w) Executive means an individual who (i) is employed by the Company as a common-law employee on the Company's United States payroll, and (ii) is designated by the Company as an Executive.
- (x) <u>Investment Fund</u> means one or more deemed investment alternatives offered to Participants from time to time. The Company may compute deemed investment gain or loss under the Investment Funds based on the actual investment performance of assets that it has deposited in a grantor trust (as described in Plan Section 10).
- (y) <u>Participant</u> means an individual presently employed by the Company who meets one or more of the requirements of Plan Section 3(a). Individuals who were formerly employed by the Company and have an Account under the Plan will remain Participants in the Plan, but only for purposes of Plan Sections 6, 7 and 8.
- (z) Plan means the Capital One Financial Corporation Voluntary Non-Qualified Deferred Compensation Plan.
- (aa) Plan Year means a calendar year. However, the first Plan Year shall mean the period beginning with May 1, 2004 and ending December 31, 2004.
- (bb) <u>**Retirement**</u> means either (i) that a Participant has attained age 55 and has ten (10) years of service with the Company, or the Participant has attained age 65.
- (cc) **Terminate** or **Termination**, with respect to a Participant, means the cessation of the Participant's employment with the Company on account of death, Disability, severance or any other reason.

(4)

2. <u>PURPOSE</u>. The Plan is intended to benefit a "select group of management or highly compensated employees," as that term is used under Title I of the Employee Retirement Income Security Act of 1974, as amended. The Plan is intended to permit Executives to defer their Compensation.

3. PARTICIPATION.

- (a) An individual presently employed by the Company is a Participant if he or she is with respect to any Plan Year, an Executive who executes a valid Deferral Election Form for that Plan Year as provided in Plan Section 3(b).
- (b) An Executive may become a Participant for any Plan Year by filing a valid Deferral Election Form according to Plan Section 4 on or before the Election Date for that Plan Year.
- (c) An individual remains a Participant as long as the Participant is entitled to a Benefit under the Plan. A Participant who ceases to be an Executive during a Plan Year shall cease making Deferrals as of the date on which he or she ceases to be an Executive.

4. DEFERRAL ELECTION. An Executive may elect on or before the Election Date to defer receipt of a portion of the Executive's Compensation. Except as provided in Plan Section 4(a), an Executive may elect a Deferral for any Plan Year only if he or she is an Executive on the Election Date for that Plan Year. The following provisions apply to deferral elections:

- (a) A Participant may defer a percentage (in 1% increments) of up to 50% of the Participant's Base Salary and a percentage (in 1% increments) of up to 100% of the Participant's Bonus. Compensation for deferrals under the Capital One Financial Corporation Associate Savings Plan shall be based on a Participant's compensation after any Deferrals made under this Plan, or any other deferred compensation plan of the Company. Any Deferrals made by the Participants to this Plan are totally independent of any election a Participant may or may not make in order to participate or receive benefits under the Capital One Financial Corporation Associate Savings Plan.
- (b) Before each Plan Year's Election Date, each Executive shall be provided with a Deferral Election Form. Except as provided below, a deferral election shall be valid only when the Deferral Election Form is completed, signed by the electing Executive, and received by the Committee on or before the Election Date for that Plan Year. In the year in which an individual becomes an Executive for purposes of this Plan, the Executive may make a deferral election by completing a Deferral Election Form within 30 days of his or her being notified by the Committee that he or she is eligible under the Plan. The deferral election will be effective for periods and Compensation earned after the Committee receives it. Subject to certain exceptions below, the period over which the amounts will be deferred

(5)

pursuant to the Plan may not be less than two years from the end of the calendar year in which the amounts are deferred.

- (c) An Executive must complete an investment election form for all amounts in the Executive's Deferral Account. The Investment Election Form may be in hard copy written form or electronic form. The Compensation deferred under a Deferral Election Form shall be allocated among available Investment Funds in percentages as specified on the investment election form. If a Participant fails to designate among which Investment Funds the deferred Compensation is to be allocated, the Committee shall allocate the deferred Compensation to the most conservative Investment Fund.
- (d) An Executive must complete a Distribution Election Form for the distribution of the Executive's Deferral Account. The Distribution Election Form may be in hard copy written form or electronic form. The Participant may receive distribution of his or her Benefit at Retirement, Termination, death, Disability or a year certain. For each year in which a Participant makes a Deferral, a Distribution Election Form must be completed and accompany the Deferral Election Form indicating the manner in which that year's deferral is to be distributed; provided however that each Deferral must be deferred for a minimum of two years.
- (e) The Committee may reject any Deferral Election Form or any Distribution Election Form or both that does not conform to the provisions of the Plan. The Committee may modify any Distribution Election Form at any time to the extent necessary to comply with any federal securities laws or regulations. The Committee's rejection or modification must be made on a uniform basis with respect to similarly situated Executives. If the Committee rejects a Deferral Election Form, the Executive shall be paid the amounts the Executive would have been entitled to receive if the Executive had not submitted the rejected Deferral Election Form.
- (f) An Executive may not revoke a Deferral Election Form after the Plan Year (or period for which the election is made). Any revocation before the beginning of the Plan Year (or period for which the election is made) has the same effect as a failure to submit a Deferral Election Form. Any writing signed by an Executive expressing an intention to revoke the Executive's Deferral Election Form and delivered to the Committee before the close of business on the relevant Election Date shall be a revocation.
- (g) An Executive may revoke or modify an existing Distribution Election Form in accordance with the requirements of Plan Section 7.
- (h) An Executive 100% vested in his or her Deferral Account subject to the requirements of Section 9.

(6)

5. EFFECT OF NO ELECTION. Except as provided in Plan Section 4(b), an Executive who has not submitted a valid Deferral Election Form to the Committee on or before the relevant Election Date may not defer any part of the Executive's Compensation for the Plan Year to this Plan. The Deferred Benefit of an Executive who submits a valid Deferral Election Form but fails to submit a valid Distribution Election Form (either as to the form or commencement of payment) before the relevant Election Date shall be distributed in a lump sum on or before the February 28 following the calendar year of the Executive's Termination.

6. INVESTMENT FUNDS.

- (a) Each Participant shall have the right to direct the deemed investment of the Participant's Deferral Account among the Investment Funds. The Committee shall determine the number and type of Investment Funds that will be available for investment in any Plan Year. At its sole discretion, the Committee may change the number and type of Investment Funds at any time and may establish procedures for the transition between Investment Funds.
- (b) Deferrals shall be credited to an Investment Fund as of the date on which the deferred Compensation would have been paid to the Participant or as soon as administratively practicable, thereafter. A separate bookkeeping account shall be established for each Participant who has directed a deemed investment in an Investment Fund. Deemed transfers between Investment Funds in the Participant's Deferral Account shall be charged and credited as the case may be to each Investment Fund account. The Investment Fund account shall be charged or credited with net earnings, gains, losses and expenses, as well as any appreciation or depreciation in market value during each Plan Year for the deemed investment in the Investment Fund. The Committee may charge or credit such earnings, gains, losses, appreciation and depreciation based on the actual investment performance of assets that it has deposited in a grantor trust (as described in Plan Section 9).

7. DISTRIBUTIONS.

(a) All Benefits, less withholding for applicable income and employment taxes, shall be paid in cash by the Company or its designee. A Participant may elect to receive a distribution of all or a portion of the Participant's Benefits subject to the provisions of this Section. Payment of each distribution of Benefits shall be made in one lump sum or in installments as provided in this Section.

(i) Unless otherwise provided herein or specified in a Participant's Distribution Election Form, any lump sum payment shall be paid, or installment payments shall begin, on or before February 28 of the calendar year after the Participant's Termination. The Participant may elect on the Participant's Distribution Election Form to begin payments (A) on or before the February 28 of

(7)

the calendar year following the calendar year of the Participant's Termination; or (B) if the Participant does not Terminate, on the January 30th of a specified calendar year; provided such amounts have been deferred for a period of at least two years.

(ii) Installment payments will be made in five (5), ten (10) or fifteen (15) year annual payments as specified in the Participant's Distribution Election Form. Benefits will not be paid more often than once a year. For a Benefit payable in a form other than a lump sum, the unpaid balance of a Participant's Deferral Account, if any, shall continue to be maintained in Investment Funds. For purposes of calculating a Participant's installment payments, the total Deferral Account for a particular year shall be divided by the remaining number of installment payments to be paid to the Participant. The installment payment for a particular year will equal the quotient obtained from the foregoing calculation. The remaining amounts credited to a Participant's Deferral Account will continue to be invested in accordance with a Participant's investment selections.

(iii) Participants may change with respect to each Deferral, the form of payment and the date on which the Deferral is to be distributed to the Participant; provided written notice of such change is received by the Committee not less than twelve months prior to the date on which the Deferral was scheduled to begin to be distributed to the Participant; and provided further such date on which the Participant subsequently elects to receive his or her distribution is the same or a later date than the payment date originally elected. The election is subject to the Committee's approval, in its absolute discretion, and the election will be effective no less than 30 days after notice is provided to the Committee.

- (b) Benefits paid on account of Termination due to Retirement death or disability shall be paid in a lump unless the Participant's Distribution Election Form specifies annual installment payments over a period of five (5), ten (10) or fifteen (15) years according to the Participants Distribution Election Form.
- (c) Benefits paid on account of a Participant's death shall be paid shall be paid in the method designated on the Participant's Distribution Election Form unless the Plan Committee in its sole discretion, determines that the Benefits shall be paid in a lump sum.
- (d) Benefits paid on account of Termination due to Disability shall begin to be paid as soon as administratively practicable following the Participant's Termination. The Benefits shall be paid in the method designated on the Participant's Distribution Election Form, or annual installment payments over a period of five (5) ten (10) or fifteen (15) years, (as determined by the Committee) if the Participant made no election on the Participant's Distribution Election Form. If a Disabled Participant begins to receive Benefits and thereafter recovers and returns to employment before the balance of the Participant's Accounts is fully paid, distributions shall

(8)

cease and any remaining Benefits under the Plan shall be governed by this Plan Section 7 and the Participant's Distribution Election Form.

- (e) Notwithstanding any other provision of this Plan or a Participant's Distribution Election Form, the Committee in its sole discretion may postpone the distribution of all or part of a Benefit to the extent that the payment would not be deductible under Section 162(m) of the Internal Revenue Code of 1986, as amended (the Code) or any successor thereto. A Benefit distribution that is postponed pursuant to the preceding sentence shall be paid as soon as it is possible to do so within the deduction limitations of Section 162(m) of the Code.
- (f) A Participant or Beneficiary may not assign Benefits. A Participant may use only one Beneficiary Designation Form to designate one or more Beneficiaries for all of the Participant's Benefits under the Plan. Such designations are revocable. Each Beneficiary shall receive the Beneficiary's portion of the Participant's Accounts on or before February 28 of the year following the Participant's death. The Committee may require that multiple Beneficiaries agree upon a single distribution method.

8. HARDSHIP DISTRIBUTIONS.

- (a) At its sole discretion and at the request of a Participant before or after the Participant's Termination, or at the request of any of the Participant's Beneficiaries after the Participant's death, the Committee may accelerate and pay all or part of any amount attributable to a Participant's Benefits. The Committee may accelerate distributions only in the event of Hardship as defined in Plan Section 8(b). An accelerated distribution under this Section shall be limited to the amount necessary to satisfy the Hardship.
- (b) Hardship is a severe financial hardship to the Participant resulting from a sudden and unexpected illness or accident of the Participant or of a dependent of the Participant, loss of the Participant's property due to casualty, or other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant. The circumstances that will constitute a Hardship will depend upon the facts of each case, but, in any case, payment will not be made to the extent that the Hardship is or may be relieved: (i) through reimbursement or compensation by insurance or otherwise, (ii) by liquidation of the Participant's assets, to the extent that the liquidation of such assets would not itself cause severe financial hardship, or (iii) by cessation of Deferrals under the Plan.
- (c) Distributions under this Plan Section 8 shall be made in one lump sum payment in cash. Distributions shall be made proportionately from all of the Investment Funds in the Participant's Accounts. The Investment Funds in the Participant's Accounts shall be valued as of the last business day prior to the distribution, or as of such other date as may be determined in the discretion of the Committee.

(d) A distribution under this Plan Section 8 shall be in lieu of that portion of a Participant's Benefit that would have been paid otherwise. A Benefit shall be adjusted by reducing the balance of the Participant's Accounts by the amount of the distribution. If a Hardship is made pursuant to this Section, the Participant shall not be permitted to again make Deferrals to the Plan until the second January 1st following the date on which the distribution is made.

9. COMPANY'S OBLIGATION.

- (a) The Plan shall be unfunded. The Company shall not be required to segregate any assets that at any time may represent a Benefit. The Company may establish a grantor trust (within the meaning of Sections 671 through 679 of the Code) for Participants and Beneficiaries and may deposit funds with the trustee of such trust to provide the Deferred Benefits to which Participants and Beneficiaries may be entitled under the Plan. The funds deposited with the trustee or trustees of such trust, and the earnings thereon, will be dedicated to the payment of Benefits under the Plan but shall remain subject to the claims of the general creditors of the Company. Any liability of the Company to a Participant or Beneficiary under this Plan shall be based solely on any contractual obligations that may be created pursuant to this Plan. No such obligation of the Company shall be deemed to be secured by any pledge of, or other encumbrance on, any property of the Company.
- (b) Notwithstanding the foregoing, if a Participant is Terminated within twenty-four months following a Change of Control, the Company shall as soon as practicable following the Termination pay in one lump sum to such Participant or Beneficiary his or her Benefit under the Plan.

10. <u>CONTROL BY PARTICIPANT</u>. A Participant shall have no control over the Participant's Benefit except according to the Participant's Deferral Election Forms, Distribution Election Forms, Investment Election Form and Beneficiary Designation Form.

11. <u>CLAIMS AGAINST PARTICIPANT'S BENEFIT</u>. An Account shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, or charge, and any attempt to do so shall be void. A Benefit shall not be subject to attachment or legal process for a Participant's debts or other obligations. Nothing contained in this Plan shall give any Participant any interest, lien, or claim against any specific asset of the Company. A Participant or the Participant's Beneficiary shall have no rights other than as a general creditor of the Company. Nothing contained in the Plan shall be deemed to give any Participant or employee the right to be retained in the service of the Company or to interfere with the right of the Company to discharge any Participant or employee at any time, regardless of the effect which such discharge shall have upon him or her as a Participant or Beneficiary shall have any further rights under the Plan upon receipt by the appropriate person of all benefits. The Committee may require the

(10)

Participant or Beneficiary to execute a receipt and release therefore in such form as shall be determined by the Committee.

12. <u>AMENDMENT OR TERMINATION</u>. Except as otherwise provided, this Plan may be altered, amended, suspended or terminated at any time by the Committee.

13. ADMINISTRATION.

- (a) This Plan shall be administered by the Administrator. The Administrator shall interpret the Plan, establish regulations to further the purposes of the Plan and take any other action necessary to the proper operation of the Plan. To the extent authorized by the Administrator, any action required to be taken by a Participant may be taken in writing, by electronic transmission, by telephone, or by facsimile. Prior to paying a Benefit under the Plan, the Administrator may require the Participant, former Participant or Beneficiary to provide such information or material as the Administrator, in its sole discretion, shall deem necessary to make any determination it may be required to make under the Plan. The Administrator may withhold payment of a Benefit under the Plan until it receives all such information and material and is reasonably satisfied of its correctness and genuineness. The Administrator may delegate all or any of its responsibilities and powers to any persons selected by it, including designated officers of employees of the Company or to the Committee.
- (b) If for any reason a Benefit payable under this Plan is not paid when due, the Participant or Beneficiary may file a written claim with the committee appointed by the Committee to review claims for benefits under the Plan (the "Claims Committee"). If the claim is denied or no response is received within forty-five (45) days after the date on which the claim was filed with the Claims Committee (in which case the claim will be to have been denied), the Participant or Beneficiary may appeal the denial to the Committee within sixty (60) days of receipt of written notification of the denial or the end of the forty-five day period, whichever occurs first. In pursuing an appeal, the Participant or Beneficiary may request that the Committee review the denial, may review pertinent documents, and may submit issues and documents in writing to the Committee. A decision on appeal will be made within sixty (60) days after the appeal is made, unless special circumstances require the Committee to extend the period for another sixty (60) days.

14. <u>NOTICES</u>. All notices or election required under the Plan must be in writing. A notice or election shall be deemed delivered if it is delivered personally or sent registered or certified mail to the person at the person's last known business address.

15. WAIVER. The waiver of a breach of any provision in this Plan does not operate as and may not be construed as a waiver of any later breach.

16. <u>CONSTRUCTION</u>. This Plan shall be adopted and maintained according to the laws of the Commonwealth of Virginia (except its choice-of-law rules and except to the



extent that such laws are preempted by applicable federal law). Headings and captions are only for convenience; they do not have substantive meaning. If a provision of this Plan is not valid or enforceable, the validity or enforceability of any other provision shall not be affected. Use of one gender includes all, and the singular and plural include each other.

IN WITNESS WHEREOF, this instrument has been executed this 28th day of May, 2004.

CAPITAL ONE FINANCIAL CORPORATION

By /s/ Matthew Schuyler

Matthew Schuyler Executive Vice President, Human Resources and Chair, Benefits Committee

CAPITAL ONE FINANCIAL CORPORATION

CAPITAL ONE BANK

CAPITAL ONE, F.S.B.

CAPITAL ONE BANK (EUROPE) PLC

\$750,000,000

CREDIT AGREEMENT

Dated as of June 29, 2004

J.P. MORGAN SECURITIES INC. as Book Manager and Lead Arranger

BANK OF AMERICA, N.A., BARCLAYS BANK PLC, CITIBANK, N.A., CREDIT SUISSE FIRST BOSTON, DEUTSCHE BANK AG, NEW YORK BRANCH, LEHMAN COMMERCIAL PAPER INC., MORGAN STANLEY BANK and WACHOVIA BANK, NATIONAL ASSOCIATION as Syndication Agents

> JPMORGAN CHASE BANK as Administrative Agent

Page

	—
SECTION 1. Definitions and Accounting Matters	1
1.01 Certain Defined Terms	1
1.02 Accounting Terms and Determinations	20
1.03 Currencies and Types of Loans	21
1.04 EMU	21
SECTION 2. Commitments, Loans, and Prepayments	22
2.01 Syndicated Loans	22
2.02 Borrowings of Syndicated Loans	23
2.03 Money Market Loans	23
2.04 Changes of Commitments	28
2.05 Fees	28
2.06 Lending Offices	29
2.07 Several Obligations; Remedies Independent	29
2.08 Evidence of Debt	30
2.09 Prepayments	31
2.10 Increases in Commitments	31
2.11 Undertaking of COB	33
2.12 Extensions of Commitments	34
SECTION 3. Payments of Principal and Interest	36
3.01 Repayment of Loans	36
3.02 Interest	36
SECTION 4. Payments; Pro Rata Treatment; Computations; Etc.	37
4.01 Payments	37
4.02 Pro Rata Treatment	38
4.03 Computations	38
4.04 Minimum Amounts	38
4.05 Certain Notices	39
4.06 Non-Receipt of Funds by the Administrative Agent	39
4.07 Sharing of Payments, Etc.	40
SECTION 5. Yield Protection, Etc.	41
5.01 Additional Costs	41
5.02 Limitation on Types of Loans	44
5.03 Illegality; Agreed Alternative Currencies	44
5.04 Treatment of Affected Loans	45
5.05 Compensation	45
5.06 Taxes	46
5.07 Replacement of Lenders	49
	15

SECTION 6. Conditions Precedent	49
6.01 Conditions to Effectiveness	49
6.02 Initial and Subsequent Loans	51
SECTION 7. Representations and Warranties	52
7.01 Corporate Existence	52
7.02 Financial Condition	52
7.03 Litigation	52
7.04 No Breach	52
7.05 Action	53
7.06 Approvals	53
7.07 Use of Credit	53
7.08 ERISA	53
7.09 Taxes	53
7.10 Investment Company Act	54
7.11 Public Utility Holding Company Act	54
7.12 Environmental Matters	54
7.13 True and Complete Disclosure	54
SECTION 8. Covenants	55
8.01 Financial Statements Etc.	55
8.02 Litigation	59
8.03 Existence, Etc.	60
8.04 Insurance	61
8.05 Prohibition of Fundamental Changes	61
8.06 Limitation on Liens	62
8.07 Financial Covenants	63
8.08 Regulatory Capital	63
8.09 Lines of Business	64
8.10 Use of Proceeds	64
SECTION 9. Events of Default	64
SECTION 10. The Administrative Agent	68
10.01 Appointment, Powers and Immunities	68
10.02 Reliance by Administrative Agent	69
10.03 Defaults	69
10.04 Rights as a Lender	70
10.05 Indemnification	70
10.06 Non-Reliance on Administrative Agent and Other Lenders	70
10.07 Failure to Act	71
10.08 Resignation or Removal of Administrative Agent	71
10.09 Co-Agents; Etc.	71

Credit Agreement

ii

SECTION 11. M	iscellaneous	72	
11.01 Waiv	ver in the second s	72	
11.02 Notic	ces	72	
11.03 Expe	enses, Etc.	73	
11.04 Ame	ndments, Etc.	73	
11.05 Succ	essors and Assigns	74	
11.06 Assig	gnments and Participations	75	
11.07 Surv	77		
11.08 Capti	ions	78	
11.09 Counterparts			
11.10 Governing Law; Submission to Jurisdiction		78	
11.11 Waiver of Jury Trial		78	
11.12 Treat	tment of Certain Information; Confidentiality	78	
11.13 Judg	ment Currency	79	
11.14 USA	PATRIOT Act	80	
SCHEDULE 2.0	1 — Commitments		
SCHEDULE 7.03			
EXHIBIT A-1	— Form of Note		
EXHIBIT A-2	— Form of Money Market Note		
EXHIBIT B-1	 Form of Opinion of Clifford Chance, special U.S. counsel to the 		
	Borrowers		
EXHIBIT B-2	 Form of Opinion of Clifford Chance, special English counsel to the Borrowers 		
EXHIBIT B-3	— Form of Opinion of John G. Finneran, Jr., Esq., or Frank R. Borchert III,		
	Esq., counsel to the Borrowers		
EXHIBIT C	 Form of Opinion of Milbank, Tweed, Hadley & McCloy LLP, special 		
	New York counsel to JPMorgan		
EXHIBIT D	 Form of Notice of Borrowing of Syndicated Loans 		
EXHIBIT E	— Form of Money Market Quote Request		

Credit Agreement

_ _

EXHIBIT F

EXHIBIT G

EXHIBIT H

EXHIBIT I EXHIBIT J Form of Money Market Quote Form of Confidentiality Agreement

Form of Assignment and Assumption

Form of Commitment Increase Letter

Form of Drawing Certificate

iii

CREDIT AGREEMENT dated as of June 29, 2004 among:

CAPITAL ONE FINANCIAL CORPORATION, a corporation organized under the laws of the State of Delaware ("COFC");

CAPITAL ONE BANK, a bank organized under the laws of the Commonwealth of Virginia ("COB");

CAPITAL ONE, F.S.B., a Federal savings bank organized under the laws of the United States of America ("FSB");

CAPITAL ONE BANK (EUROPE) PLC, a corporation organized under the laws of England ("<u>COBE</u>"; each of COFC, COB, FSB and COBE is herein referred to as a "<u>Borrower</u>" and, collectively, as the "<u>Borrowers</u>");

each lender that is a signatory hereto identified under the caption "LENDERS" on the signature pages hereto and each lender that becomes a "Lender" after the date hereof pursuant to Section 11.06(b) hereof (individually, a "Lender" and, collectively, the "Lenders"); and

JPMORGAN CHASE BANK, as agent for the Lenders (in such capacity, together with its successors in such capacity, the "Administrative Agent").

The Borrowers have requested that the Lenders agree to make loans to them in an aggregate amount, subject to Section 2.10 hereof, up to but not exceeding \$750,000,000 at any one time outstanding, to be used for general corporate purposes, and the Lenders and the Administrative Agent are willing to make such loans, on and subject to the terms and conditions provided herein.

Accordingly, the parties hereto hereby agree as follows:

SECTION 1. Definitions and Accounting Matters.

1.01 <u>Certain Defined Terms</u>. As used herein, the following terms shall have the following meanings (all terms defined in this Section 1.01 or in other provisions of this Agreement in the singular to have the same meanings when used in the plural and <u>vice versa</u>):

"<u>Administrative Agent's Account</u>" shall mean (a) in respect of (i) Dollars, the account of the Administrative Agent most recently designated by the Administrative Agent for such purpose by notice to the Lenders, (ii) Pounds Sterling, account number 36254290, maintained by JPMorgan with J.P. Morgan Europe Limited, London SWIFT CHASGB22, Sort Code: 40-52-06C, or (iii) Euro, account number 6001600037, maintained by JPMorgan with JPMorgan Chase Bank, Frankfurt, SWIFT CHASDEFX, Favour: J.P. Morgan Europe Limited,

London SWIFT CHASGB22, or (b) any other account in respect of any Alternative Currency as the Administrative Agent shall designate in a notice to the Borrowers and the Lenders.

"Administrative Questionnaire" shall mean an Administrative Questionnaire in a form supplied by the Administrative Agent.

"<u>Affiliate</u>" shall mean, with respect to any specified Person, any other Person that directly or indirectly controls, or is under common control with, or is controlled by, the specified Person. As used in this definition, "<u>control</u>" (including, with its correlative meanings, "<u>controlled by</u>" and "<u>under common control</u> with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise). Notwithstanding the foregoing, (a) no individual shall be an Affiliate of a specified Person solely by reason of his or her being a director, officer or employee of such specified Person or any of its Subsidiaries and (b) a Person and its Subsidiaries shall not be Affiliates of one another.

"<u>Agreed Alternative Currency</u>" shall mean, at any time, either of Euros and Pounds Sterling, so long as at such time, (a) such currency is dealt with in the London interbank deposit market, (b) such currency is freely transferable and convertible into Dollars in the London foreign exchange market and (c) no central bank or other governmental authorization in the country of issue of such currency is required to permit use of such currency by any Lender for making any Loan hereunder and/or to permit the relevant Borrower to borrow and repay the principal thereof and to pay the interest thereon, unless such authorization has been obtained.

"<u>Alternative Currency</u>" shall mean, at any time, any Agreed Alternative Currency and any other currency (other than Dollars) so long as at such time, (a) such currency is dealt with in the London interbank deposit market, (b) such currency is freely transferable and convertible into Dollars in the London foreign exchange market and (c) no central bank or other governmental authorization in the country of issue of such currency is required to permit use of such currency by any Lender for making any Loan hereunder and/or to permit the relevant Borrower to borrow and repay the principal thereof and to pay the interest thereon, unless such authorization has been obtained.

Credit Agreement

"<u>Applicable Facility Fee Percentage</u>", "<u>Applicable Margin</u>" with respect to Base Rate Loans, "<u>Applicable Margin</u>" with respect to Eurocurrency Loans and "<u>Applicable Utilization Fee Percentage</u>" shall mean, for any day, the respective rate per annum set forth in the table below opposite the Rating Level prevailing on such day under the caption "Applicable Facility Fee Percentage", "Applicable Margin with respect to Base Rate Loans" "Applicable Margin with respect to Eurocurrency Loans" or "Applicable Utilization Fee Percentage", as the case may be:

Rating Level	Applicable Facility Fee Percentage	Applicable Margin with respect to Eurocurrency Loans	Applicable Margin with respect to Base Rate Loans	Applicable Utilization Fee Percentage
Rating Level 1	0.100%	0.400%	0.000%	0.125%
Rating Level 2	0.125%	0.500%	0.000%	0.125%
Rating Level 3	0.150%	0.600%	0.000%	0.125%
Rating Level 4	0.175%	0.800%	0.000%	0.250%
Rating Level 5	0.225%	1.000%	0.250%	0.250%
Rating Level 6	0.350%	1.375%	0.750%	0.500%
Rating Level 7	0.500%	1.750%	1.250%	0.500%

Each change in the Applicable Facility Fee Percentage, Applicable Margin with respect to Eurocurrency Loans, Applicable Margin with respect to Base Rate Loans and the Applicable Utilization Fee Percentage resulting from a change in the Debt Rating shall become effective on the date of announcement or publication by the relevant Rating Agency of a change in the Debt Rating or, in the absence of such announcement or publication, on the effective date of such change.

With respect to any utilization fee payable under Section 2.05(b) hereof, the Applicable Utilization Fee Percentage on any date of determination shall be computed by reference to the Rating Level of the Borrower having outstanding borrowings on such day, weighted proportionately to the aggregate outstanding principal amount of such borrowings by the respective Borrowers.

"<u>Applicable Lending Office</u>" shall mean, for each Lender and for each Type and Currency of Loan, the "Lending Office" of such Lender (or of an affiliate of such Lender) designated for such Type and Currency of Loan in its Administrative Questionnaire or such other offices of such Lender (or of an affiliate of such Lender) as such Lender may from time to time specify to the Administrative Agent and the Borrowers as the office by which its Loans of such Type and Currency are to be made and maintained.

Credit Agreement

"<u>Assignment and Assumption</u>" shall mean an assignment and assumption agreement entered into by a Lender and an assignee (with the consent of any Person whose consent is required by Section 11.06(b) hereof), and accepted by the Administrative Agent, in the form of Exhibit H or any other form approved by the Administrative Agent.

"<u>Average</u>", as used in Section 2.05 hereof with respect to the aggregate outstanding principal amount of any Loans or the aggregate amount of any Commitments, shall mean, for any Computation Period, the average aggregate outstanding principal amount of such Loans or the average aggregate amount of such Commitments, as the case may be, over such Computation Period (excluding the last day of such Computation Period).

"<u>Bank Regulatory Authority</u>" shall mean the Board of Governors of the Federal Reserve System, the Comptroller of the Currency, the Federal Deposit Insurance Corporation, the FSA and all other relevant bank regulatory authorities (including, without limitation, relevant state bank regulatory authorities) having jurisdiction over a Borrower.

"Bankruptcy Code" shall mean the Federal Bankruptcy Code of 1978, as amended from time to time.

"<u>Base Rate</u>" shall mean, for any day, a rate per annum equal to the higher of (a) the Federal Funds Rate for such day <u>plus</u> 1/2 of 1% and (b) the Prime Rate for such day. Each change in any interest rate provided for herein based upon the Base Rate resulting from a change in the Base Rate shall take effect at the time of such change in the Base Rate.

"Base Rate Loans" shall mean Syndicated Loans that bear interest at rates based upon the Base Rate.

"Basic Documents" shall mean this Agreement and the Notes.

"<u>Basle Accord</u>" shall mean the proposals for risk-based capital framework described by the Basle Committee on Banking Regulations and Supervisory Practices in its paper entitled "International Convergence of Capital Measurement and Capital Standards" dated July 1988, as amended, modified and supplemented and in effect from time to time or any replacement thereof.

"Business Day" shall mean any day (a) on which commercial banks are not authorized or required to close in New York City or London, (b) if such day relates to the giving of notices or quotes in connection with a LIBOR Auction in respect of a Loan denominated in Dollars or to a borrowing of, a payment or prepayment of principal of or interest on, or the Interest Period for, a Eurocurrency Loan or LIBOR Market Loan denominated in Dollars or a notice by a Borrower with respect to any such borrowing, payment, prepayment or Interest Period, that is also a day on which dealings in Dollar deposits are carried out in the London interbank market, (c) if such day relates to the giving of notices or quotes in connection with a LIBOR Auction in respect of a Loan denominated in an Alternative Currency other than the Euro or to a borrowing of, a payment or prepayment of principal of or interest on, or the Interest

Credit Agreement

Period for, a Eurocurrency Loan or LIBOR Market Loan denominated in an Alternative Currency other than the Euro or a notice by a Borrower with respect to any such borrowing, payment, prepayment or Interest Period, that is also a day on which commercial banks and foreign exchange markets settle payments in the Principal Financial Center for the Currency in which such Loan is denominated and (d) if such day relates to the giving of notices or quotes in connection with a LIBOR Auction in respect of a Loan denominated in Euros or to a borrowing of, a payment or prepayment of principal of or interest on, or the Interest Period for, a Eurocurrency Loan or LIBOR Market Loan denominated in Euros, or a notice by a Borrower with respect to any such borrowing, payment, prepayment or Interest Period, that is also a TARGET Business Day.

"<u>Capital Lease Obligations</u>" shall mean, for any Person, all obligations of such Person to pay rent or other amounts under a lease of (or other agreement conveying the right to use) Property to the extent such obligations are required to be classified and accounted for as a capital lease on a balance sheet of such Person under GAAP, and, for purposes of this Agreement, the amount of such obligations shall be the capitalized amount thereof, determined in accordance with GAAP.

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time.

"<u>COFC Cumulative Equity Proceeds</u>" shall mean, as of any date of determination, the aggregate amount of all cash received on or prior to such date of determination by COFC and its Subsidiaries in respect of any Equity Issuance effected after March 31, 2004, net of expenses incurred by COFC and its Subsidiaries in connection therewith.

"<u>COFC Cumulative Net Income</u>" shall mean, as of any date of determination, the aggregate net operating income of COFC and its consolidated Subsidiaries (determined on a consolidated basis without duplication in accordance with GAAP) for each fiscal quarter of COFC (a) commencing with the fiscal quarter ended June 30, 2004 and (b) ending with the fiscal quarter most recently ended on or prior to such date of determination; <u>provided</u> that COFC Cumulative Net Income shall be determined exclusive of any fiscal quarter of COFC for which the net operating income of COFC and its consolidated Subsidiaries (determined on a consolidated basis without duplication in accordance with GAAP) is less than zero.

"<u>Commitment</u>" shall mean, with respect to each Lender, the commitment of such Lender to make Loans, as such commitment may be (a) reduced from time to time pursuant to Section 2.04 hereof, (b) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.06 hereof, (c) increased pursuant to Section 2.10 hereof or extended from time to time pursuant to Section 2.12 hereof. The initial amount of each Lender's Commitment is set forth on Schedule 2.01, or in the Assignment and Assumption pursuant to which such Lender shall have assumed its Commitment, as applicable. The initial aggregate amount of the Commitments is \$750,000,000.

Credit Agreement

"Commitment Increase Date" shall have the meaning assigned to such term in Section 2.10(b) hereof.

"Commitment Increase Letter" shall have the meaning assigned to such term in Section 2.10(b) hereof.

"<u>Commitment Termination Date</u>" shall mean June 29, 2007, as may be extended pursuant to Section 2.12 hereof; <u>provided</u> that if any such day is not a Business Day, then the Commitment Termination Date shall be the immediately preceding Business Day.

"<u>Computation Period</u>" shall mean, with respect to any utilization fee payable under Section 2.05(b) hereof, (a) the period from and including the date hereof to and including the first day on which such utilization fee is payable under Section 2.05(c) hereof and (b) thereafter, each period from and including the last day of the immediately preceding Computation Period to and including the next succeeding day on which such utilization fee is payable under Section 2.05(c) hereof.

"Currency" shall mean Dollars or any Alternative Currency.

"Debt Rating" shall mean, as of any date of determination thereof and with respect to any Borrower, the ratings most recently published by the Rating Agencies relating to the unsecured, unsupported senior long-term debt obligations of such Borrower; <u>provided</u> that (a) the Debt Rating on any date of determination with respect to FSB or COBE shall be deemed to be the higher of (i) the Debt Rating on such date applicable to FSB or COBE, respectively, and (ii) the Debt Rating on such date applicable to COB, (b) if a rating is not at any time assigned by a Rating Agency to the unsecured, unsupported senior long-term debt obligations of COB and (c) if a rating is not at any time assigned by at least two Rating Agencies to the unsecured, unsupported senior long-term debt obligations of COB, the Debt Rating of COB will be deemed to fall in Rating Level 7.

"Default" shall mean an Event of Default or an event that with notice or lapse of time or both would become an Event of Default.

"Defaulting Lender" shall have the meaning assigned to such term in Section 11.04 hereof.

"Delinquency Ratio" shall mean, on any date and with respect to COFC, the ratio of (a) all Past Due Receivables with respect to COFC on such date to (b) the aggregate amount of all Managed Receivables with respect to COFC on such date.

"Dollar Equivalent" shall mean, with respect to any Loan denominated in an Alternative Currency, the amount of Dollars that would be required to purchase the amount of the Alternative Currency of such Loan on the date such Loan is requested (or, in the case of

Credit Agreement

Money Market Loans, the date of the related Money Market Quote Request) or (with respect to any determination made under Section 2.01(c) hereof) on the date of any determination referred to in said Section, based upon the arithmetic mean (rounded upwards, if necessary, to the nearest four decimal places), as determined by the Administrative Agent, of the spot selling rate at which the Reference Lenders offer to sell such Alternative Currency for Dollars in the London foreign exchange market at approximately 11:00 a.m. London time for delivery two Business Days later.

"Dollars" and "§" shall mean lawful money of the United States of America.

"<u>Double Leverage Ratio</u>" shall mean, on any date, the ratio of (a) the sum of (i) Intangibles (other than goodwill) with respect to COFC on such date <u>plus</u> (ii) the aggregate investment of COFC on such date in the capital stock of its Subsidiaries as reported pursuant to Section 8.01(a) or 8.01(b) hereof (including COFC's interest in undistributed earnings of its Subsidiaries), to (b) Net Worth on such date.

"Effective Date" shall mean the first date on which the Administrative Agent notifies the Borrowers and the initial Lenders that all of the conditions set forth in Section 6.01 hereof have been satisfied.

"EMU" shall mean Economic and Monetary Union as contemplated in the Treaty on European Union, as amended and in effect from time to time.

"<u>EMU Legislation</u>" shall mean legislative measures of the European Council (including without limitation European Council regulations) for the introduction of, changeover to or operation of a single or unified European currency, being in part the implementation of the third stage of EMU.

"<u>Environmental Laws</u>" shall mean any and all present and future Federal, state, local and foreign laws, rules or regulations, and any orders or decrees, in each case as now or hereafter in effect, relating to the regulation or protection of the environment or to emissions, discharges, releases or threatened releases of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes into the indoor or outdoor environment, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes.

"Equity Issuance" shall mean (a) any issuance or sale by COFC or any of its Subsidiaries of (i) any of its capital stock, (ii) any warrants or options exercisable in respect of its capital stock (other than any capital stock of COFC or any warrants or options to purchase any capital stock of COFC that are issued to directors, officers or employees of COFC or any of its Subsidiaries pursuant to employee benefit plans established in the ordinary course of business, or any capital stock of COFC issued upon the exercise of any such warrants or options) or (iii) any other security or instrument representing an equity interest (or the right to obtain any equity interest) in COFC or any of its Subsidiaries or (b) the receipt by COFC or any of its Subsidiaries from any Person not a shareholder of COFC of any capital contribution (whether or not evidenced by any equity security issued by the recipient of such contribution); provided that

Credit Agreement

Equity Issuance shall not include (i) any such issuance or sale by any Subsidiary of COFC to COFC or any Wholly Owned Subsidiary of COFC or (ii) any capital contribution by COFC or any Wholly Owned Subsidiary of COFC to any Subsidiary of COFC.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time.

"<u>ERISA Affiliate</u>" shall mean any corporation or trade or business that is a member of any group of organizations (i) described in Section 414(b) or (c) of the Code of which any Borrower is a member and (ii) solely for purposes of potential liability under Section 302(c)(11) of ERISA and Section 412(c)(11) of the Code and the lien created under Section 302(f) of ERISA and Section 412(n) of the Code, described in Section 414(m) or (o) of the Code of which any Borrower is a member.

"Euro" shall mean the single currency of Participating Member States of the European Union.

"Eurocurrency Loans" shall mean Syndicated Loans that bear interest at rates based on rates referred to in the definition of "Fixed Base Rate" in this Section 1.01.

"<u>Eurocurrency Rate</u>" shall mean, for any Eurocurrency Loan for the Interest Period therefor, a rate per annum (rounded upwards, if necessary, to the nearest four decimal places) determined by the Administrative Agent to be equal to the Fixed Base Rate for such Loan for such Interest Period.

"Events of Default" shall have the meaning assigned to such term in Section 9 hereof.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended from time to time.

"<u>Excluded Representations</u>" shall mean the representations and warranties made in (a) the last sentence of Section 7.02 hereof and (b) Section 7.03 hereof (but only insofar as the representation and warranty in Section 7.03 hereof relates to proceedings that could have a Material Adverse Effect of the type referred to clause (a), (b) or (e) of the definition thereof in this Section 1.01, but not of the type referred to in clause (c) or (d) of the definition thereof in this Section 1.01).

"<u>Existing Credit Agreements</u>" shall mean, collectively, (i) the Credit Agreement dated as of May 5, 2003, as heretofore amended, among COB, FSB, COFC, COBE, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent, and (ii) the Multicurrency Credit Agreement made on 11 August 2000 among COFC, COB, certain financial institutions and Barclays Bank plc, as Agent.

"FDIA" shall mean the Federal Deposit Insurance Act, as amended from time to time.

Credit Agreement

"<u>Federal Funds Rate</u>" shall mean, for any day, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day, <u>provided</u> that (a) if the day for which such rate is to be determined is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day and (b) if such rate is not so published for any Business Day, the Federal Funds Rate for such Business Day on such transactions as determined by the Administrative Agent.

"Fixed Base Rate" shall mean, with respect to any Fixed Rate Loan denominated in any Currency for the Interest Period therefor, the rate for deposits in such Currency for a period comparable to such Interest Period which appears on Page 3750 of the Telerate Service (or on any successor or substitute page of such Service, or any successor to or substitute for such Service, providing rate quotations comparable to those currently provided on such page of such Service, as determined by the Administrative Agent from time to time for purposes of providing quotations of interest rates applicable to deposits in such Currency in the London interbank market) as of 11:00 a.m., London time, on the day that is two London Banking Days preceding the first day of such Interest Period; provided that, if such rate does not appear on the relevant Telerate Service Page, the "Fixed Base Rate" shall be the arithmetic mean (rounded upwards, if necessary, to the nearest four decimal places), as determined by the Administrative Agent, of the rates per annum quoted by the respective Reference Lenders at approximately 11:00 a.m. London time (or as soon thereafter as practicable) on the day that is two London Banking Days prior to (or in the case of a Fixed Rate Loan denominated in Euros, on such other date as is customary in the relevant interbank market) the first day of such Interest Period for the offering by the respective Reference Lenders to leading banks in the London interbank market of deposits denominated in such Currency having a term comparable to such Interest Period and in an amount comparable to the principal amount of such Fixed Rate Loan to be made by the respective Reference Lenders. If any Reference Lender is not participating in any Fixed Rate Loans during the Interest Period therefor, the Fixed Base Rate for such Loans for such Interest Period shall be determined by reference to the amount of such Loans that such Reference Lender would have made or had outstanding had it been participating in such Loan; provided that in the case of any LIBOR Market Loan, the Fixed Base Rate for such Loan shall be determined with reference to deposits of \$25,000,000 (or its equivalent in any Alternative Currency). If any Reference Lender does not timely furnish such information for determination of any Fixed Base Rate, the Administrative Agent shall determine such Fixed Base Rate on the basis of the information timely furnished by the remaining Reference Lenders.

"Fixed Rate Loans" shall mean Eurocurrency Loans and, for the purposes of the definition of "Fixed Base Rate" in this Section 1.01 and in Section 5 hereof, LIBOR Market Loans.

"<u>Foreign Currency Equivalent</u>" shall mean, with respect to any amount in Dollars, the amount of any Alternative Currency that could be purchased with such amount of Dollars using the reciprocal of the foreign exchange rate(s) specified in the definition of the term "Dollar Equivalent", as determined by the Administrative Agent.

"FSA" shall mean the Financial Services Authority in the United Kingdom.

"GAAP" shall mean generally accepted accounting principles in the United States of America, applied on a basis consistent with those that, in accordance with the second sentence of Section 1.02(a) hereof, are to be used in making the calculations for purposes of determining compliance with this Agreement.

"Guarantee" shall mean a guarantee, an endorsement, a contingent agreement to purchase or to furnish funds for the payment or maintenance of, or otherwise to be or become contingently liable under or with respect to, the Indebtedness, other obligations, net worth, working capital or earnings of any Person, or a guarantee of the payment of dividends or other distributions upon the stock or equity interests of any Person, or an agreement to purchase, sell or lease (as lessee or lessor) Property, products, materials, supplies or services primarily for the purpose of enabling a debtor to make payment of such debtor's obligations or an agreement to assure a creditor against loss, and including, without limitation, causing a bank or other financial institution to issue a letter of credit or other similar instrument for the benefit of another Person, but excluding endorsements for collection or deposit in the ordinary course of business. The terms "Guarantee" and "Guaranteed" used as a verb shall have a correlative meaning.

"Indebtedness" shall mean, for any Person: (a) obligations created, issued or incurred by such Person for borrowed money (whether by loan, the issuance and sale of debt securities or the sale of Property to another Person subject to an understanding or agreement, contingent or otherwise, to repurchase such Property from such Person); (b) obligations of such Person to pay the deferred purchase or acquisition price of Property or services, other than trade accounts payable (other than for borrowed money) arising, and accrued expenses incurred, in the ordinary course of business so long as such trade accounts payable are payable within 90 days of the date the respective goods are delivered or the respective services are rendered; (c) Indebtedness of others secured by a Lien on the Property of such Person, whether or not the respective indebtedness so secured has been assumed by such Person; (d) non-contingent obligations of such Person (and, for the purposes of Section 9(b) hereof, all contingent obligations of such Person) in respect of letters of credit, bankers' acceptances or similar instruments issued or accepted by banks and other financial institutions for account of such Person; (e) Capital Lease Obligations of such Person; and (f) Indebtedness of others Guaranteed by such Person.

"<u>Insured Subsidiary</u>" shall mean any insured depositary institution (as defined in 12 U.S.C. §1813(c) (or any successor provision), as amended, re-enacted or redesignated from time to time), that is controlled (within the meaning of 12 U.S.C. §1841 (or any successor provision), as amended, re-enacted or redesignated from time to time) by a Borrower.

"Intangibles" shall mean, as at any date and with respect to any Borrower, the aggregate amount (to the extent reflected in determining the consolidated stockholders' equity of such Borrower and its consolidated Subsidiaries) of (a) all write-ups (other than write-ups resulting from foreign currency translations and write-ups of assets of a going concern business made within 12 months after the acquisition of such business) subsequent to September 30, 1996 in the book value of any asset by such Borrower or any of its consolidated Subsidiaries, (b) all Investments in unconsolidated Subsidiaries and all equity investments in Persons that are not Subsidiaries and (c) all unamortized debt discount and expense, unamortized deferred charges, goodwill, patents, trademarks, service marks, trade names, anticipated future benefit of tax loss carry-forwards, copyrights, organization or developmental expenses and other intangible assets.

"Interest Period" shall mean:

(a) with respect to any Eurocurrency Loan, the period commencing on the date such Eurocurrency Loan is made and ending on the numerically corresponding day in the first, second, third or sixth calendar month thereafter, as the relevant Borrower may select as provided in Section 4.05 hereof, except that each Interest Period that commences on the last Business Day of a calendar month (or on any day for which there is no numerically corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month;

(b) with respect to any Set Rate Loan, the period commencing on the date such Set Rate Loan is made and ending on any Business Day not less than seven days thereafter, as the relevant Borrower may select as provided in Section 2.03(b) hereof;

(c) with respect to any LIBOR Market Loan, the period commencing on the date such LIBOR Market Loan is made and ending on the numerically corresponding day in the first, second, third or sixth calendar month thereafter, as the relevant Borrower may select as provided in Section 2.03(b) hereof, except that each Interest Period that commences on the last Business Day of a calendar month (or any day for which there is no numerically corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month; and

(d) with respect to any Base Rate Loan, the period commencing on the date such Base Rate Loan is made and ending on the earlier of the first Quarterly Date thereafter and the Commitment Termination Date.

Notwithstanding the foregoing: (i) if any Interest Period for any Loan would otherwise end after the Commitment Termination Date, such Interest Period shall end on the Commitment Termination Date; (ii) each Interest Period that would otherwise end on a day that is not a Business Day shall end on the next succeeding Business Day unless such succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day; (iii) except as provided in clause (v) below, no Interest Period for any Loan (other than a Base Rate Loan or a Set Rate Loan) shall have a duration of less than one

month and, if the Interest Period for any Eurocurrency or LIBOR Market Loan would otherwise be a shorter period, such Loan shall not be available hereunder for such period; (iv) no Borrower may select an Interest Period for a Loan in any Alternative Currency which would extend beyond the date on which such Alternative Currency ceases to be legal tender in its respective country; and (v) if each Lender shall have notified the Administrative Agent that the requested Interest Period is available (but subject to the foregoing clauses (i) and (ii)), a Eurocurrency Loan or LIBOR Market Loan may be made available for a specified Interest Period of less than one month or for an Interest Period of nine or 12 months; <u>provided</u> that no Loan shall be made to FSB or COBE with an Interest Period in excess of six months.

"Investment" shall mean, for any Person: (a) the acquisition (whether for cash, Property, services or securities or otherwise) of capital stock, bonds, notes, debentures, partnership or other ownership interests or other securities of any other Person or any agreement to make any such acquisition (including, without limitation, any "short sale" or any sale of any securities at a time when such securities are not owned by the Person entering into such sale); (b) the making of any deposit with, or advance, loan or other extension of credit to, any other Person (including the purchase of Property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such Property to such Person), but excluding any such advance, loan or extension of credit having a term not exceeding 90 days arising in connection with the sale of inventory or supplies by such Person in the ordinary course of business; or (c) the entering into of any Guarantee of, or other contingent obligation with respect to, Indebtedness or other liability of any other Person and (without duplication) any amount committed to be advanced, lent or extended to such Person.

"JPMorgan" shall mean JPMorgan Chase Bank.

"LIBO Margin" shall have the meaning assigned to such term in Section 2.03(c)(ii)(C) hereof.

"LIBOR Auction" shall mean a solicitation of Money Market Quotes setting forth LIBO Margins based on the Eurocurrency Rate pursuant to Section 2.03 hereof.

"<u>LIBOR Market Loans</u>" shall mean Money Market Loans the interest rates on which are determined on the basis of Eurocurrency Rates pursuant to a LIBOR Auction.

"Lien" shall mean, with respect to any Property, any mortgage, lien, pledge, charge, security interest, encumbrance or arrangement for priority or preference of any kind in respect of such Property. For purposes of this Agreement, a Person shall be deemed to own subject to a Lien any Property that it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement (other than an operating lease) or, in the case of any security, any third party right to purchase, in each case relating to such Property.

"Loans" shall mean Syndicated Loans and Money Market Loans.

Credit Agreement

"<u>Local Time</u>" shall mean, with respect to any Loan denominated in or any payment to be made in any Currency, the local time in the Principal Financial Center for the Currency in which such Loan is denominated or such payment is to be made.

"London Banking Day" shall mean any day on which commercial banks are open for business (including dealings in foreign exchange and foreign currency deposits) in London, England.

"<u>Majority Lenders</u>" shall mean, subject to the last paragraph of Section 11.04 hereof, Lenders having more than 50% of the aggregate amount of the Commitments or, if the Commitments shall have terminated, Lenders holding more than 50% of the aggregate unpaid principal amount of the Loans.

"<u>Managed Receivables</u>" shall mean, on any date and with respect to any U.S. Borrower, the sum for such U.S. Borrower and its consolidated Subsidiaries (determined on a consolidated basis without duplication in accordance with GAAP) of (a) all on-balance sheet credit card loans and other finance receivables <u>plus</u> (b) all on-balance sheet credit card loans and other finance receivables held for securitization <u>plus</u> (c) all securitized credit card loans and other finance receivables; <u>provided</u> that, as the term "Managed Receivables" is used in the definition of "Tier 1 Capital to Managed Receivables Ratio", clauses (a), (b) and (c) above shall be determined exclusive of securitized on-balance sheet finance receivables.

"<u>Mandatorily Convertible Securities</u>" shall mean the Upper DECS securities issued by COFC on April 23, 2002 pursuant to the Senior Indenture dated as of November 1, 1996, as supplemented by the First Supplemental Indenture, dated as of April 23, 2002, each by and between COFC and BNY Midwest Trust Company, and other securities hereafter issued providing for conversion thereof on substantially the same terms and conditions as such Upper DECS securities.

"Margin Stock" shall mean "margin stock" within the meaning of Regulations T, U and X.

"<u>Material Adverse Effect</u>" shall mean, with respect to a Borrower, a material adverse effect on (a) the Property, business, operations, financial condition or capitalization of such Borrower and its Subsidiaries taken as a whole, (b) the ability of such Borrower to perform its material obligations under the Basic Documents, (c) the validity or enforceability of the obligations of such Borrower under the Basic Documents, (d) the rights and remedies of the Lenders and the Administrative Agent against such Borrower under the Basic Documents or (e) the timely payment of the principal of or interest on the Loans or other amounts payable by such Borrower in connection therewith.

"<u>Material Subsidiary</u>" shall mean, at any time and with respect to any Borrower, any Subsidiary of such Borrower which, at the time any determination is being made, would constitute a "significant subsidiary" of such Borrower as defined in Rule 1-02 of Regulation S-X of the SEC as in effect on the date hereof.

Credit Agreement

"Money Market Borrowing" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"Money Market Loan Limit" shall have the meaning assigned to such term in Section 2.03(c)(ii) hereof.

"Money Market Loans" shall mean the loans provided for by Section 2.03 hereof.

"<u>Money Market Notes</u>" shall mean any promissory notes in substantially the form of Exhibit A-2 hereto issued pursuant to Section 2.08(d) hereof, and all promissory notes delivered in substitution or exchange therefor, in each case as the same shall be modified and supplemented and in effect from time to time.

"<u>Money Market Quote</u>" shall mean an offer in accordance with Section 2.03(c) hereof by a Lender to make a Money Market Loan with one single specified interest rate.

"Money Market Quote Request" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"<u>Multiemployer Plan</u>" shall mean a multiemployer plan defined as such in Section 3(37) of ERISA to which contributions have been made by any Borrower or any ERISA Affiliate and that is covered by Title IV of ERISA.

"<u>Net Worth</u>" shall mean, on any date, the consolidated stockholders' equity of COFC and its consolidated Subsidiaries <u>plus</u> an amount equal to 80% of the face amount of any Mandatorily Convertible securities issued by COFC, all determined as of such date on a consolidated basis without duplication in accordance with GAAP.

"Notes" shall mean the Syndicated Notes and the Money Market Notes.

"Participating Member State" shall mean each country so described in any EMU Legislation.

"Past-Due Receivables" shall mean, on any date with respect to any U.S. Borrower, the sum (determined with respect to such U.S. Borrower and its Subsidiaries on a consolidated basis without duplication in accordance with GAAP) of (a) all Managed Receivables the minimum payments on which are at least 90 days overdue on such date <u>plus</u> (b) all other assets which have been, in accordance with the relevant Borrower's credit policies with respect to such assets, classified as non-performing assets; <u>provided</u> that, Managed Receivables that are credit card loans, whether or not at least 90 days overdue, shall not constitute "Past-Due Receivables" to the extent of any cash balance of the account debtor on such loan on deposit with the creditor (but only to the extent such creditor is entitled under an agreement governing such credit card loan to set off such cash balances against the obligations of the account debtor under such loan and to the extent such cash balances are not subject to any other set-off or deduction by such creditor or any of its affiliates against a matured obligation owing by such debtor).

Credit Agreement

"PBGC" shall mean the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

"<u>Person</u>" shall mean any individual, corporation, company, voluntary association, partnership, limited liability company, joint venture, trust, unincorporated organization or government (or any agency, instrumentality or political subdivision thereof).

"<u>Plan</u>" shall mean an employee benefit or other plan established or maintained by any Borrower or any ERISA Affiliate and that is covered by Title IV of ERISA, other than a Multiemployer Plan.

"<u>Post-Default Rate</u>" shall mean a rate equal to 2% per annum <u>plus</u> the Base Rate as in effect from time to time <u>plus</u> the Applicable Margin for Base Rate Loans, <u>provided</u> that, with respect to principal of a Eurocurrency Loan or a Money Market Loan that shall become due (whether at stated maturity, by acceleration, by optional or mandatory prepayment or otherwise) on a day other than the last day of the Interest Period therefor, the "Post-Default Rate" shall be, for the period from and including such due date to but excluding the last day of such Interest Period, 2% per annum <u>plus</u> the interest rate for such Loan as provided in Section 3.02 hereof and, thereafter, the rate provided for above in this definition.

"Pounds Sterling" shall mean the lawful currency of the United Kingdom.

"Prime Rate" shall mean the rate of interest from time to time announced by JPMorgan at the Principal Office as its prime commercial lending rate.

"<u>Principal Financial Center</u>" shall mean (a) in the case of each Currency identified in Section 1.5 of the 2000 ISDA Definitions published by the International Swaps and Derivatives Association, Inc., the financial center identified in said Section opposite such Currency and (b) in the case of any other Currency, the principal financial center of the country that issues such Currency, as determined by the Administrative Agent.

"Principal Office" shall mean the principal office of JPMorgan, located on the date hereof at 270 Park Avenue, New York, New York 10017.

"Property" shall mean any right or interest in or to property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible.

"Qualifying Bank" shall mean any Lender (a) which is a bank as defined in Section 840A of the Income and Corporation Taxes Act 1988 of the United Kingdom (as such section may be amended from time to time) for the purposes of Section 349 of said Income and Corporation Taxes Act 1988 (as such section may be amended from time to time) making a Loan hereunder or in respect of a Loan made hereunder by a Person that was such a bank at the time that Loan was made and is within the charge to United Kingdom corporation tax with respect to

Credit Agreement

any interest received by it in respect of a Loan hereunder and is beneficially entitled to any interest payments made to it or (b) who is resident (as such term is defined in an appropriate double taxation treaty) in a country with which the United Kingdom has a double taxation treaty giving residents of that country an exemption from United Kingdom taxation on interest and does not carry on business in the United Kingdom through a permanent establishment with which the indebtedness under this Agreement in respect of which interest is paid is effectively connected.

"Quarterly Dates" shall mean the last Business Day of March, June, September and December in each year, the first of which shall be the first such day after the date hereof.

"Rating Agencies" shall mean Moody's Investors Service, Inc. and Standard & Poor's Ratings Services or, in each case, any successor nationally recognized statistical rating organization.

"Rating Levels" shall mean, on any date of determination, (a) Rating Level 1 if the Debt Rating by the Rating Agencies is at least equal to "A3" or "A-", (b) Rating Level 2 if the Debt Rating by the Rating Agencies is at least equal to "Baa1" or "BBB+", but does not fall within Rating Level 1, (c) Rating Level 3 if the Debt Rating Agencies is at least equal to "Baa2" or "BBB", but does not fall within Rating Level 1 or Rating Level 2, (c) Rating Level 4 if the Debt Rating by the Rating Agencies is at least equal to "Baa3" or "BBB-", but does not fall within Rating Level 1, Rating Level 2, (c) Rating Level 3, (e) Rating Level 5 if the Debt Rating by the Rating Agencies is at least equal to "Baa3" or "BBB-", but does not fall within Rating Level 1, Rating Level 2, Rating Level 3, (e) Rating Level 4, (f) Rating Level 6 if the Debt Rating by the Rating Agencies is at least equal to "Ba1" or "BB+", but does not fall within Rating Agencies is at least equal to "Ba3" or "BB-" or if none of the foregoing is applicable. If the Debt Rating of any Rating Agency is below the Debt Rating of the other Rating Agency and the two Debt Ratings shall be equal to or greater than "Baa3" and "BBB-", the "Rating Level" will be determined without regard to the Debt Rating of such Rating Agency with the lower Rating Level. If the Debt Rating of any Rating Agency is below the Debt Rating Agency and either Debt Rating shall be less than "Baa3" or "BBB-", the "Rating Level." will be determined based on the Debt Rating of such Rating Agency with the lower Rating Level."

"<u>Receivables</u>" shall mean, with respect to any Borrower, any amount owing, from time to time, with respect to a credit card, revolving or installment loan account, home equity line of credit or residential mortgage loan account or other receivable owned by such Borrower, including, without limitation, amounts owing to a Borrower or a Subsidiary of a Borrower for payment of goods and services, cash advances, convenience checks, membership fees, finance charges, late charges, credit insurance premiums and cash advance fees and fees relating to additional lending products, and any other receivables arising out of financing transactions by such Borrower; <u>provided</u> that the term "Receivables" shall not include any of the foregoing that is subject to a securitization (whether on or off balance sheet for any Borrower) effected in the

Credit Agreement

ordinary course of business, including any "seller interest" or retained portion of any securitization.

"<u>Reference Lenders</u>" shall mean JPMorgan, Citibank, N.A. and Bank of America, N.A. (or their respective Applicable Lending Offices, as the case may be).

"<u>Regulations A, D, T, U and X</u>" shall mean, respectively, Regulations A, D, T, U and X of the Board of Governors of the Federal Reserve System (or any successor), as the same may be modified and supplemented and in effect from time to time.

"<u>Regulatory Change</u>" shall mean, with respect to any Lender, any change after the date hereof in Federal, state or foreign law or regulations (including, without limitation, Regulation D) or the adoption or making after such date of any interpretation, directive or request applying to a class of banks including such Lender of or under any Federal, state or foreign law or regulations (whether or not having the force of law and whether or not failure to comply therewith would be unlawful) by any court or governmental or monetary authority charged with the interpretation or administration thereof.

"Reserve Requirement" shall mean, for the Interest Period for any Eurocurrency Loan or LIBOR Market Loan, the average maximum rate at which reserves (including, without limitation, any marginal, supplemental or emergency reserves) are required to be maintained during such Interest Period under Regulation D by member banks of the Federal Reserve System in New York City with deposits exceeding one billion Dollars against "Eurocurrency liabilities" (as such term is used in Regulation D). Without limiting the effect of the foregoing, the Reserve Requirement shall include any other reserves required to be maintained by such member banks by reason of any Regulatory Change with respect to (i) any category of liabilities that includes deposits by reference to which the Fixed Base Rate for Eurocurrency Loans or LIBOR Market Loans (as the case may be) is to be determined as provided in the definition of "Fixed Base Rate" in this Section 1.01 or (ii) any category of extensions of credit or other assets that includes Eurocurrency Loans or LIBOR Market Loans.

"<u>Restricted Shares</u>" shall mean, with respect to any Borrower, shares of stock of or other ownership interests in such Borrower or any Subsidiary thereof engaged primarily in the extension of consumer credit to third parties or securitizations of receivables related to such extension of consumer credit, excluding without limitation any such ownership interests of any Borrower in America One Communications, Inc.

"<u>Risk Adjusted Assets</u>" shall mean, on any date and with respect to any U.S. Borrower, the amount, for such U.S. Borrower and its consolidated Subsidiaries (determined on a consolidated basis) on such date, of (i) "weighted risk assets", within the meaning given to such term in the Capital Adequacy Guidelines for State Member Banks published by the Board of Governors of the Federal Reserve System (12 C.F.R. Part 208, Appendix A, as amended, modified and supplemented and in effect from time to time or any replacement thereof) <u>plus</u> (ii) "risk weighted assets", within the meaning given to such term in 12 C.F.R. Part 567.1.

"SEC" shall mean the Securities and Exchange Commission, or any successor agency charged with the administration and enforcement of the Securities Act and the Exchange Act.

"Securities Act" shall mean the Securities Act of 1933, as amended from time to time.

"Set Rate" shall have the meaning assigned to such term in Section 2.03(c)(ii)(D) hereof.

"Set Rate Auction" shall mean a solicitation of Money Market Quotes setting forth Set Rates pursuant to Section 2.03 hereof.

"Set Rate Loans" shall mean Money Market Loans the interest rates on which are determined on the basis of Set Rates pursuant to a Set Rate Auction.

"<u>Subsidiary</u>" shall mean, with respect to any Person, any corporation, partnership, limited liability company or other entity of which at least a majority of the Voting Securities issued by such corporation, partnership, limited liability company or other entity is at the time directly or indirectly owned or controlled by such Person or one or more Subsidiaries of such Person or by such Person and one or more Subsidiaries of such Person; <u>provided</u>, however, that any special purpose subsidiary of COFC or any of its Subsidiaries organized and operated solely to facilitate or conduct securitizations shall not be deemed to be a Subsidiary of COFC or any other Borrower.

"Swap Agreement" shall have the meaning given to such term in Section 101(53B) of the Bankruptcy Code (as in effect on the date hereof).

"Syndicated Loans" shall mean the loans provided for by Section 2.01(a) hereof, which may be Base Rate Loans and/or Eurocurrency Loans.

"<u>Syndicated Notes</u>" shall mean any promissory notes in substantially the form of Exhibit A-1 hereto issued pursuant to Section 2.08(d) hereof, and all promissory notes delivered in substitution or exchange thereof, in each case as the same shall be modified and supplemented and in effect from time to time.

"<u>Syndication Agents</u>" shall mean each of Bank of America, N.A., Barclays Bank plc, Citibank, N.A., Credit Suisse First Boston, Deutsche Bank AG, New York Branch, Lehman Commercial Paper Inc., Morgan Stanley Bank and Wachovia Bank, National Association.

"<u>Tangible Net Worth</u>" shall mean, on any date and with respect to any Borrower, the consolidated stockholders' equity of such Borrower and its consolidated Subsidiaries (<u>provided</u>, that the consolidated stockholders' equity of COFC shall include an amount equal to 80% of the face amount of any Mandatorily Convertible Securities issued by it so long as such Mandatorily Convertible Securities do not, at any time, comprise more than 25% of the Tangible

Credit Agreement

Net Worth of COFC) less Intangibles of such Borrower and its consolidated Subsidiaries, all determined as of such date on a consolidated basis without duplication in accordance with GAAP.

"<u>TARGET Business Day</u>" shall mean any day that is not (i) a Saturday or Sunday, or (ii) any other day on which the Trans-European Real-time Gross Settlement Express Transfer Payment System (or any successor settlement system) is not operating (as determined by the Administrative Agent).

"<u>Tier 1 Capital</u>" shall mean, on any date and with respect to any U.S. Borrower, the amount, for such U.S. Borrower and its consolidated Subsidiaries (determined on a consolidated basis) on such date, of "Tier 1 capital", within the meaning given to such term in the Capital Adequacy Guidelines for State Member Banks published by the Board of Governors of the Federal Reserve System (12 C.F.R. Part 208, Appendix A, as amended, modified and supplemented and in effect from time to time or any replacement thereof).

"<u>Tier 1 Capital to Managed Receivables Ratio</u>" shall mean, on any date and with respect to COFC, the ratio of (a) Tier 1 Capital (determined, for the purposes of this definition, in accordance with GAAP) with respect to COFC on such date to (b) Managed Receivables with respect to COFC on such date.

"<u>Tier 1 Capital to Risk Adjusted Assets Ratio</u>" shall mean, on any date and with respect to COB or FSB, the ratio of (a) Tier 1 Capital with respect to such Borrower on such date to (b) Risk Adjusted Assets with respect to such Borrower on such date.

"<u>Tier 1 Leverage Ratio</u>" shall mean, on any date and with respect to COB or FSB, the ratio of (a) Tier 1 Capital with respect to such Borrower on such date to (b) Total Assets with respect to such Borrower on such date.

"<u>Total Assets</u>" shall mean, on any date and with respect to any U.S. Borrower, the amount, for such U.S. Borrower and its consolidated Subsidiaries (determined on a consolidated basis) on such date, of "average total consolidated assets", within the meaning given to such term in the Capital Adequacy Guidelines for State Member Banks published by the Board of Governors of the Federal Reserve System (12 C.F.R. Part 208, Appendix b, as amended, modified and supplemented and in effect from time to time or any replacement thereof).

"<u>Total Capital</u>" shall mean, on any date and with respect to any U.S. Borrower, the amount, for such U.S. Borrower and its consolidated Subsidiaries (determined on a consolidated basis) on such date, of "total capital", within the meaning given to such term in the Capital Adequacy Guidelines for State Member Banks published by the Board of Governors of the Federal Reserve System (12 C.F.R. Part 208, Appendix A, as amended, modified and supplemented and in effect from time to time or any replacement thereof).

"Total Capital to Risk Adjusted Assets Ratio" shall mean, on any date and with respect to COB or FSB, the ratio of (a) Total Capital with respect to such Borrower on such date to (b) Risk Adjusted Assets with respect to such Borrower on such date.

Credit Agreement

"Type" shall have the meaning assigned to such term in Section 1.03 hereof.

"U.S. Borrower" shall mean any Borrower other than COBE.

"<u>Voting Securities</u>" shall mean, with respect to any Person, securities or other ownership interests having by the terms thereof ordinary voting power to elect a majority of the board of directors or other persons performing similar functions of such Person (irrespective of whether or not at the time securities or other ownership interests of any other class or classes of such Person shall have or might have voting power by reason of the happening of any contingency).

"<u>Wholly Owned Subsidiary</u>" shall mean, with respect to any Person, any corporation, partnership, limited liability company or other entity of which all of the Voting Securities issued by such corporation, partnership, limited liability company or other entity (other than, in the case of a corporation, directors' qualifying shares) are directly or indirectly owned or controlled by such Person or one or more Wholly Owned Subsidiaries of such Person or by such Person and one or more Wholly Owned Subsidiaries of such Person.

1.02 Accounting Terms and Determinations.

(a) Except as otherwise expressly provided herein, all accounting terms used herein shall be interpreted, and all financial statements and certificates and reports as to financial matters required to be delivered to the Lenders hereunder shall (unless otherwise disclosed to the Lenders in writing at the time of delivery thereof in the manner described in subsection (b) below) be prepared in accordance with generally accepted accounting principles in the United States of America, applied on a basis consistent with those used in the preparation of the latest financial statements furnished to the Lenders hereunder (which, prior to the delivery of the first financial statements under Section 8.01(a) or (b) hereof, shall mean the audited financial statements as at December 31, 2003 referred to in Section 7.02 hereof). All calculations made for the purposes of determining compliance with this Agreement shall (except as otherwise expressly provided herein) be made by application of generally accepted accounting principles in the United States of America applied on a basis consistent with those used in the preparation of the Lenders pursuant to Section 8.01 hereof (or, prior to the delivery of the first financial statements furnished to the Lenders pursuant to Section 8.01 hereof (or, prior to the delivery of the first financial statements under Section 8.01(a) or (b) hereof, used in the preparation of the audited financial statements as at December 31, 2003 referred to in Section 7.02 hereof) unless (i) any Borrower shall have objected to determining such compliance on such basis at the time of delivery of such financial statements or (ii) the Majority Lenders shall so object in writing within 30 days after delivery of such financial statements, in either of which events such calculations shall be made on a basis consistent with those used in the preparation of the latest financial statements as to which such objection shall not have been made (which, if objection is made in respect of the first financial statements d

Credit Agreement

statements referred to in Section 7.02 hereof). Notwithstanding the foregoing, the accounting terms "Risk-Adjusted Assets", "Tier 1 Capital", "Total Assets" and "Total Capital" defined in Section 1.01 hereof shall be interpreted by reference to the statutes or regulations referred to in said definitions, as such statutes or regulations are amended, modified, supplemented or replaced and in effect from time to time.

(b) COFC shall deliver to the Lenders at the same time as the delivery of any annual or quarterly financial statement under Section 8.01 hereof (i) a description in reasonable detail of any material variation between the application of accounting principles in the United States of America employed in the preparation of such statement and the application of accounting principles in the United States of America employed in the next preceding annual or quarterly financial statements as to which no objection has been made in accordance with the last sentence of subsection (a) above and (ii) reasonable estimates of the difference between such statements arising as a consequence thereof. COBE shall deliver to the Lenders at the same time as the delivery of any annual or quarterly financial statement under Section 8.01 hereof (i) a description in reasonable detail of any material variation between the application of accounting principles in England employed in the preparation of such statements as to which no objection has been made the application of accounting principles in England employed in the preparation of such statements as to which no objection has been made in accordance with the last sentence of subsection (a) above and (ii) reasonable estimates of the difference between such statements as to which no objection has been made in accordance with the last sentence of subsection (a) above and (ii) reasonable estimates of the difference between such statements arising as a consequence thereof.

(c) To enable the ready and consistent determination of compliance with the covenants set forth in Section 8 hereof, no Borrower will change the last day of its fiscal year from December 31 of each year, or the last days of the first three fiscal quarters in each of its fiscal years from March 31, June 30 and September 30 of each year, respectively.

1.03 <u>Currencies and Types of Loans</u>. Loans hereunder are distinguished by "Currency" and by "Type". The "Currency" of a Loan refers to the Currency in which such Loan is denominated. The "Type" of a Loan refers to whether such Loan is a Base Rate Loan, a Eurocurrency Loan, a Set Rate Loan or a LIBOR Market Loan, each of which constitutes a Type. Loans may be identified by one or more of their Currency and Type.

1.04 <u>EMU</u>.

(a) <u>Unavailability of Euro</u>. If the Administrative Agent at any time determines that: (1) the Euro has ceased to be utilized as the basic accounting unit of the European Community, (2) for reasons affecting the market in Euros generally, Euros are not freely traded between banks internationally, or (3) it is illegal, impossible or impracticable for payments to be made hereunder in Euros, then the Administrative Agent may in its discretion declare (such declaration to be binding on all the parties hereto) that any payment made or to be made thereafter which, but for this provision, would have been payable in Euros shall be made in Pounds Sterling or Dollars (as selected by the Administrative Agent (the "Selected Currency"))

Credit Agreement

and the amount to be so paid shall be calculated on the basis of the equivalent of the Euro in the Selected Currency.

(b) <u>Basis of Accrual</u>. If the basis of accrual of interest or fees expressed in this Agreement with respect to the Currency of any state that becomes a Participating Member State shall be inconsistent with any convention or practice in the relevant interbank market for the offering of deposits denominated in such Currency for the basis of accrual of interest or fees in respect of the Euro, such convention or practice shall replace such expressed basis effective as of and from the date on which such state becomes a Participating Member State, provided, that if any Loan in the Currency of such state is outstanding immediately prior to such date, such replacement shall take effect, with respect to such Advance, at the end of the then current Interest Period.

(c) <u>Additional Changes at Administrative Agent's Discretion</u>. This Section and other provisions of this Agreement relating to Euros shall be subject to such further changes as the Administrative Agent may from time to time in its reasonable discretion specify to the other parties hereto as necessary or appropriate to reflect the changeover to or operation of the Euro in Participating Member States.

SECTION 2. Commitments, Loans, and Prepayments.

2.01 Syndicated Loans.

(a) Each Lender severally agrees, on the terms and conditions of this Agreement, to make Syndicated Loans to any one or more of the Borrowers in Dollars or any Agreed Alternative Currency during the period from and including the date hereof to but not including the Commitment Termination Date in an aggregate principal amount at any one time outstanding up to but not exceeding the amount of the Commitment of such Lender as in effect from time to time. Subject to the terms and conditions of this Agreement, during such period any Borrower may borrow, repay and reborrow the amount of the Commitments; provided that (i) no more than 10 separate Interest Periods in respect of Eurocurrency Loans may be outstanding at any one time and (ii) no more than 20 different Interest Periods for both Syndicated Loans and Money Market Loans may be outstanding at the same time (for which purpose (x) Interest Periods described in different lettered clauses of the definition of the term "Interest Period" shall be deemed to be different Interest Periods even if they are coterminous and (y) Loans denominated in different Currencies shall be deemed to have different Interest Periods).

(b) Anything herein to the contrary notwithstanding, (i) no Syndicated Loan shall be made if, after giving effect to such Loan, the aggregate outstanding principal amount of all Syndicated Loans, together with the aggregate outstanding principal amount of Money Market Loans, would exceed the aggregate amount of Commitments then in effect, and (ii) the aggregate outstanding principal amount of all Loans to COFC may not at any time exceed 66- ²/₃% of the aggregate amount of the Commitments then in effect.

(c) For purposes of determining whether the amount of any borrowing of a Loan, together with all other Loans then outstanding, would exceed the aggregate amount of Commitments (including, without limitation, for the purposes of Sections 2.01(a), 2.01(b) and 2.03(a) hereof) or any other limitation hereunder, the amount of any Loan outstanding that is denominated in an Alternative Currency shall be deemed to be the Dollar Equivalent (determined as of the date of such borrowing) of the amount of any Loan outstanding that is denominated in an Alternative Currency shall be deemed to be the Dollar Equivalent (determined as of the date of such borrowing) of the amount of any Loan outstanding that is denominated in an Alternative Currency shall be deemed to be the Dollar Equivalent (determined as of the date of determination of the unused portion of the Commitments) of the amount in the Alternative Currency of such Loan. For purposes of calculating the amount of any utilization fee payable under Section 2.05(b) hereof, the amount of any date that is denominated in an Alternative Currency shall be deemed to be the Dollar Equivalent (and Alternative Currency shall be deemed to be the Dollar Equivalent (determined as of the date of the most recent borrowing of Loans) of the amount in the Alternative Currency of such Loan. For purposes of calculating the amount of any utilization fee payable under Section 2.05(b) hereof, the amount of any Loan outstanding on any date that is denominated in an Alternative Currency shall be deemed to be the Dollar Equivalent (determined as of the date of the most recent borrowing of Loans) of the amount in the Alternative Currency of such Loan.

2.02 Borrowings of Syndicated Loans. The applicable Borrower shall give the Administrative Agent notice of each borrowing of Syndicated Loans as provided in Section 4.05 hereof. Not later than 1:00 p.m. Local Time on the date specified for each borrowing of Syndicated Loans, each Lender shall make available the amount of the Syndicated Loan or Loans to be made by it on such date to the Administrative Agent, at the Administrative Agent's Account for the Currency in which such Loan is denominated, in immediately available funds, for account of the applicable Borrower. The amount so received by the Administrative Agent shall, subject to the terms and conditions of this Agreement, be made available to the applicable Borrower by depositing the same, in immediately available funds, in an account of the applicable Borrower.

2.03 Money Market Loans.

(a) In addition to borrowings of Syndicated Loans, at any time prior to the Commitment Termination Date, any Borrower may, as set forth in this Section 2.03, request the Lenders to make offers to make Money Market Loans to such Borrower in Dollars or in any Alternative Currency. The Lenders may, but shall have no obligation to, make such offers and the applicable Borrower may, but shall have no obligation to, accept any such offers in the manner set forth in this Section 2.03. Money Market Loans may be LIBOR Market Loans or Set Rate Loans (each a "Type" of Money Market Loan), <u>provided</u> that:

(i) there may be no more than 20 different Interest Periods for both Syndicated Loans and Money Market Loans outstanding at the same time (for which purpose (x) Interest Periods described in different lettered clauses of the definition of the term "Interest Period" shall be deemed to be different Interest Periods even if they are coterminous and (y) Loans denominated in different Currencies shall be deemed to have different Interest Periods); and

Credit Agreement

(ii) the aggregate principal amount of all Money Market Loans, together with the aggregate principal amount of all Syndicated Loans, at any one time outstanding shall not exceed the aggregate amount of the Commitments then in effect.

(b) When a Borrower wishes to request offers to make Money Market Loans, it shall give the Administrative Agent (which shall promptly notify the Lenders) notice (a "<u>Money Market Quote Request</u>") so as to be received no later than (x) 11:00 a.m. New York time on the fifth Business Day prior to the date of borrowing proposed therein, in the case of a LIBOR Auction, (y) 12:00 noon London time on the fourth Business Day prior to the date of borrowing proposed therein in the case of a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency or (z) the Business Day next preceding the date of borrowing proposed therein in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars. The applicable Borrower may request offers to make Money Market Loans for up to three different Interest Periods in a single notice (for which purpose (x) Interest Periods in different lettered clauses of the definition of the term "Interest Period" shall be deemed to be different Interest Periods even if they are coterminous and (y) Money Market Loans denominated in different Currencies shall be deemed to have different Interest Periods); <u>provided</u> that the request for each separate Interest Period or Currency shall be deemed to be a separate Money Market Quote Request for a separate borrowing (a "<u>Money Market Borrowing</u>"). Each such notice shall be substantially in the form of Exhibit E hereto and shall specify as to each Money Market Borrowing:

(i) the name of the Borrower, the Currency of such Borrowing and the proposed date of such borrowing, which shall be a Business Day;

(ii) the aggregate amount of such Money Market Borrowing, which shall be an integral multiple of \$1,000,000 and not less than \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)) but shall not cause the limits specified in Section 2.03(a) hereof to be violated;

(iii) the duration of the Interest Period applicable thereto;

(iv) whether the Money Market Quotes requested for a particular Interest Period are seeking quotes for LIBOR Market Loans or Set Rate Loans; and

(v) if the Money Market Quotes requested are seeking quotes for Set Rate Loans denominated in Dollars, the date on which the Money Market Quotes are to be submitted (the date on which such Money Market Quotes are to be submitted is called the "<u>Quotation Date</u>").

Except as otherwise provided in this Section 2.03(b), no Money Market Quote Request shall be given within five Business Days of any other Money Market Quote Request.

Credit Agreement

(c) (i) Each Lender may submit one or more Money Market Quotes, each constituting an offer to make a Money Market Loan in response to any Money Market Quote Request; <u>provided</u> that, if the applicable Borrower's request under Section 2.03(b) hereof specified more than one Interest Period, such Lender may make a single submission containing one or more Money Market Quotes for each such Interest Period. Each Money Market Quote must be submitted to the Administrative Agent not later than (x) 4:00 p.m. New York time on the fifth Business Day prior to the proposed date of borrowing, in the case of a LIBOR Auction, (y) 4:00 p.m. London time on the fourth Business Day prior to the date of borrowing proposed therein in the case of a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency or (z) 10:00 a.m. New York time on the Quotation Date in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars; <u>provided</u> that any Money Market Quote may be submitted by JPMorgan (or its Applicable Lending Office) only if JPMorgan (or such Applicable Lending Office) notifies such Borrower of the terms of the offer contained therein not later than (x) 3:45 p.m. New York time on the fifth Business Day prior to the proposed date of borrowing, in the case of a Set Rate Auction in respect of the date of borrowing proposed therein in the case of a Set Set prior to the date of borrowing proposed therein not later than (x) 3:45 p.m. New York time on the fifth Business Day prior to the proposed date of borrowing, in the case of a LIBOR Auction, (y) 3:45 p.m. London time on the fourth Business Day prior to the date of borrowing proposed therein in the case of a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency or (z) 9:45 a.m. New York time on the Quotation Date in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars. Subject to Sections 5.02(b), 5.03, 6.02 and 9 hereof, any Money Market Quote so made shall

(ii) Each Money Market Quote shall be substantially in the form of Exhibit F hereto and shall specify:

(A) the name of the Borrower, the Currency of such Borrowing and the proposed date of borrowing and the Interest Period therefor;

(B) the principal amount of the Money Market Loan for which each such offer is being made, which principal amount shall be an integral multiple of \$1,000,000 and not less than \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)); provided that the aggregate principal amount of all Money Market Loans for which a Lender submits Money Market Quotes (x) may be greater or less than the Commitment of such Lender but (y) may not exceed the principal amount of the Money Market Borrowing for a particular Interest Period for which offers were requested;

(C) in the case of a LIBOR Auction, the margin above or below the applicable Eurocurrency Rate (the "<u>LIBO Margin</u>") offered for each such Money Market Loan, expressed as a percentage (rounded upwards, if necessary, to the

Credit Agreement

nearest 1/10,000th of 1%) to be added to or subtracted from the applicable Eurocurrency Rate;

(D) in the case of a Set Rate Auction, the rate of interest per annum (rounded upwards, if necessary, to the nearest 1/10,000th of 1%) offered for each such Money Market Loan (the "Set Rate"); and

(E) the identity of the quoting Lender.

Unless otherwise agreed by the Administrative Agent and the applicable Borrower, no Money Market Quote shall contain qualifying, conditional or similar language or propose terms other than or in addition to those set forth in the applicable Money Market Quote Request and, in particular, no Money Market Quote may be conditioned upon acceptance by the applicable Borrower of all (or some specified minimum) of the principal amount of the Money Market Loan for which such Money Market Quote is being made, <u>provided</u> that the submission by any Lender containing more than one Money Market Quote may be conditioned on the applicable Borrower not accepting offers contained in such submission that would result in such Lender making Money Market Loans pursuant thereto in excess of a specified aggregate amount (the "<u>Money Market Loan Limit</u>").

(d) The Administrative Agent shall (x) in the case of a LIBOR Auction, by 5:00 p.m. New York time on the day a Money Market Quote is submitted, (y) in the case of a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency, by 5:00 p.m. London time or (z) in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars, as promptly as practicable after the Money Market Quote is submitted (but in any event not later than 10:15 a.m. New York time on the Quotation Date), notify the applicable Borrower of the terms (i) of any Money Market Quote submitted by a Lender that is in accordance with Section 2.03(c) hereof and (ii) of any Money Market Quote that amends, modifies or is otherwise inconsistent with a previous Money Market Quote submitted by such Lender with respect to the same Money Market Quote Request. Any such subsequent Money Market Quote shall be disregarded by the Administrative Agent unless such subsequent Money Market Quote is submitted solely to correct a manifest error in such former Money Market Quote. The Administrative Agent's notice to the applicable Borrower shall specify (A) the aggregate principal amount of the Money Market Borrowing for which offers have been received and (B) the respective principal amounts and LIBO Margins or Set Rates, as the case may be, so offered by each Lender (identifying the Lender that made each Money Market Quote).

(e) Not later than (x) 10:00 a.m. New York time on the fourth Business Day prior to the proposed date of borrowing in the case of a LIBOR Auction, (y) 10:00 a.m. London time on the third Business Day prior to the proposed date of borrowing in the case of a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency or (z) 11:00 a.m. New York time on the Quotation Date in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars, the applicable Borrower shall notify the

Credit Agreement

Administrative Agent of its acceptance or nonacceptance of the offers so notified to it pursuant to Section 2.03(d) hereof (which notice shall specify the aggregate principal amount of offers from each Lender for each Interest Period that are accepted, it being understood that the failure of the applicable Borrower to give such notice by such time shall constitute nonacceptance) and the Administrative Agent shall promptly notify each affected Lender. The notice from the Administrative Agent shall also specify the aggregate principal amount of offers for each Interest Period that were accepted and the lowest and highest LIBO Margins and Set Rates that were accepted for each Interest Period. The applicable Borrower may accept any Money Market Quote in whole or in part; provided that:

(i) the aggregate principal amount of each Money Market Borrowing may not exceed the applicable amount set forth in the related Money Market Quote Request;

(ii) the aggregate principal amount of each Money Market Borrowing shall be an integral multiple of \$1,000,000 and not less than \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)) but shall not cause the limits specified in Section 2.03(a) hereof to be violated;

(iii) acceptance of offers may, subject to clause (vi) below, be made only in ascending order of LIBO Margins or Set Rates, as the case may be, in each case beginning with the lowest rate so offered;

(iv) any Money Market Quote accepted in part shall be an integral multiple of \$1,000,000 and not less than \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency));

(v) the applicable Borrower may not accept any offer where the Administrative Agent has advised the applicable Borrower that such offer fails to comply with Section 2.03(c)(ii) hereof or otherwise fails to comply with the requirements of this Agreement (including, without limitation, Section 2.03(a) hereof); and

(vi) the aggregate principal amount of each Money Market Borrowing from any Lender may not exceed any applicable Money Market Loan Limit of such Lender.

If offers are made by two or more Lenders with the same LIBO Margins or Set Rates, as the case may be, for a greater aggregate principal amount than the amount in respect of which offers are permitted to be accepted for the related Interest Period, the principal amount of Money Market Loans in respect of which such offers are accepted shall be allocated by the applicable Borrower among such Lenders as nearly as possible (in an integral multiple of \$1,000,000 and not less than \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)) in proportion to the aggregate principal amount of such

Credit Agreement

offers. Determinations by the applicable Borrower of the amounts of Money Market Loans shall be conclusive in the absence of manifest error.

(f) Any Lender whose offer to make any Money Market Loan has been accepted in accordance with the terms and conditions of this Section 2.03 shall, not later than 11:00 a.m. Local Time (in the case of a LIBOR Auction or a Set Rate Auction in respect of Money Market Loans denominated in an Alternative Currency) or 1:00 p.m. New York time (in the case of a Set Rate Auction in respect of Money Market Loans denominated in Dollars) on the date specified for the making of such Loan, make the amount of such Loan available to the Administrative Agent at the Administrative Agent's Account for the Currency in which such Loan is denominated in immediately available funds, for account of the applicable Borrower. The amount so received by the Administrative Agent shall, subject to the terms and conditions of this Agreement, be made available to the applicable Borrower on such date by depositing the same, in immediately available funds, in an account of the applicable Borrower.

(g) Except for the purpose and to the extent expressly stated in Section 2.05(b) hereof, the amount of any Money Market Loan made by any Lender shall not constitute a utilization of such Lender's Commitment.

(h) The applicable Borrower shall pay to the Administrative Agent a fee of \$750 each time it gives a Money Market Quote Request to the Administrative Agent.

2.04 Changes of Commitments.

(a) The aggregate amount of the Commitments shall be automatically reduced to zero on the Commitment Termination Date.

(b) The Borrowers, acting jointly, shall have the right at any time or from time to time (i) to terminate the Commitments so long as no Syndicated Loans or Money Market Loans are outstanding (or any outstanding Loans are simultaneously repaid in full) and (ii) to reduce the aggregate unused amount of the Commitments (for which purpose use of the Commitments shall be deemed to include the aggregate principal amount of all Money Market Loans); <u>provided</u> that (x) the Borrowers shall give notice of each such termination or reduction as provided in Section 4.05 hereof, (y) each partial reduction shall aggregate to an integral multiple of \$1,000,000 and not less than \$10,000,000 and (z) no such termination or reduction shall be effected unless such notice shall have been given by each Borrower.

(c) The Commitments, once terminated or reduced, may not be reinstated.

2.05 Fees.

(a) <u>Facility Fee</u>. The Borrowers shall pay to the Administrative Agent for account of each Lender a facility fee on the daily average amount of such Lender's Commitment (regardless of utilization thereof), for the period from and including the date hereof to but not

Credit Agreement

including the earlier of the date such Commitment is terminated and the Commitment Termination Date, at a rate per annum equal to the Applicable Facility Fee Percentage; <u>provided</u>, that the Applicable Facility Fee Percentage on the Commitments available for borrowing by COFC pursuant to Section 2.01(b) shall be determined with reference to the lower of the Debt Ratings assigned to COFC and COB, while the Applicable Facility Fee Percentage on the Commitments not available for borrowing by COFC pursuant to Section 2.01(b) shall be determined with reference to the Debt Rating of COB; <u>provided</u>, <u>further</u>, that the facility fee shall not accrue or become payable by the Borrowers on the Commitment of any Defaulting Lender during the period that such Lender is a Defaulting Lender.

(b) <u>Utilization Fee</u>. The Borrowers shall pay to the Administrative Agent for account of each Lender a utilization fee, for each Computation Period occurring during the period from and including the date hereof to but not including the earlier of the date the Commitments are terminated and the Commitment Termination Date, on the excess, if any, of (i) the Average of the aggregate outstanding principal amount of all Loans (including Money Market Loans) outstanding for such Computation Period over (ii) 33% of the Average of the aggregate amount of all Commitments in effect for such Computation Period, at a rate per annum equal to the Applicable Utilization Fee Percentage. Utilization fee payable under this Section 2.05(b) with respect to any day shall be allocated among the Borrowers pro rata according to the respective Average aggregate outstanding principal amounts of Loans owing by such Borrowers for such Computation Period.

(c) <u>Payment</u>. Accrued facility fee and utilization fee shall be payable on each Quarterly Date and on the earlier of the date the Commitments are terminated and the Commitment Termination Date.

2.06 Lending Offices. The Loans of each Type and Currency made by each Lender shall be made and maintained at such Lender's Applicable Lending Office for Loans of such Type and Currency.

2.07 Several Obligations; Remedies Independent. The failure of any Lender to make any Loan to be made by it on the date specified therefor shall not relieve any other Lender of its obligation to make its Loan on such date, but neither any Lender nor the Administrative Agent shall be responsible for the failure of any other Lender to make a Loan to be made by such other Lender, and (except as otherwise provided in Section 4.06 hereof) no Lender shall have any obligation to the Administrative Agent or any other Lender for the failure by such other Lender to make any Loan required to be made by such other Lender. The amounts payable by each Borrower at any time hereunder and under the Notes to each Lender shall be a separate and independent debt, and each Lender shall (without prejudice to the provisions of Section 9 hereof) be entitled to protect and enforce its rights arising out of this Agreement and the Notes, and it shall not be necessary for any other Lender or the Administrative Agent to consent to, or be joined as an additional party in, any proceedings for such purposes.

²⁹

2.08 Evidence of Debt.

(a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of each Borrower to such Lender resulting from each Loan made by such Lender, including the date, amount, Currency, Type, interest rate and duration of Interest Period of each Loan made by such Lender to a Borrower, and amounts of principal and interest payable and paid to such Lender from time to time hereunder.

(b) The Administrative Agent shall maintain accounts in which it shall record (i) the name of each Borrower and amount of each Loan made to such Borrower hereunder, Currency and Type thereof and the interest rate and Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from each Borrower to each Lender hereunder and (iii) the amount of any sum received by the Administrative Agent from any Borrower hereunder for the account of the Lenders and each Lender's share thereof.

(c) The entries made in the accounts maintained pursuant to paragraph (a) or (b) of this Section shall be prima facie evidence of the existence and amounts of the obligations recorded therein; provided that the failure of any Lender or the Administrative Agent to maintain such accounts or any error therein shall not in any manner affect the obligation of any Borrower to repay its Loans in accordance with the terms of this Agreement.

(d) Any Lender may request through the Administrative Agent that Loans made by it be evidenced by one or more promissory notes. In such event, the applicable Borrower(s) shall prepare, execute and deliver to such Lender a promissory note payable to the order of such Lender in the form of Exhibit A-1 hereto (to evidence Syndicated Loans), or Exhibit A-2 hereto (to evidence Money Market Loans), as such Lender may request. Each Syndicated Note shall be dated the date hereof, duly executed by the applicable Borrower and payable to such Lender in a principal amount equal to the amount of its Commitment as originally in effect and otherwise duly completed. Each Money Market Note shall be dated the date hereof, duly executed by the applicable Borrower and payable to such Lender and otherwise duly completed.

(e) No Lender shall be entitled to have its Notes substituted or exchanged for any reason, or subdivided for promissory notes of lesser denominations, except in connection with an increase of such Lender's Commitment pursuant to Section 2.10 or 2.12 hereof or in connection with a permitted assignment of all or any portion of such Lender's Commitment, Loans and Notes pursuant to Section 5.07 or 11.06 hereof (and, if requested by any Lender, each Borrower agrees to so exchange any Note).

Credit Agreement

2.09 Prepayments.

(a) Optional Prepayments.

(i) Subject to Section 4.04 hereof, each Borrower shall have the right to prepay Syndicated Loans made to such Borrower at any time or from time to time, <u>provided</u> that: (i) such Borrower shall give the Administrative Agent notice of each such prepayment as provided in Section 4.05 hereof (and, upon the date specified in any such notice of prepayment, the amount to be prepaid shall become due and payable hereunder) and (ii) any prepayment of a Eurocurrency Loan on a day other than the last day of the Interest Period for such Loan shall be subject to the payment of any compensation payable under Section 5.05 hereof.

(ii) Money Market Loans may not be prepaid.

(b) <u>Currency Valuation</u>. In the event that any Borrower selects an Interest Period of more than three months' duration for any borrowing of Loans, on each Currency Valuation Date (as defined below), the Administrative Agent shall determine the sum of the aggregate outstanding principal amount of all Loans. For purposes of this determination, the outstanding principal amount of any Loan that is denominated in any Alternative Currency shall be deemed to be the Dollar Equivalent of the amount in the Alternative Currency of such Loan, determined as of such Currency Valuation Date. Upon making such determination, the Administrative Agent shall promptly notify the Lenders and each Borrower thereof. If, on the date of such determination, such sum of the aggregate principal amount of all Loans exceeds 105% of the aggregate of the Commitments as then in effect, the Borrowers shall, if requested by the Majority Lenders (through the Administrative Agent), prepay outstanding Loans other than Money Market Loans (ratably in accordance with the then outstanding aggregate principal amounts thereof except that the Borrowers may prepay Base Rate Loans in full before prepaying Eurocurrency Loans) in such amounts as shall be necessary so that after giving effect thereto the aggregate outstanding principal amount of all Loans does not exceed the aggregate Commitments, any such prepayment to be made, in the case of any Eurocurrency Loan, on the last day of the then current Interest Period with respect thereto. After the date of any such prepayment, the Borrowers shall not be required to make a prepayment under this Section 2.09(b) until any Borrower subsequently selects an Interest Period of more than three months' duration. Any such payment shall be accompanied by accrued interest thereon as provided in Section 3.02 hereof and by any amounts payable under Section 5.05 hereof.

For purposes of this Section 2.09(b), "<u>Currency Valuation Date</u>" shall mean, with respect to each Interest Period having an initial duration of more than three months for any borrowing of Loans, each date which occurs at intervals of three months after the first day of such Interest Period (or, if any such date is not a Business Day, the immediately preceding Business Day).

2.10 Increases in Commitments.

(a) The Borrowers, acting jointly, shall have the right at any time to increase the aggregate amount of the Commitments hereunder to an amount not to exceed \$1,250,000,000 by causing one or more banks or other financial institutions, which may include any Lender already party to this Agreement, to become a "Lender" party to this Agreement or (in the case of any Lender already party to this Agreement) to increase the amount of such Lender's

Credit Agreement

Commitment; provided that (i) the addition of any bank or other financial institution to this Agreement that is not already a Lender shall be subject to the consent of the Administrative Agent (which consent shall not be unreasonably withheld or delayed) and (ii) the Commitment of any bank or other financial institution becoming a "Lender" party to this Agreement, and any increase in the amount of the Commitment of any Lender already party to this Agreement, shall be in an amount equal to an integral multiple of \$1,000,000 and not less than \$10,000,000.

(b) Any increase in the aggregate amount of the Commitments pursuant to Section 2.10(a) hereof shall be effective only upon the execution and delivery by the Borrowers, the Administrative Agent and each relevant bank, financial institution or Lender of a commitment increase letter in substantially the form of Exhibit I hereto (a "<u>Commitment Increase Letter</u>"), which Commitment Increase Letter shall be delivered to the Administrative Agent not less than three Business Days prior to the Commitment Increase Date and shall specify (i) the amount of the Commitment of such bank or other financial institution becoming a "Lender" party to this Agreement or of any increase in the amount of the Commitment of such Lender and (ii) the date such increase is to become effective (the "<u>Commitment Increase Date</u>").

(c) Any increase in the aggregate amount of the Commitments pursuant to this Section 2.10 shall not be effective unless:

(i) no Default shall have occurred and be continuing on the Commitment Increase Date;

(ii) each of the representations and warranties made by the Borrowers in Section 7 hereof (other than the Excluded Representations) shall be true and correct in all material respects on and as of the Commitment Increase Date with the same force and effect as if made on and as of such date (or, if any such representation or warranty is expressly stated to have been made as of a specific date, as of such specific date);

(iii) no notice of borrowing of Syndicated Loans shall be outstanding on and as of such Commitment Increase Date;

(iv) such increase in the aggregate amount of the Commitments does not cause any Lender to hold a Commitment in an amount exceeding 25% of the aggregate amount of the Commitments as so increased;

(v) immediately after giving effect to such increase, the aggregate amount of Commitments available for borrowing by COFC shall not exceed $66-\frac{2}{3}$ of the aggregate amount of the Commitments as so increased; and

(vi) the Administrative Agent shall have received (with sufficient copies for each of the Lenders) each of (x) a certificate of the corporate secretary or assistant secretary of the Borrowers as to the taking of any corporate action necessary in connection with such increase and (y) an opinion or opinions of counsel to the Borrowers as to their corporate power and authority to borrow hereunder after giving effect to such increase.

Credit Agreement

Each notice requesting an increase in the aggregate amount of the Commitments pursuant to this Section 2.10 shall constitute a certification to the effect set forth in clauses (i) and (ii) of the preceding sentence.

(d) No Lender shall at any time be required to agree to a request of a Borrower to increase its Commitment or obligations hereunder.

2.11 <u>Undertaking of COB</u>. COB hereby agrees with each Lender and the Administrative Agent as follows:

(a) <u>Undertaking to Pay</u>. At the request of FSB and COBE, COB hereby irrevocably undertakes in favor of the Administrative Agent that COB will honor the Administrative Agent's sight drafts drawn on COB and payable to the order of the Administrative Agent upon presentation of such drafts to COB at the address to which notices are deliverable to COB under Section 11.02 hereof accompanied by a written certification referred to below (such undertaking to honor such drafts being herein called, the "<u>Undertaking</u>"). Each draft must be accompanied by written certification of the Administrative Agent in the form of Exhibit J to this Agreement. Each draft drawn under and in compliance with the Undertaking will be duly honored by COB forthwith upon presentation by paying the amount of such draft to the Administrative Agent at the Administrative Agent's Account in the manner specified in Section 4.01 hereof.

(b) <u>Amount Available</u>. The aggregate amount available to be drawn under this Section 2.11 shall be equal to (i) the aggregate amount of Commitments hereunder <u>plus</u> (ii) interest (computed on the basis of a year of 360 days) that would accrue on such aggregate amount for a period of 360 days at a rate per annum equal to 12%. Partial and multiple drawings under this Section 2.11 are permitted. The Undertaking shall expire on the date 100 days following the Commitment Termination Date.

(c) <u>Certain Terms and Conditions</u>. All charges and commissions incurred by COB in connection with the issuance or administration of the Undertaking (including any drawing in respect of the Undertaking) shall be for account of FSB or COBE, as the case may be. This Section 2.11 sets forth in full the terms of the Undertaking, and the Undertaking shall not in any way be amended, modified, amplified or limited by reference to any other Section or provision of this Agreement or any document, instrument or agreement referred to herein, and any such reference shall not be deemed to incorporate in this Section 2.11 by reference any document, instrument or agreement. The obligations of COB in respect of the Undertaking are independent of any of the obligations of any other party to this Agreement and of any obligations of COB under any other Section or provision of this Agreement (and accordingly the Undertaking is intended to be both a "credit" and a "letter of credit" within the meaning of Article 5 of the New York Uniform Commercial Code), and the entitlement of the Administrative Agent to draw under the Undertaking is subject only to compliance by the

Administrative Agent with the express conditions to drawing set forth in this Section 2.11. The Undertaking may not be assigned or transferred other than to a successor Administrative Agent appointed in accordance with Section 10.08 hereof.

(d) <u>Reimbursement</u>. FSB and COBE agree to reimburse COB for any drawing by the Administrative Agent under the Undertaking for their respective accounts, without notice or demand of any kind, not later than 1:00 p.m. (New York time) on the Business Day following such drawing, in an amount equal to the amount of such drawing. COB hereby agrees that until the payment and satisfaction in full of the principal of and interest on the Loans made by the Lenders to, and any Notes held by each Lender of, FSB and COBE and all other amounts from time to time owing to the Lenders or the Administrative Agent by FSB and COBE hereunder and under any Notes and the expiration or termination of the Commitments COB shall not exercise any right or remedy to collect any amount owing by FSB or COBE to COB under this Section 2.11(d).

(e) <u>UCP</u>. The Undertaking is subject to the Uniform Customs and Practice for Documentary Credits (1993 Revision), International Chamber of Commerce Publication No. 500 (the "<u>UCP</u>"). In the event of any conflict between the law of the State of New York (which, pursuant to Section 11.10 hereof, governs this Agreement) and the UCP, the UCP shall control. Notwithstanding Article 17 of the UCP, if the Undertaking expires during an interruption of business as described in said Article 17, COB shall effect payment if the Undertaking is drawn against within 30 days after the resumption of business.

(f) <u>Distribution of Proceeds of Drawing</u>. Each payment received by the Administrative Agent in connection with any drawing under the Undertaking shall be promptly applied by the Administrative Agent to the obligations of FSB or COBE, as the case may be, in respect of which such drawing was made.

2.12 Extensions of Commitments.

(a) The Borrowers may, by notice to the Administrative Agent not earlier than 90 days but not later than 60 days prior to each anniversary of the Effective Date, request that each Lender extend such Lender's Commitment for an additional 1-year or 2-year period from such anniversary of the Effective Date (the "<u>Extension Date</u>"); provided, that at no time shall any Lender's Commitment be longer than a period of three years.

(b) Each Lender, acting in its sole and individual discretion, shall, by notice to the Administrative Agent given not earlier than 75 days but not later than the date 45 days prior to the Extension Date (the "<u>Notice Date</u>"), advise the Administrative Agent whether or not such Lender agrees to such extension and each Lender that determines not to so extend its Commitment (a "<u>Non-Extending Lender</u>") shall notify the Administrative Agent (which shall notify the other Lenders) of such fact promptly after such determination (but in any event not later than the Notice Date) and any Lender that does not so advise the Administrative Agent on or before the Notice Date shall be deemed to be a Non-Extending Lender. The Commitment of

Credit Agreement

each Non-Extending Lender shall terminate on the Commitment Termination Date then in effect without extension (the "Existing Commitment Termination Date"), and each Borrower shall pay in full all amounts owing by such Borrower to each Non-Extending Lender hereunder on or before the Existing Commitment Termination Date.

(c) The Administrative Agent shall notify the Borrowers of each Lender's determination under this Section 2.12 not earlier than 60 days but not later than the 30 days prior to the Extension Date (or, if such date is not a Business Day, on the next preceding Business Day). The election of any Lender to agree to such extension shall not obligate any other Lender to so agree.

(d) The Borrowers shall have the right on or before the Extension Date to require any Non-Extending Lender to assign in full, in accordance with Section 11.06, its rights and obligations under this Agreement to an assignee designated by the Borrowers (subject to the approval of the Administrative Agent, which shall not be unreasonably withheld) that agrees to accept all of such rights and obligations (each, a "<u>Replacement Lender</u>"); <u>provided</u>, that prior to replacing any Non-Extending Lender with any Replacement Lender, the Borrower shall have given each Lender which has agreed to extend its Commitment an opportunity to increase its Commitment by all or a portion of the Non-Extending Lenders' Commitments.

(e) If (and only if) the total of the Commitments of the Lenders that have agreed so to extend their Commitments shall be not less than 51% of the aggregate amount of the Commitments in effect immediately prior to the Extension Date, then, effective as of the Extension Date, (i) each Replacement Lender shall thereupon become a "Lender" for all purposes of this Agreement and (ii) the Commitment of each Lender that has agreed so to extend its Commitment and of each Replacement Lender shall be extended to the date one or two years, as the case may be, after the Extension Date (except that, if such date is not a Business Day, such Commitment as so extended shall be the next preceding Business Day).

(f) Notwithstanding the foregoing, the extension of the Commitments beyond the Commitment Termination Date, as in effect at any given time, pursuant to this Section 2.12 shall be effective with respect to any Lender only if:

(i) no Default or Event of Default shall have occurred and be continuing on and as of such Commitment Termination Date; and

(ii) each of the representations and warranties in Section 7 hereof, including without limitation each of the Excluded Representations, are true and complete in all material respects on and as of such Commitment Termination Date with the same force and effect as if made on and as of such Commitment Termination Date (or, if any such representation or warranty is expressly stated to have been made as of a specific date, as of such date).

35

SECTION 3. Payments of Principal and Interest.

3.01 <u>Repayment of Loans</u>. Each Borrower hereby promises to pay to the Administrative Agent for account of each Lender the principal of each Loan made by such Lender to such Borrower, and each such Loan shall mature, on the last day of the Interest Period therefor. Except as set forth in Section 2.11 hereof, no Borrower shall have liability for the obligations of another Borrower.

3.02 <u>Interest</u>. Each Borrower hereby promises to pay to the Administrative Agent for account of each Lender interest on the unpaid principal amount of each Loan made by such Lender to such Borrower for the period from and including the date of such Loan to but excluding the date such Loan shall be paid in full, at the following rates per annum:

(a) if such Loan is a Base Rate Loan, the Base Rate (as in effect from time to time) plus the Applicable Margin with respect to Base Rate Loans;

(b) if such Loan is a Eurocurrency Loan, the Eurocurrency Rate for such Loan for the Interest Period therefor <u>plus</u> the Applicable Margin with respect to Eurocurrency Loans;

(c) if such Loan is a LIBOR Market Loan, the Eurocurrency Rate for such Loan for the Interest Period therefor <u>plus</u> (or <u>minus</u>) the LIBO Margin quoted by the Lender making such Loan in accordance with Section 2.03 hereof; and

(d) if such Loan is a Set Rate Loan, the Set Rate for such Loan for the Interest Period therefor quoted by the Lender making such Loan in accordance with Section 2.03 hereof.

Notwithstanding the foregoing, each Borrower hereby promises to pay to the Administrative Agent for account of each relevant Lender interest at the applicable Post-Default Rate on any principal of any Loan made by such Lender to such Borrower, and on any other amount payable by such Borrower to or for account of such Lender hereunder or under any Notes, that shall not be paid in full when due (whether at stated maturity, by acceleration, by mandatory prepayment or otherwise), for the period from and including the due date thereof to but excluding the date the same is paid in full. Accrued interest on each Loan shall be payable (i) on the last day of the Interest Period therefor and, if such Interest Period is longer than 90 days (in the case of a Set Rate Loan) or three months (in the case of a Eurocurrency Loan or a LIBOR Market Loan), at 90-day or three-month intervals, respectively, following the first day of such Interest Period, and (ii) in the case of any Loan, upon the payment or prepayment thereof (but only on the principal amount so paid or prepaid), except that interest payable at the Post-Default Rate shall be payable from time to time on demand. Promptly after the determination of any interest rate provided for herein or any change therein, the Administrative Agent shall give notice thereof to the Lenders to which such interest is payable and to the applicable Borrower.

Credit Agreement

SECTION 4. Payments; Pro Rata Treatment; Computations; Etc.

4.01 Payments.

(a) Except to the extent otherwise provided herein, all payments of principal of and interest on any Loan and of all other amounts to be made by a Borrower under this Agreement and any Notes shall be made in the Currency in which such Loan or other amount is denominated, in immediately available funds, without deduction, set-off or counterclaim, to the Administrative Agent at the Administrative Agent's Account for the Currency in which such Loan or other amount is denominated, not later than 1:00 p.m. Local Time on the date on which such payment shall become due (each such payment made after such time on such due date to be deemed to have been made on the next succeeding Business Day), provided that if a new Loan is to be made to a Borrower by any Lender on a date on which such Borrower is to repay any principal of an outstanding Loan of such Lender and in the same Currency, such Lender shall apply the proceeds of such new Loan to the payment of the principal to be repaid and only an amount equal to the difference between the principal to be borrower and the principal to be repaid shall be made available by such Lender to the Administrative Agent as provided in Section 2.02 hereof or paid by the applicable Borrower to the Administrative Agent pursuant to this Section 4.01, as the case may be. All amounts owing under this Agreement and any Notes (other than principal of and interest on Loans denominated in an Alternative Currency) are denominated and payable in Dollars.

(b) Any Lender for whose account any such payment is to be made may (but shall not be obligated to) debit the amount of any such payment that is not made by such time to any ordinary deposit account of the Borrower obligated to make such payment with such Lender (with notice to such Borrower and the Administrative Agent), provided that such Lender's failure to give such notice shall not affect the validity thereof.

(c) Each Borrower shall, at the time of making each payment under this Agreement or any Note for account of any Lender, specify to the Administrative Agent (which shall so notify the intended recipient(s) thereof) the Loans or other amounts payable by such Borrower hereunder to which such payment is to be applied (and in the event that such Borrower fails to so specify, or if an Event of Default has occurred and is continuing, the Administrative Agent may distribute such payment to the Lenders for application in such manner as it or the Majority Lenders, subject to Section 4.02 hereof, may determine to be appropriate).

(d) Each payment received by the Administrative Agent under this Agreement or any Note for account of any Lender shall be paid by the Administrative Agent promptly to such Lender, in like Currency and immediately available funds, for account of such Lender's Applicable Lending Office for the Loan or other obligation in respect of which such payment is made.

(e) If the due date of any payment under this Agreement or any Note would otherwise fall on a day that is not a Business Day, such payment shall be made on the next

succeeding Business Day unless such succeeding Business Day would fall in the next calendar month, in which case such payment shall be made on the next preceding Business Day.

4.02 Pro Rata Treatment. Except to the extent otherwise provided herein:

(a) each borrowing under Section 2.01 hereof of Syndicated Loans shall be made from the Lenders, each payment under Section 2.05(a) hereof of facility fee to the Lenders, and each termination or reduction under Section 2.04(b) hereof of the Commitments shall be applied to the respective Commitments of the Lenders, pro rata according to the amounts of their respective Commitments;

(b) except as otherwise provided in Section 5.04 hereof, Eurocurrency Loans having the same Interest Period shall be allocated among the Lenders pro rata according to the amounts of their respective Commitments;

(c) each payment or prepayment of principal of Syndicated Loans, and each payment under Section 2.05 hereof of utilization fee to the Lenders, shall be made by a Borrower (allocated between the Borrowers in accordance with the last sentence of Section 2.05(b) hereof) for account of the Lenders pro rata according to the respective unpaid principal amounts of the Loans owing by such Borrower held by such Lenders; and

(d) each payment of interest on Syndicated Loans shall be made by a Borrower for account of the Lenders pro rata according to the amounts of interest on such Loans then due and payable by such Borrower to such Lenders.

4.03 <u>Computations</u>. Interest on Money Market Loans and Eurocurrency Loans and facility and utilization fee shall be computed on the basis of a year of 360 days and actual days elapsed (including the first day but excluding the last day) occurring in the period for which payable, and interest on Base Rate Loans shall be computed on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed (including the first day but excluding the last day) occurring in the period for which payable. Notwithstanding the foregoing, (a) for each day that the Base Rate is calculated by reference to the Federal Funds Rate, interest on Base Rate Loans shall be computed on the basis of a year of 360 days and actual days elapsed and (b) interest on Eurocurrency Loans denominated in Pounds Sterling shall be computed on the basis of a year of 365 or 366 days, as the case may be.

4.04 <u>Minimum Amounts</u>. Except for prepayments made pursuant to Section 2.09(b) hereof, each borrowing and partial prepayment of principal of Loans (other than Money Market Loans) shall aggregate to an integral multiple of \$1,000,000 and not less than \$10,000,000 (borrowings or prepayments of Loans of different Types or, in the case of Eurocurrency Loans, having different Interest Periods at the same time hereunder to be deemed separate borrowings and prepayments for purposes of the foregoing, one for each Type or Interest Period), <u>provided</u> that (a) the aggregate principal amount of Eurocurrency Loans having the same Interest Period shall aggregate to an integral multiple of \$1,000,000 and not less than

\$20,000,000 (or, in the case of Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)) and (b) if any Eurocurrency Loans would otherwise be in a lesser principal amount for any period, such Loans shall be Base Rate Loans during such period.

4.05 <u>Certain Notices</u>. Except as otherwise provided in Section 2.03 hereof with respect to Money Market Loans, notices by a Borrower to the Administrative Agent of terminations or reductions of the Commitments and of borrowings and optional prepayments of Loans, Currencies and Types of Loans and of the duration of Interest Periods shall be irrevocable and shall be effective only if received by the Administrative Agent not later than 11:00 a.m. New York time on the number of Business Days prior to the date of the relevant termination, reduction, borrowing or prepayment or the first day of such Interest Period specified below:

Туре	Number of Business Days Prior
Termination or reduction of Commitments	3
Borrowing or prepayment of Base Rate Loans	same day
Borrowing or prepayment of, or duration of Interest Period for, Eurocurrency Loans denominated in Dollars	3
Borrowing or prepayment of, or duration of Interest Period for, Eurocurrency Loans denominated in an Agreed Alternative Currency	4

Each such notice of termination or reduction shall specify the amount of the Commitments to be terminated or reduced. Each such notice of borrowing or optional prepayment shall be in substantially the form of Exhibit D hereto and shall specify the Loans to be borrowed or prepaid and the amount (subject to Section 4.04 hereof), Currency and Type of each Loan to be borrowed or prepaid, the date of borrowing or optional prepayment (which shall be a Business Day), the Interest Period of the Loans to be borrowed or prepaid and the identity of the applicable Borrower; <u>provided</u> that any notice of borrowing given by FSB or COBE shall also be signed by COB. The Administrative Agent shall promptly notify the affected Lenders of the contents of each such notice.

4.06 <u>Non-Receipt of Funds by the Administrative Agent</u>. Unless the Administrative Agent shall have been notified by a Lender or a Borrower (the "<u>Payor</u>") prior to the date on which the Payor is to make payment to the Administrative Agent of (in the case of a Lender) the proceeds of a Loan to be made by such Lender hereunder or (in the case of a Borrower) a payment to the Administrative Agent for account of one or more of the Lenders hereunder (such payment being herein called the "<u>Required Payment</u>"), which notice shall be

Credit Agreement

effective upon receipt, that the Payor does not intend to make the Required Payment to the Administrative Agent, the Administrative Agent may assume that the Required Payment has been made and may, in reliance upon such assumption (but shall not be required to), make the amount thereof available to the intended recipient(s) on such date; and, if the Payor has not in fact made the Required Payment to the Administrative Agent, the recipient(s) of such payment shall, on demand, repay to the Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date (the "<u>Advance Date</u>") such amount was so made available by the Administrative Agent until the date the Administrative Agent recovers such amount at a rate per annum equal to the Federal Funds Rate for such day and, if such recipient(s) shall fail promptly to make such payment, the Administrative Agent shall be entitled to recover such amount, on demand, from the Payor, together with interest as aforesaid, <u>provided</u> that if neither the recipient(s) nor the Payor shall return the Required Payment to the Administrative Agent within three Business Days of the Advance Date, then, retroactively to the Advance Date, the Payor and the recipient(s) shall each be obligated to pay interest on the Required Payment as follows:

(i) if the Required Payment shall represent a payment to be made by a Borrower to the Lenders, such Borrower and the recipient(s) shall each be obligated retroactively to the Advance Date to pay interest in respect of the Required Payment at the Post-Default Rate (without duplication of the obligation of such Borrower under Section 3.02 hereof to pay interest on the Required Payment at the Post-Default Rate), it being understood that the return by the recipient(s) of the Required Payment to the Administrative Agent shall not limit such obligation of such Borrower under said Section 3.02 to pay interest at the Post-Default Rate in respect of the Required Payment; and

(ii) if the Required Payment shall represent proceeds of a Loan to be made by the Lenders to a Borrower, the Payor and such Borrower shall each be obligated retroactively to the Advance Date to pay interest in respect of the Required Payment pursuant to whichever of the rates specified in Section 3.02 hereof is applicable to the Type of such Loan, it being understood that the return by such Borrower of the Required Payment to the Administrative Agent shall not limit any claim such Borrower may have against the Payor in respect of such Required Payment.

4.07 Sharing of Payments, Etc.

(a) Each Borrower agrees that, in addition to (and without limitation of) any right of set-off, banker's lien or counterclaim a Lender may otherwise have, each Lender shall be entitled, at its option (to the fullest extent permitted by law), to set off and apply any deposit (general or special, time or demand, provisional or final), or other indebtedness, held by it for the credit or account of such Borrower at any of its offices, in Dollars or in any other currency, against any principal of or interest on any of such Lender's Loans to such Borrower or any other amount payable by such Borrower to such Lender hereunder, that is not paid when due (regardless of whether such deposit or other indebtedness is then due to such Borrower), in

Credit Agreement

which case it shall promptly notify such Borrower and the Administrative Agent thereof, <u>provided</u> that such Lender's failure to give such notice shall not affect the validity thereof.

(b) If any Lender shall obtain from a Borrower payment of any principal of or interest on any Syndicated Loan owing to such Lender or payment of any other amount owing under this Agreement (other than in respect of Money Market Loans) through the exercise of any right of set-off, banker's lien or counterclaim or similar right or otherwise (other than from the Administrative Agent as provided herein), and, as a result of such payment, such Lender shall have received a greater percentage of the principal of or interest on the Syndicated Loans or such other amounts due hereunder from such Borrower to such Lender than the percentage received by any other Lender, it shall promptly purchase from such other Lenders participations in (or, if and to the extent specified by such Lender, direct interests in) the Loans or such other amounts, respectively, owing to such other Lenders (or in interest due thereon, as the case may be) in such amounts, and make such other adjustments from time to time as shall be equitable, to the end that all the Lenders shall share the benefit of such excess payment (net of any expenses that may be incurred by such Lender in obtaining or preserving such excess payment) pro rata in accordance with the unpaid principal of and/or interest on the Loans or such other amounts, respectively, owing to each of the Lenders. To such end all the Lenders shall make appropriate adjustments among themselves (by the resale of participations sold or otherwise) if such payment is rescinded or must otherwise be restored. This Section 4.07(b) shall not apply to a payment to a Non-Extending Lender pursuant to Section 2.12(b) hereof.

(c) The Borrower obligated in respect of such Loans or other amounts agrees that any Lender so purchasing such a participation (or direct interest) may exercise all rights of set-off, banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans or other amounts (as the case may be) owing to such Lender in the amount of such participation.

(d) Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise, and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of a Borrower. If, under any applicable bankruptcy, insolvency or other similar law, any Lender receives a secured claim in lieu of a set-off to which this Section 4.07 applies, such Lender shall, to the extent practicable, exercise its rights in respect of such secured claim in a manner consistent with the rights of the Lenders entitled under this Section 4.07 to share in the benefits of any recovery on such secured claim.

SECTION 5. Yield Protection, Etc.

5.01 Additional Costs.

(a) Each Borrower shall pay directly to each Lender from time to time such amounts as such Lender may determine to be necessary to compensate such Lender for any costs that such Lender determines are attributable to its making or maintaining of any Fixed Rate

Credit Agreement

Loans owing by such Borrower or its obligation to make to such Borrower any Fixed Rate Loans hereunder, or any reduction in any amount receivable by such Lender hereunder in respect of any of such Loans or such obligation (such increases in costs and reductions in amounts receivable being herein called "<u>Additional</u> <u>Costs</u>"), resulting from any Regulatory Change (including without limitation, the introduction of, changeover to or operation of the Euro in a Participating Member State) that:

(i) shall subject any Lender (or its Applicable Lending Office for any of such Loans) to any Taxes; or

(ii) imposes or modifies any reserve, special deposit or similar requirements (other than, in the case of any Lender for any period as to which a Borrower is required to pay any amount under Section 5.01(d) hereof, the reserves against "Eurocurrency liabilities" under Regulation D referred to therein) relating to any extensions of credit or other assets of, or any deposits with or other liabilities of, such Lender (including, without limitation, any of such Loans or any deposits referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof), or any commitment of such Lender (including, without limitation, the Commitment of such Lender hereunder); or

(iii) imposes any other condition affecting this Agreement or (if any) its Notes (or any of such extensions of credit or liabilities) or its Commitment.

If any Lender requests compensation from a Borrower under this Section 5.01(a), such Borrower may, by notice to such Lender (with a copy to the Administrative Agent), suspend the obligation of such Lender thereafter to make Eurocurrency Loans to such Borrower until the Regulatory Change giving rise to such request ceases to be in effect (in which case the provisions of Section 5.04 hereof shall be applicable), provided that such suspension shall not affect the right of such Lender to receive the compensation so requested.

(b) Without limiting the effect of the foregoing provisions of this Section 5.01 (but without duplication), if any Lender shall have determined that any law or regulation or any interpretation, directive or request (whether or not having the force of law and whether or not failure to comply therewith would be unlawful) of any court or governmental or monetary authority, (i) following any Regulatory Change or (ii) implementing any risk-based capital guideline or other requirement (whether or not having the force of law and whether or not the failure to comply therewith would be unlawful) hereafter issued by any government or governmental or supervisory authority implementing at the national level any change in the Basle Accord, has or would have the effect of reducing the rate of return on assets or equity of such Lender (or any Applicable Lending Office of such Lender or any bank holding company of which such Lender is a subsidiary) as a consequence of such Lender's Commitment to make or maintain Loans to a Borrower or Loans made to such Borrower to a level below that which such Lender (or such Applicable Lending Office or such bank holding company) could have achieved but for such law, regulation, interpretation, directive or request, then such Borrower shall pay directly to each Lender from time to time on request such amounts as such Lender may

Credit Agreement

determine to be necessary to compensate such Lender (or, without duplication, such Applicable Lending Office or such bank holding company) for such reduction.

(c) Each Lender shall notify the relevant Borrower of any event occurring after the date hereof entitling such Lender to compensation from such Borrower under paragraph (a) or (b) of this Section 5.01 as promptly as practicable, but in any event within 45 days, after such Lender obtains actual knowledge thereof; provided that (i) if any Lender fails to give such notice within 45 days after it obtains actual knowledge of such an event, such Lender shall, with respect to compensation payable by such Borrower pursuant to this Section 5.01 in respect of any costs resulting from such event, only be entitled to payment under this Section 5.01 for costs incurred from and after the date 45 days prior to the date that such Lender does give such notice and (ii) each Lender will designate a different Applicable Lending Office for the Loans of such Lender affected by such event if such designation will avoid the need for, or reduce the amount of, such compensation and will not, in the sole opinion of such Lender, be disadvantageous to such Lender, except that such Lender shall have no obligation to designate an Applicable Lending Office located in the United States of America. Each Lender will furnish to the relevant Borrower a certificate setting forth the basis and amount of each request by such Lender for compensation under paragraph (a) or (b) of this Section 5.01, or of the effect of capital maintained pursuant to paragraph (b) of this Section 5.01, on its costs or rate of return of maintaining Loans or its obligation to make Loans, or on amounts receivable by it in respect of Loans, and of the amounts required to compensate such Lender under this Section 5.01, shall be conclusive, provided that such determinations and allocations are made on a reasonable basis.

(d) Without limiting the effect of the foregoing, each Borrower shall pay to each Lender on the last day of the Interest Period therefor so long as such Lender is maintaining reserves against "Eurocurrency liabilities" under Regulation D (or, unless the provisions of paragraph (b) above are applicable, so long as such Lender is, by reason of any Regulatory Change, maintaining reserves against any other category of liabilities that includes deposits by reference to which the interest rate on Fixed Rate Loans is determined as provided in this Agreement or against any category of extensions of credit or other assets of such Lender that includes any Fixed Rate Loans) an additional amount (determined by such Lender and notified to such Borrower through the Administrative Agent) equal to the product of the following for each Fixed Rate Loan to such Borrower for each day during such Interest Period:

(i) the principal amount of such Fixed Rate Loan outstanding on such day; and

(ii) the remainder of (x) a fraction the numerator of which is the rate (expressed as a decimal) at which interest accrues on such Fixed Rate Loan for such Interest Period as provided in this Agreement (less the Applicable Margin) and the denominator of which is one <u>minus</u> the effective rate (expressed as a decimal) at which such Reserve Requirements are imposed on such Lender on such day <u>minus</u> (y) such numerator; and

Credit Agreement

(iii) 1/360.

Notwithstanding the foregoing, this Section 5.01 does not apply to the extent that any Additional Costs are compensated for by Section 5.06 or would have been so compensated but for the application of any exclusion under Section 5.06(i).

5.02 Limitation on Types of Loans. Anything herein to the contrary notwithstanding, if, on or prior to the determination of any Fixed Base Rate for any Interest Period:

(a) the Administrative Agent determines, which determination shall be conclusive, that quotations of interest rates for the relevant deposits referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof are not being provided in the relevant amounts or Currencies or for the relevant maturities for purposes of determining rates of interest for either Type of Fixed Rate Loans as provided herein; or

(b) the Majority Lenders determine (or any Lender that has outstanding a Money Market Quote with respect to a LIBOR Market Loan determines), which determination shall be conclusive, and notify (or notifies, as the case may be) the Administrative Agent that the relevant rates of interest referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof upon the basis of which the rate of interest for Eurocurrency Loans (or LIBOR Market Loans, as the case may be) in any Currency for such Interest Period is to be determined will not adequately and fairly reflect the cost to such Lenders (or to such quoting Lender) of making or maintaining Eurocurrency Loans (or LIBOR Market Loans, as the case may be) in such Currency for such Interest Period;

then the Administrative Agent shall give each affected Borrower and Lender prompt notice thereof and, so long as such condition remains in effect, the Lenders (or such quoting Lender) shall be under no obligation to make additional Eurocurrency Loans in such Currency, and such Lender shall no longer be obligated to make any LIBOR Market Loan in such Currency that it has offered to make.

5.03 <u>Illegality</u>; <u>Agreed Alternative Currencies</u>. Notwithstanding any other provision of this Agreement, in the event that it becomes unlawful for any Lender or its Applicable Lending Office to honor its obligation to make or maintain Eurocurrency Loans or LIBOR Market Loans in any Currency hereunder (and, in the sole opinion of such Lender, the designation of a different Applicable Lending Office would either not avoid such unlawfulness or would be disadvantageous to such Lender), then such Lender shall promptly notify each affected Borrower thereof (with a copy to the Administrative Agent) and such Lender's obligation to make Eurocurrency shall be suspended until such time as such Lender may again make and maintain Eurocurrency Loans in such Currency (in which case the provisions of Section 5.04 hereof shall be applicable), and such Lender shall no longer be obligated to make any LIBOR Market Loan in such Currency that it has offered to make.

Credit Agreement

5.04 <u>Treatment of Affected Loans</u>. If the obligation of any Lender to make Eurocurrency Loans in Dollars shall be suspended pursuant to Section 5.01 or 5.03 hereof, then, unless and until such Lender gives notice as provided below that the circumstances specified in Section 5.01 or 5.03 hereof that gave rise to such suspension no longer exist (which such Lender agrees to do promptly upon such circumstances ceasing to exist), all Loans that would otherwise be made by such Lender as Eurocurrency Loans in Dollars shall be made instead as Base Rate Loans. If the obligation of any Lender to make Eurocurrency Loans denominated in any Agreed Alternative Currency shall be suspended pursuant to Section 5.01 or 5.03 hereof, then, unless and until such Lender gives notice as provided below that the circumstances specified in Section 5.01 or 5.03 hereof that gave rise to such suspension no longer exist (which such Lender agrees to do promptly upon such circumstances ceasing to exist), all Loans that would otherwise be made by such Lender as Eurocurrency Loans in such Agreed Alternative Currency shall, except as provided in the immediately preceding sentence, be made instead as Eurocurrency Loans denominated in Dollars.

5.05 <u>Compensation</u>. Each Borrower shall pay to the Administrative Agent for account of each Lender, upon the request of such Lender through the Administrative Agent, such amount or amounts as shall be sufficient (in the reasonable opinion of such Lender) to compensate such Lender for any loss, cost or expense that such Lender determines is attributable to:

(a) any payment or mandatory or optional prepayment of a Fixed Rate Loan or a Set Rate Loan made by such Lender to such Borrower (which shall not include the return by a Borrower pursuant to Section 4.06 hereof of any Required Payment previously advanced to such Borrower by the Administrative Agent on behalf of a Lender) for any reason (including, without limitation, the acceleration of the Loans pursuant to Section 9 hereof) on a date other than the last day of the Interest Period for such Loan; or

(b) any failure by such Borrower for any reason (including, without limitation, the failure of any of the conditions precedent specified in Section 6 hereof to be satisfied, but not including as a result of breach by a Lender of its obligation, subject to the terms and conditions hereof, to make its Loan) to borrow a Fixed Rate Loan or a Set Rate Loan (with respect to which, in the case of a Money Market Loan, such Borrower has accepted a Money Market Quote) from such Lender on the date for such borrowing specified in the relevant notice of borrowing given pursuant to Section 2.02 or 2.03(b) hereof.

Such compensation shall be equal to an amount equal to the excess, if any, of (i) the amount of interest that otherwise would have accrued on the principal amount so paid, prepaid or not borrowed for the period from the date of such payment, prepayment or failure to borrow to the last day of the then current Interest Period for such Loan (or, in the case of a failure to borrow, the Interest Period for such Loan that would have commenced on the date specified for such borrowing) at the Eurocurrency Rate for such Loan for such Interest Period over (ii) the amount of interest that otherwise would have accrued on such principal amount at a rate per annum equal to the interest component of the amount such Lender would have bid in the London interbank

Credit Agreement

market (if such Loan is a Eurocurrency Loan or a LIBOR Market Loan) or the United States secondary certificate of deposit market (if such Loan is a Set Rate Loan) for deposits denominated in the relevant Currency of leading banks in amounts comparable to such principal amount and with maturities comparable to such period (as reasonably determined by such Lender). Any Lender requesting compensation pursuant to this Section 5.05 will furnish to the relevant Borrower a certificate setting forth its computation of the amount of such compensation, which certificate shall be conclusive as to the amount of such compensation provided that the computations made therein are made on a reasonable basis.

5.06 <u>Taxes</u>.

(a) Any and all payments by each Borrower hereunder shall be made, in accordance with Section 4.01, free and clear of and without deduction or liability for any and all current or future taxes, levies, imposts, deductions, charges or withholdings, and all liabilities with respect thereto, <u>excluding</u> taxes (including, without limitation, taxes on net income, profits or gains) imposed on the Administrative Agent or any Lender (or any transferee or assignee thereof, including a participation holder (any such entity a "<u>Transferee</u>")) as a result of a present, former or future connection between the jurisdiction of the governmental authority imposing such tax or any political subdivision or taxing authority thereof or therein and the Administrative Agent or such Lender (other than a connection resulting from or attributable to the Administrative Agent or such Lender having executed, delivered or performed its obligations, or enforced, this Agreement or any Note) (all such nonexcluded taxes, levies, imposts, deductions, charges, withholdings and liabilities, collectively or individually, "<u>Taxes</u>"). If a Borrower shall be required to deduct any Taxes from or in respect of any sum payable hereunder to any Lender (or any Transferee) or the Administrative Agent, or any Lender (or any Transferee) or the Administrative Agent shall be required to pay such Taxes, (i) the sum payable shall be increased by the amount (an "<u>additional amount</u>") necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section 5.06) such Lender (or Transferee) or the Administrative Agent (as the case may be) shall receive an amount equal to the sum it would have received had no such deductions been made, (ii) such Borrower shall pay the full amount deducted to the relevant governmental authority in accordance with applicable law.

(b) In addition, each Borrower agrees to pay to the relevant governmental authority in accordance with applicable law any current or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies that arise from any payment made hereunder or from the execution, delivery or registration of, or otherwise with respect to, this Agreement or any Note ("<u>Other Taxes</u>").

(c) Each Borrower will indemnify each Lender (or Transferee) and the Administrative Agent on an after-tax basis for the full amount of Taxes and Other Taxes paid by such Lender (or Transferee) or the Administrative Agent, as the case may be, and any liability (including penalties, interest and expenses (including reasonable attorney's fees and expenses)) arising therefrom or with respect thereto (except in the case of gross negligence or willful

Credit Agreement

misconduct of such Lender (or Transferee) or the Administrative Agent), whether or not such Taxes or Other Taxes were correctly or legally asserted by the relevant governmental authority. A certificate as to the amount of such payment or liability prepared by a Lender (or Transferee), or the Administrative Agent, absent manifest error, shall be final, conclusive and binding for all purposes. Such indemnification shall be made within 30 days after the date such Lender (or Transferee) or the Administrative Agent, as the case may be, makes written demand therefor.

(d) If a Borrower determines in good faith that a reasonable basis exists for contesting a Tax, the relevant Lender (or Transferee), or the Administrative Agent, as applicable, shall cooperate with such Borrower in challenging such Tax in such Borrower's name at such Borrower's expense if requested by such Borrower.

(e) As soon as practicable after the date of any payment of Taxes or Other Taxes by a Borrower to the relevant governmental authority, such Borrower will deliver to the Administrative Agent, at its address referred to in Section 11.02, the original or a certified copy of a receipt issued by such governmental authority evidencing payment thereof.

(f) Without prejudice to the survival of any other agreement contained herein, the agreements and obligations contained in this Section 5.06 shall survive the payment in full of the principal of and interest on all Loans made hereunder.

(g) Each U.S. Borrower agrees to pay to each Lender that is not a U.S. Person such additional amounts as are necessary in order that the net payment of any amount due from such U.S. Borrower to such non-U.S. Person hereunder after deduction for or withholding in respect of any U.S. Taxes imposed with respect to such payment (or in lieu thereof, payment of such U.S. Taxes by such non-U.S. Person), will not be less than the amount stated herein to be then due and payable, provided that the foregoing obligation to pay such additional amounts shall not apply:

(i) to any payment to any Lender hereunder unless such Lender is, on the date hereof (or on the date it becomes a Lender hereunder as provided in Section 2.10, 2.12, 5.07 or 11.06(b) hereof) and on the date of any change in the Applicable Lending Office of such Lender, either entitled to submit a Form W-8BEN (relating to such Lender and entitling it to a complete exemption from withholding on all interest to be received by it hereunder in respect of the Loans) or Form W-8ECI (relating to all interest to be received by such Lender hereunder in respect of the Loans), or

(ii) to any U.S. Taxes imposed solely by reason of the failure by such non-U.S. Person to comply with applicable certification, information, documentation or other reporting requirements concerning the nationality, residence, identity or connections with the United States of America of such non-U.S. Person if such compliance is required by statute or regulation of the United States of America as a precondition to relief or exemption from such U.S. Taxes.

Credit Agreement

For the purposes of this Section 5.06(g), (A) "<u>U.S. Person</u>" shall mean a citizen, national or resident of the United States of America, a corporation, partnership or other entity created or organized in or under any laws of the United States of America or any State thereof, or any estate or trust that is subject to Federal income taxation regardless of the source of its income, (B) "<u>U.S. Taxes</u>" shall mean any present or future tax, assessment or other charge or levy imposed by or on behalf of the United States of America or any taxing authority thereof or therein, (C) "<u>Form W-8BEN</u>" shall mean Form W-8EBN (Certificate of Foreign Status of Beneficial Owner for United States Tax Withholding) of the Department of the Treasury of the United States of America and (D) "<u>Form W-8ECI</u>" shall mean Form W-8ECI (Certificate of Foreign Person's Claim for Exemption from Withholding of Tax on Income Effectively Connected with the Conduct of a Trade or Business in the United States) of the Department of the Treasury of the United States of America. Each of the Forms referred to in the foregoing clauses (C) and (D) shall include such successor and related forms as may from time to time be adopted by the relevant taxing authorities of the United States of America to document a claim to which such Form relates.

(h) Within 30 days after paying any amount to the Administrative Agent or any Lender from which it is required by law to make any deduction or withholding, and within 30 days after it is required by law to remit such deduction or withholding to any relevant taxing or other authority, the relevant Borrower shall deliver to the Administrative Agent for delivery to such non-U.S. Person evidence satisfactory to such Person of such deduction, withholding or payment (as the case may be).

(i) In relation to any payments of interest by COBE, there shall be no obligation on COBE to pay any additional amount as described in (a) above or to indemnify any Person pursuant to clause (c) above where the relevant Lender or Transferee was not, on the date of this Agreement, or has ceased to be (other than by reason of a change in law or the interpretation of any law) a Qualifying Bank.

(j) COBE and each Lender and the Administrative Agent and the relevant Transferee shall reasonably cooperate in completing any procedural formalities necessary to enable COBE to obtain authorization (where such authorization is necessary) to make any payment without any deduction for or on account of any Taxes.

(k) If the Administrative Agent or a Lender or Transferee determines, in its sole discretion, that it has received a refund of any Taxes or Other Taxes as to which it has been indemnified by a Borrower or with respect to which a Borrower has paid additional amounts pursuant to this Section 5.06, it shall pay over such refund to such Borrower (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section 5.06 with respect to the Taxes or Other Taxes giving rise to such refund), net of all reasonable out-of-pocket expenses of the Administrative Agent or such Lender or Transferee and without interest (other than any interest paid by the relevant governmental authority with respect to such refund); provided, that the Borrower, upon the request of the Administrative Agent or such Lender or Transferee, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant governmental authority) to the Administrative Agent or

Credit Agreement

such Lender or Transferee in the event the Administrative Agent or such Lender is required to repay such refund to such governmental authority. This Section shall not be construed to require the Administrative Agent or any Lender or Transferee to make available its tax returns (or any other information relating to its taxes which it deems confidential) to the Borrower or any other Person.

5.07 Replacement of Lenders. If any Lender requests compensation pursuant to Section 5.01 or 5.06 hereof, or any Lender's obligation to make Loans of any Type or in any Currency shall be suspended pursuant to Section 5.01 or 5.03 hereof, or any Lender becomes a Defaulting Lender pursuant to Section 11.04 hereof (any such Lender requesting such compensation, or whose obligations are so suspended, or that becomes and remains a Defaulting Lender, being herein called a "Subject Lender"), the Borrowers, upon three Business Days notice, may (jointly but not severally) require that such Subject Lender transfer all of its right, title and interest under this Agreement and such Subject Lender's Notes to any bank or other financial institution (a "Proposed Lender") identified by the Borrowers that is reasonably satisfactory to the Administrative Agent (i) if such Proposed Lender agrees to assume all of the obligations of such Subject Lender hereunder, and to purchase all of such Subject Lender's Loans hereunder for consideration equal to the aggregate outstanding principal amount of such Subject Lender's Loans, together with interest thereon to the date of such purchase, and satisfactory arrangements are made for payment to such Subject Lender of all other amounts payable hereunder to such Subject Lender on or prior to the date of such transfer (including any fees accrued hereunder and any amounts that would be payable under Section 5.05 hereof as if all of such Subject Lender's Loans were being prepaid in full on such date) and (ii) if such Subject Lender has requested compensation pursuant to Section 5.01 or 5.06 hereof, such Proposed Lender's aggregate requested compensation, if any, pursuant to said Section 5.01 or 5.06 with respect to such Subject Lender's Loans is lower than that of the Subject Lender. Subject to the provisions of Section 11.06(b) hereof, such Proposed Lender shall be a "Lender" for all purposes hereunder. Without prejudice to the survival of any other agreement of the Borrowers hereunder, the agreements of the Borrowers contained in Sections 2.11 (subject to the expiration of the Undertaking in accordance with the express terms of said Section), 5.01, 5.06, 11.03 and 11.13 hereof (without duplication of any payments made to such Subject Lender by the Borrowers or the Proposed Lender) shall survive for the benefit of such Subject Lender under this Section 5.07 with respect to the time prior to such replacement.

SECTION 6. Conditions Precedent.

6.01 <u>Conditions to Effectiveness</u>. The obligation of the Lenders to make their Loans on the occasion of the initial borrowing hereunder is subject to the conditions precedent that (i) the Effective Date shall have occurred on or before June 29, 2004 and (ii) the Administrative Agent shall have received the following documents, each of which shall be satisfactory to the Administrative Agent and special New York counsel to JPMorgan in form and substance:

(a) <u>Corporate Documents</u>. Certified copies of the charter and by-laws (or equivalent documents) of each Borrower and of all corporate authority for each Borrower (including, without limitation, board of director resolutions and evidence of the incumbency, including specimen signatures, of officers) with respect to the execution, delivery and performance of the Basic Documents and each other document to be delivered by such Borrower from time to time in connection herewith and the Loans hereunder (and the Administrative Agent and each Lender may conclusively rely on such certificate until it receives notice in writing from the relevant Borrower to the contrary).

Credit Agreement

(b) <u>Officer's Certificate</u>. A certificate of a senior officer of each Borrower, dated the Effective Date, to the effect that (i) no Default has occurred and is continuing and (ii) the representations and warranties made by the Borrowers in Section 7 hereof (including the last sentence of Section 7.02 hereof and in Section 7.03 hereof) are true and complete on and as of the Effective Date with the same force and effect as if made on and as of such date (or, if any such representation or warranty is expressly stated to have been made as of a specific date, as of such specific date).

(c) <u>Opinion of Special U.S. Counsel to the Borrowers</u>. An opinion, dated the Effective Date, of Clifford Chance, special U.S. counsel to the Borrowers, substantially in the form of Exhibit B-1 hereto and covering such other matters as the Administrative Agent may reasonably request (and the Borrowers hereby instruct such counsel to deliver such opinion to the Lenders and the Administrative Agent).

(d) <u>Opinion of Special English Counsel to COBE</u>. An opinion, dated the Effective Date, of Clifford Chance, special English counsel to COBE, substantially in the form of Exhibit B-2 hereto and covering such other matters as the Administrative Agent may reasonably request (and COBE hereby instructs such counsel to deliver such opinion to the Lenders and the Administrative Agent).

(e) <u>Opinion of Counsel to the Borrowers</u>. An opinion, dated the Effective Date, of John G. Finneran, Jr., Esq., General Counsel and Secretary of the Borrowers, or Frank R. Borchert III, Esq., Deputy General Counsel of the Borrowers, substantially in the form of Exhibit B-3 hereto and covering such other matters as the Administrative Agent may reasonably request (and the Borrowers hereby instruct such counsel to deliver such opinion to the Lenders and the Administrative Agent).

(f) Opinion of Special New York Counsel to JPMorgan. An opinion, dated the Effective Date, of Milbank, Tweed, Hadley & McCloy LLP, special New York counsel to JPMorgan, substantially in the form of Exhibit C hereto (and JPMorgan hereby instructs such counsel to deliver such opinion to the Lenders).

(g) <u>Notes</u>. If applicable, any Notes, duly completed and executed for each Lender requesting such Notes.

(h) <u>Existing Credit Agreement</u>. Evidence of the termination of the Commitments as defined in the Existing Credit Agreements and of the payment of all amounts payable to the Lenders and the Administrative Agent as defined therein.

Credit Agreement

(i) Other Documents. Such other documents as the Administrative Agent or special New York counsel to JPMorgan may reasonably request.

The effectiveness of the obligations of any Lender hereunder is also subject to the payment by the Borrowers of such fees as the Borrowers shall have agreed to pay or deliver to any Lender or the Administrative Agent in connection herewith, including, without limitation, the reasonable fees and expenses of Milbank, Tweed, Hadley & McCloy LLP, special New York counsel to JPMorgan, in connection with the negotiation, preparation, execution and delivery of this Agreement and any Notes (to the extent that statements for such fees and expenses have been delivered to the Borrowers).

6.02 Initial and Subsequent Loans. The obligation of any Lender to make any Loan (including any Money Market Loan) to a Borrower upon the occasion of each borrowing hereunder is subject to the further conditions precedent that:

(a) in the case of a Syndicated Loan, the applicable Borrower shall have given notice of such borrowing by delivery of a Notice of Borrowing in substantially the form of Exhibit D hereto to the Administrative Agent;

(b) in the case of a Money Market Loan, the applicable Borrower shall have requested that the Lenders make offers to make Money Market Loans by delivery of a Money Market Quote Request in substantially the form of Exhibit E hereto to the Administrative Agent; and

(c) both immediately prior to the making of such Loan and also after giving effect thereto and to the intended use thereof, but only if such borrowing will increase the aggregate outstanding principal amount of the Loans owing by such Borrower to any Lender hereunder:

(i) no Default shall have occurred and be continuing; and

(ii) the representations and warranties (x) made by such Borrower in Section 7 hereof (other than the Excluded Representations) or, if such borrowing will increase the outstanding aggregate principal amount of the Loans, the representations and warranties made by each Borrower in Section 7 hereof (other than the Excluded Representations), and (y) in the case of a borrowing by FSB or COBE, by COB (other than the Excluded Representations) shall be true and complete in all material respects on and as of the date of the making of such Loan with the same force and effect as if made on and as of such date (or, if any such representation or warranty is expressly stated to have been made as of a specific date, as of such specific date).

Credit Agreement

SECTION 7. Representations and Warranties. Each Borrower represents and warrants to the Administrative Agent and the Lenders that:

7.01 <u>Corporate Existence</u>. Each of such Borrower and its Subsidiaries: (a) is a corporation, partnership, limited liability company or other entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization; (b) (i) has all requisite corporate or other power and (ii) except to the extent it could not reasonably be expected to have a Material Adverse Effect, has all governmental licenses, authorizations, consents and approvals necessary to own its assets and carry on its business as now being conducted; and (c) is qualified to do business and is in good standing in all jurisdictions in which the nature of the business conducted by it makes such qualification necessary and where failure so to qualify could (either individually or in the aggregate) reasonably be expected to have a Material Adverse Effect. COB is a member in good standing of the Federal Reserve System, and COB's deposit accounts are insured by the Federal Deposit Insurance Corporation, and no proceedings for the termination or revocation of such insurance are pending or, to the knowledge of any Borrower, threatened.

7.02 <u>Financial Condition</u>. COFC has heretofore furnished to each of the Lenders a consolidated balance sheet of COFC and its Subsidiaries as at December 31, 2003 and the related consolidated statements of income, changes in stockholders'/division equity and cash flows of COFC and its Subsidiaries for the fiscal year ended on said date, with the opinion thereon of Ernst & Young LLP. Such financial statements present fairly, in all material respects, the consolidated financial condition of COFC and its Subsidiaries as at said date and the consolidated results of their operations and their cash flows for the fiscal year ended on said date, all in accordance with generally accepted accounting principles in the United States of America and practices applied on a consistent basis. Since December 31, 2003, there has been no material adverse change in the Property, business, operations or condition (financial or otherwise) or capitalization of COFC and its Subsidiaries taken as a whole from that set forth in said financial statements as at said date.

7.03 Litigation. Except as identified in Schedule 7.03 hereto, there are no legal or arbitral proceedings, or any proceedings by or before any governmental or regulatory authority or agency, now pending or (to the knowledge of any Borrower) threatened against or affecting COFC or any of its Subsidiaries as to which there is a reasonable possibility of an adverse determination that could (either individually or in the aggregate) have a Material Adverse Effect.

7.04 <u>No Breach</u>. None of the execution and delivery of this Agreement and the Notes, the consummation of the transactions herein contemplated or compliance with the terms and provisions hereof will conflict with or result in a breach of, or require any consent under, the charter or by-laws (or equivalent documents) of any Borrower, or any applicable law or regulation, or any order, writ, injunction or decree of any court or governmental authority or agency, or any agreement or instrument to which COFC or any of its Subsidiaries is a party or by which any of them or any of their Property is bound or to which any of them is subject, or

constitute a default under any such agreement or instrument, except for any such conflict, breach or default that, or consent that if not obtained, could not reasonably be expected (either individually or in the aggregate) to have a Material Adverse Effect and could not subject the Administrative Agent or any Lender to any material liability.

7.05 <u>Action</u>. Each Borrower has all necessary corporate power, authority and legal right to execute, deliver and perform its obligations under each of the Basic Documents to which it is a party and to consummate the transactions contemplated thereby; the execution, delivery and performance by each Borrower of each of the Basic Documents to which it is a party and the consummation of the transactions contemplated thereby have been duly authorized by all necessary corporate or other action on its part (including, without limitation, any required shareholder approvals); and this Agreement has been duly and validly executed and delivered by each Borrower and constitutes, and each Note when executed and delivered by it for value will constitute, its legal, valid and binding obligation, enforceable against such Borrower in accordance with its terms, except as may be limited by (a) bankruptcy, insolvency, receivership, conservatorship, reorganization, moratorium or similar laws of general applicability affecting the enforcement of creditors' rights and (b) the application of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

7.06 <u>Approvals</u>. No authorizations, approvals or consents of, and no filings or registrations with, any governmental or regulatory authority or agency, or any securities exchange, are necessary for the execution, delivery or performance by any Borrower of this Agreement or the Notes of such Borrower or for the consummation of any the transactions contemplated hereby or thereby or for the legality, validity or enforceability hereof or thereof.

7.07 Use of Credit. No part of the proceeds of the Loans hereunder will be used for the purpose, whether immediate, incidental or ultimate, of buying or carrying any Margin Stock.

7.08 <u>ERISA</u>. Each Plan, and, to the knowledge of each Borrower, each Multiemployer Plan, is in compliance with, and has been administered in compliance with, the applicable provisions of ERISA, the Code and the Age Discrimination in Employment Act, as amended except for non-compliance which could not reasonably be expected to have a Material Adverse Effect.

7.09 Taxes. COFC and its Subsidiaries are members of an affiliated group of corporations filing consolidated returns for Federal income tax purposes, of which COFC is the "common parent" (within the meaning of Section 1504 of the Code) of such group. COFC and its Subsidiaries have filed all Federal income tax returns and all other material tax returns that are required to be filed by them and have paid all taxes due pursuant to such returns or pursuant to any assessment received by COFC or any of its Subsidiaries, except in each case for taxes being contested in good faith by appropriate proceedings and as to which appropriate reserves are being maintained. The charges, accruals and reserves on the books of COFC and its Subsidiaries in respect of taxes and other governmental charges have been made in accordance with GAAP.

7.10 <u>Investment Company Act</u>. Neither COFC nor any of its Subsidiaries is an "investment company", or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended.

7.11 <u>Public Utility Holding Company Act</u>. Neither COFC nor any of its Subsidiaries is a "holding company", or an "affiliate" of a "holding company" or a "subsidiary company" of a "holding company", within the meaning of the Public Utility Holding Company Act of 1935, as amended.

7.12 Environmental Matters. Each of COFC and its Subsidiaries has obtained all environmental, health and safety permits, licenses and other authorizations required under all Environmental Laws to carry on its business as now being or as proposed to be conducted, except to the extent failure to have any such permit, license or authorization could not (either individually or in the aggregate) reasonably be expected to have a Material Adverse Effect. Each of such permits, licenses and authorizations is in full force and effect, and each of COFC and its Subsidiaries is in compliance with the terms and conditions thereof, and is also in compliance with all other limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables contained in any applicable Environmental Law or in any regulation, code, plan, order, decree, judgment, injunction, notice or demand letter issued, entered, promulgated or approved thereunder, except to the extent failure to comply therewith could not (either individually or in the aggregate) reasonably be expected to have a Material Adverse Effect.

7.13 <u>True and Complete Disclosure</u>. As of the date hereof, the information, reports, financial statements, exhibits and schedules furnished in writing by or on behalf of any Borrower to the Administrative Agent or any Lender in connection with the negotiation, preparation or delivery of this Agreement, when taken as a whole, do not contain any untrue statement of material fact or omit to state any material fact necessary to make the statements herein or therein, in light of the circumstances under which they were made, not misleading. All written information furnished after the date hereof by or on behalf of any Borrower to the Administrative Agent and the Lenders in connection with this Agreement and the transactions contemplated hereby will be true, complete and accurate in every material respect, or (in the case of projections) based on reasonable estimates, on the date as of which such information is stated or certified.

7.14 Legal Form. This Agreement and each Note of COBE is in proper legal form under the laws of England and Wales for the enforcement thereof against COBE in the English courts and an English court will give effect to those provisions that provide that such documents are to be governed by and construed in accordance with the laws of the State of New York, subject to the exceptions and qualifications described in the opinion of special English counsel delivered pursuant to Section 6.01(d) hereof. All formalities required in the United Kingdom for the validity and enforceability against COBE of each of the Basic Documents to which it is party have been accomplished, and no Taxes or Other Taxes are required to be paid, and no notarization is required, for the validity and enforceability thereof against COBE.

Notwithstanding anything in this Section 7 to the contrary, none of COB, FSB or COBE makes any representations or warranties under any of Sections 7.01, 7.04, 7.05, 7.06, 7.10, 7.11 and 7.12 as to COFC or any of its Subsidiaries (other than, in each case, with respect to COB, FSB, COBE and/or any of their respective Subsidiaries).

SECTION 8. <u>Covenants</u>. Each Borrower covenants and agrees with the Lenders and the Administrative Agent that, so long as any Commitment or Loan is outstanding and until payment in full of all accrued and outstanding amounts payable by each Borrower hereunder:

8.01 <u>Financial Statements Etc</u>. Each Borrower shall deliver or cause to be delivered or otherwise made available through electronic media (<u>provided</u> that the Borrowers shall give prior written notice to the Administrative Agent (which shall notify the Lenders) of such availability) to the Administrative Agent:

(a) as soon as available and in any event within 60 days after the end of each of the first three quarterly fiscal periods of each fiscal year of COFC, consolidated statements of income, changes in stockholders'/division equity and cash flows and the related consolidated balance sheet of COFC and its Subsidiaries as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding periods in the preceding fiscal year (except that, in the case of balance sheets, such comparison shall be to the last day of the prior fiscal year), accompanied by a certificate of a senior financial officer of COFC, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations of COFC and its Subsidiaries in accordance with GAAP, consistently applied, as at the end of, and for, such period (subject to normal year-end audit adjustments and absence of footnotes);

(b) as soon as available and in any event within 120 days after the end of each fiscal year of COFC, consolidated statements of income, changes in stockholders'/division equity and cash flows and the related consolidated balance sheet of COFC and its Subsidiaries as at the end of such fiscal year, setting forth in each case in comparative form the corresponding consolidated figures as of the end of and for the preceding fiscal year, and accompanied by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations and cash flows of COFC and its Subsidiaries as at the end of, and for, such fiscal year in accordance with GAAP (or, in lieu thereof, copies of COFC's Annual Report on Form 10-K as filed with the SEC containing such financial statements and information);

(c) as soon as available and in any event within 60 days after the end of each of the first three quarterly fiscal periods of each fiscal year of COB, consolidated statements of income of COB and its Subsidiaries for such period and for the period from

Credit Agreement

the beginning of the respective fiscal year to the end of such period, and the related consolidated balance sheet of COB and its Subsidiaries as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding periods in the preceding fiscal year (except that, in the case of balance sheets, such comparison shall be to the last day of the prior fiscal year), accompanied by a certificate of a senior financial officer of COB, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations of COB and its Subsidiaries in accordance with GAAP, consistently applied, as at the end of, and for, such period (subject to normal year-end audit adjustments and absence of footnotes);

(d) as soon as available and in any event within 120 days after the end of each fiscal year of COB, consolidated statements of income, changes in stockholders' equity and cash flows of COB and its Subsidiaries for such fiscal year and the related consolidated balance sheet of COB and its Subsidiaries as at the end of such fiscal year, setting forth in each case in comparative form the corresponding consolidated figures as of the end of and for the preceding fiscal year, accompanied by a certificate of a senior financial officer of COB, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations and cash flows of COB and its Subsidiaries as at the end of, and for, such fiscal year in accordance with GAAP;

(e) as soon as available and in any event within 60 days after the end of each of the first three quarterly fiscal periods of each fiscal year of FSB, consolidated statements of income of FSB and its Subsidiaries for such period and for the period from the beginning of the respective fiscal year to the end of such period, and the related consolidated balance sheet of FSB and its Subsidiaries as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding periods in the preceding fiscal year (except that, in the case of balance sheets, such comparison shall be to the last day of the prior fiscal year), accompanied by a certificate of a senior financial officer of FSB, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations of FSB and its Subsidiaries in accordance with GAAP, as at the end of, and for, such period (subject to normal year-end audit adjustments and absence of footnotes);

(f) as soon as available and in any event within 120 days after the end of each fiscal year of FSB, consolidated statements of income, changes in stockholders' equity and cash flows of FSB and its Subsidiaries for such fiscal year and the related consolidated balance sheet of FSB and its Subsidiaries as at the end of such fiscal year, setting forth in each case in comparative form the corresponding consolidated figures as of the end of and for the preceding fiscal year, and accompanied by a certificate of a senior financial officer of FSB, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of

Credit Agreement

operations and cash flows of FSB and its Subsidiaries as at the end of, and for, such fiscal year in accordance with GAAP;

(g) as soon as available and in any event within 60 days after the end of each of the first three quarterly fiscal periods of each fiscal year of COBE, consolidated statements of income of COBE and its Subsidiaries for such period and for the period from the beginning of the respective fiscal year to the end of such period, and the related consolidated balance sheet of COBE and its Subsidiaries as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding periods in the preceding fiscal year (except that, in the case of balance sheets, such comparison shall be to the last day of the prior fiscal year), accompanied by a certificate of a senior financial officer of COBE, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations of COBE and its Subsidiaries in accordance with GAAP, as at the end of, and for, such period (subject to normal year-end audit adjustments and absence of footnotes);

(h) as soon as available and in any event within 120 days after the end of each fiscal year of COBE, consolidated statements of income, changes in stockholders' equity and cash flows of COBE and its Subsidiaries for such fiscal year and the related consolidated balance sheet of COBE and its Subsidiaries as at the end of such fiscal year, setting forth in each case in comparative form the corresponding consolidated figures as of the end of and for the preceding fiscal year, accompanied by a certificate of a senior financial officer of COBE, which certificate shall state that said financial statements present fairly, in all material respects, the consolidated financial condition and results of operations and cash flows of COBE and its Subsidiaries as at the end of, and for, such fiscal year in accordance with GAAP;

(i) as soon as available and in any event within 60 days after the end of each quarterly fiscal period of each fiscal year of COB, the "Consolidated Reports of Condition and Income" for COB and its Insured Subsidiaries, all prepared in accordance with regulatory accounting principles prescribed by the Federal Financial Institutions Examination Counsel;

(j) as soon as available and in any event within 60 days after the end of each quarterly fiscal period of each fiscal year of FSB, the "Consolidated Reports of Condition and Income" for FSB and its Insured Subsidiaries, all prepared in accordance with regulatory accounting principles prescribed by the Federal Financial Institutions Examination Counsel;

(k) promptly, notice of the filing of all registration statements (excluding exhibits to such registration statements, and other than registration statements filed on Form S-8 or any successor form) and regular periodic reports filed on Form 10-K, Form 10-Q or Form 8-K (or any successor form), if any, that any Borrower shall have filed with the SEC or any national securities exchange;

57

(l) promptly upon the mailing thereof to the shareholders of COFC generally, copies of all financial statements, reports and proxy statements so mailed, to the extent not otherwise made available;

(m) as soon as possible, and in any event within ten days after any Borrower knows that any of the events or conditions specified below with respect to any Plan or Multiemployer Plan has occurred or exists, a statement signed by a senior financial officer of such Borrower setting forth details respecting such event or condition and the action, if any, that such Borrower or its ERISA Affiliate proposes to take with respect thereto (and a copy of any report or notice required to be filed with or given to the PBGC by such Borrower or an ERISA Affiliate with respect to such event or condition, except that a copy of any notice required to be filed for an event described in subparagraph (i) below may be provided at a later date (to be no later than the date such notice is filed) if it has not been filed as of the date of the signed statement described above):

(i) any reportable event, as defined in Section 4043(c) of ERISA and the regulations issued thereunder, with respect to a Plan, as to which the requirement to provide 30 days' notice to the PBGC under Section 4043(a) or Section 4043(b) of ERISA applies, other than a reportable event for which the requirement to provide such notice has been waived by regulation or for which the PBGC has announced in Technical Update 95-3 (or any subsequent administrative guideline) that it will not apply a penalty for failure to provide such notice (provided that a failure to meet the minimum funding standard of Section 412 of the Code or Section 302 of ERISA, including, without limitation, the failure to make on or before its due date a required installment under Section 412(m) of the Code or Section 302(e) of ERISA, shall be a reportable event regardless of the issuance of any waivers in accordance with Section 412(d) of the Code); and any request for a waiver under Section 412(d) of the Code for any Plan;

(ii) the distribution under Section 4041(c) of ERISA of a notice of intent to terminate any Plan or any action taken by any Borrower or an ERISA Affiliate to terminate any Plan under Section 4041(c) of ERISA;

(iii) the institution by the PBGC of proceedings under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by any Borrower or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan;

(iv) the complete or partial withdrawal from a Multiemployer Plan by any Borrower or any ERISA Affiliate that results in liability under Section 4201

Credit Agreement

or 4204 of ERISA (including the obligation to satisfy secondary liability as a result of a purchaser default) or the receipt by any Borrower or any ERISA Affiliate of notice from a Multiemployer Plan that it is in reorganization or insolvency pursuant to Section 4241 or 4245 of ERISA or that it intends to terminate or has terminated under Section 4041A of ERISA;

(v) the institution of a proceeding by a fiduciary of any Multiemployer Plan against any Borrower or any ERISA Affiliate to enforce Section 515 of ERISA, which proceeding is not dismissed within 30 days; and

(vi) the adoption of an amendment to any Plan that, pursuant to Section 401(a)(29) of the Code or Section 307 of ERISA, would result in the loss of tax-exempt status of the trust of which such Plan is a part if any Borrower or an ERISA Affiliate fails to timely provide security to such Plan in accordance with the provisions of said Sections;

(n) within five days after any executive officer of any Borrower obtains knowledge of the occurrence of any Default, if such Default is continuing, a notice of such Default describing the same in reasonable detail and, together with such notice or as soon thereafter as possible, a description of the action that the Borrowers have taken or propose to take with respect thereto;

(o) promptly after any executive officer of any Borrower knows that a change in the Debt Rating assigned by any Rating Agency has occurred, a notice describing the same;

(p) at the time any set of financial statements is furnished pursuant to paragraph (a), (b), (c), (d), (e) or (f) above, a certificate of a senior financial officer of the relevant Borrower (i) to the effect that no Default has occurred and is continuing (or, if any Default has occurred and is continuing, describing the same in reasonable detail and describing the action that the Borrowers have taken or propose to take with respect thereto) and (ii) setting forth in reasonable detail (including, without limitation, as to the component parts of relevant definitions of accounting terms included in Section 1.01 hereof) the computations necessary to determine whether such Borrower is in compliance with its obligations under Sections 8.07 and 8.08 hereof as of the end of the respective quarterly fiscal period or fiscal year; and

(q) from time to time such other information regarding the financial condition, operations or business of COFC or any of its Subsidiaries as any Lender or the Administrative Agent may reasonably request.

8.02 <u>Litigation</u>. Each Borrower will promptly give to the Administrative Agent (which shall forward the same to each Lender) notice of all legal or arbitral proceedings, and of all investigations or proceedings by or before any governmental or regulatory authority or agency, and any material development in respect of such legal or other proceedings, against or

affecting such Borrower or any of its Subsidiaries, except investigations or proceedings (a) as to which there is no reasonable possibility of an adverse determination or (b) that could not reasonably be expected to (either individually or in the aggregate) have a Material Adverse Effect.

8.03 Existence, Etc. Each Borrower will, and will cause each of its Subsidiaries to:

(a) (i) preserve and maintain its legal existence and (ii) preserve and maintain all of its rights, privileges, licenses and franchises necessary or desirable (in the relevant Borrower's judgment) in the normal conduct of its business except, in the case of this clause (ii), to the extent that failure to preserve and maintain the same could not reasonably be expected to have a Material Adverse Effect (provided that nothing in this Section 8.03 shall prohibit any transaction expressly permitted under Section 8.05 hereof);

(b) comply with the requirements of all applicable laws, rules, regulations and orders of governmental or regulatory authorities (including, without limitation, ERISA, all Environmental Laws and the FDIA and all rules and regulations promulgated thereunder) if failure to comply with such requirements could (either individually or in the aggregate) reasonably be expected to have a Material Adverse Effect;

(c) pay and discharge all taxes, assessments and governmental charges or levies imposed on it or on its income or profits or on any of its Property prior to the date on which penalties attach thereto, except for any such tax, assessment, charge or levy the payment of which is being contested in good faith and by proper proceedings and against which reserves are being maintained in accordance with GAAP;

(d) maintain all of its Properties used or useful in its business in good working order and condition ordinary wear and tear excepted, except to the extent that the failure to maintain any such Property in good working order and condition could not (either individually or in the aggregate) reasonably be expected to have a Material Adverse Effect and would not interfere in any material respect in the ordinary conduct of its business or operations;

(e) keep records and books of account, in which complete entries will be made in accordance with GAAP; and

(f) permit representatives of any Lender or the Administrative Agent, during normal business hours and following reasonable advance notice, to examine, copy and make extracts from its books and records, to inspect any of its Properties, and to discuss its business and affairs with its officers, all to the extent reasonably requested by such Lender or the Administrative Agent (as the case may be); <u>provided</u> that no Borrower shall be required to provide (i) the names of, or other information that could be used to identify, account holders, (ii) any proprietary strategic insights or statistical models

Credit Agreement

concerning account holders or potential account holders or other similar or related proprietary information, (iii) information regarding the specific nature or application of any of the information-based strategies employed by COFC and its Subsidiaries in the conduct of their business or (iv) any proprietary plans or other proprietary information relating to the development of the business of COFC and its Subsidiaries; <u>provided</u>, <u>further</u>, that no advance notice shall be required if an Event of Default has occurred and is continuing and Loans are outstanding.

8.04 <u>Insurance</u>. Each Borrower will, and will cause each of its Subsidiaries to, maintain (either in its own name or in the name of a Borrower) with financially sound and responsible insurance companies, insurance on all its respective properties in at least such amounts and against at least such risks (and with such risk retention) as are usually insured against in the same general area by companies of established repute engaged in the same or a similar business; and will furnish to the Lenders, upon request from the Administrative Agent, information presented in reasonable detail as to the insurance so carried.

8.05 <u>Prohibition of Fundamental Changes</u>. No Borrower will, nor will it permit any of its Subsidiaries to: (a) enter into any transaction of merger or consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution); or (b) convey, sell, lease, transfer or otherwise dispose of, in one transaction or a series of transactions (a "<u>Transfer</u>"), all or substantially all of its business or Property; <u>provided</u> that:

(i) any Subsidiary of COB may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, (x) COB if COB is the continuing, surviving or transferee corporation or (y) any other Subsidiary of COB;

(ii) any Subsidiary of FSB may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, (x) FSB if FSB is the continuing, surviving or transferee corporation or (y) any other Subsidiary of FSB;

(iii) any Subsidiary of COBE may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, (x) COBE if COBE is the continuing, surviving or transferee corporation or (y) any other Subsidiary of COBE;

(iv) the restriction set forth in clause (b) above shall apply, in the case of COB, only to a Transfer of Managed Receivables;

(v) any Subsidiary of COFC (other than COB, FSB or any of their respective Subsidiaries) may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, (x) COFC if COFC is the continuing, surviving or transferee corporation or (y) any other Subsidiary of COFC;

Credit Agreement

(vi) COFC or any of its Subsidiaries (other than COB) may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, COB; or COFC or any of its Subsidiaries (other than FSB) may be merged or consolidated with or into, or Transfer all or substantially all of its business or Property to, FSB;

(vii) any Subsidiary of COFC (other than COB) may merge or consolidate with or into, or Transfer all or substantially all of its business or Property (including, without limitation, interests in Subsidiaries) to, any Person (other than COFC or any of its Subsidiaries) so long as (x) the continuing, surviving or transferee corporation is a Subsidiary of COFC and (y) no Event of Default has occurred and is continuing immediately prior to such merger, consolidation or Transfer or would result therefrom; and

(viii) nothing in this Section 8.05 shall prohibit COFC or any of its Subsidiaries from (A) the sale of credit card loans and other finance receivables pursuant to securitizations (whether or not such securitization received off-balance sheet treatment for the entity effecting such securitization) or (B) the transfer of receivables in the ordinary course of its business.

8.06 <u>Limitation on Liens</u>. No Borrower will, nor will it permit any of its Subsidiaries to, create, incur, assume or suffer to exist any Lien upon any (1) Receivables of any Borrower or (2) Restricted Shares owned by it, in each case whether now owned or hereafter acquired, except:

(a) Liens for taxes not yet due or Liens for taxes being contested in good faith by appropriate proceedings for which adequate reserves (in the good faith judgment of the management of the relevant Borrower) have been established;

(b) Liens imposed by law (i) which are incurred in the ordinary course of business and (x) which do not in the aggregate materially detract from the value of such Receivables or Restricted Shares or materially impair the use thereof in the operation of the business of COFC or any of its Subsidiaries or (y) which are being contested in good faith by appropriate proceedings, which proceedings have the effect of preventing the forfeiture or sale of the Receivables or Restricted Shares subject to such Lien or (ii) which do not relate to material liabilities of COFC and its Subsidiaries and do not in the aggregate materially detract from the value of the Receivables or Restricted Shares of COFC and its Subsidiaries taken as a whole; <u>provided</u> that no Lien permitted under this clause (b) may secure any obligation in an amount exceeding \$50,000,000 and all Liens permitted under this clause (b) may not secure obligations in an aggregate amount exceeding \$75,000,000; and

(c) any pledge of Receivables to a Federal Reserve Bank made in the ordinary course of business to secure advances or other transactions and manage the liquidity position of such Borrower.

Credit Agreement

8.07 Financial Covenants.

(a) COFC will not permit its Delinquency Ratio as of the last day of any calendar month to exceed 6.0%.

(b) COFC will not permit its Tier 1 Capital to Managed Receivables Ratio as of the last day of any fiscal quarter to be less than 4.0%.

(c) COFC will not permit its Tangible Net Worth as of any date of determination to be less than the sum of (i) 75% of its Tangible Net Worth as of March 31, 2004 <u>plus</u> (ii) 60% of COFC Cumulative Net Income as of the last day of the fiscal quarter of COFC most recently ended after such date <u>plus</u> (iii) 60% of COFC Cumulative Equity Proceeds as of such date of determination.

(d) COFC will not permit the Double Leverage Ratio as of the last day of any fiscal quarter to exceed 1.25 to 1.

(e) Neither COB nor FSB will permit its Tier 1 Leverage Ratio as of the last day of any fiscal quarter of COB or FSB, as the case may be, to be less than 5.0%.

(f) Neither COB nor FSB will permit the Tier 1 Capital to Risk Adjusted Assets Ratio as of the last day of any fiscal quarter of COB or FSB, as the case may be, to be less than 6.0%.

(g) Neither COB nor FSB will permit its Total Capital to Risk Adjusted Assets Ratio as of the last day of any fiscal quarter of COB or FSB, as the case may be, to be less than 10.0%.

(h) COBE will not permit its Tangible Net Worth, converted to Dollars at the exchange rate between the Dollar and pounds sterling as of March 31, 2004, on any date to be less than 75% of its Tangible Net Worth as of March 31, 2004.

8.08 Regulatory Capital.

(a) Each U.S. Borrower will cause each of its Insured Subsidiaries to be (and each of COB and FSB so long as it is an Insured Subsidiary will be) at all times "well capitalized" for purposes of 12 U.S.C. §18310, as amended, re-enacted or redesignated from time to time, and at all times to maintain (and each of COB and FSB so long as it is an Insured Subsidiary will maintain) such amount of capital as may be prescribed from time to time, whether by regulation, agreement or order, by each Bank Regulatory Authority having jurisdiction over such Insured Subsidiary.

(b) COFC shall, and shall insure that each of its Insured Subsidiaries, at all times maintain compliance with any rules, regulations, orders or guidelines issued by any Bank Regulatory Authority having jurisdiction over such Insured Subsidiary related to subprime lending.

Credit Agreement

(c) COBE will at all times maintain a capital adequacy ratio not lower than the minimum from time to time required by, and otherwise comply with, the capital adequacy regulations of the FSA or any other relevant Bank Regulatory Authority.

8.09 Lines of Business.

(a) COB will not, nor will it permit any of its Subsidiaries to, engage to any extent in any line or lines of business activity other than as permitted by its charter.

(b) FSB will not, nor will it permit any of its Subsidiaries to, engage to any extent in any line or lines of business activity other than as permitted by its charter.

(c) COBE will not, nor will it permit any of its Subsidiaries to, engage to any material extent in any line or lines of business activity other than as permitted by its memorandum and articles of association.

(d) COFC will not, nor will it permit any of its Subsidiaries to, engage to any material extent in any line or lines of business activity other than as permitted by its charter.

8.10 <u>Use of Proceeds</u>. Each Borrower will use the proceeds of the Loans made to such Borrower hereunder for general corporate purposes (in compliance with all applicable legal and regulatory requirements, including, without limitation, Regulations T, U and X and the Securities Act and the Exchange Act and the regulations thereunder); <u>provided</u> that (a) neither the Administrative Agent nor any Lender shall have any responsibility as to the use of any of such proceeds and (b) no Borrower will use the proceeds of the Loans made hereunder to acquire directly or indirectly a majority of the voting stock issued by, or all or substantially all of the assets of, any Person except with the prior written consent of the Board of Directors of such Person or any controlling shareholder of such Person.

Notwithstanding anything in this Section 8 to the contrary, none of COB, FSB or COBE shall have any obligation (a) to cause COFC or any of its Subsidiaries (other than with respect to COB, FSB, COBE and/or any of their respective Subsidiaries) to take or refrain from taking any action or (b) to cause or prevent any event or circumstance from occurring with respect to COFC or any of its Subsidiaries (other than with respect to COB, FSB, COBE and/or any of their respective Subsidiaries).

SECTION 9. Events of Default. If one or more of the following events (herein called "Events of Default") shall occur and be continuing:

(a) Any Borrower shall: (i) default in the payment of any principal of any Loan when due (whether at stated maturity or at mandatory or optional prepayment); or (ii) default in the payment of any interest on any Loan, any fee or any other amount

Credit Agreement

payable by it hereunder when due and such default shall have continued unremedied for five or more Business Days; or

(b) (i) COFC shall default in the payment when due of any principal of or interest on any of its other Indebtedness aggregating \$100,000,000 (or its equivalent in any other currency or currencies) or more, or any event specified in any note, agreement, indenture or other document evidencing or relating to any such Indebtedness shall occur, and such event shall continue after any applicable grace period, if the effect of such event is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, such Indebtedness to become due, or to be prepaid in full (whether by redemption, purchase, offer to purchase or otherwise), prior to its stated maturity; or

(ii) COB, FSB and/or COBE shall default in the payment when due of any principal of or interest on any of their other Indebtedness aggregating \$75,000,000 (or its equivalent in any other currency or currencies) or more, or any event specified in any note, agreement, indenture or other document evidencing or relating to any such Indebtedness shall occur, and such event shall continue after any applicable grace period, if the effect of such event is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, such Indebtedness to become due, or to be prepaid in full (whether by redemption, purchase, offer to purchase or otherwise), prior to its stated maturity; or

(iii) COFC or any of its Material Subsidiaries shall default in the payment or delivery when due (whether upon termination or liquidation or otherwise), under one or more Swap Agreements, of amounts or property required to be paid or delivered having an aggregate fair market value of \$100,000,000 (or its equivalent in any other currency or currencies) or more; or

(c) Any representation, warranty or certification made or deemed made herein (or in any modification or supplement hereto) by any Borrower, or any certificate furnished to any Lender or the Administrative Agent pursuant to the provisions hereof, shall prove to have been false or misleading in any material respect as of the time made, deemed made or furnished; or

(d) Any Borrower shall default in the performance of any of its obligations under any of Sections 8.05, 8.06, 8.07, 8.08, 8.09 and 8.10 hereof, or shall, for 30 or more days, default in the performance of its obligations under Section 8.01(n); or any Borrower shall default in the performance of any of its other obligations in this Agreement and such default shall continue unremedied for a period of 30 or more days after notice thereof to such Borrower by the Administrative Agent or any Lender (through the Administrative Agent); or

Credit Agreement

(e) Any Borrower or any of its Material Subsidiaries shall admit in writing its inability to, or be generally unable to, pay its debts as such debts become due; or

(f) Any Borrower or any of its Material Subsidiaries shall (i) apply for or consent to the appointment of, or the taking of possession by, a receiver, conservator, custodian, trustee, examiner or liquidator of itself or of all or a substantial part of its Property, (ii) make a general assignment for the benefit of its creditors, (iii) commence a voluntary case under the Bankruptcy Code, (iv) file a petition seeking to take advantage of any other law relating to bankruptcy, insolvency, reorganization, liquidation, dissolution, arrangement or winding-up, or composition or readjustment of debts, (v) fail to controvert in a timely and appropriate manner, or acquiesce in writing to, any petition filed against it in an involuntary case under the Bankruptcy Code or (vi) take any corporate action for the purpose of effecting any of the foregoing; or

(g) A proceeding or case shall be commenced, without the application or consent of any Borrower or any of its Material Subsidiaries, in any court of competent jurisdiction, seeking (i) its reorganization, liquidation, dissolution, arrangement or winding-up, or the composition or readjustment of its debts, (ii) the appointment of a receiver, conservator, custodian, trustee, examiner, liquidator or the like of such Borrower or Material Subsidiary or of all or any substantial part of its Property or (iii) similar relief in respect of such Borrower or Material Subsidiary under any law relating to bankruptcy, insolvency, reorganization, winding-up, or composition or adjustment of debts, and such proceeding or case shall continue undismissed, or an order, judgment or decree approving or ordering any of the foregoing shall be entered and continue unstayed and in effect, for a period of 60 or more days; or an order for relief against any Borrower or any of its Material Subsidiaries shall be entered in an involuntary case under the Bankruptcy Code; or

(h) Any Insured Subsidiary shall cease accepting deposits or making commercial loans on the instruction of any Bank Regulatory Authority with authority to give such instruction other than pursuant to an instruction generally applicable to banks organized under the jurisdiction of organization of such Insured Subsidiary; or

(i) Any Insured Subsidiary shall cease to be an insured bank under the FDIA and all rules and regulations promulgated thereunder; or

(j) Any Insured Subsidiary shall be required (whether or not the time allowed by the appropriate Bank Regulatory Authority for the submission of such plan has been established or elapsed) to submit a capital restoration plan of the type referred to in 12 U.S.C. §18310(b)(2)(C), as amended, re-enacted or redesignated from time to time; or

(k) COFC shall Guarantee in writing the capital of any Insured Subsidiary as part of or in connection with any agreement or arrangement with any Bank Regulatory Authority; or

(1) A final judgment or judgments for the payment of money of \$100,000,000 ((i) exclusive of amounts covered by insurance or subject to indemnification by a solvent third party or (ii) its equivalent in any other currency or currencies) or more in the aggregate shall be rendered by one or more courts, administrative tribunals or other bodies having jurisdiction against any Borrower or any of its Material Subsidiaries and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution thereof shall not be procured, within 30 days from the date of entry thereof and the relevant Borrower or Material Subsidiary shall not, within said period of 30 days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal; or

(m) An event or condition specified in Section 8.01(m) hereof shall occur or exist with respect to any Plan or Multiemployer Plan and, as a result of such event or condition, together with all other such events or conditions, any Borrower or any ERISA Affiliate shall incur or shall be reasonably likely to incur a liability (exclusive of liabilities incurred in the ordinary course of business such as premiums) to a Plan, a Multiemployer Plan or the PBGC (or any combination of the foregoing) that, in the determination of the Majority Lenders, would (either individually or in the aggregate) have a Material Adverse Effect; or

(n) The expiration or termination of the Undertaking or the failing or ceasing of the Undertaking to be in full force and effect (in either case other than in accordance with its express terms) prior to the expiration or termination of all Commitments and the irrevocable payment in full of all amounts owing by FSB and COBE under this Agreement; or COB shall disaffirm, disclaim, repudiate or reject, in whole or in part, or challenge the validity of, the Undertaking; or

(o) Subject to the occurrence of any transaction expressly permitted by Section 8.05 hereof, COFC shall at any time fail to own and control, beneficially, directly or indirectly (free and clear of all Liens and other encumbrances), at least 95% of the issued and outstanding shares of capital stock of each class of Voting Securities issued by COB; or COFC shall at any time fail to own and control, beneficially, directly or indirectly (free and clear of all Liens and outstanding shares of capital stock of each class of Voting Securities issued by FSB; or COFC shall at any time fail to own and control, beneficially, directly or indirectly, directly or indirectly, directly or indirectly, directly or indirectly (free and clear of all Liens and other encumbrances), at least 95% of the issued and outstanding shares of capital stock of each class of Voting Securities issued by FSB; or COFC shall at any time fail to own and control, beneficially, directly or indirectly (free and clear of all Liens and other encumbrances), at least 95% of the issued and outstanding shares of capital stock of each class of Voting Securities issued by FSB; or COFC shall at any time fail to own and control, beneficially, directly or indirectly (free and clear of all Liens and other encumbrances), at least 95% of the issued and outstanding shares of capital stock of each class of Voting Securities issued by COBE; or

(p) During any period of 25 consecutive calendar months, a majority of the Board of Directors of COFC shall no longer be composed of individuals (i) who were members of said Board on the first day of such period, (ii) whose election or nomination to said Board was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of said Board or (iii) whose

67

election or nomination to said Board was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of said Board; or

(q) Any person or group of persons (within the meaning of Section 13 or 14 of the Exchange Act, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the SEC under the Exchange Act) of 20% or more of the issued and outstanding shares of voting common stock issued by COFC;

THEREUPON: (1) in the case of an Event of Default other than one referred to in clause (f) or (g) of this Section 9 with respect to any Borrower, (A) upon request of the Majority Lenders, the Administrative Agent will, by notice to the Borrowers, terminate the Commitments and they shall thereupon terminate, and (B) upon request of Lenders holding more than 50% of the aggregate unpaid principal amount of the Loans owing by a Borrower, the Administrative Agent will, by notice to such Borrower declare the principal amount then outstanding of, and the accrued interest on, the Loans and all other amounts payable by such Borrower hereunder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) to be forthwith due and payable, whereupon such amounts shall be immediately due and payable without presentment, demand, protest or other formalities of any kind, all of which are hereby expressly waived by such Borrower; and (2) in the case of the occurrence of an Event of Default referred to in clause (f) or (g) of this Section 9 with respect to any Borrower, the Commitments shall automatically be terminated and the principal amount then outstanding of, and the accrued interest on, the Loans and all other amounts payable by the Borrower; hereunder and under the Notes (including, without limitation, any amounts payable then outstanding of, and the accrued interest on the to any Borrower, the Commitments shall automatically be terminated and the principal amount then outstanding of, and the accrued interest on, the Loans and all other amounts payable by the Borrowers hereunder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) shall automatically become immediately due and payable without presentment, demand, protest or other formalities of any kind, all of which are hereby expressly waived by each Borrower.

Notwithstanding the foregoing, no Event of Default under any of paragraphs (a), (b), (c), (d) or (l) of this Section 9 solely with respect to COFC or any of its Subsidiaries (other than COB, FSB, COBE and/or any of their respective Subsidiaries) shall in and of itself permit the Administrative Agent or the Lenders to declare the principal amount then outstanding of, and the accrued interest on, the Loans owing by COB, FSB or COBE or any other amounts payable by COB, FSB or COBE hereunder or under the Notes to be forthwith due and payable.

SECTION 10. The Administrative Agent.

10.01 <u>Appointment, Powers and Immunities</u>. Each Lender hereby appoints and authorizes the Administrative Agent to act as its agent hereunder with such powers as are specifically delegated to the Administrative Agent by the terms of this Agreement, together with such other powers as are reasonably incidental thereto. The Administrative Agent (which term as used in this sentence and in Section 10.05 and the first sentence of Section 10.06 hereof shall include reference to its affiliates and its own and its affiliates' officers, directors, employees and agents):

(a) shall have no duties or responsibilities except those expressly set forth in this Agreement, and shall not by reason of this Agreement be a trustee for any Lender;

(b) shall not be responsible to the Lenders for any recitals, statements, representations or warranties made by any other Person contained in this Agreement, or in any certificate or other document referred to or provided for in, or received by any of them from any other Person under, this Agreement, or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement, any Note or any other document referred to or provided for herein or for any failure by any Borrower or any other Person to perform any of its obligations hereunder or thereunder;

(c) shall not be required to initiate or conduct any litigation or collection proceedings hereunder; and

(d) shall not be responsible for any action taken or omitted to be taken by it hereunder or under any other document or instrument referred to or provided for herein or in connection herewith, except for its own gross negligence or willful misconduct.

The Administrative Agent may employ agents and attorneys-in-fact and shall not be responsible for the negligence or misconduct of any such agents or attorneysin-fact selected by it in good faith. The Administrative Agent may deem and treat the payee of a Note as the holder thereof for all purposes hereof unless and until a notice of the assignment or transfer thereof shall have been filed with the Administrative Agent, together with the consent of the applicable Borrower to such assignment or transfer (to the extent required by Section 11.06(b) hereof).

10.02 <u>Reliance by Administrative Agent</u>. The Administrative Agent shall be entitled to rely upon any certification, notice or other communication (including, without limitation, any thereof by telephone, telecopy, telegram or cable) reasonably and in good faith believed by it to be genuine and correct and to have been signed or sent by or on behalf of the proper Person or Persons, and upon advice and statements of legal counsel, independent accountants and other experts selected by the Administrative Agent. As to any matters not expressly provided for by this Agreement, the Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder in accordance with instructions given by the Majority Lenders (or such other number of Lenders as may be expressly required hereby), and such instructions of the Majority Lenders and any action taken or failure to act pursuant thereto shall be binding on all of the Lenders.

10.03 <u>Defaults</u>. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of a Default unless the Administrative Agent has received notice from a Lender or a Borrower specifying such Default and stating that such notice is a "Notice of Default". In the event that the Administrative Agent receives such a notice of the occurrence of a Default, the Administrative Agent shall give prompt notice thereof to the Lenders. The Administrative Agent shall (subject to Sections 10.07 and 11.04 hereof) take such action with respect to such Default as shall be directed by the Majority Lenders, <u>provided</u> that,

Credit Agreement

unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default as it shall deem advisable in the best interest of the Lenders except to the extent that this Agreement expressly requires that such action be taken, or not be taken, only with the consent or upon the authorization of the Majority Lenders or all of the Lenders.

10.04 <u>Rights as a Lender</u>. With respect to its Commitment and the Loans made by it, JPMorgan (and any successor acting as Administrative Agent) in its capacity as a Lender hereunder shall have the same rights and powers hereunder as any other Lender and may exercise the same as though it were not acting as the Administrative Agent, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include JPMorgan (and any successor acting as Administrative Agent) in its individual capacity. JPMorgan (and any successor acting as Administrative Agent) and its affiliates may (without having to account therefor to any Lender) accept deposits from, lend money to, make investments in and generally engage in any kind of banking, trust or other business with any Borrower (and any of its Subsidiaries or Affiliates) as if it were not acting as the Administrative Agent, and JPMorgan (and any successor) and its affiliates may accept fees and other consideration from any Borrower for services in connection with this Agreement or otherwise without having to account for the same to the Lenders.

10.05 <u>Indemnification</u>. The Lenders agree to indemnify the Administrative Agent (to the extent not reimbursed under Section 11.03 hereof, but without limiting the obligations of the Borrowers under said Section 11.03), ratably in accordance with their respective Commitments, for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever that may be imposed on, incurred by or asserted against the Administrative Agent (including by any Lender) arising out of or by reason of any investigation in connection with or in any way relating to or arising out of this Agreement or any other documents contemplated by or referred to herein or the transactions contemplated hereby (including, without limitation, the costs and expenses that any Borrower is obligated to pay under Section 11.03 hereof, but excluding, unless a Default has occurred and is continuing, normal administrative costs and expenses incident to the performance of its agency duties hereunder) or the enforcement of any of the terms hereof or of any such other documents, <u>provided</u> that no Lender shall be liable for any of the foregoing to the extent they arise from the gross negligence or willful misconduct of the Administrative Agent.

10.06 <u>Non-Reliance on Administrative Agent and Other Lenders</u>. Each Lender agrees that it has, independently and without reliance on the Administrative Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own credit analysis of each Borrower and its Subsidiaries and decision to enter into this Agreement and that it will, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own analysis and decisions in taking or not taking action under this Agreement. The Administrative Agent shall not be required to keep itself informed as to the

performance or observance by any Borrower of this Agreement or any other document referred to or provided for herein or to inspect the Properties or books of any Borrower or any of its Subsidiaries. Except for notices, reports and other documents and information expressly required to be furnished to the Lenders by the Administrative Agent hereunder, the Administrative Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the affairs, financial condition or business of any Borrower or any of its Subsidiaries (or any of their Affiliates) that may come into the possession of the Administrative Agent or any of its Affiliates.

10.07 <u>Failure to Act</u>. Except for action expressly required of the Administrative Agent hereunder, the Administrative Agent shall in all cases be fully justified in failing or refusing to act hereunder unless it shall receive further assurances to its satisfaction from the Lenders of their indemnification obligations under Section 10.05 hereof against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action.

10.08 <u>Resignation or Removal of Administrative Agent</u>. Subject to the appointment and acceptance of a successor Administrative Agent as provided below, the Administrative Agent may resign at any time by giving notice thereof to the Lenders and the Borrowers, and the Administrative Agent may be removed at any time with or without cause by the Majority Lenders. Upon any such resignation or removal, the Majority Lenders (in consultation with the Borrowers) shall have the right to appoint a successor Administrative Agent. If no successor Administrative Agent shall have been so appointed by the Majority Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent's giving of notice of resignation or the Majority Lenders' removal of the retiring Administrative Agent, then the retiring Administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent, that shall be a bank with a combined capital and surplus of at least \$500,000,000 that has an office in New York, New York. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent, the provisions of this Section 10 shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as the Administrative Agent.

10.09 <u>Co-Agents; Etc</u>. Neither the Book Manager and Lead Arranger nor any Syndication Agent named on the cover page hereof shall have any obligations under this Agreement except (a) in its capacity as a "Lender" hereunder and (b) if and so long as such Person is the "Administrative Agent" hereunder, in its capacity as Administrative Agent hereunder.

Credit Agreement

SECTION 11. Miscellaneous.

11.01 <u>Waiver</u>. No failure on the part of the Administrative Agent or any Lender to exercise and no delay in exercising, and no course of dealing with respect to, any right, power or privilege under this Agreement or any Note shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege under this Agreement or any Note preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The remedies provided herein are cumulative and not exclusive of any remedies provided by law.

11.02 <u>Notices</u>. (a) Except in the case of notices and other communications expressly permitted to be given by telephone (and subject to paragraph (b) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopy, as follows:

(i) if to any Borrower, to it at 1680 Capital One Dr., McLean, VA 22102-2980, Attention of the Director of Capital Markets (Telephone No. 703-720-1000, Facsimile No. 703-720-3169);

(ii) if to the Administrative Agent, to JPMorgan Chase Bank, Loan & Agency Services, 1111 Fanin, 10th Floor, Houston, TX 77002, Attention of Mr. Jeremy M. Jones (Facsimile No. 713-750-2223), with a copy to JPMorgan Chase Bank, 270 Park Avenue, New York, New York 10017, Attention of Financial Institutions Corporate Banking (Telephone No. 212-270-6261, Facsimile No. 212-270-0670) or, if in respect of any Loan made in any Alternative Currency, to J.P. Morgan Europe Limited, 125 London Wall, 9th Floor, London EC2Y5AJ, UK, Attention of Nichola Hall, (Facsimile No. +44-207-777-2360); and

(iii) if to any Lender, to it at the address(es) (or telecopy number(s)) set forth in its Administrative Questionnaire.

(b) Notices and other communications to the Lenders hereunder (including all documents delivered pursuant to Section 8.01 hereof, with the exception of documents delivered pursuant to Section 8.01(n)) may be delivered or furnished by electronic communications pursuant to procedures approved by the Administrative Agent; provided, that the foregoing shall not apply to notices pursuant to Section 2 unless otherwise agreed by the Administrative Agent and the applicable Lender. The Administrative Agent or any Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; provided, that approval of such procedures may be limited to particular notices or communications.

(c) Any party hereto may change its address or telecopy number for notices and other communications hereunder by notice to the other parties hereto (or, in the case of any such change by a Lender, by notice to the Borrower and the Administrative Agent). All notices and other communications given to any party hereto in accordance with the provisions of this Agreement shall be deemed to have been given on the date of receipt.

Credit Agreement

11.03 Expenses, Etc. Each Borrower agrees to pay or reimburse each of the Lenders and the Administrative Agent for: (a) all reasonable out-of-pocket costs and expenses of the Administrative Agent (including, without limitation, the reasonable fees and expenses of Milbank, Tweed, Hadley & McCloy LLP, special New York counsel to JPMorgan) in connection with (i) the negotiation, preparation, execution and delivery of this Agreement and the making of the Loans hereunder and (ii) the negotiation or preparation of any modification, supplement or waiver of any of the terms of this Agreement or any of the other Basic Documents requested by any Borrower (whether or not consummated); (b) all out-of-pocket costs and expenses of the Lenders and the Administrative Agent (including, without limitation, the fees and expenses of legal counsel) in connection with (i) any Default and any enforcement or collection proceedings resulting therefrom, including, without limitation, all manner of participation in or other involvement with (x) bankruptcy, insolvency, receivership, foreclosure, winding up or liquidation proceedings, (y) judicial or regulatory proceedings and (z) workout, restructuring or other negotiations or proceedings (whether or not the workout, restructuring or transaction contemplated thereby is consummated) and (ii) the enforcement of this Section 11.03; and (c) all transfer, stamp, documentary or other similar taxes, assessments or charges levied by any governmental or revenue authority in respect of this Agreement or any of the other Basic Documents or any other document referred to herein; provided that COB shall have no such payment or reimbursement obligation in connection with Loans made to COFC.

Each Borrower hereby agrees to indemnify the Administrative Agent and the Lenders and their affiliates and the respective directors, officers, employees, attorneys and agents thereof from, and hold each of them harmless against, any and all losses, liabilities, claims, damages or expenses incurred by any of them (including, without limitation, any and all losses, liabilities, claims, damages or expenses incurred by the Administrative Agent to any Lender) arising out of or by reason of any investigation or litigation or other proceedings (including any threatened investigation or litigation or other proceedings, and whether or not the Administrative Agent or any Lender is a party to such litigation or other proceedings) relating to this Agreement or the Loans hereunder or any actual or proposed use by any Borrower or any of its Subsidiaries of the proceeds of any of the Loans hereunder, including, without limitation, the reasonable fees and disbursements of counsel incurred in connection with any such investigation or litigation or other proceedings (but excluding any such losses, liabilities, claims, damages or expenses incurred by reason of the gross negligence or willful misconduct of the Person to be indemnified); provided that COB shall have no liability under the foregoing indemnity in connection with events or circumstances relating solely to COFC or any of its Subsidiaries (other than COB or any of its Subsidiaries). No party shall have any liability to any other party for any indirect, consequential or punitive damages in connection with any matter relating hereto.

11.04 <u>Amendments, Etc</u>. Except as otherwise expressly provided in this Agreement, any provision of this Agreement may be modified or supplemented only by an instrument in writing signed by the Borrowers and the Majority Lenders, or by the Borrowers and the Administrative Agent acting with the consent of the Majority Lenders, and any provision

Credit Agreement

of this Agreement may be waived only by an instrument in writing signed by the Majority Lenders or by the Administrative Agent acting with the consent of the Majority Lenders; <u>provided</u> that: (a) no modification, supplement or waiver shall, unless by an instrument signed by all of the Lenders or by the Administrative Agent acting with the consent of all of the Lenders: (i) increase or extend the term of the Commitments, or extend the time or waive any requirement for the reduction or termination of the Commitments, (ii) extend the date fixed for the payment of principal of or interest on any Loan or any fee payable hereunder, (iii) reduce the amount of any such payment of principal, (iv) reduce the rate at which interest is payable on such principal or any such fee is payable, (v) alter the obligations of any Borrower to prepay Loans or (vi) alter the percentage set forth in Section 2.01(b) or 2.10(c)(v) hereof; (b) no modification, supplement or waiver shall, unless by an instrument signed by all of the Lenders or by the Administrative Agent acting with the consent of all of the Lenders: (i) alter the manner in which payments or prepayments of principal, interest or other amounts hereunder shall be applied as between the Lenders or as between Syndicated Loans or Money Market Loans, (ii) alter the terms of this Section 11.04 or Section 2.11, 4.02, 4.07 or 10.09 hereof, (iii) modify the definition of the term "Majority Lenders" or modify any provision hereof, or (iv) waive any of the conditions precedent set forth in Section 6.01 hereof; and (c) any modification or supplement of Section 11.04 and Section 11.06(c) hereof, no modification, supplement or waiver relating to any of Sections 7, 8 and 9 of this Agreement shall be deemed to increase, or extend the term of, the Commitments.

Anything in this Agreement to the contrary notwithstanding, if at a time when the conditions precedent set forth in Section 6 hereof to any Loan hereunder are, in the opinion of the Majority Lenders, satisfied, any Lender shall fail to fulfill its obligations to make such Loan (any such Lender, a "Defaulting Lender") then, for so long as such failure shall continue, the Defaulting Lender shall (unless the Borrowers and the Majority Lenders, determined as if the Defaulting Lender were not a "Lender" hereunder, shall otherwise consent in writing) be deemed for all purposes relating to amendments, modifications, waivers or consents under this Agreement (including, without limitation, under this Section 11.04) to have no Loans or Commitments, shall not be treated as a "Lender" hereunder when performing the computation of Majority Lenders, and shall have no rights under the preceding paragraph of this Section 11.04; provided that any action taken by the other Lenders pursuant to this paragraph with respect to the matters referred to in clause (a) or (b) of the preceding paragraph shall not be effective as against the Defaulting Lender.

11.05 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

11.06 Assignments and Participations.

(a) <u>Assignments Generally</u>. No Borrower may assign any of its rights or obligations hereunder or under the Notes without the prior consent of all of the Lenders and the Administrative Agent (and any attempted assignment or transfer by any Borrower without such consent shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby and, to the extent expressly contemplated hereby, the Affiliates and directors, officers, employees, counsel and agents of each of the Administrative Agent and the Lenders and their Affiliates) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Assignments by Lenders.

(i) Subject to the conditions set forth in paragraph (b)(ii) below, any Lender may assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans at the time owing to it) with the prior written consent (such consent not to be unreasonably withheld (it being understood that it will not be unreasonable for the Borrowers to withhold consent to an assignment to any assignee whose long term debt obligations are then rated below Baa3 by Moody's Investor Service, Inc. or below BBB- by Standard & Poor's Rating Services)) of:

(A) the Borrowers, <u>provided</u> that no consent of the Borrowers shall be required for an assignment to a Lender, an Affiliate of a Lender or, if an Event of Default has occurred and is continuing, any other assignee; and

(B) the Administrative Agent, provided that no consent of the Administrative Agent shall be required for an assignment to a Lender or an Affiliate of a Lender.

(ii) Assignments shall be subject to the following additional conditions:

(A) except in the case of an assignment to a Lender or an Affiliate of a Lender or an assignment of the entire remaining amount of the assigning Lender's Commitment or Loans, the amount of the Commitment or Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$5,000,000 unless each Borrower and the Administrative Agent otherwise consent, provided that no such consent of the Borrowers shall be required if a Default has occurred and is continuing;

(B) each partial assignment with respect to Syndicated Loans shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations with respect to Syndicated Loans under this Agreement;

(C) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and

Credit Agreement

recordation fee of \$3,500, except in the case of an assignment referred to in Section 2.12 or 5.07, in which case the Borrower or the assignee shall pay such fee;

(D) the assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire; and

(E) any assignee or prospective assignee shall execute a confidentiality agreement pursuant to Section 11.12(b) prior to receiving any Confidential Information.

(iii) Subject to acceptance and recording thereof pursuant to paragraph (b)(iv) of this Section, from and after the effective date specified in each Assignment and Assumption the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be subject to its obligations under Section 11.12 for a period of two years following the effective date specified in such Assignment and Assumption and entitled to the benefits of Sections 5.01, 5.05, 5.06 and 11.03). Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this Section 11.06 shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with paragraph (c) of this Section.

(iv) The Administrative Agent, acting for this purpose as an agent of the Borrowers, shall maintain at one of its offices a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitment of, and principal amount of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive absent manifest error, and the Borrowers, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrowers and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

(v) Upon its receipt of a duly completed Assignment and Assumption executed by an assigning Lender and an assignee, the assignee's completed Administrative Questionnaire (unless the assignee shall already be a Lender hereunder), the processing and recordation fee referred to in paragraph (b) of this Section and any written consent to such assignment required by paragraph (b) of this Section, the Administrative Agent shall accept such Assignment and Assumption and record the information contained therein in the Register. No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this paragraph.

(c) <u>Participations</u>. (i) Any Lender may, without the consent of the Borrower or the Administrative Agent, sell participations to one or more banks or other entities (a "<u>Participant</u>") in all or a portion of such Lender's rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans owing to it); <u>provided</u> that (A) such Lender's obligations under this Agreement shall remain unchanged, (B) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (C) the Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement and (D) any Participant or prospective Participant shall execute a confidentiality agreement pursuant to Section 11.12(b) prior to receiving any Confidential Information. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; <u>provided</u> that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in the first proviso to Section 11.04 that affects such Participant; and <u>provided</u>, further, that such sales may not cause any Loan to become a "plan asset" of an employee benefit plan or ther plan subject to ERISA or Section 4975 of the Code under Section 2510.3-101 of the Regulations of the U.S. Department of Labor or otherwise. Subject to paragraph (C)(ii) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 5.01, 5.05 and 5.06 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to paragraph (b) of this Section. To the extent permitted by law, each Participant also shall be entitle

(ii) A Participant shall not be entitled to receive any greater payment under Section 5.01, 5.05 or 5.06 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrowers' prior written consent.

(d) <u>Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including without limitation any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section shall not apply to any such pledge or assignment of a security interest; <u>provided</u> that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto.

11.07 <u>Survival</u>. The obligations of each Borrower under Sections 2.11 (subject to the last sentence of Section 2.11(b) hereof), 5.01, 5.05, 5.06, 11.03 and 11.13 hereof, and the obligations of the Lenders under Section 10.05 hereof, shall survive the repayment of the Loans and the termination of the Commitments and, in the case of any Lender that may assign any

interest in its Commitment or Loans hereunder, shall, in the case of any event or circumstance that occurred prior to the effective date of such assignment, survive the making of such assignment, notwithstanding that such assigning Lender may cease to be a "Lender" hereunder. In addition, each representation and warranty made, or deemed to be made by a notice of any Loan, herein or pursuant hereto shall survive the making of such representation and warranty, and no Lender shall be deemed to have waived, by reason of making any Loan, any Default that may arise by reason of such representation or warranty proving to have been false or misleading, notwithstanding that such Lender or the Administrative Agent may have had notice or knowledge or reason to believe that such representation or warranty was false or misleading at the time such Loan was made.

11.08 <u>Captions</u>. The table of contents and captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.

11.09 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and any of the parties hereto may execute this Agreement by signing any such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by telecopy shall be effective as delivery of a manually executed counterpart of this Agreement.

11.10 <u>Governing Law; Submission to Jurisdiction</u>. This Agreement and the Notes shall be governed by, and construed in accordance with, the law of the State of New York. Each Borrower hereby submits to the nonexclusive jurisdiction of the United States District Court for the Southern District of New York and of the Supreme Court of the State of New York sitting in New York County (including its Appellate Division), and of any other appellate court in the State of New York, for the purposes of all legal proceedings arising out of or relating to this Agreement or the transactions contemplated hereby. Each Borrower hereby irrevocably waives, to the fullest extent permitted by applicable law, any objection that it may now or hereafter have to the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court has been brought in an inconvenient forum.

11.11 <u>Waiver of Jury Trial</u>. EACH OF THE BORROWERS, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE NOTES OR THE TRANSACTIONS CONTEMPLATED HEREBY.

11.12 Treatment of Certain Information; Confidentiality.

(a) Each Borrower acknowledges that from time to time financial advisory, investment banking and other services may be offered or provided to such Borrower or one or

more of its Subsidiaries (in connection with this Agreement or otherwise) by any Lender or by one or more subsidiaries or affiliates of such Lender, and each Borrower hereby authorizes each Lender to share any information delivered to such Lender by such Borrower and its Subsidiaries pursuant to this Agreement, or in connection with the decision of such Lender to enter into this Agreement, to any such subsidiary or affiliate, it being understood that any such subsidiary or affiliate receiving such information shall be bound by the provisions of paragraph (b) below as if it were a Lender hereunder. Such authorization shall survive the repayment of the Loans and the termination of the Commitments.

(b) Each Lender and the Administrative Agent agrees (on behalf of itself and each of its affiliates, directors, officers, employees and representatives) to keep confidential any non-public information supplied to it by any Borrower pursuant to this Agreement, <u>provided</u> that nothing herein shall limit the disclosure of any such information (i) after such information shall have become public (other than through a violation of this Section 11.12), (ii) to the extent required by statute, rule, regulation or judicial process, (iii) to counsel for any of the Lenders or the Administrative Agent, (iv) to bank examiners (or any other regulatory authority having jurisdiction over any Lender or the Administrative Agent), or to auditors or accountants, (v) to the Administrative Agent or any other Lender, (vi) in connection with the enforcement of rights or remedies hereunder, (vii) to a subsidiary or affiliate of such Lender as provided in paragraph (a) above or (viii) to any assignee or participant (or prospective, assignee or participant), so long as such assignee or participant (or prospective assignee or participant), so long as such assignee or participant (or prospective assignee or participant) first executes and delivers to the respective Lender a Confidentiality Agreement substantially in the form of Exhibit G hereto (or executes and delivers to such Lender an acknowledgement to the effect that it is bound by the provisions of this Section 11.12(b), which acknowledgement may be included as part of the respective assignment or participation agreement pursuant to which such assignee or participant acquires an interest in the Loans hereunder); <u>provided, further</u>, that in no event shall any Lender or the Administrative Agent be obligated or required to return any materials furnished by any Borrower. The obligations of any assignee that has executed a Confidentiality Agreement in the form of Exhibit G hereto shall be superseded by this Section 11.12 upon the date upon which such assignee becomes a Lender hereunder pu

11.13 Judgment Currency. This is an international loan transaction in which the specification of Dollars or an Alternative Currency, as the case may be (the "<u>Specified Place</u>"), and any payment in New York City or the country of the Specified Currency, as the case may be (the "<u>Specified Place</u>"), is of the essence, and the Specified Currency shall be the currency of account in all events relating to Loans denominated in the Specified Currency. The payment obligations of the Borrowers under this Agreement and the Notes shall not be discharged by an amount paid in another currency or in another place, whether pursuant to a judgment or otherwise, to the extent that the amount so paid on conversion to the Specified Currency and transfer to the Specified Place under normal banking procedures does not yield the amount of the Specified Currency at the Specified Place due hereunder. If for the purpose of obtaining judgment in any court it is necessary to convert a sum due hereunder in the Specified Currency into another currency (the "<u>Second Currency</u>"), the rate of exchange which shall be

Credit Agreement

applied shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the Specified Currency with the Second Currency on the Business Day next preceding that on which such judgment is rendered. The obligation of each Borrower in respect of any such sum due from it to the Administrative Agent or any Lender hereunder shall, notwithstanding the rate of exchange actually applied in rendering such judgment, be discharged only to the extent that on the Business Day following receipt by the Administrative Agent or such Lender, as the case may be, of any sum adjudged to be due hereunder or under the Notes in the Second Currency to the Administrative Agent or such Lender, as the case may be, may in accordance with normal banking procedures purchase and transfer to the Specified Place the Specified Currency with the amount of the Second Currency so adjudged to be due; and each Borrower hereby, as a separate obligation and notwithstanding any such judgment, agrees to indemnify the Administrative Agent or such Lender, as the case may be, on demand in the Specified Currency, any difference between the sum originally due to the Administrative Agent or such Lender, as the case may be, in the Specified Currency and the amount of the Specified Currency so purchased and transferred.

11.14 <u>USA PATRIOT Act</u>. Each Lender hereby notifies each Borrower that pursuant to the requirements of the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Act</u>"), it is required to obtain, verify and record information that identifies each Borrower, which information includes the name and address of such Borrower and other information that will allow such Lender to identify each Borrower in accordance with the Act.

Credit Agreement

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered as of the day and year first above written.

BORROWERS

CAPITAL ONE FINANCIAL CORPORATION

By /s/ Gary Perlin

Name: Gary Perlin Title: Executive Vice President and Chief Financial Officer

Credit Agreement

CAPITAL ONE BANK

By /s/ Thomas A. Feil

Name: Thomas A. Feil Title: Vice President, Capital Markets and Assistant Treasurer

Credit Agreement

CAPITAL ONE, F.S.B.

By /s/ Stephen P. Theobald

Name: Stephen P. Theobald Title: Vice President and Chief Financial Officer

Credit Agreement

CAPITAL ONE BANK (EUROPE) PLC

By /s/ Stephen Linehan Name: Stephen Linehan Title: Duly Authorized Attorney

Credit Agreement

ADMINISTRATIVE AGENT

JPMORGAN CHASE BANK

By /s/ Christine Herrick Name: Christine Herrick Title: Vice President

LENDER

JPMORGAN CHASE BANK

By /s/ Christine Herrick

Name: Christine Herrick Title: Vice President

Credit Agreement

SCHEDULE 2.01

COMMITMENTS

NAME OF LENDER	COMMITMENT
IDM and an Object Deals	¢ _ (0,000,000
JPMorgan Chase Bank	\$ 60,000,000
Bank of America, N.A.	\$ 45,000,000
Barclays Bank plc	\$ 45,000,000
Citibank, N.A.	\$ 45,000,000
Credit Suisse First Boston	\$ 45,000,000
Deutsche Bank AG New York Branch	\$ 45,000,000
Lehman Commercial Paper Inc.	\$ 45,000,000
Morgan Stanley Bank	\$ 45,000,000
Wachovia Bank, National Association	\$ 45,000,000
Bank of Montreal	\$ 35,000,000
HSBC Bank USA	\$ 35,000,000
Merrill Lynch Bank USA	\$ 35,000,000
The Royal Bank of Scotland plc	\$ 35,000,000
Societe Generale New York	\$ 35,000,000
William Street Commitment Corporation	\$ 35,000,000
ABN AMRO Bank N.V.	\$ 20,000,000
The Bank of New York	\$ 20,000,000
BNP Paribas	\$ 20,000,000
Calyon New York Branch	\$ 20,000,000
Dresdner Bank AG, New York Branch	\$ 20,000,000
Royal Bank of Canada	\$ 20,000,000
TOTAL	\$ 750,000,000

Securities Litigation

Beginning in July 2002, COFC was named as a defendant in twelve putative class action securities cases. All twelve actions were filed in the United States District Court for the Eastern District of Virginia. Each complaint also named as "Individual Defendants" several of COFC's executive officers.

On October 1, 2002, the Court consolidated these twelve cases. Pursuant to the Court's order, Plaintiffs filed an amended complaint on October 17, 2002, which alleged that COFC and the Individual Defendants violated Section 10(b) of the Exchange Act, Rule 10b-5 promulgated thereunder, and Section 20(a) of the Exchange Act. The amended complaint asserted a class period of January 16, 2001, through July 16, 2002, inclusive. The amended complaint alleged generally that, during the asserted class period, COFC misrepresented the adequacy of its capital levels and loan loss allowance relating to higher risk assets. In addition, the amended complaint alleged generally that COFC failed to disclose that it was experiencing serious infrastructure deficiencies and systemic computer problems as a result of its growth.

On December 4, 2002, the Court granted defendants' motion to dismiss plaintiffs' amended complaint with leave to amend. Pursuant to that order, plaintiffs filed a second amended complaint on December 23, 2002, which asserted the same class period and alleged violations of the same statutes and rule. The second amended complaint also added a new Individual Defendant and asserted violations of GAAP. Defendants moved to dismiss the second amended complaint on January 8, 2003, and plaintiffs filed a motion on March 6, 2003, seeking leave to amend their complaint. On April 10, 2003, the Court granted defendants' motion to dismiss plaintiffs' second amended complaint, denied plaintiffs' motion for leave to amend, and dismissed the consolidated action with prejudice. Plaintiffs appealed the Court's order, opinion, and judgment to the United States Court of Appeals for the Fourth Circuit on May 8, 2003, and briefing on the appeal concluded in September 2003. Oral argument was held on February 25, 2004.

COFC believes that it has meritorious defenses with respect to this case and intends to defend the case vigorously. At the present time, management is not in a position to determine whether the resolution of this case will have a material adverse effect on either the consolidated financial position of COFC or COFC's results of operations in any future reporting period.

Schedule 7.03

[Form of Syndicated Note]

PROMISSORY NOTE

\$

, 200____, 200___, 200___, 200___, 200____, 200____, 200____, 200____, 200_

FOR VALUE RECEIVED, [CAPITAL ONE BANK, a bank chartered under the laws of the Commonwealth of Virginia][CAPITAL ONE, F.S.B., a Federal savings bank chartered under the laws of the United States of America] [CAPITAL ONE FINANCIAL CORPORATION, a corporation organized under the laws of the State of Delaware] [CAPITAL ONE BANK (EUROPE) PLC, a corporation organized under the laws of England] (the "<u>Borrower</u>"), hereby promises to pay to the order of ______ (the "<u>Lender</u>"), for account of its respective Applicable Lending Offices provided for by the Credit Agreement referred to below, at the Administrative Agent's Account, the principal sum of ______ Dollars (or such lesser amount as shall equal the aggregate unpaid principal amount of the Dollar Syndicated Loans made by the Lender to the Borrower under the Credit Agreement and, in the case of Syndicated Loans made in an Alternate Currency, such amount as shall equal the aggregate unpaid principal amount of such Loans made by the Lender to the Borrower under the such Loans made by the Lender to the Borrower under durin immediately available funds, on the dates and in the principal amounts provided in the Credit Agreement, and to pay interest on the unpaid principal amount of each such Loan, to such Account, in like money and funds, for the period commencing on the date of such Loan until such Loan shall be paid in full, at the rates per annum and on the dates provided in the Credit Agreement.

The date, amount, Type, Currency, interest rate and duration of Interest Period of each Syndicated Loan made by the Lender to the Borrower, and each payment made on account of the principal thereof, shall be recorded by the Lender on its books and, prior to any transfer of this Note, endorsed by the Lender on the schedule attached hereto or any continuation thereof, <u>provided</u> that the failure of the Lender to make any such recordation (or any error in making any such recordation) or endorsement shall not affect the obligations of the Borrower to make a payment when due of any amount owing under the Credit Agreement or hereunder in respect of the Syndicated Loans made by the Lender.

This Note is one of the Syndicated Notes referred to in the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto (including the Lender) and JPMorgan Chase Bank, as Administrative Agent, and evidences Syndicated Loans made by the Lender thereunder. Terms used but not defined in this Note have the respective meanings assigned to them in the Credit Agreement.

Syndicated Note

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for prepayments of Syndicated Loans upon the terms and conditions specified therein.

Except as permitted by Section 11.06 of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall be governed by, and construed in accordance with, the law of the State of New York.

[CAPITAL ONE BANK]

[CAPITAL ONE, F.S.B.]

[CAPITAL ONE FINANCIAL CORPORATION]

[CAPITAL ONE BANK (EUROPE) PLC]

By ______ Title:

Syndicated Note

SCHEDULE OF LOANS

This Note evidences Loans made under the within-described Credit Agreement to the Borrower, on the dates, in the principal amounts, of the Types, bearing interest at the rates and having Interest Periods of the durations set forth below, subject to the payments and prepayments of principal set forth below:

Principal Amount of Loan (and Currency thereof)	Type of Loan	Interest Rate	Interest Period	Amount Paid or Prepaid	Unpaid Principal Amount	Notation Made By

Syndicated Note

[Form of Money Market Note]

PROMISSORY NOTE

, 200_ New York, New York

FOR VALUE RECEIVED, [CAPITAL ONE BANK, a bank chartered under the laws of the Commonwealth of Virginia][CAPITAL ONE, F.S.B., a Federal savings bank chartered under the laws of the United States of America][CAPITAL ONE FINANCIAL CORPORATION, a corporation organized under the laws of the State of Delaware] [CAPITAL ONE BANK (EUROPE) PLC, a corporation organized under the laws of England] (the "<u>Borrower</u>"), hereby promises to pay to the order of _______ (the "<u>Lender</u>"), for account of its respective Applicable Lending Offices provided for by the Credit Agreement referred to below, at the Administrative Agent's Account, the aggregate unpaid principal amount of the Money Market Loans made by the Lender to the Borrower under the Credit Agreement, in the respective Currencies in which such Loans are denominated and in immediately available funds, on the dates and in the principal amounts provided in the Credit Agreement, and to pay interest on the unpaid principal amount of each such Money Market Loan, to such Account, in like money and funds, for the period commencing on the date of such Money Market Loan until such Money Market Loan shall be paid in full, at the rates per annum and on the dates provided in the Credit Agreement.

The date, amount, Type, Currency, interest rate and maturity date of each Money Market Loan made by the Lender to the Borrower, and each payment made on account of the principal thereof, shall be recorded by the Lender on its books and, prior to any transfer of this Note, endorsed by the Lender on the schedule attached hereto or any continuation thereof, <u>provided</u> that the failure of the Lender to make any such recordation (or any error in making any such recordation) or endorsement shall not affect the obligations of the Borrower to make a payment when due of any amount owing under the Credit Agreement or hereunder in respect of the Money Market Loans made by the Lender.

This Note is one of the Money Market Notes referred to in the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto (including the Lender) and JPMorgan Chase Bank, as Administrative Agent, and evidences Money Market Loans made by the Lender thereunder. Terms used but not defined in this Note have the respective meanings assigned to them in the Credit Agreement.

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for prepayments of Money Market Loans upon the terms and conditions specified therein.

Money Market Note

Except as permitted by Section 11.06 of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall be governed by, and construed in accordance with, the law of the State of New York.

[CAPITAL ONE BANK] [CAPITAL ONE, F.S.B.] [CAPITAL ONE FINANCIAL CORPORATION] [CAPITAL ONE BANK (EUROPE) PLC]

By ______ Title:

Money Market Note

SCHEDULE OF MONEY MARKET LOANS

This Note evidences Money Market Loans made under the within-described Credit Agreement to the Borrower, on the dates, in the principal amounts, and of the Types and Currencies, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and prepayments of principal set forth below:

Principal Amount of Loan	Type and Currency of Loan	Interest Rate	Maturity Date of Loan	Amount Paid or Prepaid	Unpaid Principal Amount	Notation Made By

Money Market Note

June ___, 2004

Each of the Lenders Party to the Credit Agreement Referred to Below

JPMorgan Chase Bank, as Administrative Agent 270 Park Avenue New York, NY 10017

Ladies and Gentlemen:

We have acted as special New York and Virginia counsel to Capital One Financial Corporation ("<u>COFC</u>"), Capital One Bank ("<u>COB</u>"), Capital One, F.S.B. ("<u>FSB</u>") and Capital One Bank (Europe) PLC ("<u>COBE</u>" and, collectively with COFC, COB and FSB, the "<u>Borrowers</u>" and COFC, COB and FSB collectively being the "<u>U.S. Borrowers</u>") in connection with the Credit Agreement (the "<u>Credit Agreement</u>") dated as of June _____ among the Borrowers, the Lenders party thereto and JPMorgan Chase Bank, as Administrative Agent. Capitalized terms not otherwise defined herein shall have the meanings set forth in the Credit Agreement.

As to facts relevant to the opinions expressed herein, we have examined and relied upon the representations and warranties made in (or pursuant to) the Credit Agreement and the certificates and other instruments delivered pursuant thereto. We have assumed that all such representations and warranties are accurate. We have not reviewed the dockets or other records of any governmental or regulatory body.

In connection with the opinions herein, we have assumed the legal capacity of all natural persons, the genuineness of all signatures, the authenticity of all items submitted to us as originals and the conformity with originals of all items submitted to us as copies.

We have assumed for purposes of our opinions hereinafter set forth that the Credit Agreement has been duly authorized, executed and delivered by each of the parties thereto (other than the U.S. Borrowers) and is a binding obligation of each such party (other than the U.S. Borrowers) and that each such party (other than the U.S. Borrowers) is duly organized and validly existing under the jurisdiction in which it is organized and has full power, authority and legal right to execute, deliver and perform the Credit Agreement.

Opinion of Special U.S. Councel to the Borrowers

Based on and subject to the foregoing and the other paragraphs hereof, and such examination of law as we have deemed necessary, we are of the opinion that:

- 1. Each U.S. Borrower has all requisite power and authority to execute and deliver each Basic Document and perform its obligations thereunder and has taken all necessary action to authorize the execution, delivery and performance thereof.
- 2. The Credit Agreement has been duly executed and delivered by each Borrower and constitutes a legal, valid and binding obligation of such Borrower, enforceable in accordance with its terms.
- 3. Neither the execution, delivery and performance by any Borrower of any Basic Document, nor the compliance by any Borrower with the terms and provisions thereof: (a) violates the General Corporation Law of the State of Delaware, any law, statute or regulation of the State of New York or the Commonwealth of Virginia or any federal law of the United States (including Regulations T, U and X of the Board of Governors of the Federal Reserve System) that, in each case, is applicable to such Borrower and which, in our experience, would normally apply to transactions of the type contemplated by the Basic Documents, or (b) any provision of the organizational documents of any U.S. Borrower.
- 4. No consent, approval or authorization of, or filing with, any governmental authority of the State of New York, the Commonwealth of Virginia or the United States that, in each case, is applicable to any Borrower is required for (a) the due execution, delivery and performance by any Borrower of any Basic Document or (b) the validity, binding effect or enforceability of any Basic Document, except (i) in each case as have previously been made or obtained, and (ii) consents, approvals, authorizations or filings as may be required to be obtained or made by any Lender as a result of its involvement in the transactions contemplated by the Credit Agreement.
- 5. A Virginia court should, if properly presented with the issue, give effect to those provisions of the Basic Documents providing that such documents are to be governed by and construed in accordance with the laws of the State of New York insofar as such provisions relate to the substantive law of the State of New York and to the validity, nature, interpretation and effect of the documents, except (i) to the extent, if any, that federal law applies, (ii) to the extent procedural (as opposed to substantive) laws are involved or (iii) to the extent that the applicable laws of the State of New York violate a public policy of Virginia.

The opinions set forth above are subject to the following limitations and qualifications:

(a) Our opinions are limited to the federal law of the United States, the laws of the State of New York, General Corporation Law of the State of Delaware and the laws of the Commonwealth of Virginia, and we do not express any opinion concerning any other law.

(b) Our opinions are subject to (i) the effect of bankruptcy, insolvency, reorganization, conservatorship, receivership, liquidation, fraudulent conveyance, fraudulent transfer, equitable subordination, readjustment of debt, moratorium or other similar laws now or hereafter in effect relating to or affecting the rights or remedies of creditors generally, and (ii) the effect of general principles of equity, including, without limitation, concepts of materiality, reasonableness, good faith and fair dealing (regardless of whether considered in a proceeding in equity or at law) and by limitations on he availability of specific performance, injunctive relief or other equitable remedies.

(c) We express no opinion as to any provision of any Basic Document that (i) provides for a manner of service of process different than that prescribed or permitted by law; (ii) relates to severability or separability; (iii) relates to indemnification or reimbursement obligations to the extent any such provision violates public policy; (iv) purports to require that all amendments, waivers and terminations be in writing

Opinion of Special U.S. Councel to the Borrowers

or to require disregard of any course of dealing between parties; (v) purports to waive the right to object to venue or to assert forum non conveniens; or (vi) purports to grant rights of set-off to participants.

(d) With respect to our opinion in paragraph 4(a) as it relates to Regulations T, U and X of the Board of Governors of the Federal Reserve System, we have assumed that the Borrowers will comply with the provisions of the Credit Agreement relating to the use of proceeds.

(e) We express no opinion as to Section 11.13 of the Credit Agreement.

This opinion is delivered only to you in connection with the Credit Agreement and is solely for your benefit and may not be relied upon or used by, circulated, quoted or referred to, nor may copies be delivered to, any other person, other than your assignees, transferees and prospective assignees and transferees, without our prior written consent. We disclaim any obligation to update this opinion letter for events occurring or coming to our attention after the date hereof.

Very truly yours,

Opinion of Special U.S. Councel to the Borrowers

[FORM OF OPINION OF SPECIAL ENGLISH COUNSEL TO COBE]

To: Each of the Lenders Party to the Loan Agreement referred to below

[•] 2004

JPMorgan Chase Bank (as Administrative Agent) 270 Park Avenue New York, NY 10017

Dear Sirs

We have acted as English legal advisers to Capital One Bank (Europe) PLC ("**COBE**") in connection with the US\$750,000,000 Credit Agreement dated [•] 2004 (the "**Loan Agreement**"), among COBE as a Borrower, the other Borrowers party thereto, the Lenders party thereto and JPMorgan Chase Bank, as Administrative Agent (the Administrative Agent and the Lenders are collectively referred to as the "**Lender Parties**" and individually as the "**Lender Party**") (the "**Transaction**").

1. INTRODUCTION

1.1 Transaction Documents

The opinions given in this letter relate only to the following documents in connection with the Transaction (the "Transaction Documents"):

- 1.1.1 the Loan Agreement (including the form of the notes in Exhibit A-1 and Exhibit A-2 thereof, the "Notes");
- 1.1.2 such other documents as we have deemed necessary or appropriate as a basis for the opinions set forth below.

1.2 Interpretation

- 1.2.1 Terms defined in or given a particular construction in the Loan Agreement shall have the same meaning when used in this letter, unless a contrary indication appears.
- 1.2.2 Headings in this letter are for ease of reference only and shall not affect its interpretation.

Opinion of Special English Councel to COBE

1.3 Legal Review

For the purpose of issuing this letter, we have reviewed only those documents and conducted only those searches and enquiries referred to in Schedule 1.

1.4 Assumptions and Reservations

The opinions given in this letter are given on the basis of the assumptions set out in Schedule 2 and are subject to the reservations set out in Schedule 3 to this letter.

1.5 Scope of Opinions

The opinions given in this letter are only given in connection with the Transaction solely for the purposes of Section 6.01(d) of the Loan Agreement, and are strictly limited to the matters stated in paragraph 2. They may not be relied on for any other purpose and may not be read as extending by implication to any other matters. We express no opinion as to any liability to tax which may arise or be suffered as a result of or in connection with the Transaction.

1.6 Applicable Law

This letter and the opinions given in it are governed by and relate only to English law as applied by the English courts at the date of this letter. For the purpose of this letter, we have made no investigation of the laws of any jurisdiction other than England and Wales and, accordingly, in this letter we express no opinion on the laws of any other jurisdiction.

2. **OPINION**

On the basis of assumptions, reservations and qualifications set out herein, we are of the opinion that:

- 2.1 the choice of New York law as the governing law of the Loan Agreement (and the Notes, if issued as at the date hereof) and the submission of COBE in the Loan Agreement to the non-exclusive jurisdiction of the Courts of the State of New York is a valid and binding selection which will be upheld, recognised and given effect by the Courts of England;
- 2.2 COBE is duly incorporated and registered under the laws of England and Wales;
- 2.3 Searches made on [•] 2004 at the Companies Registry against COBE and an oral enquiry made to the Central Registry of Winding Up Petitions on [•] revealed no petition, order or resolution for the winding up of COBE and no application for or administration order in respect of COBE, and no notice of intention to appoint an administrator in respect of COBE or notice of appointment of a receiver or administrator in respect of COBE;
- 2.4 COBE has all requisite corporate power and authority to enter into and perform its obligations under the Loan Agreement (including, as at the date hereof, the issuing of the Notes if requested by any Lender pursuant to Section 2.08(d) of the Loan Agreement);
- 2.5 The execution and delivery of the Loan Agreement by COBE has been duly authorised by all necessary corporate action on its part and has been duly executed and delivered by it which

Opinion of Special English Councel to COBE

execution and delivery does not and will not result in any violation by it of any term of its Memorandum or Articles of Association;

- 2.6 The obligations of COBE under the Loan Agreement (and the Notes, if requested by any Lender pursuant to Section 2.08(d) of the Loan Agreement as if issued as at the date hereof) constitutes legal, valid, binding and enforceable obligations of COBE.
- 2.7 There are no provisions of the laws of England and Wales which would be contravened by the execution or delivery of the Loan Agreement (and the Notes, if issued as at the date hereof).
- 2.8 No consents, approvals or authorisations of, or filings with, any governmental authority of England and Wales are required by law by COBE in relation to the execution and delivery by COBE of the Loan Agreement (and the Notes, if issued as at the date hereof), other than the registration of correctly completed Forms 395 and original executed copies of any security documents with the Registrar of Companies.

3. RELIANCE AND DISCLOSURE

This letter is delivered to you in connection with the proposed Transaction and is for your sole benefit. It may not be relied on by, and must not be transmitted to, any other person nor are its contents to be quoted or referred to in any document or filed with any applicable authority, governmental agency or body or any other person whatsoever without our prior written consent. We disclaim any obligation to update this opinion letter for events occurring or coming to our attention after the date hereof. Furthermore, this letter is given on the basis that any limitation on the liability of any other adviser to all or any of the persons to whom this letter is addressed, whether or not we are aware of that limitation, will not adversely affect our position in any circumstances.

Yours faithfully

Clifford Chance Limited Liability Partnership

Opinion to Special English Councel to COBE

SCHEDULE 1 DOCUMENTS AND SEARCHES

1. **DOCUMENTS**

We have reviewed only the following documents for the purposes of this letter:

- 1.1 An execution copy dated [•] 2004 of the Loan Agreement;
- 1.2. Certified true copies of the following documents, certified by [•] as at the date hereof, in his capacity as director of COBE:
 - 1.2.1 the Certificate of Incorporation of COBE;
 - 1.2.2 the memorandum and articles of association of COBE;
 - 1.2.3 the minutes of the meeting of the board of directors of COBE held on [•] 2004 to approve the Loan Agreement (the "Board Resolutions"); and
 - 1.2.4 the power of attorney (the "Power of Attorney") executed by COBE dated [•] 2004 authorising the attorneys (as defined therein) to execute and deliver the Loan Agreement on behalf of COBE;
- 1.3 a certificate (the "Solvency Certificate") dated [•] 2004 executed by [•] in his capacity as director of COBE, certifying that the Transferor is solvent as at the date of the certificate; and
- 1.4 a Companies House Direct search under the name of COBE conducted on [•] 2004 showing [•] as director of COBE,

2. EXTENT OF EXAMINATION AND INVESTIGATION

- 2.1 We have made no searches or enquiries concerning, and examined no documents entered into by or affecting COBE or any other person or any corporate records of the aforesaid, save for non-original copies of those documents expressly specified in this opinion as having been examined.
- 2.2 We have not assisted in the investigation or verification of the facts, or the reasonableness of any assumption or statement or opinion (including, without limitation, as to the solvency of any party), contained in any of the Documents or in determining whether any material fact has been omitted therefrom. To the extent that the accuracy of such facts and statements not so investigated or verified and of any facts stated in any other document (or orally confirmed) is relevant to the contents of this opinion, we have with your permission assumed the such facts and statements are correct. Where an assumption is stated to be made in this opinion, we have not made any investigation with respect to the matters that are the subject of such assumption.

Opinion of Special English Councel to COBE

SCHEDULE 2 ASSUMPTIONS

1. ORIGINAL AND GENUINE DOCUMENTATION

- 1.1 All signatures, stamps and seals and all original documents are genuine.
- 1.2 Any document identified in Schedule 1 as a copy is complete and conforms to the original, which itself is genuine.
- 1.3 Any document identified in Schedule 1 as a draft will be executed by all parties to it in the form examined by us.
- 1.4 All translations of any document identified in Schedule 1 are accurate.
- 1.5 Any certificate referred to in Schedule 1 is correct in all respects.
- 1.6 The Notes, if issued, are in the form set out in Exhibit A-1 or Exhibit A-2 of the Loan Agreement.

2. CORPORATE AUTHORITY OF COBE

- 2.1 The truth, accuracy and completeness at all relevant times of each of the statements of matters of fact contained in the Board Resolutions and the Power of Attorney.
- 2.2 No changes have been made to the Memorandum and Articles of Association of COBE from those copies referred to in paragraph 1.2 of Schedule 1.
- 2.3. The Board Resolutions of the directors of COBE were duly passed by duly appointed directors of COBE and none of these resolutions have been rescinded or amended and are in full force and effect.
- 2.4 The directors of COBE have acted in good faith and in the best interests of COBE in approving the Transaction and the execution of the Loan Agreement.
- 2.5 Each director has disclosed any interest which he may have in the Transaction in accordance with the provisions of the Companies Act 1985 and the Articles of Association of COBE and none of the directors of COBE has any interest in the Transaction except to the extent permitted by the Articles of Association.
- 2.6 The information disclosed by our searches on [•] 2004 at the Companies Registry and our oral enquiry at the Central Registry of Winding up Petitions on [•] 2004 in relation to COBE was then accurate and has not since been altered and that such searches did not fail to disclose any information which had been delivered for registration but did not appear from the information available at the time of our searches and that such oral enquiry did not fail to elicit any material information.
- 2.7 On [•] 2004, COBE:

2.7.1 was not insolvent;

Opinion of Special English Counsel to COBE

- 2.7.2 would not be, as a consequence of doing any act or thing which the Loan Agreement contemplates, permits or requires COBE to do, become insolvent; and
- 2.7.3 no insolvency event exists in relation to COBE or of the assets or undertaking of COBE.

In this regard we would refer you to the the Solvency Certificate of COBE.

3. LOAN AGREEMENT ASSUMPTIONS

- 3.1 The representations and warranties of COBE and any other representations and warranties by any of the parties under the Loan Agreement (other than any representations and warranties in relation to any of the matters specifically the subject of the opinions in this letter) are and will be when made or repeated or when deemed made or repeated, as the case may be, true and accurate in all respects and that such representations and warranties were at all relevant times true and accurate.
- 3.2 Any representation or warranty given by any of the parties to the Loan Agreement that it is not aware of or has no notice of any act, matter, thing or circumstance means that the same does not exist or has not occurred.
- 3.3 Each party to the Loan Agreement (other than COBE) has the capacity, power and authority to enter into and to exercise its rights and to perform its obligations under the Loan Agreement.
- 3.4 The Loan Agreement has been duly and validly executed and delivered by each of the parties thereto (other than COBE).
- 3.5 In respect of the Loan Agreement and each of the transactions contemplated by, referred to in, provided for or effected by the Loan Agreement:
 - 3.5.1 the parties to the Loan Agreement entered into the same in good faith and for the purpose of carrying on their respective businesses;
 - 3.5.2 the parties to the Loan Agreement entered into the same on arms' length commercial terms; and
 - 3.5.3 the parties to the Loan Agreement believed and had reasonable grounds for believing that the same would benefit and be in the interests of such party.
- 3.6 None of the transactions contemplated by the Loan Agreement will infringe the terms of, or constitute a default under, any agreement, indenture, contract, mortgage, deed or other instrument to which COBE is a party or by which any of its property, undertaking, assets or revenues are bound.

4. FOREIGN LAW

- 4.1 All deeds, instruments, assignments, agreements and other documents in relation to the matters contemplated by the Loan Agreement and this Opinion are within the capacity and powers of and have been validly authorised, executed and delivered by the parties thereto other than COBE.
- 4.2 All deeds, instruments, assignments, agreements and other documents in relation to the matters contemplated by the Loan Agreement and this Opinion are duly executed, valid and legal

Opinion of Special English Counsel to COBE

obligations binding on and enforceable against the parties thereto (and are not subject to avoidance by any person) under all applicable laws and in all applicable jurisdictions other than the laws of England and all acts, conditions or things required to be fulfilled, performed or effected in connection with the Loan Agreement under the laws of any jurisdiction other than England have been duly fulfilled, performed and effected.

- 4.3 All deeds, instruments, assignments, agreements and other documents in relation to the matters contemplated by the Loan Agreement and this Opinion, insofar as any of such deeds, instruments, assignments, agreements and other documents falls to be performed in any jurisdiction other than England, will not be illegal or ineffective by virtue of the laws of that jurisdiction.
- 4.4 There are no provisions of the laws of any jurisdiction outside England which would be contravened by the execution or delivery of the Loan Agreement, and that none of the opinions expressed herein will be affected by the laws (including the public policy) of any jurisdiction outside England.
- 4.5 The US law opinions dated [•] 2004 and delivered pursuant to Sections 6.01(c) and 6.01(f) of the Loan Agreement is true, complete and correct in all respects.

Opinion of Special English Counsel to COBE

SCHEDULE 3 RESERVATIONS

The opinions given in this letter are subject to the following reservations:

1. LIMITS ON SCOPE OF OPINION

- 1.1 No opinion is given as to the proper execution or delivery of the Loan Agreement by any party thereto (other than COBE).
- 1.2 No opinion is given as to the validity and enforceability of the Loan Agreement or any other document to which COBE is a party.
- 1.3 No opinion is given on any issues which may arise out of or relate to the giving of financial assistance pursuant to section 151 (et seq.) of the Companies Act 1985.

2. SEARCHES

2.1 Search at Companies Registry for insolvency events

The search at the Companies Registry referred to in paragraph 2.3 above is not conclusively capable of revealing whether or not:

- 2.1.1 a winding up order has been made in respect of a company or a resolution passed for the winding up of a company; or
- 2.1.2 an administration order has been made in respect of a company or notice of the appointment of an administrator has been filed with the Court; or

2.1.3 a receiver, administrative receiver, administrator or liquidator has been appointed in respect of a company,

since notice of these matters might not be filed with the Registrar of Companies immediately and, when filed, might not be entered on the public microfiche of the relevant company immediately. In addition, that search is not capable of revealing, prior to the making of the relevant order, whether or not a winding up petition or an application for an administration order has been presented or notice of intention to appoint an administrator under paragraphs 14 or 22 of Schedule B1 to the Insolvency Act has been filed with the Court.

2.2 Search at Central Registry for insolvency events

The enquiry at the Central Registry of Winding Up Petitions referred to in paragraph 2.3 above relates only to a resolution for winding up, a petition for the making of a winding up order or the making of a winding up order by the Court and the presentation, at the High Court of Justice in London, of an application for the making of an administration order or the making by such court of an administration order. It is not capable of revealing conclusively whether or not such a winding up petition or application for an administration order has been presented or winding up or administration order granted as:

2.2.1 details of a winding-up petition may not have been entered on the records of the Central Registry of Winding Up Petitions immediately; or

Opinion of Special English Counsel to COBE 8

- 2.2.2 in the case of a winding up petition presented to a County Court, details of the petition may not have been notified to the Central Registry and entered on such records at all;
- 2.2.3 in the case of application for the making of an administration order and such order, if such application is made to, and an order is made by, a Court other than the High Court in London, no record of such application or order will be kept by the Central Registry;
- 2.2.4 a winding up order or administration order may be made before the relevant petition has been entered on the records of the Central Registry and the making of such order may not have been entered on the records immediately;
- 2.2.5 details of a notice of intention to appoint an administrator under paragraphs 14 and 22 of Schedule B1 to the Insolvency Act may not be entered on the records immediately; and

2.2.6 the response to an enquiry only relates to the period of six months prior to the date when the enquiry was made.

Search at Central Registry: Out of court appointment of an administrator

2.3 An administrator of a company may now be appointed out of court either by the holder of qualifying floating charge or by the company or its directors. Where the appointment of an administrator is being effected by a company itself (as opposed to its directors), there is no obligation for notice of intention to appoint an administrator to be filed with the Court unless there exists a qualifying floating charge holder who is or may be entitled to appoint an administrator to the company, a person who is or may be entitled to appoint an administrative receiver, any sheriff who is charged with execution or other legal process against the company or a person who has distrained against the company (paragraph 26 of Schedule B1 and Rule 2.20 of The Insolvency (Amendment) Rules 2003). Furthermore, where the appointment is being effected by the holder of a qualifying floating charge, the appointor is only obliged to file a notice of intention to appoint an administrator with the Court where there exists a creditor who holds a prior qualifying floating charge (Rule 2.15(2) of The Insolvency (Amendment) Rules 2003). The Court may not, therefore, receive a notice of intention to appoint an administrator before an appointment of an administrator is made and a search at the Central Registry will not reveal that such an appointment is pending. We would note, however, that under paragraph 15 of Schedule B1, where a qualifying floating charge holder proposes to appoint an administrator, notice must be served on the holder of a prior qualifying floating charge and, under paragraph 26 of Schedule B1, where the company or its directors proposes to appoint an administrator, notice must be served on a person who is or may be able to appoint an administrative receiver and a qualifying floating charge holder who is or may be entitled to appoint an administrator of the company respectively.

No enquiries of County Courts

2.4 We have not made enquiries of any County Court as to whether an application for the appointment of an administrator has been presented to, or a notice of intention to appoint an administrator has been filed with, or an administration order has been made by, such County Court against COBE.

Opinion of Special English Counsel to COBE

3. BORROWING LIMITS

3.1 We express no opinion with regard to whether COBE is in compliance with any borrowing limits in any instrument by which it may be bound as to which we have no knowledge, save that no such limitation is contained in the Memorandum and Articles of Association.

4. CHOICE OF LAW

4.1 The choice of New York law to govern the Loan Agreement would not be recognised or upheld if there were reasons for avoiding the choice of law on the grounds that its application would be contrary to public policy.

5. **DEFAULT INTEREST**

Penalties unenforceable

5.1 Any provision of the Loan Agreement providing for the payment of additional moneys consequent on the breach of any provision thereof by any person expressed to be a party thereto, whether expressed to be by way of penalty, additional interest, liquidated damages or otherwise would be unenforceable if such provision were held to constitute a penalty and not a genuine and reasonable pre-estimate of the loss likely to be suffered as a result of the breach in question.

Interest increases commercially justifiable not a penalty

5.2 We believe that any provision for the payment of a modest increase in interest upon the happening of a default would not constitute a penalty provided the increase could in the circumstances be explained as commercially justifiable and that its dominant purpose was not to deter the paying party from breach and that it did not take effect retrospectively (see **Lordsvale Finance plc v. Bank of Zambia**, [1996] QB 752).

No opinion on whether default interest provisions in Transaction Documents are penalties

5.3 We express no opinion, however, as to whether any such provision in the Loan Agreement does in fact comply with the criteria set out in paragraph 7.2 (*Interest increases commercially justifiable not a penalty*).

The Late Payments of Commercial Debts (Interest) Act (1998)

5.4 If the Loan Agreement [or the Notes] do not provide a contractual remedy for the late payment of any amount payable thereunder that is a substantial remedy within the meaning of the Late Payment of Commercial Debts (Interest) Act 1998 (the "**Interest Act**"), the person entitled to that amount may have a right to statutory interest (and to payment of certain fixed sums) in respect of that late payment at the rate (and in the amount) from time to time prescribed by the Secretary of State with the consent of the Treasury. Any term of the Loan Agreement [or the Notes] may be void to the extent that it excludes or varies that right to statutory interest, or purports to confer a contractual right to interest that is not a substantial remedy for late payment of that amount, within the meaning of the Interest Act.

Opinion of Special English Counsel to COBE

6. MEANING OF "ENFORCEABLE"

6.1 No implication of enforceability

This Opinion is not to be taken to imply that any obligation would necessarily be capable of enforcement in all circumstances in accordance with its terms.

6.2 Limitations on meaning of enforceable

The term "enforceable", as used in this Opinion, means that the obligations assumed by COBE under the Loan Agreement are of the type which a Court enforces. In particular:

- 6.2.1 enforcement of obligations of a party to be performed after the date hereof may be limited by bankruptcy, insolvency, liquidation, administration, moratorium, reorganisation and other laws of general application relating to or affecting the rights of creditors as such law may be applied in the event of an insolvency event with respect to such party;
- 6.2.2 a Court will not necessarily grant any remedy, the availability of which is subject to equitable considerations or which is otherwise in the discretion of the Court; in particular, orders for specific performance and injunctions are, in general, discretionary remedies under English law and an order for specific performance is not available where damages are considered by the Court to be an adequate alternative remedy;
- 6.2.3 any obligation imposed on a secured creditor to hold certain moneys receivable by it on trust pursuant to the Loan Agreement may constitute a charge which may require to be registered in accordance with the Companies Act 1985 in order to be effective; no such charge is intended to be so registered;
- 6.2.4 where any party to the Loan Agreement is vested with a discretion or may determine a matter in its opinion, that party may be required to exercise its discretion in good faith, reasonably and for a proper purpose, and to form its opinion in good faith and on reasonable grounds;
- 6.2.5 claims may become barred under the Limitation Act 1980 or the Foreign Limitation Periods Act 1984 or may be or become subject to the defence of set-off or to counterclaim;
- 6.2.6 where obligations are to be performed in a jurisdiction outside England, they may not be enforceable in England to the extent that performance would be illegal or contrary to public policy under the laws, or contrary to the exchange control regulations, of the other jurisdiction and an English court may take into account the law of the place of performance in relation to the manner of performance and the steps to be taken in the event of defective performance;
- 6.2.7 it is uncertain whether the parties can agree in advance the governing law of claims connected with the contract but which are not claims on the contract, such as a claim in tort;
- 6.2.8 a Court may stay proceedings if concurrent proceedings are being brought elsewhere and may decline to accept jurisdiction in certain cases;

Opinion of Special English Counsel to COBE

- 6.2.9 the enforcement of the obligations of the parties to the Loan Agreement may be limited by the provisions of English law applicable to agreements held to have been frustrated by events happening after their execution;
- 6.2.10 enforcement of obligations may be invalidated by reason of fraud;
- 6.2.11 no opinion is given as to available methods of or procedures for enforcement of security interests under English law;
- 6.2.12 the effectiveness of any provision of the Loan Agreement which allows an invalid provision to be severed in order to save the remainder of its provisions will be determined by the Courts in their discretion; and
- 6.2.13 any provision of the Loan Agreement stating that a failure or delay on the part of any party in exercising any right or remedy shall not operate as a waiver of such right or remedy may not be effective.

7. GENERAL

Determinations not Final

7.1 A determination, designation, calculation or certificate by any party to the Loan Agreement as to any matter provided for therein might, in certain circumstances, be held by a Court not to be final, conclusive and binding (if, for example, it could be shown to have been made arbitrarily or in bad faith) notwithstanding the provisions of the Loan Agreement.

Other Forum Having Competent Jurisdiction

7.2 A court has power to stay an action on application by a party to proceedings if (a) the party applying for the stay establishes that there is some other forum, having competent jurisdiction, which is more appropriate for the trial of the action (that is in which the case can be tried more suitably for the interests of all the parties and the ends of justice), and (b) staying the action would not be inconsistent with the EC Regulation 44/2001 on Jurisdiction and the Recognition and Enforcement of Judgments in Civil and Commercial Matters (which has force of law in the Asset Jurisdiction under the European Communities Act 1982) or with the Brussels (EC) Convention on Jurisdiction and the Enforcement of Judgments in Civil and Commercial Matters of 1968 (as amended) or the Lugano Convention on Jurisdiction and the Enforcement of Judgments in Civil and Conventions have the force of law in the Asset Jurisdiction under the Civil Jurisdiction and Judgments Act 1982 (as amended).

Court judgment could supersede agreements

7.3 There is some possibility that a Court would hold that a judgment on a particular agreement or instrument, whether given in an English Court or elsewhere, would supersede such agreement or instrument to all intents and purposes, so that any obligation thereunder which by its terms would survive such judgment might not be held to do so.

Opinion of Special English Counsel to COBE

Stamp duty undertakings and indemnities

7.4 Any undertaking or indemnity in relation to stamp duties may be void under the provisions of Section 117 of the Stamp Act 1891.

Judgments in non-sterling currencies

7.5 As a result of recent cases and legislation, that in the event of any proceedings being brought in a Court in respect of a monetary obligation expressed to be payable in a currency other than Sterling, a Court would give judgment expressed as an order to pay such currency or its sterling equivalent at the time of payment or enforcement of judgment even though in the past the Courts have considered themselves only able to give a monetary judgment expressed in sterling. However, with respect to an insolvency event, English law may require that all claims or debts are converted into Sterling at an exchange rate determined by the Court at a date related thereto, such as the date the company went into liquidation (that is, the date of the winding up order or the date of the resolution to wind up or where the administrator proposes to make a distribution to creditors pursuant to Chapter 10 part 2 of the Insolvency Rules 1986, the date the company entered into administration).

Partial Invalidity

7.6 In some circumstances a Court will not give effect to any provision of the Loan Agreement which provides that, in the event of any invalidity, illegality or unenforceability of any provision of the Loan Agreement, the remaining provisions thereof shall not be affected or impaired, particularly if to do so would not accord with public policy or would require that the Court make a new contract for the parties.

Enforcement by Court of provisions of Reviewed Initial Primary Transaction Documents

- 7.7 A Court may refuse to give effect to any provision of the Loan Agreement:
- 7.7.1 for the payment of expenses in respect of the costs of enforcement (actual or contemplated) or of unsuccessful litigation brought before a Court or where the Court has itself made an order for costs; or
- 7.7.2 which would involve the enforcement of foreign revenue or penal laws; or
- 7.7.3 which would be inconsistent with English public policy.

Rules and procedures of Courts

7.8 Any action brought against any company in the Courts would be subject to the rules and procedures of the Courts, including the power of a Court, at its discretion to order a claimant in an action, being a party not ordinarily resident in some part of the United Kingdom to provide security for costs.

Oral amendments to Reviewed Initial Primary Transaction Documents

7.9 The parties to the Loan Agreement may be able to amend the Loan Agreement by oral agreement despite any provision to the contrary.

Opinion of Special English Counsel to COBE

Restrictions on exercise of statutory power

7.10 Except as otherwise referred to in this Opinion any provision of the Loan Agreement which constitutes, or purports to constitute, a restriction on the exercise of any statutory power by any party to the Loan Agreement or any other person may be ineffective.

Future agreement

7.11 To the extent that any matter is expressly to be determined by future agreement or negotiation in the Loan Agreement, the relevant provision may be unenforceable or void for uncertainty.

Unreasonable or Arbitrary Basis

7.12 A determination, designation, calculation or certificate of any party in relation to any of the Loan Agreement as to any matter provided for therein might, in certain circumstances, be held by a court not to be final, conclusive and binding (for example, if it could be shown to have an unreasonable or arbitrary basis or not to have been reached in good faith) notwithstanding the provisions of the relevant documents.

United Nations Sanctions

Obligation, transfer or payment, unenforceable or void

7.13 If any party to the Loan Agreement [or any party to any transfer of, or payment in respect of, a Note] is controlled by or otherwise connected with a person or is itself resident in, incorporated in or constituted under the laws of or exercising public functions in a country which is the subject of United Nations, European Community or UK sanctions implemented or effective in the United Kingdom under the United Nations Act 1946 or the Emergency Laws (Reenactments and Repeals) Act 1964 or the Anti-terrorism, Crime and Security Act 2001 or under the Treaty establishing the European Community, as amended, or is otherwise the target of any such sanctions, then the obligations to that party under the Loan Agreement, or in respect of the relevant transfer or payment, may be unenforceable or void or otherwise affected.

Discretion exercised reasonably

7.14 Where any person is vested with a discretion or may determine a matter in its opinion, English law may require that such discretion is exercised reasonably or that such opinion is based on reasonable grounds.

Disapplication of delays resulting in waiver

7.15 Any provision of the Loan Agreement stating that a failure or delay, on the part of any party to the Loan Agreement, in exercising any right or remedy under the Loan Agreement shall not operate as a waiver of such right or remedy may not be effective.

Opinion of Special English Counsel to COBE

Each of the Lenders party to the Credit Agreement referred to below JPMorgan Chase Bank, as Administrative Agent 270 Park Avenue New York, New York 10017

Ladies and Gentlemen:

I am the General Counsel and Secretary of Capital One Financial Corporation ("<u>COFC</u>"), Capital One Bank ("<u>COB</u>") and Capital One, F.S.B. ("<u>FSB</u>" and, collectively with COFC and COB, the "<u>U.S. Borrowers</u>"), and, together with other attorneys under my supervision, have acted as counsel to the U.S. Borrowers and Capital One Bank (Europe) plc ("<u>COBE</u>" and, collectively with the U.S. Borrowers, the "<u>Borrowers</u>") in connection with (i) the Credit Agreement (the "<u>Credit Agreement</u>") dated as of June [__], 2004 among the Borrowers, the Lenders party thereto and JPMorgan Chase Bank, as Administrative Agent, providing for loans to be made by the Lenders to the Borrowers in an aggregate initial principal amount not exceeding \$750,000,000 (or, to the extent specified in the Credit Agreement, its equivalent in certain foreign currencies and as such amount may be increased pursuant to Section 2.10 of the Credit Agreement) and (ii) the various other agreements, instruments and other documents referred to in the next following paragraph. Capitalized terms used but not defined herein have the respective meanings given to such terms in the Credit Agreement. This opinion letter is being delivered pursuant to Section 6.01(e) of the Credit Agreement.

In rendering the opinions expressed below, we have examined executed or conformed counterparts, or copies otherwise identified to our satisfaction of the following agreements, instruments and other documents, copies of which have been delivered to you:

(a) the Credit Agreement;

(b) the Notes; and

Opinion of Counsel to the Borrowers

(c) such records of the Borrowers and such other documents as we have deemed necessary or appropriate as a basis for the opinions set forth below.

The agreements, instruments and other documents referred to in clauses (a) and (b) above are collectively referred to as the "Credit Documents."

We have obtained or have been furnished with, and as to factual matters have relied upon, such certificates, advices and assurances from representatives of the Borrowers, public officials and others and the representations and warranties and covenants and agreements of the parties contained in the Credit Documents as we have deemed necessary or appropriate for the purposes of this opinion. In connection with the opinions herein, we have assumed the legal capacity of all natural persons, the authenticity of all items submitted to us as originals and the conformity with originals of all items submitted to us as copies. We have also assumed the genuineness of all signatures and the due execution and delivery pursuant to due authorization of the Credit Documents by, and the valid and binding nature of such Credit Documents on, the parties thereto other than the Borrowers.

Based on and subject to the foregoing and the other paragraphs hereof, it is my opinion that:

1. COFC is a corporation validly existing and in good standing under the laws of the State of Delaware. COB is a banking corporation validly existing and in good standing under the laws of the Commonwealth of Virginia. FSB is a savings bank validly existing and in good standing under the laws of the United States of America.

2. Each U.S. Borrower has the corporate power to execute and deliver the Credit Documents to which it is a party and to consummate the transactions set forth therein. Each U.S. Borrower has, in all material respects, the corporate power to borrow under the Credit Agreement.

3. Each Credit Document has been duly authorized by all necessary corporate action on the part of each U.S. Borrower which is a party to such Credit Document.

4. The Credit Agreement has been duly executed and delivered by each U.S. Borrower party thereto.

5. Neither the execution and delivery by each U.S. Borrower of the Credit Documents to which it is a party nor the performance by each U.S. Borrower of its obligations under the Credit Documents to which it is a party, result in a breach of the terms of, or constitute a default under the applicable organizational documents of each U.S. Borrower, each as amended, and do not result in a material breach of the terms of, or constitute a material default under (i) any rule, order (known to me), statute or regulation, to the extent the foregoing relate to Delaware corporate, or other Delaware laws, Virginia banking, or other Virginia laws or United

Opinion of Counsel to the Borrowers

States federal law, of any Delaware, Virginia or United States court, regulatory body, or administrative or governmental agency having jurisdiction over each U.S. Borrower, as applicable, or (ii) the terms of any material indenture or other material agreement or instrument to which each U.S. Borrower is a party (and, for purposes of clauses (i) and (ii) of this paragraph 5, a breach or default is "material" solely if it would materially adversely affect the validity or enforceability of, or the rights and remedies of the Lenders or the Administrative Agent under, the Credit Agreement).

6. Neither the execution and delivery by COBE of the Credit Documents to which it is a party nor the performance by COBE of its obligations under the Credit Documents to which it is a party, result in a material breach of the terms of, or constitute a material default under (i) any rule, order (known to me), statute or regulation, to the extent the foregoing relate to Delaware corporate, or other Delaware laws, Virginia banking, or other Virginia laws or United States federal law, of any Delaware, Virginia or United States court, regulatory body, or administrative or governmental agency having jurisdiction over COBE, as applicable, or (ii) the terms of any material indenture or other material agreement or instrument to which COBE is a party.

7. Except as set forth in Schedule 7.03 to the Credit Agreement, to my knowledge there are no actions, proceedings or investigations pending or threatened in writing before any court, governmental or regulatory authority or agency against or affecting any Borrower or any of its Subsidiaries or any of their respective Properties, except proceedings that, if adversely determined, would not have a Material Adverse Effect.

Insofar as the foregoing opinions relate to (i) the good standing of COB, they are based solely on a certificate of good standing dated June [·], 2004 received from the State Corporation Commission of the Commonwealth of Virginia and attached hereto as Exhibit A, (ii) the good standing of COFC, they are based solely on a certificate of good standing dated June [·], 2004 received from the Secretary of State of the State of Delaware and attached hereto as Exhibit B, and (iii) the good standing of FSB, they are based solely on a certificate of corporate existence dated June [·], 2004 received from the Office of Thrift Supervision, attached hereto as Exhibit C.

The foregoing opinions are subject to (i) applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or other similar law affecting the rights and remedies of creditors generally as may be in effect from time to time and (ii) general principles of equity, regardless of whether such is considered in a proceeding at law or in equity. We note that the rights to indemnity and contribution under the Credit Documents may be limited by United States federal laws or the public policy underlying such laws.

The opinions expressed herein are limited to, and I express no opinion as to any jurisdiction other than, the laws of the Commonwealth of Virginia, the Delaware General Corporation Law and the federal law of the United States of America.

Opinion of Counsel to the Borrowers

This opinion letter is delivered to you in connection with the above described transaction and is solely for your benefit and may not be relied upon or used by, circulated, quoted or referred to, nor may copies be delivered to, any other person without my prior written consent. We disclaim any obligation to update this opinion letter for events occurring or coming to our attention after the date hereof.

Very truly yours,

John G. Finneran, Jr.

Opinion of Counsel to the Borrowers

June 29, 2004

Each of the Lenders party to the Credit Agreement referred to below JPMorgan Chase Bank, as Administrative Agent 270 Park Avenue New York, New York 10017

Ladies and Gentlemen:

We have acted as special New York counsel to JPMorgan Chase Bank ("<u>JPMorgan</u>") in connection with (i) the Credit Agreement dated as of June 29, 2004 (the "<u>Credit Agreement</u>") among Capital One Financial Corporation ("<u>COFC</u>"), Capital One Bank ("<u>COB</u>"), Capital One, F.S.B ("<u>FSB</u>"), Capital One Bank (Europe) plc ("<u>COBE</u>" and, collectively with COFC, COB and FSB, the "<u>Borrowers</u>"), the Lenders party thereto and JPMorgan, as Administrative Agent, providing for loans to be made by the Lenders to the Borrowers in an aggregate principal amount not exceeding \$750,000,000 (or, to the extent specified in the Credit Agreement, its equivalent in certain foreign currencies and as such amount may be increased pursuant to Section 2.10 of the Credit Agreement) and (ii) the various other agreements, instruments and other documents referred to in the next following paragraph. Capitalized terms used but not defined herein have the respective meanings given to such terms in the Credit Agreement. This opinion letter is being delivered pursuant to Section 6.01(f) of the Credit Agreement.

In rendering the opinions expressed below, we have examined the following agreements, instruments and other documents:

(a) the Credit Agreement; and

(b) such other documents as we have deemed necessary as a basis for the opinions expressed below.

In our examination, we have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals and the conformity with authentic original documents of all documents submitted to us as copies, and we have assumed that all authorizations, approvals or consents of, and all filings and registrations with, any governmental or regulatory authority or agency required for the making and performance by each Borrower of

Opinion of Special New York Counsel to JPMorgan

the Credit Agreement have been obtained or made and are in effect. When relevant facts were not independently established, we have relied upon representations made in or pursuant to the Credit Agreement.

In rendering the opinions expressed below, we have assumed, with respect to all of the documents referred to in this opinion letter, that:

(i) such documents have been duly authorized by, have been duly executed and delivered by, and (except to the extent set forth in the opinions below as to the Borrowers) constitute legal, valid, binding and enforceable obligations of, all of the parties to such documents;

(ii) all signatories to such documents have been duly authorized; and

(iii) all of the parties to such documents are duly organized and validly existing and have the power and authority (corporate or other) to execute, deliver and perform such documents.

Based upon and subject to the foregoing and subject also to the comments and qualifications set forth below, and having considered such questions of law as we have deemed necessary as a basis for the opinions expressed below, we are of the opinion that the Credit Agreement constitutes, and each of the Notes, when duly executed and delivered in accordance with the Credit Agreement will constitute, the legal, valid and binding obligation of each Borrower, enforceable against each Borrower in accordance with its terms, except as may be limited by bankruptcy, fraudulent conveyance or transfer, insolvency, receivership, conservatorship, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally (as such laws would apply in the event of the insolvency, receivership, conservatorship or reorganization of, or other similar occurrence with respect to, COB or FSB) and, in the case of the obligations of COBE, subject to the possible judicial application of foreign laws or governmental action affecting the rights of creditors generally except as the enforceability of the Credit Agreement or the Notes, when duly executed and delivered in accordance with the Credit Agreement, is subject to the application of general principles of equity (regardless of whether considered in a proceeding in equity or at law), including, without limitation, (a) the possible unavailability of specific performance, injunctive relief or any other equitable remedy and (b) concepts of materiality, reasonableness, good faith and fair dealing.

The foregoing opinions are subject to the following comments and qualifications:

(A) The enforceability of Section 11.03 of the Credit Agreement may be limited by (i) laws rendering unenforceable indemnification contrary to Federal or state securities laws and the public policy underlying such laws and (ii) laws limiting the enforceability of provisions exculpating or exempting a party from, or requiring indemnification of a party for, liability for its own action or inaction, to the extent the action or inaction involves gross negligence, recklessness, willful misconduct or unlawful conduct.

Opinion of Special New York Counsel to JPMorgan 2 (B) The enforceability of provisions in the Credit Agreement to the effect that terms may not be waived or modified except in writing may be limited under certain circumstances.

(C) We express no opinion as to (i) the effect of the laws of any jurisdiction in which any Lender is located (other than the State of New York) that limit the interest, fees or other charges such Lender may impose, (ii) Section 4.07(c) of the Credit Agreement, (iii) the second sentence of Section 11.10 of the Credit Agreement, insofar as such sentence relates to the subject matter jurisdiction of the United States District Court for the Southern District of New York to adjudicate any controversy related to any of the Basic Documents, (iv) the waiver of inconvenient forum set forth in the third sentence of Section 11.10 of the Credit Agreement with respect to proceedings in the United States District Court for the Southern District of New York and (v) Section 11.13 of the Credit Agreement.

(D) We wish to point out with reference to obligations stated to be payable in a currency other than Dollars that (i) a New York statute provides that a judgment rendered by a court of the State of New York in respect of an obligation denominated in such other currency would be rendered in such other currency and would be converted into Dollars at the rate of exchange prevailing on the date of entry of the judgment and (ii) a judgment rendered by a Federal court sitting in the State of New York in respect of an obligation denominated in such other currency may be expressed in Dollars, but we express no opinion as to the rate of exchange such Federal court would apply.

The foregoing opinions are limited to matters involving the Federal laws of the United States and the law of the State of New York, and we do not express any opinion as to the laws of any other jurisdiction.

At the request of our client, this opinion letter is, pursuant to Section 6.01(f) of the Credit Agreement, provided to you by us in our capacity as special New York counsel to JPMorgan and may not be relied upon by any Person for any purpose other than in connection with the transactions contemplated by the Credit Agreement without, in each instance, our prior written consent.

Very truly yours,

WFC/WJM

Opinion of Special New York Counsel to JPMorgan 3 [Form of Notice of Borrowing of Syndicated Loans]

[Date]

					-
To:	JPMorgan Chase Bank, as Administrative Agent				
From:	[Name of Borrower]				
Re:	Notice of Borrowing				
Pursuant to Section 2.02 of the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the " <u>Credit Agreement</u> ") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent, the undersigned Borrower hereby gives notice of a borrowing of Syndicated Loans described below:					
Name of E	Borrower:				
Aggregate Amount o	Principal f Loans to be borrowed:				
Currency	of Loans to be borrowed:				
Type of Lo	oans to be borrowed:				
Business I	Day of borrowing:				
Interest Pe	eriod to be applicable:		1		

This notice of borrowing constitutes a certification by the undersigned Borrower to the effect set forth in Section 6.02(c) of the Credit Agreement, both as of the date of this notice of borrowing and, unless the undersigned notifies the Administrative Agent prior to the date of such borrowing, as of the date of such borrowing.

If the undersigned Borrower is FSB or COBE, then COB has signed this notice of borrowing on the line provided below.

¹/ No Loan may be made to FSB or COBE with an Interest Period in excess of six months.

Notice of Borrowing

Terms used herein have the meanings assigned to them in the Credit Agreement.

[NAME OF BORROWER]

By

Title:

[COB hereby confirms its obligations under Section 2.11 of the Credit Agreement after giving effect to the borrowing of Loans by [FSB] [COBE] requested in this notice of borrowing:

CAPITAL ONE BANK

By ______ Title:]²

 $\frac{2}{2}$ Insert if FSB or COBE is the Borrower.

Notice of Borrowing

EXHIBIT E

[Form of Money Market Quote Request]

To: JPMorgan Chase Bank, as Administrative Agent

From: [Name of Borrower]

Re: Money Market Quote Request

Pursuant to Section 2.03 of the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent, we hereby give notice that we request Money Market Quotes from the Lenders for the following proposed Money Market Borrowing(s):

Borrowing Date	Quotation Date ^[1]	Amount ^[2]	Type and Currency ^[3]	Interest Period ^[4]

If the undersigned Borrower is FSB or COBE, then COB has signed this Money Market Quote Request on the line provided below.

Terms used herein have the meanings assigned to them in the Credit Agreement.

[NAME OF BORROWER]

[Date]

By ____

Title:

* All numbered footnotes appear on the last page of this Exhibit.

Money Market Quote Request

[COB hereby confirms its obligations under Section 2.11 of the Credit Agreement after giving effect to the borrowing of Loans by [FSB] [COBE] requested in this Money Market Quote Request:

CAPITAL ONE BANK

By

- Title:]³
- ^[1] In the case of Set Rate Loans to be denominated in Dollars, for use if a Set Rate in a Set Rate Auction is requested to be submitted before the Borrowing Date.
- ^[2] Each amount must be an integral multiple of \$1,000,000 and at least \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)).
- ^[3] Insert either "LIBO Margin" (in the case of LIBOR Market Loans) or "Set Rate" (in the case of Set Rate Loans).
- ^[4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Loan, a period of not less than seven days after the making of such Set Rate Loan and ending on a Business Day. No Loan may be made to FSB or COBE with an Interest Period in excess of six months.

 $\frac{3}{4}$ Insert if FSB or COBE is the Borrower.

Money Market Quote Request 2

[Form of Money Market Quote]

To:	JPMorgan Chase Bank, as Administrative Agent
Attention:	Loan & Agency Services
Re:	Money Market Quote to [Name of Borrower] (the " <u>Borrower</u> ")

This Money Market Quote is given in accordance with Section 2.03(c) of the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent. Terms defined in the Credit Agreement are used herein as defined therein.

In response to the Borrower's invitation dated ______, we hereby make the following Money Market Quote(s) on the following terms:

1. Quoting Lender:

2. Person to contact at Quoting Lender:

3. We hereby offer to make Money Market Loan(s) in the following principal amount[s], for the following Interest Period(s) and at the following rate(s):

Borrowing Date	Quotation Date ^[1]	Amount ^[2]	Type and Currency ^[3]	Interest Period ^[4]	Rate ^[5]

<u>provided</u> that the Borrower may not accept offers that would result in the undersigned making Money Market Loans pursuant hereto in excess of \$______ in the aggregate (the "<u>Money Market Loan Limit</u>").

* All numbered footnotes appear on the last page of this Exhibit.

Money Market Quote

We understand and agree that the offer(s) set forth above, subject to the satisfaction of the applicable conditions set forth in the Credit Agreement, irrevocably obligate[s] us to make the Money Market Loan(s) for which any offer(s) (is/are) accepted, in whole or in part (subject to the third sentence of Section 2.03(e) of the Credit Agreement and any Money Market Loan Limit specified above).

Very truly yours,

[NAME OF LENDER]

By

Authorized Officer

Dated: _____

- ^[2] The principal amount bid for each Interest Period may not exceed the principal amount requested. Bids must be made for an integral multiple of \$1,000,000 and at least \$5,000,000 (or, in the case of a Borrowing of Money Market Loans denominated in an Alternative Currency, the Foreign Currency Equivalent thereof (rounded to the nearest 1,000 units of such Alternative Currency)).
- ^[3] Indicate "LIBO Margin" (in the case of LIBOR Market Loans) or "Set Rate" (in the case of Set Rate Loans).
- ^[4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Loan, a period of not less than seven days after the making of such Set Rate Loan and ending on a Business Day, as specified in the related Money Market Quote Request. No Loan may be made to FSB or COBE with an Interest Period in excess of six months.
- ^[5] For a LIBOR Market Loan, specify margin over or under the Eurocurrency Rate determined for the applicable Interest Period. Specify percentage (rounded to the nearest 1/10,000 of 1%) and specify whether "PLUS" or "MINUS". For a Set Rate Loan, specify rate of interest per annum (rounded to the nearest 1/10,000 of 1%).

Money Market Quote

^[1] As specified in the related Money Market Quote Request.

[Date]

[Form of Confidentiality Agreement]

CONFIDENTIALITY AGREEMENT

[Insert Name and Address of Prospective Participant or Assignee]

Re: Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent.

Ladies and Gentlemen:

As a Lender party to the Credit Agreement, we have agreed with the Borrowers pursuant to Section 11.12 of the Credit Agreement to keep confidential, except as otherwise provided therein, all non-public information delivered to us pursuant to the Credit Agreement.

As provided in said Section 11.12, we are permitted to provide you, as a prospective [holder of a participation in the Loans (as defined in the Credit Agreement)] [assignee Lender], with certain of such non-public information subject to the execution and delivery by you, prior to receiving such non-public information, of a Confidentiality Agreement in this form. Such information will not be made available to you until your execution and return to us of this Confidentiality Agreement.

Accordingly, in consideration of the foregoing, you agree (on behalf of yourself and each of your affiliates, directors, officers, employees and representatives and for the benefit of us and the Borrowers) that (A) such information will not be used by you except in connection with the proposed [participation][assignment] mentioned above and (B) you shall keep such information confidential, <u>provided</u> that (x) nothing herein shall limit the disclosure of any such information (i) after such information shall have become public (other than through a violation of Section 11.12 of the Credit Agreement), (ii) to the extent required by statute, rule, regulation or judicial process, (iii) to your counsel or to counsel for any of the Lenders or the Administrative Agent, (iv) to bank examiners (or any other regulatory authority having jurisdiction over any Lender or the Administrative Agent), or to auditors or accountants, (v) to the Administrative

Confidentiality Agreement

Agent or any other Lender, (vi) in connection with the enforcement of rights or remedies under the Credit Agreement, (vii) to a subsidiary or affiliate of yours as provided in Section 11.12(a) of the Credit Agreement or (viii) to any assignee or participant (or prospective assignee or participant) so long as such assignee or participant (or prospective assignee or participant) first executes and delivers to you a Confidentiality Agreement substantially in the form hereof and (y) in no event shall you be obligated to return any materials furnished to you pursuant to this Confidentiality Agreement.

If you are a prospective assignee, your obligations under this Confidentiality Agreement shall be superseded by Section 11.12 of the Credit Agreement on the date upon which you become a Lender under the Credit Agreement pursuant to Section 2.12, 5.07 or 11.06(b) thereof. This Confidentiality Agreement shall be governed by, and construed in accordance with, the law of the State of New York without reference to choice of law doctrine.

Please indicate your agreement to the foregoing by signing as provided below the enclosed copy of this Confidentiality Agreement and returning the same to us.

Very truly yours,

[INSERT NAME OF LENDER]

By ______ Title:

The foregoing is agreed to as of the date of this letter:

[INSERT NAME OF PROSPECTIVE PARTICIPANT OR ASSIGNEE]

By

Title:

Confidentiality Agreement

[Form of Assignment and Assumption]

ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (the "<u>Assignment and Assumption</u>") is dated as of the Effective Date set forth below and is entered into by and between [*Insert name of Assignor*] (the "<u>Assignor</u>") and [*Insert name of Assignee*] (the "<u>Assignee</u>"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (as amended, the "<u>Credit Agreement</u>"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below (i) all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the respective facilities identified below (including any letters of credit included in such facilities) and (ii) to the extent permitted to be assigned under applicable law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clauses (i) and (ii) above being referred to herein collectively as the "<u>Assigned Interest</u>"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

1. Assignor:

2. Assignee:

Assignment and Assumption

		[and is an Affiliate/Approved	d Fund of [<i>identify Lender</i>] ⁴]		
3.	Borrower:				
4.	Administrative Agent:	JPMorgan Chase Bank, as th Agreement	e administrative agent under th	e Credit	
5.	Credit Agreement:	The \$750,000,000 Credit Agreement dated as of June 29, 2004 among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the Lenders parties thereto and JPMorgan Chase Bank, as Administrative Agent			
6.	Assigned Interest:				
	Facility Assigned		Aggregate Amount of Commitment/Loans for all Lenders	Amount of Commitment/Loans Assigned	Percentage Assigned of Commitment/Loans ⁵
			\$	\$	%

Effective Date (herein, the "<u>Effective Date</u>"): ______, 20__ [TO BE INSERTED BY ADMINISTRATIVE AGENT AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER IN THE REGISTER THEREFOR.]

\$ \$ \$ \$ %

%

⁴ Select as applicable.

⁵ forth, to at least 9 decimals, as a percentage of the Commitment/Loans of all Lenders thereunder.

Assignment and Assumption

- 2 -

The terms set forth in this Assignment and Assumption are hereby agreed to:

ASSIGNOR

[NAME OF ASSIGNOR]

By: ______ Title:

ASSIGNEE

[NAME OF ASSIGNEE]

Assignment and Assumption

- 3 -

[Consented to and]⁶ Accepted:

JPMORGAN CHASE BANK, as Administrative Agent

By

Title: [Consented to:]⁷

CAPITAL ONE FINANCIAL CORPORATION

By

Title:

CAPITAL ONE BANK

By ______ Title:

⁶ To be added only if the consent of the Administrative Agent is required by the terms of the Credit Agreement.

⁷ To be added only if the consent of the Borrowers is required by the terms of the Credit Agreement.

Assignment and Assumption

- 4 -

CAPITAL ONE, F.S.B.

By ______ Title:

CAPITAL ONE BANK (EUROPE) PLC

By ______ Title:

Assignment and Assumption

- 5 -

\$750,000,000 CREDIT AGREEMENT DATED AS OF JUNE 29, 2004

AMONG CAPITAL ONE FINANCIAL CORPORATION, CAPITAL ONE BANK, CAPITAL ONE, F.S.B., CAPITAL ONE BANK (EUROPE) PLC, CERTAIN LENDERS PARTY THERETO AND JPMORGAN CHASE BANK, AS ADMINISTRATIVE AGENT

STANDARD TERMS AND CONDITIONS FOR

ASSIGNMENT AND ASSUMPTION

1. Representations and Warranties.

1.1 <u>Assignor</u>. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Basic Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Basic Documents or any collateral thereunder, (iii) the financial condition of any Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Basic Document or (iv) the performance or observance by any Borrower, any of its Subsidiaries or Affiliates or any other Person of any of its respective obligations under any Basic Document.

1.2. <u>Assignee</u>. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it satisfies the requirements, if any, specified in the Credit Agreement that are required to be satisfied by it in order to acquire the Assigned Interest and become a Lender, (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 8.01 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender, and (v) if it is not a U.S. Person, attached to this Assignment and Assumption is any documentation required

Assignment and Assumption

to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Basic Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Basic Documents are required to be performed by it as a Lender.

2. <u>Payments</u>. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.

3. <u>General Provisions</u>. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the State of New York.

Assignment and Assumption

- 2 -

[Form of Commitment Increase Letter]

COMMITMENT INCREASE LETTER

Capital One Bank Capital One, F.S.B. Capital One Financial Corporation Capital One Bank (Europe) plc 1680 Capital One Dr. McClean, VA 22101-2980

JPMorgan Chase Bank, as Administrative Agent Loan & Agency Services 1111 Fanin, 10th Floor Houston, TX 77002 Attention: Mr. Jeremy M. Jones

Ladies and Gentlemen:

Reference is made to the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit</u> <u>Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B., Capital One Bank (Europe) plc, the lenders party thereto and JPMorgan Chase Bank, as Administrative Agent. Terms used but not defined herein have the respective meanings given to such terms in the Credit Agreement.

This Commitment Increase Letter is delivered pursuant to Section 2.10 of the Credit Agreement.

If, prior to the execution and delivery of this Commitment Increase Letter, the undersigned is a Lender already party to the Credit Agreement, then the undersigned hereby

Commitment Increase Letter

133

[Date]

agrees that, effective as of the Commitment Increase Date set forth below, the Commitment of such Lender set forth below is increased by an amount equal to the "Commitment Increase Amount" set forth below.

If, prior to the execution and delivery of this Commitment Increase Letter, the undersigned is not a Lender already party to the Credit Agreement, then the undersigned hereby agrees that, effective as of the Commitment Increase Date set forth below, the undersigned shall have a Commitment in an amount equal to the "Commitment Increase Amount" set forth below.

 Commitment Increase Date:
 ______, ____

 Commitment Increase Amount:
 \$ _______

The undersigned agrees with the Borrowers and the Administrative Agent that the undersigned will, from and after the Commitment Increase Date, be a "Lender" under the Credit Agreement (if not already a "Lender" thereunder) and perform all of the obligations of the undersigned as a "Lender" under the Credit Agreement in respect of the Commitment Increase Amount (together with, if already a "Lender" under the Credit Agreement, the Commitment of the Lender in effect immediately prior to the execution and delivery of this Commitment Increase Letter).

This Commitment Increase Letter shall be governed by and construed in accordance with the law of the State of New York without reference to choice of law doctrine.

Very truly yours,

[INSERT NAME OF LENDER]

By

Title:

Agreed:

CAPITAL ONE FINANCIAL CORPORATION

By

Title:

Commitment Increase Letter

- 2 -

CAPITAL ONE BANK

By ______ Title:

CAPITAL ONE, F.S.B.

By ______ Title:

CAPITAL ONE BANK (EUROPE) PLC

By ______ Title:

JPMORGAN CHASE BANK, as Administrative Agent

By ______ Title:

Commitment Increase Letter

- 3 -

[Form of Drawing Certificate] DRAWING CERTIFICATE

Capital One Bank

Ladies and Gentlemen:

Reference is made to the Undertaking entered into by Capital One Bank ("<u>COB</u>") pursuant to Section 2.11 of the Credit Agreement dated as of June 29, 2004 (as modified and supplemented and in effect from time to time, the "<u>Credit Agreement</u>") among Capital One Financial Corporation, Capital One Bank, Capital One, F.S.B. ("<u>FSB</u>"), Capital One Bank (Europe) plc ("<u>COBE</u>"), the lenders party thereto and the Administrative Agent party thereto. Terms used but not defined herein have the respective meanings given to such terms in the Credit Agreement.

The undersigned, a duly authorized representative of the Administrative Agent (the "Administrative Agent"), hereby certifies that:

1. The Administrative Agent is the beneficiary of the Undertaking.

2. The Administrative Agent hereby requests payment in an amount equal to the amount of the draft accompanying this Certificate (the "<u>Draft</u>"), which amount is not greater than the aggregate amount due and payable by [FSB] [COBE] on the date of this Certificate in respect of the principal of or interest on the Loans made by the Lenders to, and the Notes held by each Lender of, [FSB] [COBE] or any other amount owing by [FSB] [COBE] to any Lender or the Administrative Agent under the Credit Agreement or any of the Notes.

3. The amount represented by the Draft has not been paid by [FSB] [COBE] and has not been the subject of and paid pursuant to a prior drawing by the Administrative Agent under the Undertaking.

4. The date of the Draft is the date of this Certificate.

IN WITNESS WHEREOF, the undersigned has executed this Certificate on [insert date of draft accompanying this Certificate].

[NAME OF ADMINISTRATIVE AGENT], as Administrative Agent

By

Authorized Representative

Drawing Certificate

CERTIFICATION FOR QUARTERLY REPORT ON FORM 10-Q OF CAPITAL ONE FINANCIAL CORPORATION AND CONSOLIDATED SUBSIDIARIES

I, Richard D. Fairbank, certify that:

- 1. I have reviewed this quarterly report of Capital One Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 1. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 5, 2004

CAPITAL ONE FINANCIAL CORPORATION

By: /s/ RICHARD D. FAIRBANK

Richard D. Fairbank Chairman of the Board, Chief Executive Officer and President

CERTIFICATION FOR QUARTERLY REPORT ON FORM 10-Q OF CAPITAL ONE FINANCIAL CORPORATION AND CONSOLIDATED SUBSIDIARIES

I, Gary L. Perlin, certify that:

- 1. I have reviewed this quarterly report of Capital One Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 1. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 5, 2004

CAPITAL ONE FINANCIAL CORPORATION

By: /s/ GARY L. PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer

Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Richard D. Fairbank, Chairman and Chief Executive Officer of Capital One Financial Corporation, a Delaware corporation ("Capital One"), do hereby certify that:

The Quarterly Report on Form 10-Q for the period ended June 30, 2004 (the "Form 10-Q") of Capital One fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Capital One.

Dated: August 5, 2004

By: /s/ RICHARD D. FAIRBANK

Richard D. Fairbank Chairman of the Board, Chief Executive Officer and President

Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Gary L. Perlin, Executive Vice President and Chief Financial Officer of Capital One Financial Corporation, a Delaware corporation ("Capital One"), do hereby certify that:

The Quarterly Report on Form 10-Q for the period ended March 31, 2004 (the "Form 10-Q") of Capital One fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Capital One.

Dated: August 5, 2004

By: /s/ GARY L. PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer